ECONOMIC STUDIES

DEPARTMENT OF ECONOMICS SCHOOL OF BUSINESS, ECONOMICS AND LAW UNIVERSITY OF GOTHENBURG 189

Essays in Industry Dynamics on Imperfectly Competitive Markets

Florin G. Maican



UNIVERSITY OF GOTHENBURG

ISBN 978-91-85169-49-8 ISSN 1651-4289 print ISSN 1651-4297 online

Printed in Sweden, Geson Hylte Tryck 2010



To my wife, Iuliana, and my son, Rares.

Contents

Acknowledgements	iii
Short Introduction	v

Essay 1: Productivity Dynamics and the Role of "Big-Box" Entrants in T	Retailing
1. Introduction	2
2. The retail food market and data	5
3. Productivity estimation	10
4. Large entrants and productivity	21
5. Conclusions	27
References	29
Appendix A	42
Appendix B	43
Appendix C	46

Essay 2: Productivity Dynamics, R&D, and Competitive Pressure

1. Introduction	2
2. Overview of the industries	7
3. Modeling framework	9
4. Productivity estimation	14
5. Empirical results	21
6. Discussion and conclusions	29
References	31
Appendix A	43
Appendix B	43

Essay 3: Industry Dynamics and Format Repositioning in Retail	
1. Introduction	2
2. Overview of the Swedish retail food industry	6
3. Modeling approach	10
4. Estimation	12
6. Results	23
7. Conclusions	28
References	29
Appendix A	43
Appendix B	43
Appendix C	44

Essay 4: From Boom to Bust: A Dynamic Analysis of IT Services

1. Introduction	2
2. Overview of the Swedish IT services industry	6
3. Modeling approach	10
4. Estimation	12
5. Results	20
6. Conclusions	30
References	31
Appendix A	47
Appendix B	47
Appendix C	48

Acknowledgements

I would like to express my deepest gratitude and thanks to:

Lennart Hjalmarsson, my supervisor. Without your help, this thesis would never have been written. Thank you for your generosity and incredible support, and for all early meetings (before 8:30 am).

Richard Sweeney, my co-supervisor and co-author on my Ph.Licentiate on Empirical International Finance. I admire your professionalism and your desire to make a difference. I have learned a lot from you.

The following institutions for providing financial support: The Jan Wallander and Tom Hedelius Foundation, Swedish Competition Authority, Handelns Utvecklingsråd, and The Knut and Alice Wallenberg Foundation.

Sorin Maruster and Maria Risberg, my friends. Thank you for generosity of spirit. I appreciate your immense support before entering the PhD program.

Matilda Orth, my co-author and friend. Thank you for your true friendship and incredible support, and for reading every new version of my papers. I admire your attention to detail and your commitment to excellence.

The following people, who, in their courses, have influenced my thinking about empirical Industrial Organization: Daniel Ackerberg, Victor Aguirregabiria, Marcus Asplund, and Ariel Pakes. Thank you for finding time after long course hours to discuss my research. I truly appreciate your help. A special thank you to the Nordic Network in Economics (NNE) for providing excellent IO courses.

Rune Stenbacka, who read several versions of the first two essays. Thanks for your generosity. I appreciate your professionalism.

Johan Stennek and Måns Söderbom, with whom I have been worked in IO courses. Thanks for your support.

The following people, who have given me valuable advice during the research process: Mats Bergman, Arne Bigsten, Hans Bjurek, Fredrik Carlsson, Evert Carlsson, Dick Durevall, Lennart Flood, Douglas Hibbs, Olof Johansson-Stenman, Per Lundborg, Catalin Starica, Roger Wahlberg, and Joakim Westerlund. Thank you all for your support.

Cristian Huse, who read the final version of the thesis. Thank you for valuable comments and suggestions.

Eva Jonason and Eva-Lena Neth-Johansson for invaluable administrative support over the years. I admire how fast you can solve administrative problems. Åsa Adin for helping me with wage contracts. Mona Jönefors for making sure that we stay within budget. Jeanette Saldjoughi for helping me with practicalities concerning Licentiate and PhD thesis. Thank you for keeping my life in order.

All my friends and colleagues at the Department of Economics and Center for Finance and former colleagues at the Department of Economic Cybernetics for your support. You are too many to list here. Thank you so much!

My family for their love and support. Thanks for understanding long hours of work, and sacrificed weekends and vacations. Iuliana, my wife, I appreciate how well you understand me and your love, support, and humor. Words can never express how much I appreciate your support. Rares, my son, who was born when I started this PhD program. Thank you for boundless energy, great humor, and being so supportive. Thank you also for decorating my office and updating me about Harry Potter and Star Wars. My parents and grandparents, for providing me with values and a work ethic that have truly helped me in life. My sister Adriana for all your support.

Radio Swiss Classic (http://www.radioswissclassic.ch) for broadcasting an amazing selection of the best classical music.

April, 2010 Gothenburg "Not everything that counts can be counted, and not everything that can be counted counts." Einstein

Short Introduction

This thesis consists of four empirical essays on imperfectly competitive markets. The essays are grouped by the methodology used.¹ Essay 1 studies the effect of large entrants on productivity dynamics in Swedish food retail. Essay 2 studies the productivity dynamics in high R&D spending manufacturing industries where competitive pressure plays a key role in the choice of R&D spending. Essay 3 analyses store format repositioning in Swedish food retail. Essay 4 studies the impact of the 2001 dot-com bust, a natural experiment, on productivity dynamics and cost structure in Swedish IT services. Essays 1 and 2 use a single agent dynamic framework, whereas Essays 3 and 4 use truly dynamic games.

Essay 1: Productivity Dynamics and the Role of Big-Box Entrants in Retailing

(with Matilda Orth)

Products from the food sector fulfill our needs for basic survival and are purchased by almost everyone in society. Entry of large (big-box) stores along with a drastic fall in the total number of stores is a striking trend in retail markets in both US and many European countries. In retail, there is a lack of knowledge regarding the market structure effects caused by large entrants (Swedish Competition Authority, 2004:2). An interesting economic issue is whether entrants influence the productivity of incumbent stores. The question posed is of certain importance due to the existing entry regulation (common across European countries), giving the local governments the power to decide whether or not a store is allowed to enter the market. Essay 1 uses a dynamic structural model to estimate total factor productivity in retail. Then it assesses whether entry of large stores drives exit and growth in the productivity distribution of incumbents. Using detailed data on all retail food stores in Sweden, the paper finds that local market characteristics, selection, and nonlinearities in the productivity process are important when estimating retail productivity. We find that large entrants force low productivity stores to exit and surviving stores to increase their productivity growth. Growth increases most among incumbents in the bottom part of the productivity distribu-

¹Ackerberg et al. (2008) and Pakes (2008) survey recent developments in the empirical analysis of imperfectly competitive markets.

tion, and then declines with the productivity level of incumbents. The essay uses political preferences in local markets to control for endogeneity of large entrants. The findings suggest that large entrants play a crucial role for driving productivity growth.

Essay 2: Productivity Dynamics, R&D, and Competitive Pressure

The link between investment in research and development (R&D) and firm performance is one of the most studied topics in industrial organization. Early literature on this relationship largely focused on estimating the average or expected returns (private or social) to R&D spending. However, R&D spending not only increases a firms productivity, it also affects the entire productivity distribution of the industry through exit of firms and reallocations as well as displacements of labor and capital. From a policy perspective, the analysis of the entire productivity distribution enhances our understanding of the dynamics of firms investment in R&D and physical capital.²

Essay 2 proposes a dynamic structural model to estimate productivity when productivity evolves as an endogenous process and firms decide how much to invest depending on the competitive pressure they face. Using data on all manufacturing firms in Sweden, this paper finds that open market policies and entrepreneurship policies complement R&D policies and are important drivers of the competitiveness of established firms. Conservative estimates suggest that the optimal investment is at least 0.7 to 2.5 times the actual investment in R&D for a median firm and 2 to 4 times for a firm located in the upper part of the productivity growth distribution.

Essay 3: Industry Dynamics and Format Repositioning in Retail

Powerful chains dominate the retail food markets in both Europe and US due to increasing importance of for example information technology, distribution systems, and demand. Each chain operates a number of well-defined store formats and continuously considers a trade-off between repositioning of store formats, entry of new stores, and exit of existing stores. Recent investment strategies aim to increase product differentiation in store formats. Each investment implies, however, a sunk cost. Since both entry and repositioning of formats are regulated, insights about the trade-off between entry and repositioning and its link to competition closely

 $^{^{2}}$ In the theoretical firm dynamics models proposed by Ericson and Pakes (1995), Hopenhayn (1992), and Jovanovic (1982), the stochastic evolution of firm productivity determines the success or failure of the firm in an industry.

connect to competition policy. A large variety of store formats can ensure that consumers gain access to low prices and wide and attractive product assortments (The Nordic Competition Authorities 2005:1). In Sweden, this is particularly important as municipalities have the obligation to evaluate the competitive impact of new stores (Swedish Competition Authority 2008:5).

Essay 3 proposes a fully dynamic oligopoly model to estimate the costs of repositioning store formats together with sunk costs of entry and sell-off values of exit in the retail industry. In differentiated product markets, when firms are affected by demand shocks, they may react by repositioning their products, which in turn affects market structure. The model gives important information about driving forces behind format changes and how such repositioning can be linked to entry and exit. Using data from Sweden, the results indicate that both repositioning and entry costs increase with market size, and their growth decreases when moving to larger markets. Small markets have higher sell-off values than repositioning costs, but large entry costs. The difference between higher entry and lower repositioning costs explains why the number of observed repositionings is higher than the number of entrants. Since entry is regulated in most OECD countries, repositioning costs and their link to competition have important implications for competition policy.

Essay 4: From Boom to Bust: A Dynamic Analysis of IT Services

The IT industry contributes significantly to increased productivity and improved service quality in virtually all sectors of the economy (Jorgenson et al., 2008). The lower adoption rate and small size of IT investments in Europe have been found to have been responsible for the lower productivity growth in Europe than in US in the 1990s (van Ark et al., 2008).

Essay 4 proposes a fully dynamic structural model to analyze the impact of the 2001 dot-com bust on the productivity dynamics and the cost structure for IT services. Aggregate demand shocks such as the burst of the 2001 dot-com bubble affect firms behavior and, therefore, the market structure. The empirical application builds on an eight year panel dataset that includes every IT service firm in Sweden and is representative for many other European countries. Incumbents are more productive than entrants and net exit contributed the most to productivity growth in the IT services after the dot-com bust. Essay 4 finds higher fixed investment and labor costs for software but lower for operational services after the dot-com bust. Finding the relative importance of fixed costs is a step closer to being able to link policies that affect adjustment costs in IT services.

References

- ACKERBERG, D., L. BENKARD, S. BERRY, AND A. PAKES (2008): Handbook of Econometrics, vol. 6, chap. Econometric Tools for Analyzing Market Outcomes, pp. 4171–4276. Elsevier.
- ERICSON, R., AND A. PAKES (1995): "Markov-Perfect Industry Dynamics: A Framework for Empirical Work," *Review of Economic Studies*, 62, 53–83.
- HOPENHAYN, H. A. (1992): "Entry, Exit and Firm Dynamics in Long Run Equilibrium," *Econometrica*, 60(5), 1127–1160.
- JORGENSON, D. W., M. S. HO, AND K. J. STIROH (2008): "A Retrospective Look at the U.S. Productivity Growth Resurgence," *Journal of Economic Perspective*, 22, 1.
- JOVANOVIC, B. (1982): "Selection and the Evolution of Industry," *Econometrica*, 50(5), 649–670.
- PAKES, A. (2008): "Theory and Empirical Work on Imperfectly Competitive Markets," Discussion Paper 14117, NBER.
- SWEDISH COMPETITION AUTHORITY (2004:2): "Konsumenterna, Matpriserna och Konkurrensen (Consumers, Retail Food Prices, and Competition)," Technical Report 2, Stockholm.
- (2008:5): "Action for Better Competition (Åtgärder För Bättre Konkurrens)," Technical Report 5, Stockholm.
- THE NORDIC COMPETITION AUTHORITIES (2005:1): "Nordic Food Markets A Taste for Competition," Technical Report.
- VAN ARK, B., M. O'MAHONY, AND M. P. TIMMER (2008): "The Productivity Gap between Europe and the United States: Trends and Causes," *Journal of Economic Perspectives*, 22(1), 25–44.

Paper I

Productivity Dynamics and the Role of "Big-Box" Entrants in Retailing*

Florin Maican[†]and Matilda Orth[‡]

November 9, 2009

Abstract

Entry of large ("big-box") stores along with a drastic fall in the total number of stores is a striking trend in retail markets. We use a dynamic structural model to estimate total factor productivity in retail. Then we assess whether entry of large stores drives exit and growth in the productivity distribution of incumbents. Using detailed data on all retail food stores in Sweden, we find that local market characteristics, selection, and nonlinearities in the productivity process are important when estimating retail productivity. Large entrants force low productive stores to exit and surviving stores to increase their productivity growth. Growth increases most among incumbents in the bottom part of the productivity distribution, and then declines with the productivity level of incumbents. We use political preferences in local markets to control for endogeneity of large entrants. Our findings suggest that large entrants play a crucial role for driving productivity growth.

Keywords: Retail markets; imperfect competition; industry dynamics; TFP; dynamic structural model.

JEL Classification: O3, C24, L11.

[†]Department of Economics, University of Gothenburg, Box 640, SE 405 30, Göteborg, Sweden, Phone +46-31-786 4866, Fax: +46-31-786 4154, *E-mail*: florin.maican@handels.gu.se

 ‡ Department of Economics, University of Gothenburg, Box 640, SE 405 30, Göteborg, Sweden, Phone +46-31-786 5984, Fax: +46-31-786 4154, *E-mail*: matilda.orth@economics.gu.se

^{*}We thank Daniel Ackerberg, Victor Aguirregabiria, Mats Bergman, Jan De Loecker, Pierre Dubois, Martin Dufwenberg, Lennart Hjalmarsson, Jordi Jaumandreu, Vincent Réquillart, Rune Stenbacka, Johan Stennek, Måns Söderbom, and seminar participants at Toulouse School of Economics and the University of Gothenburg for valuable comments and discussions. In addition, we thank participants at EEA 2008 (Milano), EARIE 2007 (Valencia), the Nordic Workshop in Industrial Organization 2007 (Stockholm), the Conference of the Research Network on Innovation and Competition Policy 2007 (Mannheim), and the Swedish Workshop on Competition Research 2007 (Stockholm) for helpful comments and suggestions. Special thanks to the Trade Union Institute for Economic Research (FIEF) and the Swedish Retail Institute (HUI) for providing the data. Financial support from the Swedish Competition Authority is gratefully acknowledged.

1 Introduction

Recent methods for structural estimation of production functions have almost only been applied to manufacturing industries.¹ There have been few attempts to estimate multi-factor productivity in retail markets, where entry and exit have been found to play a more crucial role for labor productivity growth than in manufacturing (Foster et al. 2006). The major structural change in retail markets during the last few decades is in fact the entry of large ("big-box") stores, along with a drastic fall in the number of stores. The most striking example is the expansion of Wal-Mart, which has been found to greatly lower retail prices, and increase exit of retail stores in the U.S., the "Wal-Mart effect".² For instance, the number of single-store retailers in the U.S. declined 55% from 1963 to 2002 (Basker 2007). Retail markets in Europe also follow the "big-box" trend, though on a smaller scale, with for example Carrefour, Metro, Schwartz, and Tesco. Although there is an emerging literature about retail markets, the impact of this structural change on productivity has not been given much attention.³ Our goal is to combine recent extensions of the Olley and Pakes' (1996) framework to estimate total factor productivity (TFP) in retail markets, and to investigate the impact of increased competition from large entrants on exit and productivity growth of incumbents. That is, do large entrants drive reallocation of inputs and outputs, i.e., exit of low productive stores and growth of surviving stores with different positions in the productivity distribution? Detailed data on all retail food stores in Sweden give us unique opportunities to analyze the questions at hand.

Productivity analysis in retailing is more complex than in many other industries because of the problem of measuring output (Griffith and Harmgart 2005, Reynolds et al. 2005). We use a dynamic structural model to estimate productivity, which has the advantage of allowing stores to have heterogenous responses to industry shocks (Ackerberg et al. 2007). In detail, our model is based on the following key features of retail markets: First, the most common characteristics of retail

¹Olley and Pakes (1996), Pavcnik (2002), Levinsohn and Petrin (2003), Ackerberg et al. (2006), Buettner (2004), De Loecker (2009), Doraszelski and Jaumandreu (2009).

 $^{^2}Basker$ (2005), Basker (2007), Basker and Noel (2009), Holmes (2008), and Jia (2008). Fishman (2006) and Hicks (2007) provide a general discussion on the Wal-Mart effect.

 $^{^{3}}$ Two European contributions are Bertrand and Kramarz (2002), who find that retail markets in France have lower labor growth and higher concentration as a consequence of regulation, and Sadun (2008), who finds that regulation in the UK reduces employment in independent stores.

data are lumpy investments and a weak measure of intermediate inputs.⁴ Because labor and capital are key inputs in retail markets, we recover productivity from the optimal choice of labor (Doraszelski and Jaumandreu 2009). Second, because retail stores operate in local markets we control for local market characteristics, i.e. for large entrants and population density. We control for endogeneity of large entrants by using political preferences in local markets as instruments. Third, because large store types are more likely than smaller ones to survive larger shocks to productivity we control for selection, as do Olley and Pakes (1996). Fourth, recent literature emphasizes the importance of controlling for prices when estimating production functions in imperfect competitive markets (Foster et al. 2008, De Loecker 2009). Since store prices and quantities are rarely observed in retail data we control for unobserved prices by introducing a simple demand system as in Klette and Griliches (1996), and thus obtain mark-up estimates at the industry level.⁵ Compared to two-step estimators (Olley and Pakes 1996, Ackerberg et al. 2006), our one-step estimator has the advantages of increased efficiency and reduced computational burden. Identification comes from that we recover unobserved productivity from the labor demand function of known parametric form using a good measure of store wages. The assumption that labor is a static input abstracts from training, hiring and firing costs. We argue that this assumption is less restrictive in retail food than in many other industries because part time working is common, the share of skilled labor is low, and stores frequently adjust labor due to variation in customer flows. We also test the validity of this assumption.

The role of large entrants has a direct link to competition policy because the majority of OECD countries have entry regulations, though much more restrictive in Europe than in the U.S. The main rationale is that new entrants generate both positive and negative externalities, which require careful evaluation by local authorities. Advantages, such as productivity gains, lower prices, and wider product assortments, stand in contrast to drawbacks, in terms of fewer stores, and environmental issues. Because we anticipate large entrants to have an extensive impact

 $^{^4}$ While Olley and Pakes (1996) assume strict monotonicity of the investment function to recover unobserved productivity, Levinsohn and Petrin (2003) use the intermediate input of materials.

⁵Other studies that introduce prices in the production function are Melitz (2000), Levinsohn and Melitz (2002), Katayama et al. (2003), and Doraszelski and Jaumandreu (2009). In contrast to Doraszelski and Jaumandreu, who observe prices, we account for unobserved store prices.

on market structure, they are carefully evaluated in the planning process. The consequences of regulation (e.g. supermarket dominance) are frequently debated among policy makers in Europe (European Parliament 2008). Our primary objective is not to quantify the magnitude of inter-firm reallocations over time, i.e., how (large) entrants, exits, and incumbents contribute to aggregate productivity growth.⁶ Instead we provide evidence for how large entrants influence exit and the productivity growth of incumbents in local markets.

We focus on food retailing because it accounts for a large (15%) share of consumers' budgets (Statistics Sweden 2005) and thus constitutes a large share of retailing. Besides, many other service sectors follow similar trends as retail food. The Swedish market is appropriate to analyze because it follows two crucial trends common among nearly all OECD countries: There has been a structural change towards larger but fewer stores; in fact, the total number of stores in Sweden declined from 36,000 in the 1950s to below 6,000 in 2003 (Swedish National Board of Housing, Building, and Planning 2005). And there is an entry regulation that gives municipalities power to decide over the land use and, consequently, whether or not a store is allowed to enter the market.

Our study connects to the literature of dynamic models with heterogenous firms (Jovanovic 1982, Hopenhayn 1992, and Ericson and Pakes 1995). In particular, we build on the growing literature on productivity heterogeneity within industries that use dynamic structural models (Olley and Pakes 1996, Pavcnik 2002, Levinsohn and Petrin 2003, Ackerberg et al. 2006). Recent studies on productivity dynamics show two important facts: large and persistent productivity differences among firms, and substantial reallocation across firms in the same industry.⁷ They found that the key mechanism to foster growth is reallocation from less to more productive firms, either through less productive firms exiting and more productive firms entering or through increased productivity among incumbents, or both. Thus, increased competition forces low productive firms to exit, increasing the market shares of more productive firms. The productivity distribution is thus truncated from below, increasing the mean, and decreasing dispersion (Melitz 2003, Asplund and Nocke 2006). Using a local market approach, Syverson (2004)

 $^{^{6}}$ We estimate the contribution of *all* entrants to aggregate productivity growth using various TFP decompositions (Griliches and Regev 1995, Foster et al. 2001, Melitz and Polanec 2009) but, due to data constraints, we cannot quantify the exact contribution of large entrants.

⁷Caves (1998) and Bartelsman and Doms (2000) provide surveys, mainly on manufacturing.

emphasizes that demand density result in similar improvements in productivity distribution. In retail, entry and exit have been found to contribute to almost all labor productivity growth in the U.S., where chains replace low productive firms with high productive entrants (Foster et al. 2006). In Sweden, large food stores have been found to offer lower prices than others (Asplund and Friberg 2002). However, how large entrants influence local market competition and changes in the productivity distribution of incumbents has not been analyzed in detail.⁸

The empirical results show that it is important to control for local market characteristics, prices, selection, and to allow for nonlinearities in the productivity process when estimating retail productivity. Large entrants force low productive stores to exit, and surviving stores to increase their productivity growth. Growth increases most among incumbents in the bottom part of the productivity distribution, and then declines with the productivity level of incumbents. Large entrants thus spur reallocation of resources towards more productive stores. Aggregate productivity growth was 8% from 1997 to 2001, of which most is due to incumbents that increase their productivity, and exit of stores with lower productivity than incumbents. From a policy perspective, we claim that a more liberal design and application of entry regulations would support productivity growth in the Swedish retail food market.

The next section describes the retail food market and the data. Section 3 presents the modeling approach for estimating productivity, and those results. Section 4 reports the link between large entrants and exit and productivity growth. Section 5 summarizes and draws conclusions.

2 The retail food market and data

Retail food markets in the OECD countries are fairly similar, consisting of firms operating uniformly designed store types. In Sweden, the food market is dominated by four firms that together had 92% of the market shares in 2002: ICA(44%), Coop(22%), Axfood(23%), and Bergendahls(3%). Various independent owners

⁸The paper also relates to the vast literature on how competition affects productivity, emphasizing both positive and negative effects theoretically, but often positive effects empirically. Recent theoretical contributions are Nickell (1996), Schmidt (1997), Boone (2000), Melitz (2003), and Raith (2003); whereas recent empirical contributions include Porter (1990), MacDonald (1994), Nickell (1996), Blundell et al. (1999), Sivadasan (2004), and Aghion et al. (2009).

make up the remaining 8% market share.⁹ ICA consists mostly of independently owned stores with centralized decision making. Coop, on the other hand, consists of centralized cooperatives with decisions made at national or local level. Axfood and Bergendahls each have a mix of franchises and centrally owned stores, the latter mainly in the south and southwest of Sweden.¹⁰

A majority of OECD countries have entry regulations that give power to local authorities. The regulations differ substantially across countries, however (Hoj et al. 1995, Boylaud and Nicoletti 2001, Griffith and Harmgart 2005, Pilat 2005). While some countries strictly regulate large entrants, more flexible zoning laws exist, for instance, in the U.S. (Pilat 1997). The Swedish Plan and Building Act (PBA) gives power to the 290 municipalities to decide over applications for new entrants. In case of inter-municipality questions of entry, they are handled by the 21 county administrative boards. PBA is claimed to be one of the major barrier to entry, resulting in diverse outcomes, e.g., in price levels, across municipalities (Swedish Competition Authority 2001:4). Several reports stress the need to better analyze how regulation affects market outcomes (Pilat 1997, Swedish Competition Authority 2001:4, 2004:2). Large entrants are often newly built stores in external locations, making regulation highly important.¹¹ Appendix A describes PBA in greater detail.

■ Data. In order to cover various store productivity measures and define large entrants, we use two micro-data sets. The first data set, collected by Delfi Marknadsparter AB (DELFI), defines a unit of observation as a store based on its geographical location, i.e., its physical address (sources are described in Appendix A). This data, covering all retail food stores in the Swedish market during 1995-2002, include store type, chain, revenue class, and sales space (in square meters). The store type classification (12 different) depends on size, location, product assortment etc. An advantage with DELFI is that it contains all stores and their physical locations; shortcomings are a lack of input/output measures and the fact that revenue information is collected by surveys and reported in classes. There-

⁹International firms with hard discount formats entered the Swedish market after the study period: Netto in 2002, and Lidl in 2003.

 $^{^{10}\}mathrm{In}$ 2000, Axel Johnson and the D-group (D&D) merged, initiating more centralized decision making and more uniformly designed store concepts.

¹¹Possibly, firms can adopt similar strategies as their competitors and buy already established stores. As a result, more productive stores can enter without involvement of PBA and, consequently, the regulation will not work as an entry barrier that potentially affects productivity. Of course, we cannot fully rule out the opportunity that firms buy already established stores.

fore, we use DELFI only to define large entrants.

The most disaggregated level for which more accurate input and output measures exist is organization number (Statistics Sweden, SCB). SCB provides data at this level based on tax reporting. But due to anonymous codes, the two data sets cannot be linked. Financial Statistics (FS) provides input and output measures, and Regional Labor Statistics (RAMS) comprises data on wages for all organization numbers from 1996 to 2002 belonging to SNI code 52.1, "Retail sales in non-specialized stores", which covers the four dominant firms (ICA, Coop, Axfood, and Bergendahls).¹² This FS-RAMS data, at the organization number level, consist of "multi-store" units, which may be one store or more with the same organization number (e.g., due to having the same owner).¹³ Over 80% of the stores in DELFI each have their own organization number, so that less than 20%of the observations in FS-RAMS consist of two or more stores (discussed in detail below). If a firm consists of two stores, we observe total, not average, inputs and outputs. Note that all stores are reported in both data sets. Appendix A gives more information about the FS-RAMS data. Finally, we connect demographic information (population, population density, average income, and political preferences) from SCB to FS-RAMS and DELFI.

■ Local markets. Food products fulfill daily needs, are often of relatively short durability, and stores are thus located close to consumers. The travel distance when buying food is relatively short (except if prices are sufficiently low), and nearness to home and work are thus key aspects for consumers choosing where to shop, though distance likely increases with store size.¹⁴ The size of the local market for each store depends on its type. Large stores attract consumers from a wider area than do small stores, but the size of the local market also depends on the distance between stores. We assume that retail markets are isolated geographic units, with stores in one market competitively interacting only with other stores in the same local market. A complete definition of local markets requires information about the exact distance between stores. Without this information

¹²SNI (Swedish National Industry) classification codes build on the EU standard NACE.

 $^{^{13}{\}rm FS}\text{-}{\rm RAMS}$ does not rely on addresses like DELFI, so we could not do a more detailed investigation of TFP and geographical distance (location).

¹⁴The importance of these factors is confirmed by discussions with representatives from ICA, COOP, and Bergendahls. According to surveys made by the Swedish Institute for Transport and Communication Analysis, the average travel distance for trips with the *main* purpose of buying retail food products is 9.83 kilometers (1995-2002).

we must rely on already existing measures. The 21 counties in Sweden are clearly too large to be considered local markets for our purposes, while the 1,534 postal areas are probably too small, especially for large stores (on which we focus). Two intermediate choices are the 88 local labor markets or the 290 municipalities. Local labor markets take into account commuting patterns, which are important for the absolutely largest types such as hypermarkets and department stores, while municipalities seem more suitable for large supermarkets. As noted, municipalities are also the location of local government decisions regarding new entrants. We therefore use municipalities as local markets.

■ Large entrants and endogeneity. DELFI relies on geographical location (address) and classifies store types, making it appropriate for defining large entrants. Because of a limited number of large stores, we need to analyze several of the largest store types together. We define the five largest types (hypermarkets, department stores, large supermarkets, large grocery stores, and other¹⁵) as "large" and four other types (small supermarkets, small grocery stores, convenience stores, and mini markets) as "small".¹⁶ Gas station stores, seasonal stores, and stores under construction are excluded due to these types not belonging in the SNI-code 52.1 in FS-RAMS. From the point of view of the Swedish market, we believe that these types are representative of being large.

A key problem when analyzing the link between large entrants and productivity growth is the endogeneity of large entry. We hence need to bring exogenous variation in large entry using instruments. No major policy reforms changing the conditions for large entrants have taken place in Sweden during the study period (see Appendix A for details about PBA).¹⁷ Local authorities in Sweden decide however about entry of big-box stores. Following Bertrand and Kramarz (2002) and Sadun (2008) we use political preferences in municipalities as instruments for large entrants.¹⁸ We use variation in political preferences across local markets

¹⁵Stores classified as other stores are large and externally located.

¹⁶Alternatively, we define observations in FS-RAMS with sales above the 5th percentile of large stores' sales in DELFI as large; otherwise as small. Even though the available data do not allow a perfect match, the number of large entrants in FS-RAMS (so defined) follows a trend over time similar to that of the large entrants in DELFI. The empirical results (available from the authors upon request) are consistent with those reported here.

 $^{^{17}{\}rm Studies}$ based on UK data have used major policy reforms to handle endogeneity of entry (Sadun 2008, Aghion et al. 2009).

 $^{^{18}\}mathrm{Data}$ on the number of formal applications for entry, and rejections, is unfortunately not available in Sweden.

throughout the election periods 1994-1998 and 1999-2002 to add exogenous variation in the number of large entrants. We expect non-socialist local governments to have a more liberal view of large entrants, though the number of applications and rejections to each municipality is unfortunately not available in Sweden. Our instruments are valid if political preferences capture decision-making about large entrants and are uncorrelated with unobserved shocks.

■ Descriptive statistics. Table 1 presents descriptive statistics of the Swedish retail food industry from the two data sets DELFI and FS-RAMS for 1996-2002. As noted, over 80% of the observation units in FS-RAMS are identical to the stores in DELFI. The rest (20% in the beginning and 14% in the end) are multi-store units in FS-RAMS. The number of stores in DELFI decreases over the period from 4,664 to 3,585, i.e., a reduction of over 23%, indicating that many stores closed. In FS-RAMS, the number of observations decreases by about 17% (from 3,714 to 3,067).¹⁹ The share of large stores in DELFI increases from 19% to nearly 26%. While total sales space is virtually constant, mean sales space increases by 33%. Thus there has been a major structural change towards larger but fewer stores in the Swedish retail food market. Total wages (in FS-RAMS) increase over 22% (in real terms), while the number of employees increases only 9%.²⁰ Total sales increase about 26% (in FS-RAMS). Total sales in DELFI are lower and increase only 10% due to survey collection and interval reporting.

Table 2 shows median characteristics of local markets (municipalities) with and without large entrants during 1997-2002. The median number of stores varies between 22 and 54 in large entry markets, compared to 13-15 in non-entry markets. The number of markets with at least one large entrant varies between 6 and 23. Among these, up to 3 large entrants establish in the same market in the same year. As we expect, median entry and exit are higher in large entry than in non-entry markets, with median population, population density, and income also higher there. Large entry markets also have a lower concentration; the median four store concentration ratio is about 0.5 in these markets while it is over 0.7 in markets without large entrants.

¹⁹This indicates that entry and exit based on changes in organization numbers in FS-RAMS in some cases differ from entry and exit based on addresses in DELFI due to, e.g., re-organizations. ²⁰The aggregate growth of real wages in Sweden was 24% during the period.

3 Productivity estimation

Our model of competition among retail stores is based on Ericson and Pakes' (1995) dynamic oligopoly framework. A store is described by a vector of state variables consisting of productivity $\omega \in \Omega$, capital stock $k \in \mathbb{R}_+$, and local market characteristics $z \in Z$. Incumbent stores maximize the discounted expected value of future net cash flows. Stores compete in the product market and collect their payoffs. At the beginning of each time period, incumbents decide whether to exit or continue to operate in the local market.²¹ If the store exits, scrap value ϕ is received. If the store continues, it chooses optimal levels of labor l and investment $i \geq 0$. We assume capital is a dynamic input that accumulates according to $k_{t+1} = (1 - \delta)k_t + i_t$, where δ is the depreciation rate. Labor, on the other hand, is a non-dynamic input chosen based on current productivity. Changes in investment do not guarantee a more favorable state tomorrow, but do guarantee more favorable distributions over future states. As in Olley and Pakes (1996)(hereafter OP), the transition probabilities of productivity follow a first order Markov process with $P(d\omega|\omega)$. We denote $V(\omega_{it}, k_{jt}, z_{mt})$ to be the expected discounted value of all future net cash flows for store j in market m at period t. $V(\omega_{it}, k_{it}, z_{mt})$ is defined by the solution to the following Bellman equation with the discount factor $\beta < 1$

$$V(\omega_{jt}, k_{jt}, z_{mt}) = max \left\{ \phi, \sup_{i_{jt}} [\pi(\omega_{jt}, k_{jt}, z_{mt}) - c_i(i_{jt}, k_{jt}) - c_l(l_{jt}) + \beta E[V(\omega_{jt+1}, k_{jt+1}, z_{mt+1})|\omega_{jt}, i_{jt}] \right\}$$
(1)

where $\pi(\omega_{jt}, k_{jt}, z_{mt})$ is the profit function, increasing in both ω_{jt} and k_{jt} ; $c_i(i_{jt}, k_{jt})$ is investment cost in new capital, where $c_i(\cdot)$ is increasing in investment choice i_{jt} and decreasing in capital stock k_{jt} ; and $c_l(l_{jt})$ is the cost of labor, where $c_l(\cdot)$ is increasing in labor l_{jt} . Incumbent stores are assumed to know their scrap value ϕ prior to making exit and investment decisions. The solution of the store's optimization problem (1) gives optimal policy functions for labor $l_{jt} = \tilde{l}_{jt}(\omega_{jt}, k_{jt}, z_{mt})$, investment $i_{jt} = \tilde{i}_{jt}(\omega_{jt}, k_{jt}, z_{mt})$, and exit decision $\chi_{jt+1} = \tilde{\chi}_{jt}(\omega_{jt}, k_{jt}, z_{mt})$. The exit rule χ_{jt+1} depends on the threshold productivity $\underline{\omega}_{mt}(k_{jt}, z_{mt})$, where z_{mt} is a vector of local market characteristics such as the number of large entrants e_{mt}^L ,

 $^{^{21}}$ The decision to exit or continue is made at the store level, though firms that operate several stores can influence the decision of each store through possible chain effects. However, the firm takes the decision based on store performance.

and population density p_{mt}^{dens} .

■ Value added generating function. We assume Cobb-Douglas technology where stores sell a homogeneous product, and that the factors underlying profitability differences among stores are neutral efficiency differences.²² We follow the common notation of capital letters for levels and small letters for logs. The production function can be specified as

$$q_{jt} = \beta_0 + \beta_l l_{jt} + \beta_k k_{jt} + \omega_{jt} + u_{jt}^p \tag{2}$$

where q_{jt} is the log of quantity sold by store j at time t; l_{jt} is the log of labor input; and k_{jt} is the log of capital input. The unobserved ω_{jt} is productivity, and u_{jt}^p is either a measurement error (which can be serially correlated) or a shock to productivity that is not predictable during the period in which labor can be adjusted. Since physical output is complex to measure in retail markets and therefore not observed, we use deflated value added as a proxy for output.

Imperfect competition. Equation (2) assumes that prices are constant across stores.²³ Foster et al. (2008) analyze the relation between physical output, revenues, and firm-level prices in the context of market selection. They find that productivity based upon physical quantities is negatively correlated with establishment-level prices, but productivity based upon revenues is positively correlated with those prices. The retail food market is characterized by imperfect competition, and product differentiation is a key factor. When a store has some market power, its price influences its productivity. If a store cuts its price, then more inputs are needed to satisfy increasing demand. This negative correlation between inputs and prices leads to underestimation of the labor and capital parameters in the production function (Klette and Griliches 1996, Melitz 2000, De Loecker 2009).²⁴ Following this literature, we consider a standard horizontal product differentiation demand system

$$p_{jt} = p_{mt} + \frac{1}{\eta}q_{jt} - \frac{1}{\eta}q_{mt} - \frac{1}{\eta}\lambda_{jt} - \frac{1}{\eta}u_{jt}^d$$
(3)

 $^{^{22} \}rm We$ can easily apply another specification; for example, translog with neutral efficiency across stores would do equally well.

 $^{^{23} \}mathrm{Under}$ perfect competition, productivity of the price-taking stores is not influenced by store level prices.

 $^{^{24}{\}rm If}$ the products are perfect substitutes, then deflated sales are a perfect proxy for unobserved quality adjusted output.

where p_{jt} is output price, p_{mt} and q_{mt} are output price and quantity in local market m, λ_{jt} is demand shifters (observed and unobserved), and u_{jt}^d is a simple i.i.d. shock to demand. The parameter η (< -1 and finite) captures the elasticity of substitution among stores.²⁵ Due to data constraints the demand system is quite restrictive, implying a single elasticity of substitution for all stores, so that there are no differences in cross price elasticities, i.e., we have a constant markup over marginal cost $(\frac{\eta}{1+\eta})$, and the Learner index is $(\frac{1}{|\eta|})$. We can however allow the elasticity of substitution to differ across local market groups such as counties (21 in total). The Learner index for county g is then $\frac{1}{|\eta_{n}|}$.

We decompose demand shifters λ_{jt} into observed local market characteristics z_{mt} , i.e., number of large entrants e_{mt}^L , population density p_{mt}^{dens} , and unobserved demand shocks v_{jt} as

$$\lambda_{jt} = z_{mt}^{'}\beta_z + v_{jt}$$

where v_{jt} are either correlated unexpected shocks to demand or i.i.d. The unobserved demand shocks v_{jt} are unobserved by the econometrician but known to or predictable by the stores when they make their input, price or exit decisions.

Since we have unobserved store prices and quantities, we use the deflated value added y_{jt} , defined as $q_{jt} + p_{jt} - p_{mt}$, as output in the estimation of the sales (value added) generating function. However, if p_{mt} is unobserved, the consumer price index for food products p_{It} can be used as a proxy. Controlling for unobserved store price p_{jt} in the value added generating function in (2), we then have

$$y_{jt} \equiv \left(1 + \frac{1}{\eta}\right) \left[\beta_0 + \beta_l l_{jt} + \beta_k k_{jt}\right] - \frac{1}{\eta} q_{mt} - \frac{1}{\eta} z'_{mt} \beta_z + \left(1 + \frac{1}{\eta}\right) \omega_{jt} - \frac{1}{\eta} \upsilon_{jt} - \frac{1}{\eta} u^d_{jt} + \left(1 + \frac{1}{\eta}\right) u^p_{jt}$$
(4)

Assuming that store productivity follows an exogenous first order Markov process, actual productivity can be written as the sum of expected productivity given the store information set \mathcal{F}_{t-1} , $E[\omega_{jt}|\mathcal{F}_{t-1}]$, and the i.i.d. productivity shock ξ_{jt}

$$\omega_{jt} = E[\omega_{jt}|\mathcal{F}_{t-1}] + \xi_{jt}.$$
(5)

The conditional expectation function $E[\omega_{jt}|\mathcal{F}_{t-1}]$ is unobserved by the econometrician (though known to the store). The shock ξ_{jt} may be thought of as the

 $^{^{25}}$ The vertical dimension is to some extent also captured since deflated output measures both quantity and quality, which is correlated with store type (size).

realization of uncertainties that are naturally linked to productivity. Therefore, the value added generating function becomes

$$y_{jt} = \begin{pmatrix} 1+\frac{1}{\eta} \end{pmatrix} [\beta_0 + \beta_l l_{jt} + \beta_k k_{jt}] - \frac{1}{\eta} q_{mt} - \frac{1}{\eta} z'_{mt} \beta_z + \left(1+\frac{1}{\eta}\right) E[\omega_{jt-1}|\mathcal{F}_{t-1}] + \\ \left(1+\frac{1}{\eta}\right) \xi_{jt} - \frac{1}{\eta} v_{jt} - \frac{1}{\eta} u^d_{jt} + \left(1+\frac{1}{\eta}\right) u^p_{jt}$$
(6)

We face a trade-off between a flexible approximation of the ω_{jt} process and separation of demand shocks from productivity.²⁶ The estimation strategy chosen depends on whether demand shocks v_{jt} are thought to be correlated over time and on whether we use a linear or nonlinear approximation of the conditional expectation $E[\cdot]$ (Ackerberg et al. 2007). We first present Case (1) when v_{jt} is correlated over time, which includes ω_{jt} and v_{jt} following either a general Markov process or an AR(1). The Markov processes can be either dependent or independent. Under AR(1), ω_{jt} and v_{jt} can follow either the same or different processes and no further assumptions are needed to estimate the parameters. Then we present Case (2) when v_{jt} is i.i.d.

Case (1): v_{jt} are correlated over time

First, if ω_{jt} and v_{jt} follow dependent Markov processes then v_{jt-1} will enter as a separate variable in the conditional expectation $E[\omega_{jt}|\omega_{jt-1}, v_{jt-1}]$. To solve the identification problem in (6) we need an estimate of v_{jt-1} . The Berry et al. (1995) (BLP) literature produces estimates of a set of "unobserved product characteristics" that might be used as v_{jt} (Ackerberg et al. 2007 discuss this in detail), which we might interpret as unobserved store quality. But in our case, it is impossible to back out v_{jt} using this method because it requires more firm specific data such as prices and advertisement.

Second, if ω_{jt} and v_{jt} follow independent Markov processes then expected productivity at time t conditional on information set \mathcal{F}_{t-1} does not depend on v_{jt-1} . But in this case v_{jt} is an important determinant of optimal labor or investment, and thus affects actual productivity. Since we have two unobservables (ω_{jt} and v_{jt}) and no other control variable for v_{jt} , identification in (6) requires one of the following assumptions:

²⁶The alternative of not controlling for prices at all requires even stronger assumptions.

(a) $\tilde{\omega}_{jt} \equiv (1 + \frac{1}{\eta})(\omega_{jt} - \frac{1}{\eta}v_{jt})$, i.e., quality adjusted productivity, follows a first order nonlinear Markov process: $\tilde{\omega}_{jt} = E[\tilde{\omega}_{jt}|\mathcal{F}_{t-1}] + \xi_{jt} = \tilde{h}(\tilde{\omega}_{jt-1}) + \xi_{jt}$, where \tilde{h} is an approximation of the conditional expectation (Melitz 2000, Levinsohn and Melitz 2002). In other words, a positive shock in either productivity or demand makes the store sell more but the exact source of the shock does not matter.

(b) ω_{jt} and v_{jt} follow different AR(1) processes.²⁷ We assume that $\omega_{jt} = \rho_1 \omega_{jt-1} + \xi_{jt}$ and $v_{jt} = \rho_2 v_{jt-1} + \mu_{jt}$. One way to eliminate the unobserved demand shock from the value added generating function (6) is to take the first difference $\tilde{y}_{jt} = y_{jt} - \rho_1 y_{jt-1}$. If $\rho_1 = \rho_2$, this is sufficient for identification. If $\rho_1 \neq \rho_2$, the unobserved demand shock v_{jt} is completely removed if we apply the difference $\tilde{y}_{jt} - \rho_2 \tilde{y}_{jt-1}$ in (6). Note that $\tilde{y}_{jt} - \rho_2 \tilde{y}_{jt-1}$ is stationary if $\rho_1 > \rho_2$, i.e., if productivity is more persistent than the demand shock (the roots of $\tilde{y}_{jt} - \rho_2 \tilde{y}_{jt-1}$ are $\rho_2 - \rho_1$ and $-\rho_2$).

The advantage of (a) is that it allows for nonlinearities in the productivity process and the possibility of controlling for selection (see Case (2)). The drawbacks of (a) are that we observe quality-adjusted productivity and that we need more assumptions to back out productivity. The advantage of (b) is that we can sort out persistent demand shocks from productivity and that no more assumptions are needed for identification. A drawback of allowing for two AR(1) processes in (b) is that it is more data demanding, because we need two lags and thus dropping two years of data, to make sure that we have removed the persistent unobserved demand shocks. Since a store needs to be present in the data for at least three years, this severely restricts the dynamics.

Case (2) v_{jt} are *i.i.d.*

In this case, demand shocks are not correlated with inputs or with exit decisions. Therefore v_{jt} collapses into the i.i.d. demand shocks from the price equation u_{jt}^d . Below we describe the estimation strategy when productivity follows a general Markov process.

■ Inverse labor demand function. A central feature of retail data is lumpy

 $^{^{27}\}mathrm{See}$ the dynamic panel model of Blundell and Bond (2000).

investment and a weak measure of intermediate inputs. We recover productivity from the optimal choice of labor using a good measure of store specific wages (Doraszelski and Jaumandreu 2009).²⁸ The idea relies on Levinsohn and Petrin (2003) who recover unobserved productivity from the demand for static intermediate input of materials. We assume that labor is a static and variable input chosen based on current productivity. The functional form of the value added generating function provides a parametric form of the labor demand function, unlike Levinsohn and Petrin (2003) and Ackerberg et al. (2006) that are non-parametric in materials. The advantage is that we can include many stores with zero investment while not making any assumptions about the stores' dynamic programming problem. In abstract of store level wages it may however be hard to estimate the coefficients of static inputs in the Cobb-Douglas case (Bond and Söderbom 2005).

Our assumption that labor is a static and variable input abstracts from costs of training, hiring and firing employees, though for several reasons this is less restrictive in retail than in many other industries. Part time workers are common. As much as 40% of the employees in retail food work part time, compared to 20% for the Swedish economy as a whole (Statistics Sweden). The share of skilled labor is low. Only 15% of the retail employees had a university education in 2002, compared to 32% for the total Swedish labor force (Statistics Sweden). Stores have long opening hours and adjust their labor due to variations in customer flows over the day, week, month and year. Moreover, the training process might be shorter than in many other industries. We use the number of full-time adjusted employees as our measure of labor.

Our assumption that each store chooses labor based on its productivity implies that labor l_{jt} is correlated with the random productivity shock ξ_{jt} . In year t, stores chose current labor l_{jt} based on current productivity ω_{jt} , which gives labor demand as

$$l_{jt} = \frac{1}{1 - \beta_l} [\beta_0 + ln(\beta_l) + \alpha + \beta_k k_j + \omega_{jt} - (s_{jt} - p_{jt})]$$
(7)

 $^{^{28}}$ The average wage contains both price of labor and its composition, e.g., ages, gender, and skill groups. Our measure of wage is a good reflection of exogenous changes in the price of labor because the 22% growth in total retail wages during the period (Table 1) is in line with the 24% growth in aggregate real wages in Sweden (Statistics Sweden).

where $\alpha = ln E[e^{\xi_{jt}}]$ and s_{jt} is the log of wage rate paid by store j in period t. Under the functional form assumption on the value added generating function, we have a known functional form for the labor demand and inverse labor demand functions. Solving for ω_{jt} in Equation (7) yields the inverse labor demand function from which we can recover unobserved productivity

$$\omega_{jt} = \frac{\eta}{1+\eta} \left[\delta_1 + \left[(1-\beta_l) - \frac{1}{\eta} \beta_l \right] l_{jt} + s_{jt} - p_{It} - \left(1 + \frac{1}{\eta} \right) \beta_k k_{jt} + \frac{1}{\eta} q_{mt} + \frac{1}{\eta} z'_{mt} \beta_z \right]$$
(8)

where $\delta_1 = -ln(\beta_l) - ln(1 + \frac{1}{\eta}) - \beta_0(1 + \frac{1}{\eta}) - \ln E[e^{u_{jt}^p}] + \frac{1}{\eta} \ln E[e^{u_{jt}^d}] + \frac{1}{\eta} \ln E[e^{v_{jt}}]$. We then substitute the inverse demand function (8) in the value added generating function (6).²⁹

It is important to stress again that we can estimate the value added generating function coefficients (6) because we have assumed that labor is a static variable. Comparing with non-parametric approaches, our estimator is more transparent in how real wages and unobserved demand shocks affect labor demand. Ackerberg et al. (2006) (ACF) is an alternative estimator for which we show results in the empirical part. We use OLS and ACF estimators as benchmarks, i.e., without controlling for unobserved prices and local market characteristics. In ACF, labor has dynamic implications and labor demand is assumed to be a non-parametric function. It possible to control for unobserved prices and local market conditions a similar way in ACF. Dorazelski and Jamandreu (2009) discuss the relative merits of the parametric and non-parametric approaches.

Selection. Selection can be essential in retail markets because large stores are more likely to survive larger shocks to productivity than are small stores. Stores' decisions to exit in period t depend directly on ω_{jt} , and therefore the decision is correlated with the productivity shock ξ_{jt} . The threshold productivity takes local market characteristics such as large entrants and population density into account (Appendix B gives a detailed description of selection). To estimate β_l and β_k while controlling for selection, we use predicted survival probabilities \mathcal{P}_{t-1} . Substituting

 $^{^{29}}$ The condition for identification is that the variables in the parametric part of the model are not perfectly predictable (in the least square sense) by the variables in the non-parametric part (Robinson 1988). Including additional variables that affect productivity guarantees identification, i.e., there cannot be a functional relationship between the variables in the parametric and non-parametric parts (Newey et al. 1999). For example, z_{mt} cannot be perfectly predicted from ω_{jt} .

the survival probabilities and the inverse labor demand function (8) into (6) yields the final value added generating function that we estimate:

$$y_{jt} = \begin{pmatrix} 1+\frac{1}{\eta} \end{pmatrix} [\beta_0 + \beta_l l_{jt} + \beta_k k_{jt}] - \frac{1}{\eta} q_{mt} - \frac{1}{\eta} z'_{mt} \beta_z + \left(1+\frac{1}{\eta}\right) h(\mathscr{P}_{t-1}, \omega_{jt-1}) + \\ \left(1+\frac{1}{\eta}\right) \xi_{jt} - \frac{1}{\eta} v_{jt} - \frac{1}{\eta} u_{jt}^d + \left(1+\frac{1}{\eta}\right) u_{jt}^p.$$
(9)

Estimation strategy. The estimation of our extended Olley and Pakes model adjusted for retailers (EOP) consists of two parts. First, we use a probit model with a third order polynomial to estimate survival probabilities, which are then substituted into (9). Then, we estimate (9) using the sieve minimum distance procedure proposed by Ai and Chen (2003) and Newey and Powell (2003) for i.i.d. data (see Ackerberg et al. (2009) for a discussion of semiparametric inference to IO models). The goal is to obtain an estimable expression for the unknown parameters $\boldsymbol{\beta}$ and h_H , where H indicates all parameters in $h(\cdot)$. We approximate $h(\cdot)$ by a third order polynomial expansion in ω_{jt-1} , given by (8).³⁰ We use a tensor product polynomial series of labor (l_{it-1}) , capital (k_{it-1}) , total wages (s_{it-1}) , the consumer price index for food products (p_{It}) , and local market conditions (z_{mt-1}) including large entrants (e_{mt-1}^L) and population density (p_{mt-1}^{dens}) , plus local political preferences (pol_{mt}) as instruments. This set of instruments is also used to estimate the optimal weighting matrix. Using GMM, the parameters (β, h_H) are then jointly estimated. Since there are non-linearities in the coefficients, we use the Nelder-Mead numerical optimization method to minimize the GMM objective function³¹

$$\min_{\boldsymbol{\beta},h_H} Q_N = \left[\frac{1}{N} W' \psi(\cdot;\boldsymbol{\beta},h_H)\right]' A \left[\frac{1}{N} W' \psi(\cdot;\boldsymbol{\beta},h_H)\right]$$
(10)

where A is the weighting matrix defined as $A = \left[\frac{1}{N}W'\psi\psi'W\right]^{-1}$, W is the matrix of instruments, and $\psi_{jt}(\cdot;\beta,h_H) = \left[\left(1+\frac{1}{\eta}\right)\xi_{jt}-\frac{1}{\eta}v_{jt}-\frac{1}{\eta}u_{jt}^d+\left(1+\frac{1}{\eta}\right)u_{jt}^p\right]$. Estimation is done at the industry level, controlling for local conditions. Estimation results at county level (21 municipality groups) are available from authors. An

 $^{^{30}\}mathrm{As}$ a robustness check, we also expand $h(\cdot)$ using a fourth order polynomial, and the results are similar.

 $^{^{31}}$ This simplex method converges quickly and is more robust to the starting values than quasi-Newton methods such as BFGS. Our EOP estimation procedure is written in R (<u>www.r-project.org</u>). The procedure is more computationally demanding when controlling for selection. However, estimation at the county level reduces computing time substantially.

advantage of estimating at county level is that we obtain the mark-ups at the county level.³² The major disadvantage is that we loose efficiency in estimation in the small counties. Another advantage of using counties is that they are responsible for inter-municipality implementation of entry regulation. However, we control for municipality characteristics in the estimation. Appendix B presents a detailed description of the estimation procedure.

■ Results: store TFP. We estimated coefficients of the value added generating function using OLS, the ACF two-stage estimator, and five specifications of our model. These five are: DP1 - productivity and persistent demand shocks follow the same AR(1) process, i.e., an updated version of the Blundell and Bond (2000) estimator; DP2 - productivity and persistent demand shocks follow different AR(1) processes; EOPs - productivity follows a nonlinear Markov process, and we control for selection, but not for prices or local market characteristics; EOPm productivity follows a nonlinear Markov process, and we control for prices and local market characteristics but not for selection; and EOPms - productivity follows a nonlinear Markov process and we control for prices, selection and local market characteristics.³³ We include number of large entrants and population density, as local market covariates in the demand equation. We control for endogeneity of large entrants by using political preferences in local markets as instruments.³⁴

³²Another reason for estimating at county level is that our method requires more observations than is available at municipality level.

³³To make our DP1 and DP2 estimators comparable with ACF and the dynamic panel estimator (Blundell and Bond 2000), we assume that productivity ω_{jt} is an AR(1) process, i.e., $\omega_{jt} = \rho_1 \omega_{jt-1} + \xi_{jt}$. We use k_{jt} and l_{jt-1} as instruments, i.e., they are assumed to be uncorrelated with the shocks ξ_{jt} and υ_{jt} . However, we need an additional moment in DP to identify ρ_1 and therefore assume that the shock ξ_{jt} is uncorrelated with $(\omega_{jt-1} + u_{jt-1}^d)$. In ACF, we use k_{jt} and l_{jt-1} as second stage instruments, i.e., labor is chosen with full knowledge of ω_{t-1} . Ackerberg et al. (2006) provide a detailed comparison between OP-type estimators and dynamic panel estimators.

 $^{^{34}}$ As noted earlier, we base on the political preferences in each municipality because no major policy reforms took place in Sweden during the study period, and we do not have access to the number of applications and rejections in the planning process. The Social Democrats are the largest party nationally with 40.6% of seats on average, collaborating with the Left Party (8%) and the Green Party (4.2%). The non-socialist group consists of the Moderate Party (18%), most often together with the Center Party (13.2%), Christian Democrats (5.9%), and the Liberals (5.6%). 22% of the municipalities had a non-socialist majority during 1996-1998, increasing to 32% during 1999-2002. The non-socialists had 8.6%-85%, averaging 40.7% (1996-1998) and 44.1% (1999-2002). The correlation between the non-socialist share of seats and the number of large entrants is 0.005, or 0.086 if we exclude the Center Party, which is typically strong in the countryside where there is less likely to be large entrants. As we expect, we observe more large entrants in municipalities with non-socialist local government. We use a dummy for non-socialist majority in the estimations.

EOPs, EOPm, and EOPms require estimation of one non-parametric function, in contrast to ACF, which requires two. A major advantage of DP1, DP2, EOPm and EOPms is that they control for unobserved prices which otherwise might create a downward bias in the scale estimator (Klette and Griliches 1996). Another advantage is that the correction for omitted prices also yields an estimate of market output, which makes it possible to compute the elasticity of substitution η and an average industry mark-up.

Table 3 has two columns for each of the DP and EOP specifications. Column (1) shows the coefficients including elasticities, and Column (2) the larger true estimated coefficients, without elasticity. Since all specifications use deflated value added, we use Column (1) to compare OLS and ACF with DP and EOP.

The elasticity of scale estimate in the DP and EOP regressions is greater than in OLS (1.115) and ACF (0.931), it varies between 1.140 (EOPs) and 1.426 (DP2). The minimum point estimate of labor is 0.686 (DP2) and the maximum is 0.948(OLS). The minimum point estimate of capital is 0.116 (EOPm) and the maximum is 0.426 (DP2). Controlling for local market characteristics is important: Including the number of large entrants and population density in the price equation change the demand elasticity and capital estimates substantially, making both smaller. When we allow productivity to follow an AR(1) process (DP1, DP2), estimates of capital are over 3 times larger than in EOP. The estimated productivity transition (ρ_1) is about 0.4 in both DP1 and DP2, i.e., a rather low persistency in over time. Furthermore, the estimated demand elasticity in DP1 (-5.674) seems unreasonably high in absolute value for retail food (Hall 1988). To test the assumption of linearity in productivity, we regress current productivity, recovered from DP1 and DP2, on a third order polynomial extension of previous productivity. The coefficients of ω_{it-1}^2 and ω_{it-1}^3 are statistically different from zero, indicating that productivity does not follow an AR(1) process. This might be one of the reasons for the large values of capital (over 0.4) in the DP specifications. We therefore recognize that it is important to allow for a nonlinear Markov process in productivity.

In ACF, EOPs, EOPm, and EOPms productivity follows a nonlinear Markov process. As noted, comparing with DP, the capital coefficients are smaller and the labor coefficients larger. As theory suggests the coefficients on both capital and labor decrease when controlling for prices in EOPm and EOPms, comparing with

OLS.35

EOPs and EOPms (as well as ACF) control for selection. Theory and empirical investigations then predict a lower labor and higher capital coefficients (Ackerberg et al. 2007).³⁶ The capital coefficient in EOPms (0.145) is in fact larger than in EOPm (0.116), but smaller than in ACF (0.163). Controlling for selection in EOPms yields a smaller labor coefficient 0.840 than in EOPs (0.945). Those results are in line with the OP literature.

The coefficients on large entrants and population density are negative and statistically significant in all specifications. The lowest demand elasticity (-2.96) is in EOPms, i.e., when we allow productivity to follow a nonlinear process and control for selection. Thus, the implicit assumption $\eta = -\infty$, often used in empirical studies, does not hold. In EOPms the the mark-up, defined as price over marginal cost, is 1.509. Our mark-up is consistent with previous findings based on retail data (see, e.g., Hall 1988).

We test the validity of our assumption that labor is static. If the inverse labor demand function is misspecified, the labor coefficient in the value added generating function differs from the one in the inverse labor demand function. We estimate the restricted and unrestricted models. Then we compute the GMM distance statistic, $D_N = N * [Q_N(\beta_{\text{restricted}}) - Q_N(\beta_{\text{unrestricted}})]$, to test the null of equal labor coefficients. Note that we could estimate only the unrestricted model and test the equality of the labor coefficients directly by Wald test. The two statistics are however asymptotically equivalent under the null hypothesis (Newey and West 1987). The results indicate that the null of equal coefficients is accepted for EOPms, i.e., our assumption of static labor is valid. For EOPm and EOPs, the unrestricted models are rejected by the Sargan test of overidentified restrictions. Although the labor coefficients are weakly identified, their values are very similar (we need additional moments for labor).

Summarizing, it seems important to allow for nonlinearities in the productivity process and to control for prices, local market characteristics and selection when

³⁵If we do not control for unobserved demand shocks, we expect the coefficients of labor and capital to be upper biased. The reason is the positive correlation between inputs and demand shocks. In case that demand shocks are still present the coefficients would thus decrease even more. Nevertheless, our results when controlling for prices show that both coefficients decrease.

 $^{^{36}}$ Since stores with large capital stock can survive even if they have low productivity, we expect selection to induce a negative correlation between capital and the disturbance term in the selected sample.

estimating productivity in retail markets. It is thus central to deal with omitted price bias, unobserved demand characteristics and selection.

4 Large entrants and productivity

Next we proceed to investigate whether large entrants influence exit and productivity growth of surviving stores. Our goal is to evaluate whether large entrants have a greater impact on one part of the productivity distribution than another. To do this, we use TFP estimated by EOPms that allows for a general Markov process in productivity and selection, and DP2 that guarantees to clear out persistent demand shocks from productivity. Based on theories using dynamic models with heterogenous firms, our hypothesis for how increased competition from large entrants influences reallocation of resources is: Exit of low productive stores and higher productivity growth among surviving stores (Hopenhayn 1992, Melitz 2003, Syverson 2004, Asplund and Nocke 2006). We consider the role of large entrants for productivity levels, transitions in the productivity distribution in local markets, exit and productivity growth.³⁷ Finally, we decompose aggregate productivity growth of all entrants, exits and incumbents (due to data constraints we cannot measure the contribution of large entrants to aggregate productivity growth).

■ **Productivity levels.** Figure 1 shows kernel density estimates of TFP (estimated by EOPms) in markets the year of, and the year after, large entry. Though the differences are small, both the upper- and lower tails of the distribution are greater after large entry. However, productivity is to some extent lower in the middle of the distribution following large entry. Mean TFP of incumbents is higher the year after entry (-0.016) than the year of entry (-0.291) and the standard deviation is smaller (Table 4, panel A). Using t-test, the null of equal means is rejected at the 1% significance level. Using F-test, the null of equal standard deviations cannot be rejected.³⁸

Transitions in the productivity distribution. To explore changes in pro-

 $^{^{37}}$ We primarily focus on changes after large entry because several permanent reasons might explain differences between markets with and without large entrants.

 $^{^{38}\}text{Defining entry markets as municipalities with at least one large entrant, mean TFP is smaller in markets with entrants (-0.016) than in markets without (0.100) and the standard deviation is larger (Table 4, panel B). The null of equal means and equal standard deviations are both rejected and significant at the 1% level.$

ductivity distributions in local markets we classify incumbents into six percentile bins (p10, p10-25, p25-50, p50-75, p75-90, p90) each year, based on their productivity. Then we follow movements between percentile bins or exit over time.

In markets with large entrants, more incumbents stay in the same percentile from one year to another, i.e., the diagonal shares are larger (Table 5). Large entrants thus make the percentile movements more persistent. The shares that stay in the same percentile are 33-49% in entry markets, but 32-42% in non-entry markets. The most pronounced difference occurs in the upper tail. That is, high productive incumbents stay high productive in entry markets. Almost 50% stay in p90, comparing with only 35% in markets without large entry. In the bottom part of the distribution, incumbents in entry markets either stay in their productivity percentile or exit. In contrast, bottom part incumbents in markets without large entry decrease their productivity without being forced to exit. The total share of stores that exit is higher in entry markets (17.3%) than non-entry markets (15.5%). Regardless of large entry, more stores increase their productivity in the bottom part of the distribution, while more stores decrease their productivity in the top, except p90. Finally, entry markets have less movements between extreme percentiles. For example, only about 4% move from p10 to an above median percentile in markets with large entry and over 6% in markets without. We discuss productivity growth in detail in Section 4.2, but first we analyze exit.

4.1 Exit

Over 50% of the exits come from the two lowest percentiles (p10, p10-25) in markets with large entrants, but less than 42% in markets without large entrants (Table 5, panel A). Large entrants thus result in more exit among low productive stores. In markets without large entrants, more stores have lower productivity and yet continue to operate. While exit mainly occurs from the bottom part of the distribution, entrants are found across the whole distribution (not reported) as in previous findings in retail markets (Foster et al. 2006).

According to our model, the survival probabilities imply that the decision to exit depends on productivity, capital stock and local market characteristics, i.e., large entrants and population density (see Section 3). Stores decide whether to exit or continue in the beginning of each period based on information regarding market conditions and we thus use large entrants in the previous year. Based on the stopping rule we show probit regressions of exit

$$Pr(exit_{jt} = 1 | \omega_{jt}, e_{mt-1}^{L}, k_{jt}, p_{mt-1}^{dens}) = \phi(\gamma_0 + \gamma_e e_{mt-1}^{L} + D_{jt-1} * e_{mt-1}^{L} \Gamma + \gamma_k k_{jt} + \gamma_p p_{mt-1}^{dens} + \alpha_t)$$
(11)

where $exit_{jt}$ is equal to one if a store exit and otherwise zero; k_{jt} is log of capital; e_{mt-1}^{L} is the number of large entrants; $D_{jt-1} * e_{mt-1}^{L}$ are interaction terms between productivity percentile dummies and the number of large entrants; p_{mt-1}^{dens} is log of population density; ϕ is the cumulative distribution function of the standard normal; and α_t is a vector of year dummies; p_{mt-1}^{dens} and e_{mt-1}^{L} constitute z_{mt-1} in our model in Section 3.

Table 6 shows regression results for the probability of exit. The first specification (columns 1 and 3) relies on the pure stopping rule and includes productivity, capital, large entrants and population density. In line with both theory and previous empirical studies (Olley and Pakes 1996, Pavcnik 2002), exit is less likely if productivity and capital stock are high for both the nonlinear (column 1) and linear (column 3) productivity process. That is, stores with lower productivity and capital stock are more likely to exit. Moreover, exit is more common from markets if population density is high whereas the coefficient on large entrants is positive but not significant at conventional significance levels.

The expanded specification (columns 2 and 4) includes interaction terms of large entrants with the six productivity dummies, using the middle group (p50-75) as reference. In the nonlinear estimation (column 2), the coefficient on large entrants is now negative and statistically significant at the 10 percent level. The coefficients on the interaction terms are all positive and jointly significant with the coefficient of large entry for p10 and p25-50. Exit is 0.29 percentage points more likely after large entry for stores is in the bottom part of the productivity distribution (p10 or p25-50) than for those in the middle. For the linear productivity process (column 4), the interaction terms are not significant, most likely because of lack of data (a store needs to be at least three years in the data).

To summarize, we find evidence that exit occurs from the bottom part of the productivity distribution after large entry which truncates the distribution from below in line with our hypothesis (Hopenhayn 1992, Olley and Pakes 1996, Melitz 2003, Syverson 2004, Asplund and Nocke 2006).
4.2 Productivity growth of incumbents

Productivity growth is given by the difference between log of productivity in time t and productivity in t-1: $\omega_{jt} - \omega_{jt-1}$. We only consider productivity growth of incumbents, and thus exclude stores that enter or exit. Figure 2 shows that incumbents' TFP growth (estimated by EOPms) is higher in markets with large entrants than in market without. While the largest difference occurs in the bottom part of the distribution, the top parts are similar. Mean productivity growth of incumbents is larger in markets with large entrants (15.9%) than in markets without (13.8%) and the standard deviation is smaller (Table 7, panel B). The t-test cannot reject the null of equal mean values but the F-test can reject the null of equal standard deviations at the 5% significance level.

In markets with large entrants, Figure 3 shows a striking improvement in incumbents' productivity growth between the year of, and the year after, entry: TFP growth is higher in all parts of the distribution after entry. Mean productivity growth of incumbents is -11% the year of large entry, whereas it is 15.9% the year after (Table 7, panel A). The t-test of equal mean values is rejected at the 1% significance level. The standard deviation is larger after entry, 0.56 compared to 0.53. Using F-test, the null of equal standard deviations is rejected at the 5% significance level.

Although Figure 2 and 3 indicate that large entrants might have an impact on the distribution of incumbents' productivity growth, we need to isolate the role of large entrants from store and market characteristics. Therefore, we regress the number of large entrants on incumbents' productivity growth the year after large entry,

$$\theta_{jt} = \alpha_0 + \alpha_e e_{mt-1}^L + D_{jt-1} * e_{mt-1}^L \boldsymbol{\alpha} + \alpha_p p_{mt-1}^{dens} + \alpha_m + \alpha_t + \varepsilon_{jt}$$
(12)

where $\theta_{jt} = \omega_{jt} - \omega_{jt-1}$ is incumbents' productivity growth between periods *t-1* and t; e_{mt-1}^L is the number of large entrants; $D_{jt-1} * e_{mt-1}^L$ are six interaction terms between percentile productivity dummies and large entrants; p_{mt-1}^{dens} is population density; α_t and α_m are vectors of time and market dummies; and ε_{jt} is an i.i.d. error term.

To isolate the impact of large entrants, we control for unobserved local market heterogeneity by using fixed effects for local markets and years. To control for endogeneity because large entry depends on the productivity of incumbents, we use different specifications of the one-step GMM estimator. Table 8 shows the regression results. GMM specification (1) uses lagged political preferences, lagged population density, and lagged income as instruments for large entrants; GMM specification (2) uses lagged large entrants (e_{mt-2}^L) plus lagged population density and lagged income as instruments. It is important to note that adding income as demand shifter does not change our results. Since we get consistent results with all estimators, we primarily discuss the results of GMM (1) with TFP estimated by EOPms.

The coefficient on large entrants is positive and significant at the 1% level when we estimate with large entrants and productivity dummies but no interaction terms (Table 8). On average, large entrants thus increase productivity growth among incumbents. More importantly, large entrants have a greater impact on some parts on the incumbents' productivity distribution than others. The coefficient on large entry is then negative, whereas those on the interaction terms are all positive, and all significant at the 1% level. The coefficients of large entrants and the interaction terms are jointly significant. As a result of large entry, low productive incumbents increase their productivity growth, by 14% for those in p10 instead of in the middle group (p50-75), by 5% (for p10-25) and by 4% (for p25-50). On the other hand, large entry reduces productivity growth of incumbents in the upper distribution percentiles by -3% (for p75-90) and -7% (for p90), relative to the middle group. The growth increase is thus largest for surviving stores with low productivity, and then declines with survivors' productivity.

In a previous version of the paper we investigate how large entrants affect the distribution of local market productivity, without controlling for large entry and unobserved demand shocks when estimated productivity. We found that productivity dispersion increases as a result of large entrants; the most productive incumbents become more productive, and the least productive become less productive (results are available from the authors). One explanation why the least productive stores become less productive is that a demand shock hit them after large entry, but they still find demand to survive. Controlling for unobserved demand shocks when estimate productivity we find that the least productive become more productive. These results indicate importance of unobserved demand shocks when estimating productivity.

The results for the linear TFP process (DP2) are consistent with the ones we find for the nonlinear process (EOPms). The marginal effects of large entrants on productivity growth is larger for DP2 than for EOPms. One likely explanation is that DP2 only captures strong incumbents due to that stores need to be at least three years in the data. Our findings are in line with our hypothesis that competition increases productivity growth of incumbents.

The coefficient on population density is positive and significant at the 1% level for EOPms using GMM (1), but negative for DP2.

■ Decomposition of aggregate productivity growth. Because of data constraints we can not decompose the contribution of large entrants to aggregate TFP growth but only the contribution of all entrants, exits, and incumbents. We use three recent decompositions, the one by Foster et al. (2001) (FHK), Griliches and Regev (1995) (GR), and Melitz and Polanec (2009) (MP) which is a dynamic version of the static decomposition by Olley and Pakes (1996). All decompositions are discussed in detail in Appendix C.

Aggregate TFP growth was 8% from 1997 to 2001 (Tables 9 and 10). While overall industry growth is the same in all decompositions, the relative contribution of incumbents, entrants and exits differ. In both FHK and GR, incumbents that continue for the whole period contribute about 6%. Net entry stand for 1.85% in GR and 1.47% in FHK. Incumbents that increase both productivity and market shares stand for 0.75% of growth in FHK.

In MP, entrants and exits only have a positive contribution when their aggregate productivity is larger than that of continuing stores in the same period. As we expect, incumbents' contribution is larger in MP (9.53%) than in GR and FHK. Incumbents are more productive than both entrants (-3.67%) and exits (2.14%). Among incumbents, those that obtain productivity improvements are central (11.2%), whereas reallocation of market shares among them are not (-1.71%). The direct effect of exits is the largest component (11.7%) showing that exits with lower productivity than incumbents play a key role for growth. The indirect effects show that the covariance between market shares and productivity is greater for entrants and exits than for incumbents. The decomposition results confirm our findings based on large entrants, i.e., incumbents that increase their productivity, and low productive stores that exit foster productivity growth in retail.

5 Conclusions

The present study gives new insights into competition and productivity differences among retail stores. Net entry is found to foster almost all labor productivity growth in the U.S. retail sector (Foster et al. 2006). Multi-factor productivity in retail markets has however rarely been studied, contrary to manufacturing. We provide a first attempt to use recent advances in structural estimation of production functions to estimate total factor productivity in retail markets. Based on recent extensions of the Olley and Pakes' (1996) framework, we provide a model that takes key features of retail markets into account. In particular, we investigate one of the most crucial trends in retail markets: entry by large ("big-box") stores. On both sides of the Atlantic, the pros and cons of the big-box format have been widely debated (the Wal-Mart effect). We analyze whether large entrants force low productive stores out from the market and increase productivity growth among surviving stores with different positions in the productivity distribution. We use political preferences in local markets to control for endogeneity of large entrants. Our empirical application relies on detailed data on all retail food stores in Sweden, which is representative to many European markets in terms of market structure and regulation.

The results show that when estimating retail productivity, it is central to control for local market characteristics, and for selection, and to allow for nonlinearities in the productivity process. We recognize that large entrants clearly drive reallocation of resources towards more productive stores. After large entry, low productive stores are more likely to exit. In addition, large entrants increase productivity growth of incumbent stores. The magnitude of the growth increase varies however with an incumbent's position in the productivity distribution. The increase in growth declines with productivity, implying that growth increases relatively more among low productive survivors than among high productive ones. The productivity distribution thus gets truncated from below and dispersion decreases. Decompositions of aggregate productivity growth, 8% from 1997 to 2001 in the Swedish retail food market, confirm importance of incumbents and low productive exits. We conclude that entry by big-box stores spurs reallocation of resources towards more productive stores, and thus works as a catalyst for retail productivity growth.

Our findings contribute with knowledge to competition policy because entry regulation issues greatly concern policy makers in Europe, where such regulations are generally much more restrictive than in the U.S. As an example, the European Parliament recently highlighted an investigation of supermarket dominance (European Parliament 2008). We argue that a more restrictive design and application of entry regulations can hinder reallocation towards more productive units and thus hinder aggregate productivity growth. Note however that we clarify the indirect link between regulation, large entrants, and productivity because the numbers of approvals and rejections are not available. Besides productivity, entry regulations compound a wide range of other aspects. How to balance potential productivity growth against increased traffic and broader environmental issues are interesting issues for future research. It would also be interesting to apply our extended Olley and Pakes (1996) framework to other service markets such as banking and health care services. Future work would also benefit from using fully dynamic models (Dunne et al. 2005, Beresteanu and Ellickson 2006, Aguirregabiria et al. 2007, Holmes 2008) that would more carefully consider the importance of sunk costs, chain effects, and market adjustments.

References

- ACKERBERG, D., L. BENKARD, S. BERRY, AND A. PAKES (2007): "Econometric Tools for Analyzing Market Outcomes," *Handbook of Econometrics*, 6, 4171– 4276.
- ACKERBERG, D., K. CAVES, AND G. FRASER (2006): "Structural Identification of Production Functions," Mimeo, UCLA.
- ACKERBERG, D., X. CHEN, AND J. HAHN (2009): "A Practical Asymptotic Variance Estimator for Two-Step Semiparametric Estimators," Mimeo, UCLA.
- AGHION, P., R. BLUNDELL, R. GRIFFITH, P. HOWITT, AND S. PRANTL (2009): "The Effects of Entry on Incumbent Innovation and Productivity," *Review of Economics and Statistics*, 91(1), 20–32.
- AGUIRREGABIRIA, V., R. MIRA, AND H. ROMAN (2007): "An Estimable Dynamic Model of Entry, Exit and Growth in Oligopoly Retail Markets," *American Economic Review*, 97(2), 449–454.
- AI, C., AND X. CHEN (2003): "Efficient Estimation of Models with Conditional Moment Restrictions Containing Unknown Functions," *Econometrica*, 71(6), 1795–1843.
- ASPLUND, M., AND R. FRIBERG (2002): "Food Prices and Market Structure in Sweden," Scandinavian Journal of Economics, 104(4), 547–567.
- ASPLUND, M., AND V. NOCKE (2006): "Firm Turnover in Imperfectly Competitive Markets," *Review of Economic Studies*, 73, 295–327.
- BAILY, M., C. HULTEN, AND D. CAMPBELL (1992): "Productivity Dynamics in Manufacturing Establishments," *Brooklyns Papers on Economic Activity*, pp. 187–249.
- BARTELSMAN, E., AND M. DOMS (2000): "Understanding Productivity: Lessons from Longitudinal Microdata," *Journal of Economic Literature*, 38(3), 569–594.
- BASKER, E. (2005): "Job Creation or Destruction? Labor-Market Effects of Wal-Mart Expansion," *Review of Economics and Statistics*, 87(1), 174–183.
 - (2007): "The Causes and Consequences of Wal-Mart's Growth," *Journal* of *Economic Perspectives*, 21(3), 177–198.
- BASKER, E., AND M. NOEL (2009): "The Evolving Food Chain: Competitive Effects of Wal-Mart's Entry into the Supermarket Industry," *Journal of Eco*nomics and Management Strategy, 18(4), 177–198.

- BERESTEANU, A., AND P. ELLICKSON (2006): "The Dynamics of Retail Oligopoly," Mimeo, Duke University.
- BERRY, S., J. LEVINSOHN, AND A. PAKES (1995): "Automobile Prices in Market Equilibrium," *Econometrica*, 63(4), 841–890.
- BERTRAND, M., AND F. KRAMARZ (2002): "Does Entry Regulation Hinder Job Creation? Evidence from the French retail industry," *Quarterly Journal of Economics*, 117(4), 1369–1413.
- BLUNDELL, R., AND S. BOND (2000): "GMM Estimation with Persistent Panel Data: An Application to Production Functions," *Econometric Reviews*, 31(3), 549–569.
- BLUNDELL, R., R. GRIFFITH, AND J. VAN REENEN (1999): "Market Share, Market Value, and Innovation in a Panel of British Manufacturing Firms," *Review of Economic Studies*, 66(3), 529–554.
- BOND, S., AND M. SÖDERBOM (2005): "Adjustment Costs and the Identification of Cobb-Douglas Production Functions," Working Paper 05/04, Institute of Fiscal Studies.
- BOONE, J. (2000): "Competitive Pressure: The Effects on Investments in Product and Process Innovation," *RAND Journal of Economics*, 31(3), 549–569.
- BOYLAUD, O., AND G. NICOLETTI (2001): "Regulatory reform in retail distribution," OECD Working Paper 32.
- BUETTNER, T. (2004): "R&D and the Dynamics of Productivity," Mimeo, London School of Economics.
- CAVES, R. (1998): "Industrial Organization and New Findings on the Turnover and Mobility of Firms," *Journal of Economic Literature*, 36(4), 1947–1982.
- CHEN, X., AND S. LUDVIGSON (2007): "Land of Addicts? An Empirical Investigation of Habit-Based Asset Pricing Models," Mimeo, New York University.
- DE LOECKER, J. (2009): "Product Differentiation, Multi-Product Firms and Structural Estimation of Productivity," Mimeo, Princeton University.
- DORASZELSKI, U., AND J. JAUMANDREU (2009): "R&D and Productivity: Estimating Production Functions when Productivity is Endogenous," Mimeo, Harvard University.
- DUNNE, T., S. KLIMEK, M. ROBERTS, AND Y. XU (2005): "Entry and Exit in Geographical Markets," Mimeo, Pennsylvania State University.

- ERICSON, R., AND A. PAKES (1995): "Markov-Perfect Industry Dynamics: A Framework for Empirical Work," *Review of Economic Studies*, 62(1), 53–83.
- EUROPEAN PARLIAMENT (2008): "On investigating and remedying the abuse of power by large supermarkets operating in the European Union," Written Declaration 0088, European Parliament.

FISHMAN, C. (2006): The Wal-Mart Effect. Penguin Press, New York.

- FOSTER, L., J. HALTIWANGER, AND C. KRIZAN (2001): "The Link Between Aggregate and Micro Productivity growth: Evidence from Retail Trade," NBER Working Paper 13783.
 - (2006): "Market Selection, Reallocation, and Restructuring in the U.S. Retail Trade Sector in the 1990s," *Review of Economics and Statistics*, 88(4), 748–758.
- FOSTER, L., J. HALTIWANGER, AND C. SYVERSON (2008): "Reallocation, Firm Turnover, and Efficiency: Selection on Productivity or Profitability?," *American Economic Review*, 98(1), 394–425.
- GRIFFITH, R., AND H. HARMGART (2005): "Retail Productivity," International Review of Retail, Distribution and Consumer Services, 15(3), 281–290.
- GRILICHES, Z., AND H. REGEV (1995): "Productivity and Firm Turnover in Israeli Industry: 1979-1988," Journal of Econometrics, 65, 175–203.
- HALL, R. (1988): "The Relation between Price and Marginal Cost in U.S. Industry," Journal of Political Economy, 96(5), 921–947.
- HICKS, M. J. (2007): *The Local Economic Impact of Wal-Mart*. Cambria Press, Youngstown, New York.
- HOJ, J., T. KATO, AND D. PILAT (1995): "Deregulation and privatisation in the service sector," OECD Working Paper 25.
- HOLMES, T. (2008): "The Diffusion of Wal-Mart and Economies of Density," NBER Working Paper 13783.
- HOPENHAYN, H. (1992): "Entry, Exit and Firm Dynamics in Long Run Equilibrium," *Econometrica*, 60(5), 1127–1160.
- HULTEN, C., AND F. WYKOFF (1981): chap. The measurement of Economic Depreciation, in C. Hulten (ed.), *Depreciation, Inflation, and Taxation of Income* from Capital, Urban Institute Press, Washington, D.C.
- JIA, P. (2008): "What Happens when Wal-Mart comes to Town: An Empirical Analysis of the Discount Retailing Industry," *Econometrica*, 76(6), 1263–1316.

- JOVANOVIC, B. (1982): "Selection and the Evolution of Industry," *Econometrica*, 50(5), 649–670.
- KATAYAMA, H., S. LU, AND J. TYBOUT (2003): "Why Plant-Level Productivity Studies are Often Misleading, and an Alternative Approach to Inference," NBER Working Paper 9617.
- KLETTE, T., AND Z. GRILICHES (1996): "The Inconsistency of Common Scale Estimators when Output Prices are Unobserved and Endogenous," *Journal of Applied Econometrics*, 11(4), 343–361.
- LEVINSOHN, J., AND M. MELITZ (2002): "Productivity in a Differentiated Products Market Equilibrium," Mimeo, Harvard University.
- LEVINSOHN, J., AND A. PETRIN (2003): "Estimating Production Functions Using Inputs to Control for Unobservables," *Review of Economic Studies*, 70(2), 317–341.
- MACDONALD, J. (1994): "Does Import Competition Force Efficient Production?," Review of Economics and Statistics, 76(4), 721–727.
- MELITZ, M. (2000): "Estimating Firm Level Productivity in Differentiated Product Industries," Mimeo, Harvard University.

(2003): "The Impact of Trade on Intra-Industry Reallocations and Aggregate Industry Productivity," *Econometrica*, 71(6), 1695–1725.

- MELITZ, M., AND S. POLANEC (2009): "Dynamic Olley-Pakes Decomposition with Entry and Exit," Mimeo, Princeton University.
- NEWEY, W., AND J. POWELL (2003): "Instrumental Variables Estimation of Nonparametric Models," *Econometrica*, 71(3), 1557–1569.
- NEWEY, W., J. POWELL, AND F. VELLA (1999): "Nonparametrric Estimation of Triangular Simultaneous Equations Models," *Econometrica*, 67(3), 565–603.
- NEWEY, W., AND K. WEST (1987): "A simple, positive definite, heteroscedasticity and autocorrelation consistent covariance matrix," *Econometrica*, 55, 1201– 1228.
- NICKELL, S. (1996): "Competition and Corporate Performance," Journal of Political Economy, 104(4), 724–746.
- OLLEY, S., AND A. PAKES (1996): "The Dynamics of Productivity in the Telecommunications Equipment Industry," *Econometrica*, 64(6), 1263–1297.
- PAVCNIK, N. (2002): "Trade Liberalization Exit and Productivity Improvements: Evidence from Chilean Plants," *Review of Economic Studies*, 69(1), 245–276.

PILAT, D. (1997): "Regulation and Performance in the Distribution Sector," OECD Working Papers 180.

(2005): "Assessing the Productivity of the UK Retail Sector: Some Further Reflections," *The International Review of Retail, Distribution and Consumer Research*, 15(3), 291–296.

- PORTER, M. (1990): The Competitive Advantage of Nations. Macmillan Press.
- RAITH, M. (2003): "Competition, Risk, and Managerial Incentives," American Economic Review, 93, 1425–1436.
- REYNOLDS, J., E. HOWARD, D. DRAGUN, B. ROSEWELL, AND P. ORMEROD (2005): "Assessing the Productivity of the UK Retail Sector," *International Review of Retail, Distribution and Consumer Research*, 15(3), 237–280.
- ROBINSON, P. (1988): "Root-N-Consistent Semiparametric Regression," Econometrica, 56(4), 931–954.
- SADUN, R. (2008): "Does Planning Regulation Protect Independent Retailers?," Center for Economic Performance Discussion Paper No 888, London School of Economics.
- SCHMIDT, K. (1997): "Managerial Incentives and Product Market Competition," *Review of Economic Studies*, 64, 191–213.
- SIVADASAN, J. (2004): "Barriers to Entry and Productivity: Micro-Evidence from Indian Manufacturing Sector Reforms," Mimeo, University of Chicago.
- STATISTICS SWEDEN (2005): "Hushållens ekonomi vem lånar, vem konsumerar? (The Economy of Households: Who lends, who comsumes?)," Mimeo, Stockholm.
- SWEDISH COMPETITION AUTHORITY (2001:4): "Kan Kommunerna Pressa Matpriserna? (Can the Municipalities put Pressure on Prices?)," Technical Report 4, Stockholm.
 - (2004:2): "Konsumenterna, Matpriserna och Konkurrensen (Consumers, Retail Food Prices, and Competition)," Technical Report 2, Stockholm.
- SWEDISH NATIONAL BOARD OF HOUSING, BUILDING, AND PLANNING (2005): "Shopping martters! Retailing and sustainable urban development," Technical Report.
- SYVERSON, C. (2004): "Market Structure and Productivity: A Concrete Example," Journal of Political Economy, 112(2), 1181–1222.

Table 1: Characteristics of the Swedish Retail Food Market

A. DEL	.FI					
Year	No. of	Large	Large	Mean sales	Total sales	Total
	stores	stores	entry	space (m^2)	space (m^2)	sales
1996	4,664	905	21	538	2,510,028	129,326,000
1997	4,518	925	8	550	2,483,248	126,732,397
1998	4,351	926	9	587	2,552,794	130, 109, 604
1999	4,196	936	18	604	2,514,367	133, 156, 023
2000	3,994	948	23	654	2,587,952	138,314,044
2001	3,656	942	28	689	2,471,510	139, 352, 920
2002	3,585	932	5	718	2,525,084	142,532,944
B. FS-F	RAMS					
Year	No. of	No. of	Total	Value	Total	
	"multi-	employees	wages	added	sales	
	stores"					
1996	3,714	74,100	9,882,234	18,319,407	141,743,876	
1997	3,592	73,636	10,322,136	18,838,130	142,840,611	
1998	3,482	74,696	10,766,043	19,185,120	147,726,647	
1999	3,398	74,758	11,110,785	19,570,472	152,160,949	
2000	3,287	77,180	11,536,063	20,389,492	154,106,865	
2001	3,094	76,905	11,522,482	20,748,902	158, 512, 132	
2002	3.067	80,931	12.081.931	22.473.696	179.335.162	

NOTE: DELFI is provided by Delfi Marknadspartner AB and contains all retail food stores based on their geographical location (address). FS-RAMS is provided by Statistics Sweden and consists of all organization numbers in SNI-code 52.1, i.e., "multi-store" units that contain one store or several (e.g., due to the same owner). Sales (incl. 12% VAT), value-added, and wages are measured in thousands of 1996 SEK (1USD=6.71SEK, 1EUR=8.63 SEK). Sales in DELFI are collected by surveys and reported in classes, while sales are based on tax reporting in FS-RAMS. Therefore, total sales are lower in DELFI than in FS-RAMS. From 1996 to 2002, the total population in Sweden increased from 8,844,499 to 8,940,788.

Year	1997	1998	1999	2000	2001	2002
A. Markets with large entr	rants					
No. of stores	37.00	54.00	29.00	32.00	33.00	22.00
No. of all entrants	2.00	2.00	3.00	2.00	1.00	2.00
No. of all exits	3.00	2.00	2.00	3.00	1.00	
Population	57,441.00	60,429.00	37,195.00	48,250.00	58,361.00	22,907.00
Population density	80.88	57,92.00	68.03	79.38	77.29	52.77
Per capita income	149.10	157.60	161.60	170.30	179.10	177.60
Store concentration (C_4)	0.53	0.49	0.62	0.60	0.53	0.70
Total no. of markets	10	9	20	20	23	6
B. Markets without large	entrants					
No. of stores	15.00	15.00	15.00	14.00	13.00	14.00
No. of all entrants	0.00	0.00	1.00	0.00	0.00	0.00
No. of all exits	0.00	1.00	1.00	1.00	0.00	
Population	14,827.00	15, 133.00	14,322.00	14,154.00	14,068.00	15,207.00
Population density	25.80	25.78	25.22	25.60	24.75	26.20
Per capita income	143.30	149.10	155.90	162.50	168.40	175.90
Per capita income Store concentration (C_4)	143.30 0.71	$ \begin{array}{r} 149.10 \\ 0.71 \end{array} $	155.90 0.72	162.50 0.73	168.40 0.75	175.90 0.76

Table 2: Medians of local market characteristics

NOTE: 1996 is left out because entrants are not observed. Municipalities, considered as local markets, increase from 288 to 290 due to three municipality break-ups during the period. Stores, entrants and exits come from DELFI. Population density is defined as total population per square kilometer in the municipality. Concentrations (C_4) show the market share captured by the top four stores.

Table 3: Value added g	generating	function e	estimates									
	OLS	ACF	DP	1	DP	2	EOF	S	EOI	m	EOP	ms
			(1)	(2)	(1)	(2)	(1)	(2)	(1)	(2)	(1)	(2)
Log no. of labor	0.948	0.768	0.754	0.916	0.686	0.900	0.845	0.921	0.945	1.205	0.840	1.269
- - - -	(cn0.0)	(7.60.0)	(100.0)	107	(0.004)	007 0	(100.0)	0000	(100.0)	1	(7.00.0)	.00.0
Log of capital	0.167	0.163	0.400	0.485	(0.003)	0.400	(0.001)	0.232	0.116	0.147	0.145 (0.001)	0.201
Market output $\left(-\frac{1}{2}\right)$	(20010)	(00000)	0.176		0.313		0.082		0.216		0.338	
			(0.002)		(0.001)		(0.001)		(0.001)		(0.001)	
Number of large entrants			-0.945	-5.371	-0.031	-0.098			-0.093	-0.430	0.034	0.100
			(0.002)		(0.001)				(0.001)		(0.004)	
Population density			-0.103 (0.002)	-0.421	-0.166 (0.001)	-0.529			-0.054 (0.001)	-0.252	-0.033 (0.001)	-0.186
Productivity transition (ρ_1)			0.417		0.449							
			(200.0)		(100.0)							
Demand shock transition (ρ_2)	-				0.353 (0.106)							
Scale $(\beta_l + \beta_k)$	1.115	0.931	1.402		1.426		1.140		1.362		1.420	
Demand elasticity (η)			-5.674		-3.198		-12.192		-4.659		-2.962	
Mark-up $\left(\frac{\eta}{1+\eta}\right)$			1.214		1.455		1.089		1.273		1.509	
Sargan (p-value)		0.081					0.296		0.990		0.935	
No. of obs.	23,521	16,186	15,6	40	15,6	40	15,64	10	15,6	40	15,6	40
NOTE: The dependent variab OLS is ordinary least square when ω_{jt} and v_{jt} follow the the semi-parametric estimatic equation (9) with control for in Section 3, i.e., we control fo	le is log of de regression. same AR(1) on of equation prices and for prices, lo	flated value a ACF is Acker process; DF on (9) withou local market cal market ch	dded. Labo rberg, Cave 2 is linear 1t local ma characterist aracteristic	or is measu s, and Fra- estimation rket charac ics but no s, and sele	red as num ser's (2006 t of equation t for select t for select ction. In t	ber of full-) two-stage on (6) whe ut controll ion; EOPr he EOP sp	time adjust e estimation in ω_{jt} and ing for sele ing is the so	ed employ a method; v_{jt} follow sction; EC smi-param t, columns	rees. All re DP1 is lii v two diffe DPm is the netric estim s (1) show	gressions in near estim- rent AR(1 semi-para nation of e estimated	nclude year ation of eq) processes umetric esti quation (9 coefficients	dummies. uation (6) ; EOPs is imation of) specified including
elasticity (see Equation 6); co. In ACF, current capital stock for estimation. Market output defined as arrice over marginal	lumns (2) sh t and previor t is measure 1 cost	iow estimated us labor are u d as the mark	coefficients ised as inst cet share we	s without e ruments, a sighted out	lasticity. F nd standaı put in the	teported st d errors ar municipali	andard erre re compute ty. Demane	ors (in par d using bo d refers to	entheses) a ootstrap. I o the elastic	are robust n EOP, tw city of sub	to heterosc ro-step GM stitution. N	edasticity. M is used Mark-up is
nemited as price over man.	1 coor.											

35



Figure 1: TFP kernel density estimates, incumbent stores in markets the year of, and the year after, large entry

Table 4: TFP and large entrants

A. Markets with large entrants	Mean	Std. Dev.	
Year of entry	-0.291	1.064	
Year after entry	-0.016	1.051	
Test (p-value)	0.001	0.671	
B. All markets			
With entry _{t-1}	-0.016	1.051	
Without entry _{t-1}	0.100	0.909	
Test (p-value)	0.001	0.001	

NOTE: This table summarizes TFP levels of incumbents in markets before and after large entrants, and in markets with and without large entrants. T-test is used for mean, and F-test is used for standard deviation (p-values reported). TFP is estimated using the semi-parametric EOPms method described in Section 3. Large entrants are defined as the five largest store types in the DELFI data (hypermarkets, department stores, large supermarkets, large grocery stores, and other stores).

Table 5: Transition matrix from t-1 (column) to t (row) in percentage

	`	/	· ·	/ *	0		
Percentile	<p10< td=""><td>p10-p25</td><td>p25-p50</td><td>p50-p75</td><td>p75-p90</td><td>>p90</td><td>Exit</td></p10<>	p10-p25	p25-p50	p50-p75	p75-p90	>p90	Exit
A. Markets with large entrants in t-1							
<p10< td=""><td>35.59</td><td>22.88</td><td>10.17</td><td>1.69</td><td>1.69</td><td>0.85</td><td>27.12</td></p10<>	35.59	22.88	10.17	1.69	1.69	0.85	27.12
p10-p25	14.05	33.47	23.55	4.13	1.65	0.00	23.14
p25-p50	2.53	12.64	42.30	19.77	4.14	1.38	17.24
p50-p75	0.85	2.34	21.44	44.37	13.80	2.97	14.23
p75-p90	0.00	1.77	6.38	26.24	37.23	13.48	14.89
>p90	0.63	1.25	2.50	11.88	20.00	48.75	15.00
B. Markets without large entrants in t-	1						
<p10< td=""><td>31.84</td><td>26.02</td><td>12.55</td><td>4.08</td><td>0.61</td><td>1.53</td><td>23.37</td></p10<>	31.84	26.02	12.55	4.08	0.61	1.53	23.37
p10-p25	15.32	34.75	23.92	5.63	1.39	0.82	18.16
p25-p50	4.11	13.72	40.75	20.17	3.97	1.77	15.52
p50-p75	0.86	3.15	20.52	42.06	16.04	5.50	11.87
p75-p90	0.39	1.34	8.52	25.83	33.05	16.81	14.06
>p90	1.09	0.84	5.13	15.63	25.55	35.38	16.39

NOTE: TFP is estimated using the semi-parametric EOPms method described in Section 3. Municipalities are considered as local markets. Large entrants in period t-1 are defined as the five largest store types in the DELFI data (hypermarkets, department stores, large supermarkets, large grocery stores, and other stores).

Table 6: Regression results: Exit

· · · · · · · · · · · · · · · · · · ·	TFP nor	linear (EOPms)	TFP 1	inear (DP2)
	(1)	(2)	(3)	(4)
	(1)	(2)	(3)	(*)
Log of productivity,	-0.058		-0.013	
	(0.015)		(0.007)	
Large entrants _{t-1}	0.022	-0.152	0.014	0.061
0 01	(0.037)	(0.088)	(0.057)	(0.109)
p10*Large entrants _{t-1}	· · · ·	0.293	· · · ·	0.121
01		(0.118)		(0.155)
p10-p25*Large entrants _{t-1}		0.190		0.128
		(0.126)		(0.167)
p25-p50*Large entrants _{t-1}		0.293		-0.216
		(0.113)		(0.160)
p75-p90*Large entrants _{t-1}		0.071		-0.275
		(0.117)		(0.197)
p90*Large entrants _{t-1}		0.209		-0.036
		(0.129)		(0.192)
Log of capital _t	-0.083	-0.087	-0.077	-0.070
	(0.009)	(0.009)	(0.014)	(0.014)
Log of population density _t	0.018	0.020	-0.004	-0.004
00	(0.008)	(0.008)	(0.011)	(0.011)
Year dummies	yes	yes	yes	yes
No. of obs.	11,132	11,132	7,376	7,376

NOTE: This table shows probit regressions on exit. TFP is estimated using the semi-parametric EOP method described in Section 3 (EOPms) and linear panel specification (DP2). Reported standard errors (in parentheses) are robust to heteroscedasticity. Large entrants in period t-1 are defined as the five largest store types in the DELFI data (hypermarkets, department stores, large supermarkets, large grocery stores, and other stores). We use six percentile bins for productivity in each local market and year, with p50-75 used as reference group.



Figure 2: TFP growth kernel density estimates, incumbent stores in markets with and without large entrants

Table 7: TF	P Growth	and 1	arge	entrants
-------------	----------	-------	------	----------

0		
Mean	Std. Dev.	
-0.110	0.535	
0.159	0.568	
0.001	0.049	
0.159	0.568	
0.138	0.598	
0.227	0.021	
	Mean -0.110 0.159 0.001 0.159 0.138 0.227	Mean Std. Dev. -0.110 0.535 0.159 0.568 0.001 0.049

NOTE: This table summarizes TFP growth of incumbents in markets before and after large entrants, and in markets with and without large entrants. T-test is used for mean, and F-test is used for standard deviation (p-values reported). TFP is estimated using the semi-parametric EOPms method described in Section 3. Large entrants are defined as the five largest store types in the DELFI data (hypermarkets, department stores, large supermarkets, large grocery stores, and other stores).



Figure 3: TFP growth kernel density estimates, incumbent stores in markets the year of, and the year after, large entrants

		TFP nc	onlinear (F	(OPms)				FP linear	· (DP2)	
	Ö	LS	Within	GN	IM	õ	S	Within		GMM
	(1)	(2)	_	(1)	(2)	(1)	(2)		(1)	(2)
Large entrants _{t-1}	0.040	-0.009	-0.021	-0.227	0.451	0.084	0.006	-0.073	-1.001	0.252
	(0.008)	(0.014)	(0.017)	(0.004)	(0.001)	(0.023)	(0.045)	(0.058)	(0.001)	(0.001)
p10 [*] Large entrants _{t-1}		0.161	0.188	0.371	0.343		0.275	0.372	1.241	1.400
		(0.022)	(0.028)	(0.001)	(0.001)		(0.063)	(0.081)	(0.001)	(0.001)
p10-p25* Large entrants _{t-1}		0.079	0.107	0.281	0.593		0.154	0.273	1.087	1.840
		(0.020)	(0.025)	(0.001)	(0.001)		(0.060)	(0.081)	(0.001)	(0.001)
p25-p50* Large entrants _{t-1}		0.071	0.096	0.263	0.422		0.127	0.235	1.019	1.301
		(0.021)	(0.024)	(0.001)	(0.001)		(0.055)	(0.074)	(0.001)	(0.001)
p75-p90*Large entrants _{t-1}		0.004	0.017	0.196	-0.462		-0.084	0.005	0.798	-1.618
		(0.020)	(0.024)	(0.001)	(0.001)		(0.065)	(0.093)	(0.001)	(0.001)
p90*Large entrants _{t-1}		-0.031	-0.022	0.161	-0.314		-0.245	-0.090	0.625	-1.196
		(0.023)	(0.030)	(0.001)	(0.001)		(0.070)	(0.101)	(0.001)	(0.001)
Log of population density _t	0.180	0.110	-0.038	0.151	-1.166	-0.113	-0.264	-0.031	-0.244	-5.206
	(0.164)	(0.163)	(0.015)	(0.001)	(0.001)	(0.505)	(0.503)	(0.052)	(0.003)	(0.004)
Year fixed effects	yes	yes	yes	yes	yes	yes	yes	yes	yes	yes
Market fixed effects	yes	yes	no	yes	yes	yes	yes	no	yes	yes
Adjusted R^2	0.745	0.746				0.253	0.258			
J-test (p-value)				0.994	0.998				0.997	0.998
No. of obs.	13,626	13,626	13,626	13,626	9,774	7,376	7,376	7,376	7,376	4,884
NOTE: TFP is estimated t	using the se	emi-param	etric EOP	method d	escribed in	Section 3 (E	OPms) an	d linear p	anel speci	ication (DP2).
Standard errors reported in	n parenthes	es, and one	e-step GM	M estimat	or is used.	GMM (1) us	es lagged	political p	references	as instruments
for large entry, GMM (2) a	also adds la	gged large	entrants	(in t-2), la	gged popula	ation density.	, and lagge	ed income.	. J-test re:	fers to the test
for overidentified restriction	ns in GMM	models.]	Large entr	ants in per	riod t-1 are	defined as tl	ne five larg	est store t	types in th	ie DELFI data
(hypermarkets, department	stores, larg	ge superma	urkets, larg	e grocery s	stores, and c	other stores).	We use si	k percentil	le bins for	productivity in
each market and year, with	ı p50-75 use	d as refere	ance group							

 Table 8: Regression results: TFP growth

Table 9: Decomposition of retail food	productivity	growth,	1997	to 2001
---------------------------------------	--------------	---------	------	---------

			Percenta	ge of growth	from	
Overall	Within	Between	Cross	Entry	Exit	Net entry
industry	stores	stores	stores			
growth	(1)	(2)	(3)	(4)	(5)	(4) - (5)
A. Baily et	al (1992) / F	oster et al (200	1)			
0.08	0.0572	0.0006	0.0075	0.0025	-0.0121	0.0147
B. Griliches	s and Regev (1995)				
0.08	0.0609	0.0005		0.0294	-0.0109	0.0185

NOTE: Decomposition using Equation (18) in Section 3; TFP is estimated using the semi-parametric estimation (EOPms) described in Section 3. Shares of local market sales are used as weights. Appendix C describes the decompositions in detail.

Table 10: Dynamic Olley and Pakes decomposition of TFP growth 1997-2001: Melitz and Polanec (2009)

			Percen	tage of growth	n from	
-	Surviv	ving	Entra	ints		Exits
Overall	Unweigh.	Cov	Unweigh.	Weigh.	Unweigh.	Weigh.
Industry						
Growth						
0.08	0.1124	-0.0171	-0.0893	-0.0367	0.117	0.0214

NOTE: TFP is estimated using the semi-parametric estimation EOP described in Section 3 and 4. Shares of local market sales are used as weights. Appendix C describes the decomposition in detail.

Appendix A: PBA and data sources

Entry regulation (PBA). On July 1, 1987, a new regulation was imposed in Sweden, the Plan and Building Act (PBA). Compared to the previous legislation, the decision process was decentralized, giving local governments power over entry in their municipality and citizens a right to appeal the decisions. Since 1987, only minor changes have been implemented in PBA. From April 1, 1992 to December 31, 1996, the regulation was slightly different, making explicit that the use of buildings should not counteract efficient competition. Since 1997, PBA has been more or less the same as prior to 1992. Long time lags in the planning process make it impossible to directly evaluate the impact of decisions. In practice, differences because of the policy change seem small (Swedish Competition Authority 2001:4). Nevertheless, PBA is claimed to be one of the major entry barriers, resulting in different outcomes, e.g., price levels, across municipalities (Swedish Competition Authority 2001:4, Swedish Competition Authority 2004:2). Municipalities are then, through the regulation, able to put pressure on prices. Those that constrain entry have less sales per capita, while those where large and discount stores have a higher market share also have lower prices.

■ The DELFI data. DELFI Marknadspartner AB collects daily data on retail food stores from a variety of channels: (1) public registers, the trade press, and daily press; (2) the Swedish retailers association (SSLF); (3) Kuponginlösen AB (which deals with rebate coupons collected by local stores); (4) the chains' headquarters; (5) matching customer registers from suppliers; (6) telephone interviews; (7) yearly surveys; and (8) the Swedish Retail Institute (HUI). Location, store type, owner, and chain affiliation are double-checked in corporate annual reports.

Each store has an identification number linked to its geographical location (address). The twelve store types, based on size, location, product assortment, etc., are hypermarkets, department stores, large supermarkets, large grocery stores, other stores, small supermarkets, small grocery stores, convenience stores, gasstation stores, mini markets, seasonal stores, and stores under construction.

Sales and sales space are collected via yearly surveys. Revenues (including VAT) are recorded in 19 classes. Due to the survey collection, a number of missing values are substituted with the median of other stores of the same type in the same local market. In total, 702 stores have missing sales: 508 in 1996, and 194 in later years. For sales space, all 5,013 values are missing for 1996, and are therefore replaced with the mean of each stores' 1995 and 1997 values. In addition, 2,810 missing sales space values for later years are replaced similarly. In total, 698 observations are missing both sales and sales space.

■ The FS-RAMS data. FS-RAMS contains all registered organization numbers in the different Swedish industries from 1996 to 2002. *Value added* is defined as total shipments, adjusted for inventory changes, minus costs of materials. *Labor* is the total number of employees. We deflated sales, value added, wages, and investment by the consumer price index (CPI) from IMF-CDROM 2005.

Capital is constructed using a perpetual inventory method, $k_{t+1}(1-\delta)k_t + i_t$. Since the data distinguishes between buildings and equipment, all calculations of the capital stock are done separately for buildings and equipment. In the paper, we include equipment in the capital stock. Including both equipment and buildings in the capital stock does not change the results, however. As suggested by Hulten and Wykoff (1981), buildings are depreciated at a rate of 0.0361, and equipment at 0.1179. In order to construct capital series using the perpetual inventory method, an initial capital stock is needed. We set initial capital stock to its first occurrence in FS-RAMS, defining entry as the first year in FS (some of the stores have been in FS since 1973).

Appendix B: Selection and estimation strategy

Selection. A store's decision to exit in period t depends directly on productivity ω_{jt} , so that the decision will be correlated with the productivity shock ε_{jt} . To identify β_l and β_k , we use estimates of survival probabilities, given by

$$Pr(\chi_{t} = 1 | \underline{\omega}_{t}(k_{jt}, z_{mt-1}), \mathcal{F}_{t-1}) = Pr(\omega_{jt} \geq \underline{\omega}_{t}(k_{jt}, z_{mt-1}) | \\ \underline{\omega}_{t}(k_{jt}, z_{mt-1}), \omega_{jt-1}) \\ = P_{t-1}(i_{jt-1}, l_{jt-1}, k_{jt-1}, s_{jt-1}, p_{mt-1}, q_{mt-1}, \\ z_{mt-1}) \\ \equiv \mathscr{P}_{t-1}$$

$$(13)$$

where the second equality follows from (8). Controlling for selection, we can express the non-parametric function $h(\cdot)$ (the approximation of the conditional expectation $E[\omega_{jt}|\mathcal{F}_{t-1}]$) as a function of threshold market productivity $\underline{\omega}_t$ and the information set \mathcal{F}_{t-1} . As a result, threshold market productivity can be written as a function of \mathscr{P}_{t-1} and \mathcal{F}_{t-1} . Substituting Equations (8) and (13) into (6) yields

$$y_{jt} = \left(1 + \frac{1}{\eta}\right) \left[\beta_0 + \beta_l l_{jt} + \beta_k k_{jt}\right] - \frac{1}{\eta} q_{mt} - \frac{1}{\eta} z'_{mt-1} \beta_z + h\left(\mathscr{P}_{t-1}, \frac{\eta}{1+\eta} \left(\delta_1 + \left[(1 - \beta_l) - \frac{1}{\eta} \beta_l\right] l_{jt-1} - \left(1 + \frac{1}{\eta}\right) \beta_k k_{jt-1} + s_{jt-1} - p_{It-1} + \frac{1}{\eta} q_{mt-1} + \frac{1}{\eta} z'_{mt-1} \beta_z\right)\right) + \left(1 + \frac{1}{\eta}\right) \xi_{jt} - \frac{1}{\eta} v_{jt} - \frac{1}{\eta} u_{jt}^d + \left(1 + \frac{1}{\eta}\right) u_{jt}^p.$$
(14)

Estimation strategy. We first use a probit model with a third order polynomial to estimate the survival probabilities in (13). The predicted survival probabilities are then substituted into (9), which is estimated in the second step. We now turn to details about the estimation procedure of the latter step. The semiparametric regression (9) is estimated using the sieve minimum distance (SMD) procedure proposed in Newey and Powell (2003) and Ai and Chen (2003) for i.i.d. data.³⁹ The goal is to obtain an estimable expression for the unknown parameter of interest, $\boldsymbol{\alpha} = (\boldsymbol{\beta}, h)'$. We denote the true value of the parameters with the subscript "a", so that $\boldsymbol{\alpha}_a = (\boldsymbol{\beta}_a, h_a)'$. The moment conditions could then be written more compactly as

$$E[\psi_{jt}(\boldsymbol{X}_{jt},\boldsymbol{\beta}_a,h_a)|\boldsymbol{\mathcal{F}}_t^*] = 0 \quad j = 1,\cdots, N \quad t = 1,\cdots, T$$
(15)

where N is the total number of stores, \mathcal{F}_t^* is the information set at time t, and $\psi_{jt}(\cdot)$ is defined as

$$\psi_{jt}(\boldsymbol{X}_{jt}, \boldsymbol{\beta}_{a}, h_{a}) \equiv \left[\left(1 + \frac{1}{\eta} \right) \xi_{jt} - \frac{1}{\eta} v_{jt} - \frac{1}{\eta} u_{jt}^{d} + \left(1 + \frac{1}{\eta} \right) u_{jt}^{p} \right] \\ = y_{jt} - \left(1 + \frac{1}{\eta} \right) \left[\beta_{0} + \beta_{l} l_{jt} + \beta_{k} k_{jt} \right] + \frac{1}{\eta} q_{mt} + \frac{1}{\eta} z_{mt}^{'} \beta_{z} - h(\omega_{jt-1})$$

Let \mathcal{F}_t be an observable subset of \mathcal{F}_t^* . Then Equation (15) implies

$$E[\psi_{jt}(\boldsymbol{X}_{jt},\boldsymbol{\beta}_a,h_a)|\mathcal{F}_t] = 0 \quad j = 1,\cdots,N \quad t = 1,\cdots,T$$
(16)

If the information set \mathcal{F}_t is informative enough, such that $E[\psi_{jt}(\boldsymbol{X}_{jt},\boldsymbol{\beta},h)|\mathcal{F}_t] = 0$ for all j and for any $0 \leq \beta < 1$, then $(\boldsymbol{\beta},h)' = (\boldsymbol{\beta}_a,h_a)'$. The true parameter

³⁹Chen and Ludvigson (2007) show that the SMD procedure and its large sample properties can be extended to stationary ergotic time series data.

values must satisfy the minimum distance relation

$$\boldsymbol{\alpha}_{a} = (\boldsymbol{\beta}_{a}, h_{a})' = \arg\min_{\boldsymbol{\alpha}} E[m(\boldsymbol{\mathfrak{F}}_{t}, \boldsymbol{\alpha})'m(\boldsymbol{\mathfrak{F}}_{t}, \boldsymbol{\alpha})]$$

where $m(\mathcal{F}_t, \boldsymbol{\alpha}) = E[\psi(\boldsymbol{X}_t, \boldsymbol{\alpha})|\mathcal{F}_t], \psi(\boldsymbol{X}_t, \boldsymbol{\alpha}) = (\psi_1(\boldsymbol{X}_t, \boldsymbol{\alpha}), \cdots, \psi_N(\boldsymbol{X}_t, \boldsymbol{\alpha}))'$ for any candidate values $\boldsymbol{\alpha} = (\boldsymbol{\beta}, h)'$. The moment conditions are used to describe the SMD estimation of $\boldsymbol{\alpha}_a = (\boldsymbol{\beta}_a, h_a)'$. The SMD procedure has three parts. First, we can estimate the function $h(\cdot)$, which has an infinite dimension of unknown parameters, by a sequence of finite-dimensional unknown parameters (sieves) denoted h_H . Approximation error decreases as the dimension H increases with sample size N. Second, the unknown conditional mean $m(\mathcal{F}_t, \boldsymbol{\alpha}) = E[\psi(\boldsymbol{X}_t, \boldsymbol{\alpha})|\mathcal{F}_t]$ is replaced with a consistent nonparametric estimator $\hat{m}(\mathcal{F}_t, \boldsymbol{\alpha})$ for any candidate parameter values $\boldsymbol{\alpha} = (\boldsymbol{\beta}, h)'$. Finally, the function h_H is estimated jointly with the finite dimensional parameters $\boldsymbol{\beta}$ by minimizing a quadratic norm of estimated expectation functions,

$$\hat{\boldsymbol{\alpha}} = \arg \min_{\boldsymbol{\beta}, h_H} \frac{1}{T} \sum_{t=1}^{T} \hat{m}(\boldsymbol{\mathcal{F}}_t, \boldsymbol{\beta}, h_H)' \hat{m}(\boldsymbol{\mathcal{F}}_t, \boldsymbol{\beta}, h_H)$$
(17)

We approximate $h(\cdot)$ by a third order polynomial and substitute it in (16) as if it were the true model. Since the errors $\psi_t(\cdot)$ are orthogonal to the regressors $\mathcal{F}_t = (1, l_{jt-1}, k_{jt}, q_{mt-1}, z_{mt-1})$, we use a third order power series of \mathcal{F}_t , denoted \boldsymbol{P} , as instruments. We estimate $m(\mathcal{F}, \boldsymbol{\alpha})$ as the predicted values from regressing the errors $\psi_t(\cdot)$ on the instruments. Using \boldsymbol{P} , we specify the weighting matrix as $\boldsymbol{A} = I_N \otimes (\boldsymbol{P}'\boldsymbol{P})^{-1}$, making the estimation a GMM case. The weighting matrix \boldsymbol{A} gives greater weight to moments that are highly correlated with the instruments. Using the specified GMM implementation, the parameter values $(\boldsymbol{\beta}, h_H)$ are jointly estimated.

Appendix C: Productivity decompositions

Though we cannot determine the exact contribution of large entrants, our data allow us to decompose aggregate productivity growth due to entrants, exits, and incumbents. Industry level productivity (Ω_t) can then be expressed as the weighted average productivity: $\Omega_t \equiv \sum_{j \in N} ms_{jt} \omega_{mt}$, where N is the number of stores, and $ms_{jt} = sales_{jt}/sales_t$.

The change in retail food productivity from year t to year t' can be written as

$$\Delta\Omega_{t,t'} = \sum_{j \in C_{t,t'}} ms_{jt} \Delta\omega_{jt,t'} + \sum_{j \in C_{t,t'}} \Delta ms_{jt,t'}(\omega_{jt} - \Omega_t) + \sum_{j \in C_{t,t'}} \Delta ms_{jt,t'} \Delta\omega_{jt,t'} + \sum_{j \in E_{t,t'}} ms_{jt'}(\omega_{jt'} - \Omega_t) - \sum_{j \in X_{mt,t'}} ms_{jt}(\omega_{jt} - \Omega_t)$$
(18)

where Δ is the difference operator $(\Delta\Omega_{t,t'} = \Omega_{t'} - \Omega_t)$; $C_{t,t'}$ is the set of continuing stores, i.e., operating in both t and t'; $E_{t,t'}$ is the set of entering stores, i.e., that operated in t' but not in t; and $X_{t,t'}$ is the set of exiting stores, i.e., that operated in t but not in t'. This decomposition, derived by Foster et al. (2001)(FHK), is a modified version of the decomposition by Baily et al. (1992).

The decomposition (18) thus consists of five terms. The first term (Within) is the increase in productivity when the continuing stores increase their productivity at initial sales. The second term (Between) is the increase in productivity when continuing stores with above-average productivity expand their share of sales relative to stores with below-average productivity. The third term (Cross) captures the increase in productivity when continuing stores increase their market shares, while the fourth and fifth terms (Entry and Exit) are productivity increases due to entry and exit, respectively.

The second productivity decomposition used is given by Griliches and Regev (1995) (GR) and modified by FHK to allow for entry and exit

$$\Delta\Omega_{t,t'} = \sum_{j \in C_{t,t'}} \overline{ms_j} \Delta\omega_{jt,t'} + \sum_{j \in C_{t,t'}} \Delta ms_{jt,t'}(\overline{\omega_j} - \overline{\Omega}) + \sum_{j \in E_{t,t'}} ms_{jt'}(\omega_{jt'} - \overline{\Omega}) - \sum_{j \in X_{t,t'}} ms_{jt}(\omega_{jt} - \overline{\Omega})$$
(19)

where bars over a variable indicate the average of the variable across t and t'. The within term in the GR decomposition consists of the growth rates of continuing stores' TFP weighted by the average of their shares across t and t'. Both decom-

positions compare aggregate productivity of entering and existing stores, either to the aggregate productivity of all stores (FHK) or to the unweighted average of aggregate productivity of all stores (GR).

Olley and Pakes (1996) (OP) proposes a static decomposition of aggregate productivity, in which the weighted productivity of continuing stores, Ω_t , has two components: (1) contribution of productivity improvements, $\overline{\Omega}_t$; and (2) market share reallocations for the continuing stores $cov(ms_{jt}, \omega_{jt}) \equiv \sum_j (ms_{jt} - \overline{ms}_t)(\omega_{jt} - \overline{\Omega}_t)$. The difference in productivity index, $\Delta\Omega_{t,t'}$, can be written as

$$\Delta\Omega_{t,t'} = \Delta\overline{\Omega}_{t,t'} + \Delta cov_{t,t'}.$$
(20)

The OP decomposition ignore the entry and exit. However, Melitz and Polanec (2009) (MP) suggest a dynamic OP decomposition where there is a positive contribution for entering and exiting stores only when the aggregate productivity of these stores is larger than that of continuing stores in corresponding periods. The aggregate productivity in periods t and t' can be decomposed as

$$\Omega_t = ms_{C_t}\Omega_{C_t} + ms_{X_t}\Omega_{X_t}$$

$$\Omega_{t'} = ms_{C_{t'}}\Omega_{C_{t'}} + ms_{E_{t'}}\Omega_{E_{t'}}$$
(21)

where $m_{S_{t}}$, $m_{S_{t'}}$, $m_{S_{t'}}$, and $m_{S_{t}}$ are the aggregate market shares of incumbents (in period t and t'), entrants and exits, respectively. The change in aggregate productivity can be written as

$$\Delta\Omega_{t,t'} = \Delta\overline{\Omega}_{C_{t,t'}} + \Delta cov_{C_{t,t'}} + ms_{E_{t'}}(\Omega_{E_{t'}} - \Omega_{C_{t'}}) + ms_{X_t}(\Omega_{C_t} - \Omega_{X_t}).$$
(22)

where the contribution of continuing firms is divided into within-firm productivity improvements $(\Delta \overline{\Omega}_{C_{t,t'}})$ and market share reallocations $(\Delta cov_{C_{t,t'}})$ as in OP. The contribution of entrants and exits contains two parts, unweighted average productivity (direct effect) and the covariance term (indirect effect). For entrants: $ms_{E_{t'}}(\overline{\Omega}_{E_{t'}} - \overline{\Omega}_{C_{t'}})$, and $ms_{E_{t'}}(cov(\Omega_{E_{t'}}) - cov(\Omega_{C_{t'}}))$. For exits: $ms_{X_t}(\overline{\Omega}_{C_t} - \overline{\Omega}_{X_t})$, and $ms_{X_t}(cov(\Omega_{C_t}) - cov(\Omega_{X_t}))$.

Paper II

Productivity Dynamics, R&D, and Competitive Pressure^{*}

Florin G. Maican[†]

Draft: April 30, 2010

Abstract

This paper proposes a dynamic structural model to estimate productivity when productivity evolves as an endogenous process and firms decide how much to invest depending on the competitive pressure they face. Using data from Sweden, this paper finds that open market policies and entrepreneurship policies complement R&D policies and are important drivers of the competitiveness of established firms. Conservative estimates suggest that optimal investment is at least 0.7 to 2.5 times the actual investment in R&D for a median firm and 2 to 4 times for a firm located in the upper part of the productivity growth distribution.

Keywords: R&D; productivity; production function; selection; competitive pressure; market dynamics.

JEL Classification: O3, C51, L11, L13, D24.

^{*}I would like to thank Lennart Hjalmarsson, Magnus Henrekson, Cristian Huse, Georg Licht, Per Lundborg, Anton Nivorozhkin, Matilda Orth, Lars Persson, Maria Risberg, Rune Stenbacka, Roger Svensson, and Richard Sweeney for comments and suggestions. I would also like to thank the team Trade Union Institute for Economic Research (FIEF) for data access and seminar participants at the University of Gothenburg; Research Institute of Industrial Economics, Stockholm; Conference of the International J. A. Schumpeter Society (Nice); and EARIE (Amsterdam); Knowledge for Growth: Role and Dynamics of Corporate R&D (Seville).

[†]Box 640 SE 405 30, Göteborg, Sweden, Phone +46-31-786 4866, Fax: +46-31-786 4154, E-mail: florin.maican@handels.gu.se

1 Introduction

The link between investment in research and development (R&D) and firm performance is one of the most studied topics in industrial organization. Early literature on this relationship largely focused on estimating the average or expected returns (private or social) to R&D spending.¹ However, even if R&D spending increases a firm's productivity, it also affects the entire productivity distribution of the industry through the exit of firms and reallocations as well as displacements of labor and capital. From a policy perspective, the analysis of the entire productivity distribution enhances our understanding of the dynamics of firms' investment in R&D and physical capital.² The paper's objective is to investigate the impact of R&D spending and competitive pressure on the industry-wide distribution of productivity.

The analysis is based on a dynamic model that allows for the effect of competitive pressure on R&D spending and productivity. The model is an extension of Olley and Pakes (1996)' (OP) semiparametric framework for estimating production function parameters, which accounts for the selection induced by liquidation as well as for the simultaneity induced by the endogeneity of input demands. Recent production function estimation studies extend the OP framework by endogenizing productivity. For example, Buettner (2004) extends the OP method by allowing the distribution of future productivity to evolve endogenously over time - a firm's R&D spending affects the distribution of future productivity conditional on the current productivity. Ackerberg et al. (2007) (ABBP) suggest introducing a technological indicator, i.e., they introduce two Markov processes: one controlled and one exogenous. Muendler (2005) suggests that firm-level capital investment interacted with sector-level competition variables is a superior model to capture a firm's individual market expectations and to correct for transmission bias. Aw et al. (2009) and Doraszelski and Jaumandreu (2009) also endogenize productivity, allowing it to depend on the amount of R&D investment.

In many industries firms engage in R&D with the aim of improving future productivity, and they decide how much to spend depending on the competitive pressure they face. If one believes that the true underlying model of firm dynamics should include R&D spending and competitive pressure, then without an explicit

¹Griliches (1998) provides a survey of the effect of R&D on productivity.

²In the theoretical firm dynamics models proposed by Ericson and Pakes (1995), Hopenhayn (1992), and Jovanovic (1982), the stochastic evolution of firm productivity determines the success or failure of the firm in an industry.

model it is unclear whether a framework with exogenous productivity process approach can be applied. A further improvement to previous work is that the present paper explicitly incorporate the effect of competitive pressure on productivity and its link to R&D and discuss the identification when a dynamic framework is used.

Does competition affect productivity? In the Schumpeterian view causality between R&D and market structure goes in both directions.³ Although there exists a theoretical basis for the conjecture that competition enhances productivity. the empirical evidence is somewhat ambiguous. Aghion et al. (2005) develop a theoretical growth model where competition may increase the incremental profit from innovation and reduce the innovation incentive for firms with low productivity. Using U.K. firm data, they find an inverted U-shape between innovation and competition, i.e., increasing competition has a positive impact on innovation at low levels of competition but a negative impact when competition is already high.⁴ However, their findings of a positive net impact of competition on innovation are in line with the previous literature. Geroski (1990) and Nickell (1996) find empirical evidence that increases in competition are good for innovation. Comparing firms' productivity, R&D investment, and survival in the same industry in Korea and Taiwan, Aw et al. (2003) emphasize selection effects based on productivity induced by the high competitive pressure in Taiwan, e.g., a less concentrated market structure and low dispersion in productivity among survivors, and explain the low productivity of Korean firms by lack of entry and exit. The impact of competition on productivity is also emphasized by Syverson (2004a,b), who analyzes how product substitutability (demand-side aspects) affects performance and market structure in the U.S. ready-mixed concrete industry. He finds that an increase in product substitutability, i.e., an increase in competitive pressure, increases median productivity and decreases productivity dispersion in the market.

Competitive pressure faced by firms affects their choice of R&D and then both R&D and competitive pressure influence the stochastic evolution of a firm's productivity. In my setting the decision to invest and how much to invest in R&D and

 $^{^{3}}$ Vives (2004) analyzes the effects of competition on R&D effort for a variety of market structures. His findings are: an increasing number of firms tend to reduce R&D spending-provided that the total market for varieties does not shrink; an increasing total market size increases both the R&D effort and the number of varieties.

⁴Analyzing the effects of competitive pressure on a firm's incentive to invest in product and process innovations, Boone (2000) derives the conditions under which a rise in competitive pressure increases each firm's investments in process innovations to improve efficiency. He finds that the effects of a rise in competitive pressure on firm's incentive to innovate depend on the firm's type, which is determined by its efficiency level relative to that of its opponents.

physical capital depends on the competitive pressure faced by firms. In the OP setting two firms with the same current productivity and different capital stock will have different distributions of future productivity, while in the Buettner (2004) setting, current capital influences R&D spending, which affects future productivity. The present paper endogenizes the productivity process highlighting two channels through which competitive pressure affects productivity. It shows that under few restrictions on the model primitives, the policy function for capital investments generated by the structural model is still invertible (Pakes, 1994). The unobserved productivity state can be expressed as a function of capital, investment, and competitive pressure. The endogenous productivity choice model justifies the retention of observations with non-positive investment when competitive pressure is included. I use four measures, computed using five-digit information, as proxies for competitive pressure: the number of small (fewer than 100 employees) firms, median R&D spending at the industry level, change in concentration (C4), and foreign demand, i.e., total sales to foreign firms.

Accounting for competitive pressure when estimate productivity, the paper also links to the recent trade literature. There is a well documented positive correlation between productivity and export market participation. Costantini and Melitz (2007) provide a theoretical dynamic model of firm-level adjustment to trade liberalization modeling the joint entry, exit, export, and innovation decisions of heterogeneous firms. Their model captures the following channels for productivity improvements: (i) the selection effect of more productive firms into export markets and (ii) the effect of trade on productivity resulting in improvements in firms' productivity.⁵ Recent empirical studies point out that R&D investment and access to new technology increase productivity as well as the pay-off to exporting (Aw et al., 2008). Investment decisions depend on the expected future profitability and fixed and sunk costs. Aw et al. (2009) provide a dynamic structural model of firms' decision to invest in R&D and to export.⁶ My model framework does not explicitly model the choice of R&D, it does it in an indirect way. Jones and

⁵They find that anticipation of liberalization and a gradual path of liberalization induce firms to innovate ahead of export market entry. Using a general equilibrium model of the decision to innovate and export, Atkeson and Burstein (2007) find analytically that a decline in marginal trade costs raises process innovation (higher productivity) in exporting firms relative to non-exporting firms (comparative advantages).

⁶Using Taiwanese plant-level data, they find that (i) self-selection of high productive plants is the dominant channel driving participation in the export market and R&D investment, and that (ii) both R&D investment and exporting have a positive direct effect on a plant's future productivity.

Williams (1998) link the gap between the recent growth literature and the empirical productivity literature by constructing a relationship between social rate of return to R&D and the coefficient estimates of the productivity literature. They show that the latter represent a lower bound.

My model is applied to three Swedish manufacturing industries. In an international perspective, Swedish firms are big spenders on R&D.⁷ Little work, however, has been done on the impact of R&D spending on the distribution of firm performance in Sweden.⁸ The data used covers 1996-2002, a period of significant adjustment, and include all Swedish firms in three important manufacturing industries: machinery and equipment (MME), electrical and optical equipment (EOE), and transport equipment (MTE). The comprehensive nature of the data allows analysis of the dynamics of small plants that are often unobserved due to data limitations. I find that both selection bias and simultaneity bias induced by firm dynamics affect the magnitude of the capital coefficient in the value-added generating function. Structural dynamic models with R&D and capital investments based on the value added-generating function approach neglect competitive pressure which may lead to inconsistent coefficient estimates. A failure to adequately account for the dynamics of non-technical labor or/and technical labor can lead to severe under-estimations of capital stock when endogenizing productivity process (Ackerberg et al., 2006). Since the measure of productivity depends on these estimates, their consistency is crucial for the analysis.

My analysis yields several important findings. First, I find support for productivity improvements related to R&D and competitive pressure. Second, not endogenizing productivity when accounting for competitive pressure might result in high rates of return to R&D, which implies underinvestment. I find that a positive change in concentration has a negative effect on firms' productivity growth, and this effect is larger in the upper part of the productivity growth distribution. On the other hand, my results indicate that an increase in the number of small firms (fewer than 100 employees) has a positive impact on all parts of productivity distribution in the MME and EOE industries. Therefore, my findings suggest that entrepreneurship policies can complement policies that promote R&D spending.

⁷There are sixteen Swedish manufacturing firms in the 2009 EU Scoreboard rank, which is a list of the EU-1000 group of firms ranked by their R&D spending in the 2008 financial year (EU, 2009). R&D investments were around 4 percent of GDP in 2001. Even if Sweden is ranked in the top in terms of national R&D intensity, the R&D content of Swedish production was found to be low in previous studies (Blomström and Kokko, 1994).

⁸Svensson (2008) provides a survey of the research on R&D in Sweden.

Foreign market penetration has a positive impact on productivity growth only for MME and MTE firms that are located in the upper tail of the productivity growth distribution. An increase in the median R&D spending at the subsector level has a positive impact only for median growth firms, i.e., firms that have very low or high productivity growth are not affected by more R&D spending at the subsector level. Third, the paper finds that the aggregate productivity gains from 1996 to 2002 are around 8 percent in the MME industry and around 22 percent in the EOE and MTE industries. The continuing firms that have increased both their productivity and market share are responsible for most of the productivity growth in the MME industry. These firms also contributed to the productivity growth in the EOE industry, where the entrants have a contribution of about 8 percent. In the MTE industry, the productivity growth is driven by the continuing firms that have increased their productivity. Fourth, the study finds that the private rate of return to R&D depends on the firm's location in the productivity growth distribution. Looking at the median firms, there is a rate of return to R&D around 20 percent in the MME industry, around 10 percent in the EOE industry, and around 21 percent in the MTE industry. The firms in the upper part of the distribution have a higher private rate of return, which implies underinvestment. Fifth, the paper tries to find whether the chosen manufacturing industries engage too much or too little in R&D. Using Jones and Williams (1998) relation between social rate of return to R&D and productivity estimates, I find that optimal R&D investment for a median firm is at least around 1.3 to 2.5 times the actual spending in the MME and MTE industries, and at least around 0.7 to 1.3 in the EOE industry. The ratio between optimal and actual R&D spending is higher for firms located in the upper part of the productivity growth distribution: at least 2 to 4 times the actual spending in the MME and the MTE industries, and 1 to 2 times in the EOE industry.⁹ My results also suggest that an estimate of average rate of return to R&D might give an under-evaluation of the actual investment.

In the reminder of this paper, Section 2 describes the data and presents an overview of three Swedish industries, and documents some changes in their structures. The dynamic modeling framework used to compute productivity is outlined in Section 3, while Section 4 discusses econometric implementation. Section 5 presents results of productivity estimation and rates of return to R&D. It also dis-

 $^{^{9}}$ Analyzing the Swedish multinational plants, Fors (1997) finds that around four-fifths of the value added is attributed to home R&D and R&D in foreign plants seems to not be used as input in home plants.

cusses the optimal R&D investment and identifies the factors behind productivity growth at the industry level. Section 6 summarizes and concludes the paper.

2 Overview of the Industries

This section provides an overview of the selected industries and helps motivate the empirical strategy. The empirical strategy was chosen based on the information provided by entries, exits, and R&D-to-sales ratios.

Data. The paper draws on a census of all Swedish manufacturing firms provided by Statistics Sweden, Financial Statistics(FS) and Regional Labor Statistics(RAMS). While FS contains annual information about firm input and output, RAMS contains annual information on employee education and wages. The panel data set covers the period from 1996-2002 belonging to Swedish Standard Industrial Classification (SNI) code 29 ("Manufacture of machinery and equipment"), codes 30-33 ("Manufacture of electrical and optical equipment"), and codes 34-35 ("Manufacture of transport equipment").¹⁰ The unit of observation is a firm; over 99 percent of the firms are single-plant establishments. Appendix A gives more information about the data as well as variable definitions.

Table 1 presents characteristics of the chosen manufacturing industries. The MME industry is the largest, and MTE is the smallest. In all industries, the largest amount of R&D spending occurred after 2000. In 2000 and 2001 the Swedish economy had entered a cyclical downturn. The slowdown was partially explained by weaker international demand. Another impact on the Swedish economy during this period was the bursting of the IT bubble on the stock exchanges.

International companies like Atlas Copco (mining and construction equipment) and Tetra Laval (liquid food packaging and dairy equipment) dominate the MME industry. In 2002, the industry produced a value added of SEK 47.6 billion, employed 87,741 people in Sweden, and spent SEK 4.6 billion on R&D. The EOE industry is dominated by international companies like ABB (power and automation equipment) and Electrolux (appliances). In 2002, this industry produced a value added of SEK 30.3 billion, employed 86,156 people in Sweden and spent SEK 34.7 billion on R&D. The MTE industry is one of the most important manufacturing industries in Sweden. It includes cars, trucks and buses, aircraft, trains, and

 $^{^{10}{\}rm The}$ SNI standard builds on the Statistical Classification of Economic Activities in the European Community (NACE). The SNI standard is maintained by Statistics Sweden (http://www.scb.se).

marine and aircraft engines. Volvo, Saab, and Scania dominate final vehicle assembly. The existence of a large number of subcontractors underscores the importance of the industry. In the past few years, the industry has undergone rapid restructuring. The Volvo trademark is used by two separate companies: Volvo Group, a manufacturer of construction and farm machinery as well as heavy trucks and Volvo Cars, a manufacturer of automobiles owned by Ford Motor Company since 1998. In 1989, General Motors (GM) acquired 50 percent of Saab Automobile AB, the second manufacturer of automobiles (Investor AB controlled 50 percent). GM acquired the remaining Saab shares in 2000, turning the company into a whollyowned subsidiary. Subcontractors suffer from extensive restructuring as well since final vehicle makers tend to cut down on the number of suppliers when introducing new models. In 2002, the industry produced a value added of SEK 46.9 billion, employed 91,474 people in Sweden and spent SEK 24.2 billion on R&D. This industry was seriously affected by the global downturn in 2008. While a job in this industry was not long ago considered to be a secure position this changed in 2009. Entry. Table 2 presents an analysis of the entrants in all three industries. Around 8 percent of the firms active in 2001 in the MME industry entered in 1973 or before, and accounted for 31 percent of the technical employment and 63 percent of the industry's R&D spending in 2001. Of the firms active in 2001, the proportion that entered after 1996 is constant around 3 percent. Their share of R&D spending is smaller than 1 percent after 1997. In the EOE industry, around 2 percent of the firms active in 2001 entered in the 1980s, and they account for 33 percent of the sales and 60 percent of R&D spending. The highest share of the industry's employment in 2001, 20 percent, is linked to firms that entered in 1997. These firms have 7 percent of the technical employment and around 1.5 percent of the industry's R&D spending in 2001. Around 15 percent of the MTE industry's R&D spending in 2001 cames from firms that entered in 1983 or earlier. The firms that entered after 1997 and that were still active in 2001 seem to not be R&D incubators since they have almost no R&D spending in 2001. They might be subsidiaries of the larger firms in this industry. In all three industries, the high share of 1996 entry firms that were active in 2001 is due to the sample selection prior to 1996. Most of the post-1996 entrants in the database are small firms, accounting for no more than 8 percent of all employment in 2001. The MTE industry is the only industry where the large share of R&D spending does not imply a large share of technical employment.

Exit. Table 3 provides information about the exit process. Exit seems to play

an important role in the adjustment process after 1996. Around 29 percent of the firms in all industries that were active in 1997 did not survive until 2001. These firms spent about 20 percent of the 1997 R&D and produced about 30 percent of the 1997 output in the MME and the EOE industries. The lowest amount of R&D spent in 2000 by firms that did not survive until 2001 was in the MTE industry (1 percent of the 2000 R&D spending). Most likely, those firms are subcontractors of large firms.

R&D spending. Table 4 shows the evolution of R&D-to-sales ratios for firms with sales below and above the median, respectively. The scale effect (R&D-to-sales ratios) analysis gives us information about the advantages of the industry newcomers. In all three industries, firms that are larger than the median (based on sales) tend to spend a higher share of sales on R&D than those that are smaller than the median, except for in 2001, when the opposite occurred. The EOE and the MTE industries differ largely in spending. In the MTE industry, the larger-than-median firms spent more than double the proportion of sales on R&D than the smaller-than-median ones. In the EOE industry, firms spent more than double the proportion of sales on R&D than in the MME industry.

What does high R&D spending yield in the three Swedish industries? Spending more does not necessarily help, while spending too little will hurt. My data emphasize that R&D budget levels vary substantially, even within sub-industries. There is not one specific approach to spending money on innovation development, but there are some successful stories in the discussed industries. The aim of the paper is to investigate whether there exists a statistical relation between R&D spending and future productivity at the industry level.

3 Modeling Framework

This section presents the structure of the behavioral model of firms. I begin by introducing the assumptions and structural properties of the stochastic dynamic model, and then derive theoretical results that justify the empirical work in the rest of the paper. I assume a stochastic dynamic single-agent model for the industry. A firm maximizes the expected discounted value of future net cash flows. Firm's state variables are productivity, ω , capital stock, k, and competitive pressure, θ .

The dynamic model is formulated by the following Bellman equation with the
discount factor β ($\beta < 1$):¹¹

(1)

$$V(\omega,k,\theta) = max \left\{ \phi, \sup_{\psi',i} \left[\pi(\omega,k,\theta) - c(i,k) - z(\psi',\omega) + \beta \int V(\omega',k',\theta') P(d\omega'|\psi',\theta) \right] \right\}$$

where (ω', k', θ') denotes the random next-period state, where the probabilities associated with the next-period state are conditioned on the starting state (ω, k, θ) and choosing action (ψ', i) ; $z(\psi', \omega)$ is the R&D spending function; c(i, k) is the cost of physical capital; and i is the investment choice of the firm.

The firm makes a discrete decision whether to exit or stay in business after observing its state variables at the beginning of each period. If it exits, the firm receives a termination value, ϕ . If the firm stays in business, it earns the net profit $r(\tilde{\psi}, \tilde{i}, \omega, k, \theta) = \pi(\omega, k, \theta) - c(i, k) - z(\psi', \omega)$ in state (ω, k, θ) when action $(\tilde{\psi}, \tilde{i})$ is selected. The action represents the choice of the next period's productivity through R&D spending, ψ' , and the decision to invest in capital, *i*.

I assume that the profit function, $\pi(\omega, k, \theta)$, is bounded from above; nondecreasing in ω and k; strict supermodular in (ω, k) and (ω, θ) ; and continuously differentiable. A rise in competitive pressure truncates the distribution of profits, i.e., low profitable firms exit and surviving firms increase their marginal profits. The firm adapts to increased competitive pressure by raising its productivity. The cost of physical capital c(i, k) is bounded from below; non-decreasing in i and decreasing in k; submodular; and continuously differentiable. The R&D spending function $z(\psi', \omega)$ is non-negative, non-decreasing in ψ' and decreasing in ω ; submodular; and strictly submodular on the set $\{(\psi', \omega, \theta) | z(\psi', \omega) > 0\}$.

Investment in capital has a deterministic effect on future capital stock. Spending on R&D influences future productivity stochastically. Both investments depend on competitive pressure, θ . In each period, the firm chooses how much to invest in capital stock (and indirectly in the next period's capital stock), the quantity of intermediate inputs, labor, and distribution of the next period's productivity through its R&D spending. The accumulation equation for capital is given by

 $k' = (1 - \delta)k + i,$ ¹¹See Ericson and Pakes (1995).

where k' denotes the next period's capital stock; δ is the rate of capital depreciation; and *i* is the investment choice of the firm at the beginning of the current period that enters in capital stock at the end of the current period.

The firm invests in R&D to improve its productivity in the future years, yet the outcomes of the research process are uncertain. The distribution of future productivity ω' conditional on information at time t depends on actual productivity, ω ; R&D spending, z; physical investment in capital, i; and competitive pressure, θ . R&D spending and current productivity affect the distribution of future productivity only through a single index, $\psi' = (\omega, z)$ (Buettner, 2004). For simplicity we introduce a single index, $\psi' = (\omega, z)$, which implies that both productivity and R&D spending affect the distribution for ω' only through ψ' . The productivity process $\{\omega\}$ is a controlled first order Markov process and its primitives are given by the family of conditional distributions,

$$\mathbb{P} = \{ P(\cdot | \psi', \theta, i), (\psi', \theta, i) \in \Psi \times \Theta \times \mathbb{R}_+ \subset \mathbb{R}^3 \}$$

The family \mathbb{P} is assumed be stochastically increasing in *i* for each value (ψ', θ) (increases in investment lead to better, in a stochastic dominance sense, distribution for future efficiency), stochastically increasing in ψ' for each given (θ, i) (conditional on *i*, the higher the choice ψ' , the better the distribution of tomorrow's ω). It is also assumed to be stochastically increasing in θ for each given (ψ', i) (conditional on *i* and ψ' , the higher pressure θ the better the distribution of tomorrow's ω), and continuous in the sense that when integrated against a continuous bounded function of ω' , it produces a continuous bounded function of *i*, θ , and ψ' .

The return to R&D is uncertain, and the probability distribution over future productivity is parametrized by competitive pressure, θ . Competitive pressure, θ , indexes the *sensitivity* of the probability distribution to future distribution choice ψ' : higher values of θ correspond to probability distributions where future distribution choice is more effective at shifting probability weights towards high realizations of productivity.

The optimal policies of exit, investment, and choice distribution of future productivity are $\{\tilde{\chi}(\omega, k, \theta), \tilde{i}(\omega, k, \theta), \tilde{\psi}(\omega, k, \theta)\}$. Solving the dynamic model, we obtain the following optimal policy functions:

(2) Exit rule:
$$\chi' = \tilde{\chi}(\omega, k, \theta) = \begin{cases} 1 & (\text{continue}) \text{ if } \omega \ge \underline{\omega}(k, \theta) \\ 0 & (\text{exit}) \text{ otherwise} \end{cases}$$

- (3) Physical investment choice: $i = \tilde{i}(\omega, k, \theta)$
- (4) Distribution choice: $\psi' = \tilde{\psi}(\omega, k, \theta)$

The function $\underline{\omega}(k,\theta)$ denotes the threshold productivity. For each capital stock, k, and competition pressure, θ , there exists an exit threshold productivity: if the value of productivity is below $\underline{\omega}(k,\theta)$, then the firm exits; otherwise it stays in business.

Competitive pressure affects the investment demand function and R&D spending. Pakes (1994) and Buettner (2004) prove the monotonicity for physical investment function in a model that does not allow for the effect of competitive pressure. This present paper takes the next step and demonstrates that the optimal physical investment choice is non-decreasing in choice of distribution, capital stock, and competitive pressure.

Lemma 1 The value function $V(\omega, k, \theta)$ is bounded above, non-decreasing in productivity ω and capital k, supermodular in (ω, k) and (ω, θ) , and unique.

Proof: see appendix B.

Lemma 2 The optimal **physical investment choice** conditional on ψ' , k, and θ

$$\tilde{i}(\psi',k,\theta) = \arg \sup_{i} \left[-c(i,k) + \beta \int V(\omega',k',\theta') P(d\omega'|\psi',\theta) \right]$$

is non-decreasing in ψ' , k, and θ .

Proof: see appendix B.

Lemma 3 The policy function for the choice of distribution

$$\tilde{\psi}(\omega,k,\theta) = \arg \sup_{\psi'} \left[\pi(\omega,k,\theta) - c(\tilde{i}(\psi',k,\theta),k) - z(\psi',\omega) + \beta \int V(\omega',k',\theta') P(d\omega'|\psi',\theta) \right]$$

is non-decreasing in ω and strictly non-decreasing in ω on the sets

$$\left\{(\omega,k,\theta)|z(\tilde{\psi}'(\omega,k,\theta),\omega)>0\right\}\cup\left\{(\omega,k,\theta)|\pi(\omega,k,\theta)\quad is \ supermodular \ in \quad (\omega,\theta)\right\}.$$

Proof: see appendix B.

Theorem 1 The policy function for the **investment choice** $\tilde{i}(\omega, k, \theta) = \tilde{i}(\tilde{\psi}(\omega, k, \theta), k, \theta)$ is non-decreasing in ω and strictly non-decreasing in ω on the sets

$$\left\{ (\omega, k, \theta) | \tilde{i}(\omega, k, \theta) > 0 \land z(\tilde{\psi}(\omega, k, \theta), \omega) > 0 \right\} \cup \\ \left\{ (\omega, k, \theta) | \pi(\omega, k, \theta) \quad is \ supermodular \ in \quad (\omega, \theta) \right\}.$$

Proof: see appendix B.

The results from Theorem 1 suggest that the data with zero physical investment can be used when controlling for competitive pressure. Muendler (2005) finds the same result, but he uses a particular dynamic framework with a quadratic adjustment cost including fixed adjustment cost, but without R&D data. My theoretical results indicate that the productivity function is strictly non-decreasing when competitive pressure increases and the firm invests in R&D. Syverson (2004a) argues that demand-side features also play a role in creating the observed productivity variation. Investigating the effect of spatial substitutability on productivity distribution in the US cement industry, he finds that increases in substitutability truncate the productivity levels, and less productivity dispersion. Increasing product substitutability can be seen as in an increase in competitive pressure.

Endogenous productivity and competition. This study extends the OP framework by endogenizing the productivity process (ABBP; and Doraszelski and Jaumandreu, 2009). I propose an extension of previous estimators, including the effect of competitive pressure on R&D spending and on physical investment. While ABBP and Doraszelski and Jaumandreu (2009) discuss endogeneity of the productivity process, their proposed estimators omit the link between productivity and competition affects productivity, emphasizing both positive and negative effects theoretically, but often positive effects empirically. Recent theoretical contributions are Nickell (1996), Schmidt (1997), Boone (2000), Melitz (2003), and Raith (2003); whereas recent empirical contributions include Porter (1990), MacDonald (1994), Nickell (1996), Blundell et al. (1999), Sivadasan (2004), Syverson (2004a), and Aghion et al. (2009).

4 Productivity estimation

This section discusses the estimation of a value-added generating function including competitive pressure when the strict monotonicity of the optimal investment or intermediate inputs choice is used to recover the parameters of this function (Olley and Pakes, 1996).

Value-added generating function. I assume a Cobb Douglas production technology:

(5)
$$Q_j = A_j K_j^{\beta_k} L_j^{\beta_l},$$

where Q_j is physical output, K_j is capital stock, and L_j is labor. The variable A_j represents the Hicksian neutral efficiency level of firm j, and it is not observed by the econometrician. The physical output Q_{jt} is not observed and is usually replaced by deflated value-added or sales using an industry price deflator. Taking the natural logs in expression (5) and indexing my variables by time t yield

(6)
$$y_{jt} = \beta_0 + \beta_k k_{jt} + \beta_l l_{jt} + \varepsilon_{jt}$$

where lowercase symbols represent natural logs of variables and $\ln A_{jt} = \beta_0 + \varepsilon_{jt}$, and y_{jt} represents the deflated value-added. The coefficient β_0 is the mean efficiency level across firms, and ε_{jt} is the deviation from that mean for firm j in period t. The unobserved ε_{jt} is divided into two components: ω_{jt} and η_{jt} . The unobservables that are neither observed nor predictable by the firm before its input and exit decisions at time t are represented by η_{jt} . The component ω_{jt} is observed by the firm when it chooses inputs or makes exit decisions. It represents unobserved productivity, and the endogeneity problems are consolidated into it and not into η_{jt} . The component η_{jt} represents either a serially uncorrelated additional productivity shock or a measurement error that can be serially correlated. Output, input factors, productivity, and error terms are time and firm specific. Value-added generating function coefficients are constant across time and firms.

Ackerberg et al. (2007) discuss various approaches that can be used to solve the bias problem in estimation of value-added generating function: fixed effects, or the Blundell and Bond (2000) instrumental variable approach, and the control function approach used in the OP framework. The OP estimation framework solves the problem of firm-specific time-varying unobserved productivity in the estimation of the production function.

Marschak and Andrew (1944) point out that the endogeneity of input choices might cause problems in estimation of the value-added generating function (6). On the one hand, highly productive firms invest more in physical capital, and the future capital stock is positively correlated with ω_{jt} . On the other hand, highly productive firms have higher employment conditional on capital because they have a higher marginal product of labor in (6). Selection of firms through exit is another source of bias. A firm optimally decides to exit when its productivity is less than its exit threshold, which is a function of capital stock and competitive pressure. The exit threshold is decreasing in capital because firm's profit is strictly increasing in capital. Firms with large capital stock might operate even if they are not productive. It follows that the lower bound of the range of productivity realizations for surviving firms in the data is decreasing in capital. Therefore, average productivity among survivors is decreasing in the capital stock leading to a downward bias in the capital coefficient. Another source of bias is *omitted price* variable bias (Klette and Griliches, 1996). If the firm has some pricing power, then the estimates of (β_k, β_l) will be biased since the amount of inputs used might be correlated with the price a firm charges. First, firm-level price deviation from the industry-wide price is captured in the error term. If this price variation is correlated with the inputs, the estimated coefficients will be biased. Intermediate inputs and labor are negatively correlated with the unobserved price, yielding a downward bias in intermediate inputs and labor coefficients. Omitted price bias works in a direction opposite that of a simultaneity bias, making any prior on the direction of the bias difficult. This paper compares estimators based on different identification methods. I do not control explicitly for omitted price bias. To control for unobserved prices is straightforward by including as simple demand system (Klette and Griliches, 1996; De Loecker, 2009; Melitz, 2000; Maican and Orth, 2009; and Foster et al., 2008). In the empirical section, the paper controls for this bias in an indirect way by accounting for competitive pressure.

Timing assumptions. ABBP provide a detailed discussion about the assumptions needed to estimate a production function using the OP framework (control function approach). Three types of assumptions are important in this approach. First, there is an assumption that refers to the points in time when inputs are chosen by the firm relative to when they are used to generate value-added. Second, there is a scalar assumption that limits the dimensionality of the econometric unobservables that impact firm behavior. The third assumption is a strict monotonicity on the investment demand or on one of the intermediate inputs choice

demand-investment (intermediate inputs choice) is strictly monotonic in the scalar unobservable for a firm whose investment (intermediate inputs choice) is strictly positive.¹²

At the beginning of each period t, the firm observes its state variables: productivity state, ω_{jt} ; the capital stock, k_{jt} ; and competitive pressure, θ_{jt} . Then it decides whether it to stay in business or exit. If it stays in business, it then decides the levels of investment in capital, intermediate inputs, how much of the variable factor labor to employ, and R&D spending given competitive pressure, θ_{jt} . The shock η_{jt} is realized after those choices are made. Thus, labor l_{jt} responds to the productivity ω_{jt} , but is uncorrelated with the error term η_{jt} . The physical investment decision in period t is made in period t - 1 based on the information available, i.e., the productivity ω_{jt-1} and the values of the competitive pressure θ_{jt-1} .¹³ In other words, actual capital, k_{jt} , is not affected by current productivity shocks, ξ_{jt} . In addition, η_{jt} is uncorrelated with k_{jt} and with previous η 's.

Identification. Exploiting the monotonicity property of the investment function, unobserved productivity is a function of the current investment, i_{jt} , actual capital stock, k_{jt} , and competitive pressure, θ_{jt} :

(7) $\omega_{jt} = \tilde{\omega}_t(i_{jt}, k_{jt}, \theta_{jt}),$

where the functional form $\tilde{\omega}_t(\cdot)$ is unknown, i.e., it depends in a complex way on all the primitives of the structural model. If the investment demand function is not invertible in productivity, then another way to back out productivity is from the competitive pressure function. Competitive pressure faced by a firm in the market is a function of its productivity, investment, and capital stock, i.e., $\theta_{jt} = \tilde{\theta}_{jt}(\omega_{jt}, k_{jt}, i_{jt})$. Thus, productivity is given by $\omega_{jt} = \tilde{\theta}_t^{-1}(i_{jt}, k_{jt}, \theta_{jt})$. To avoid the collinearity problems discussed in Ackerberg et al. (2006) (ACF), I assume that labor has dynamic implications (part of state space).¹⁴ Thus, labor l_{jt} is also an unknown function of the state variables and the proxy variable (investment and/or

 $^{^{12} \}rm While OP$ assumes strict monotonicity of the investment demand function, Levinsohn and Petrin (2003) assume strict monotonicity of intermediate inputs choice demand.

¹³Levinsohn and Petrin (2003) present a detailed discussion on the timing of data collection and of the actual investment decisions. Those details are not known in my case, but the investment decision affects the proxy's implementation. Current investment is ordered before the productivity shock in period t is known.

¹⁴This assumption can be tested, however.

competition in this case):

(8)
$$l_{jt} = l_t(\omega_{jt}, k_{jt}, \theta_{jt}) = h_t(i_{jt}, k_{jt}, \theta_{jt}).$$

Rewriting the value-added generating function (6) yields:

(9)
$$y_{jt} = \phi_t(i_{jt}, k_{jt}, \theta_{jt}) + \eta_{jt},$$

where $\phi_t(i_{jt}, k_{jt}, \theta_{jt}) \equiv \beta_0 + \beta_l l_{jt} + \beta_k k_{jt} + \tilde{\omega}_t(i_{jt}, k_{jt}, \theta_{jt})$. The function $\phi_t(\cdot)$ combines all the dynamic variables (labor and capital), investment, and competitive pressure. An estimate of the unknown function $\phi_t(\cdot)$, denoted $\tilde{\phi}_t(\cdot)$, can be obtained from equation (9). Since labor and capital have dynamic implications, they cannot be estimated in the first stage (Robinson, 1988). The coefficients β_l and β_k are estimated in the second stage. The present paper assumes that productivity ω_{jt} follows a first-order endogenous Markov process, and that capital does not immediately respond to the innovation (shocks) in productivity, ξ_{jt} . The innovation in productivity over last period's expectation is given by $\xi_{jt} = \omega_{jt} - E[\omega_{jt}|\psi_{jt}, \theta_{jt-1}]$, where the index $\psi_{jt} = (\omega_{jt-1}, r_{jt-1})$ implies that the R&D investment and current productivity affect the distribution of future productivity ω_{jt} only through ψ_{jt} . For any value of β_k and β_l , the conditional expectation $E[\omega_{jt}|\psi_{jt}, \theta_{jt-1}]$ can be computed.

Identification of β_l and β_k depends on whether self-selection of firms through exit is a concern and whether there exists R&D investment data. Since productivity follows a Markov process, the value-added generating function can be written as

(10)
$$y_{jt} = \beta_0 + \beta_l l_{jt} + \beta_k k_{jt} + E[\omega_{jt}|\psi_{jt}, \chi_{jt} = 1, \theta_{jt-1}] + \xi_{jt} + \eta_{jt}.$$

By assumption, the choice distribution ψ_{jt} in t-1 is sufficient to characterize the distribution of ω_{jt} given the competitive pressure θ_{jt-1} .

The expected productivity conditional on survival and firms' information set at t-1 is an unknown function. However, it can be estimated by a non-parametric approach. The present paper presents the models where self-selection of firms through exit is an issue. In this case, the expectation of productivity conditional

on past information and survival becomes

$$E[\omega_{jt}|\psi_{jt},\chi_{jt}=1,\theta_{jt-1}] = \frac{\int_{\omega_j \ge \omega_{jt}} \omega_j P(d\omega_j|\psi_{jt},\theta_{jt-1})}{\int_{\omega_j \ge \omega_{jt}} P(d\omega_j|\psi_{jt},\theta_{jt-1})} \\ = \left[Pr(\chi_{jt}=1|\underline{\omega}_{jt},\psi_{jt},\theta_{jt-1})\right]^{-1} \cdot \int_{\omega_j \ge \omega_{jt}} \omega_j P(d\omega_j|\psi_{jt},\theta_{jt-1}) \\ = g(\psi_{jt},\underline{\omega}_{jt},\theta_{jt-1}) - \beta_0.$$

The bias term $g(\psi_{jt}, \underline{\omega}_{jt}, \theta_{jt-1})$ is a function of state variables because $\psi_{jt} = \tilde{\psi}(\omega_{jt-1}, k_{jt-1}, \theta_{jt-1})$ and $\underline{\omega}_{jt}(k_{jt}(k_{jt-1}), \theta_{jt})$. To control for the impact of the unobservable on selection, we need a measure of productivity ω_{jt} that makes the firm indifferent between continuing and selling off. In a model without R&D data, an estimate of the survival probability, which is a proxy for the threshold $\underline{\omega}_t$, can be obtained as follows:

$$Pr(\chi_{jt} = 1 | \underline{\omega}_{jt}, \mathfrak{F}_{jt-1}) = Pr(\chi_{jt}(\omega_{jt}, k_{jt}, \theta_{jt}) = 1 | \underline{\omega}_t(k_{jt}, \theta_{jt}) \\, \tilde{\psi}_t(\omega_{jt-1}, k_{jt-1}, \theta_{jt-1}), \theta_{jt}) \\= \tilde{pr}_{jt-1}(k_{jt}, k_{jt-1}, \theta_{jt}, \theta_{jt-1}) \equiv Pr_{jt}.$$

I obtain estimates for the survival probabilities, Pr_{jt} , by regressing survival in t on polynomial extension in k_{jt} , k_{jt-1} , l_{jt-1} , θ_{jt} , and θ_{jt-1} .¹⁵ The probability of survival is strictly decreasing in the exit threshold, $\underline{\omega}_{jt}$. This implies that the threshold $\underline{\omega}_{jt}$ can be obtained inverting the survival probability Pr_{jt} : $\underline{\omega}_{jt} = f(\psi_{jt}, Pr_{jt}, \theta_{jt-1})$.

Buettner (2004) demonstrates the invertibility of the investment policy function in an extended OP framework where productivity evolves as an endogenous Markov process assuming that R&D investment of a function of k_{jt} and ω_{jt} , i.e., $z_{jt} = z_t(\omega_{jt}, k_{jt})$. If the R&D spending is increasing in the capital stock k_{jt} , then k_{jt} can be obtained from the inversion of $z_t(\cdot)$ function. Thus, future capital stock k_{jt+1} can be inferred from k_{jt} and investment function, $i(\cdot)$, creating an identification problem. There might be empirical evidence that the invertibility of the investment function fails (Greenstreet, 2005). However, if this is the case, then it is more likely that the invertibility of the R&D spending function in k_{jt} does not hold. This paper shows that model can be identified when account for the competitive pressure faced by firms when making their investments. Doraszelski and Jaumandreu (2009) propose a model that endogenize the productivity by considering the effect of the R&D spending on productivity. Their one-step estimation model relies on the assumption that labor is a static variable, i.e., it has

¹⁵The future capital stock appears in the last expression because $\omega_{jt-1} = \tilde{\omega}(i_{jt-1}, k_{jt-1}, \theta_{jt-1})$.

no dynamic implications. This approach is problematic if there are training costs, strong union support, or more general large costs associated with hiring and/or laying off as in Sweden.

Since R&D investment takes place on a longer period of time, it is expected that labor has dynamic implications in intensive R&D industries. Based on one step estimation, the Doraszelski and Jaumandreu (2009) framework is more efficient than the two-step framework.¹⁶ Doraszelski and Jaumandreu (2009) discuss the relative merits of the parametric and non-parametric approaches (used here).

Competitive advantage has important implications for innovation. Competition makes firms invest in reducing their costs, and hence improve their productivity. This aspect is ignored in both the Buettner and Dorazelski-Jaumandreu frameworks. The present paper fills this gap. It also puts forth additional evidence on the link between productivity and competition.

(i) Use data on R&D investment

The distribution choice ψ_{jt} is obtained by inverting the R&D investment function $z_{t-1}(\psi_{jt}, \omega_{jt-1}, \theta_{jt-1})$, i.e., $\psi_{jt} = \tilde{z}_{t-1}^{-1}(z_{jt-1}, \omega_{jt-1}, \theta_{jt-1})$. In this case, the second-stage estimation becomes

(11)

$$\begin{split} y_{jt} &= \beta_k k_{jt} + \beta_l l_{jt} \\ &+ g(\tilde{z}_{t-1}^{-1}(z_{jt-1}, \omega_{jt-1}, \theta_{jt-1}), f(\tilde{z}_{t-1}^{-1}(z_{jt-1}, \omega_{jt-1}, \theta_{jt-1}), Pr_{jt}, \theta_{jt-1}), \theta_{jt-1}) \\ &+ \xi_{jt} + \eta_{jt} \\ &= \beta_k k_{jt} + \beta_l l_{jt} + \tilde{g}(\hat{\phi}_{jt-1} - \beta_k k_{jt-1} - \beta_l l_{jt-1}, z_{jt-1}, Pr_{jt}, \theta_{jt-1}) + \xi_{jt} + \eta_{jt}, \end{split}$$

where $\tilde{g}(\cdot)$ is an unknown non-parametric function in $\hat{\phi}_{jt-1} - \beta_k k_{jt-1} - \beta_l l_{jt-1}$, P_{jt} , z_{jt-1} , and θ_{jt-1} . I assume that R&D investment is uncorrelated with the error term in (11). R&D investment and the error term are correlated if R&D investment is used in the construction of the value-added measure y_t .

(ii) No R&D investment data

Buettner (2004) suggests the following way to proxy for ψ_{jt} without having to use k_{jt-1} . None of the terms is a function of k_{jt-1} and ψ_{jt} in the Bellman equation (1). I use the threshold function combined with the fact that $\psi_t =$

 $^{^{16}\}mathrm{Wooldridge}$ (2005) also suggests a one-step formulation of the OP methodology.

 $\tilde{\psi}(\omega_{jt-1}, k_{jt-1}, \theta_{jt-1})$. This yields the following equation for the second stage:

(12)
$$y_{jt} = \beta_k k_{jt} + \beta_l l_{jt} + g(\tilde{\psi}_t(\omega_{jt-1}, k_{jt}, \theta_{jt-1}), \underline{\omega}_{jt}(k_{jt}, \theta_{jt}), \theta_{jt-1}) + \xi_{jt} + \eta_{jt}$$

$$= \beta_k k_{jt} + \beta_l l_{jt} + \tilde{g}(\hat{\phi}_{jt-1} - \beta_k k_{jt-1} - \beta_l l_{jt-1}, k_{jt}, \theta_{jt}, \theta_{jt-1}) + \xi_{jt} + \eta_{jt},$$

which is the same equation as for the non-selection case. This is my preferred model to estimate productivity in the empirical part of the paper. However, I also control if firms invest in R&D. The advantage is that there is no need to do the inversion in the R&D spending function. It is less probable that competitive pressure θ_{jt} impacts a firm's choice of l_{jt} but does not impact choice of investment at t. To estimate labor in the first stage, competitive pressure θ_{jt} must bring additional variance that is independent of ω_{jt} and k_{jt} . If competitive pressure θ_{jt} is serially correlated and unobserved, it is part of the state space, and we are not able to do the inversion. If θ_{jt} is serially correlated and observed, we are able to do the inversion, but labor cannot be estimated in the first stage because of perfect collinearity.

Estimation. The residuals $\xi_{jt} + \eta_{jt}$ from equation (11) or (12) are functions of parameters $\boldsymbol{\beta}^* \equiv (\beta_l^*, \beta_k^*)$. To identify β_l and β_k , the following moment conditions can be used: $E[(\xi_{jt} + \eta_{jt})k_{jt}] = E[\xi_{jt}k_{jt}] = 0$, $E[(\xi_{jt} + \eta_{jt})k_{jt-1}] = E[\xi_{jt}k_{jt-1}] = 0$, and $E[(\xi_{jt} + \eta_{jt})l_{jt-1}] = E[\xi_{jt}l_{jt-1}] = 0$. The first two moment conditions help identify β_k . They imply that capital does not respond to the innovation in productivity ξ_{jt} . The third moment condition implies that previous labor must be uncorrelated with actual innovation in productivity. This is true because l_{jt-1} is apart of a firm's information set at t - 1 and should be uncorrelated with ξ_{jt} . Thus, we get estimates of $\boldsymbol{\beta}^* = (\hat{\beta}_k, \hat{\beta}_l)$ minimizing the GMM criterion function:

(13)
$$Q(\boldsymbol{\beta}^*) = \min_{\boldsymbol{\beta}^*} \sum_{h=1}^{\#\mathbf{w}} \left(\sum_{j} \sum_{t=T_{j_0}}^{T_{j_1}} \left(\xi_{jt} \hat{+} \eta_{jt} \right) (\boldsymbol{\beta}^*) \mathbf{w}_{jht} \right)^2,$$

where j indexes firms; h indexes the instruments; T_{j_0} and T_{j_1} index the first and ante last period that firm j is observed; and $\mathbf{w}_{jt} = \{k_{jt}, k_{jt-1}, l_{jt-1}\}$.

5 Empirical Results

This section presents empirical results from estimation of value-added generating function, summary statistics for both level and growth, the estimated rate of return to R&D investment, optimal R&D spending, and productivity decomposition at the industry level.

Value-added generating function estimation. Table 5 reports the coefficient estimates of value-added generating function based on OLS and semiparametric estimators. The semiparametric estimators that treat productivity as an exogenous process are OP, LP, and ACF (ACF-i and ACF-m). The estimators that treat productivity as an endogenous process are B-1, B-2, and B-3, as proposed by Buettner (2004). In addition to these, the paper proposes EP-i and EP-all, estimators that eliminate the potential identification problems in Buettner's estimators. In total, there are ten estimators used for each industry. While the main aim of the paper is not to compare of the different semiparametric twostep estimators in detail, I discuss the main findings and their implications for productivity level and growth in three R&D intensive Swedish manufacturing industries.¹⁷ Conditional on the estimator used, the study uses the following factors that generate value added: non-technical labor, technical labor, and capital. In the OP estimator, non-technical and technical labor are static, i.e., they are estimated in the first-stage, and productivity is recovered from an inverse investment demand function. In the LP estimator both labor variables are static but productivity is recovered from inverse demand function for materials. In the ACF estimators, non-technical and technical labor have dynamic implications and productivity is recovered from investment (ACF-i) or materials (ACF-m). I control for selection in the semiparametric estimators. The OLS and EP-all use the full sample. The three versions of the Buettner estimator (B-1, B-2, and B-3) that endogenize productivity are the following: B-1, - which uses capital stock as a proxy for R&D; B-2, which - uses previous R&D spending in the non-linear function that determines future productivity; and B-3, which controls for selection in addition to B-2. Value-added generating function estimation with R&D data might suffer from persistent unobserved shocks that vary within firms but resist treatment and cause bias (unobserved demand factors correlated over time) (Muendler, 2005).

The proposed estimators (EP-i and EP-all), which endogenize the productivity

¹⁷While I discuss only the robustness of semiparametric methods, Biesebroeck (2007) discusses the robustness of different methods used to measure productivity: index numbers, data envelopment analysis, stochastic frontiers, instrumental variables, and semiparametric estimation (OP).

process, capture the effects of R&D spending and competitive pressure on future productivity. In the EP-i estimator, labor has dynamic implications and productivity is recovered from the investment demand function, i.e., only the data with positive investment are used. The EP-all estimator is the EP-i estimator that uses all the data, i.e., including zero investment. In the theoretical part of the paper (Section 3), I show that identification is still possible when competitive pressure is included. The estimators ACF-i, B-1, and EP-i have in common that labor is estimated in the second stage. In contrast to ACF-i, the B-1 and EP-i estimators endogenize productivity allowing for R&D (in B-1 only through capital stock).

The degree of competitive pressure in a market is difficult to determine with precision, and cannot be captured by one variable (Geroski, 1990). The present paper uses four measures as a proxy for competitive pressure: (i) the number of small (fewer than 100 employees) firms, (ii) median R&D spending at the industry level, (iii) change in concentration (C4), and (iv) foreign demand, i.e., total sales to foreign firms. Foreign demand captures international demand and competition conditions as well as aggregate demand. All variables are computed using fivedigit information. The estimation takes place at the two-digit industry level for the following industries: machinery and equipment (MME), electrical and optical equipment (EOE), and transport equipment (MTE). This implies that firms producing various two-digit goods use the same factor proportions, but goods are imperfect substitutes in consumption, which can lead to different investment behavior in physical capital and in R&D within an industry. Firm differences in exposure to domestic and international competition might lead to differences in investment behavior and in productivity response to international shocks. According to theory and previous empirical findings, the coefficients on variable inputs, such as labor, should be biased upwards in the OLS estimation. But the direction of the bias on the capital coefficient is ambiguous. The estimates of the coefficients on labor and capital using semiparametric estimates move in a direction that points to successful elimination of simultaneity and selection bias (Section 4). In what follows, I discuss the estimates separately for each industry.

Panel A presents the estimates for the MME industry. When one of the twostep semiparametric estimators is used, the both labor coefficients (non-technical and technical) are lower than the OLS ones. The lowest value for non-technical labor (0.410) is obtained when productivity is endogenous and we account for competitive pressure (the EP-i estimator). Among the estimators that recover productivity from investment, EP-i also gives the lowest value for the technical labor coefficient (0.206). The low values for the coefficient of capital in OLS and the Buettner estimators (less than 0.100) indicates a possible selection (OLS) or identification problem in the Buettner estimator (Ackerberg et al., 2007; Doraszelski and Jaumandreu, 2009). The omission of controlling for aggregate demand shocks and competition in the market in the estimation is a possible explanation. The negative demand shocks and lack of competition imply a decrease in elasticity of capital, i.e., large firms stay in the market even if they are not productive; they do not face competition from new entrants due to the low demand. Accounting for selection and keeping productivity exogenous increases the capital coefficient value (e.g., 0.191 in OP). The largest capital coefficient (0.214) is obtained from the EP-i estimator, i.e., controlling for R&D spending and competitive pressure in the productivity process. There is a difference between a large low-productive firm that does not invest in R&D and one that does. Endogenizing productivity implies that productivity is not a simple first order Markov process, i.e., R&D spending might affect future productivity. R&D spending might have higher future productivity. Furthermore, comparing two incumbent firms with equal capital, the firm facing higher competitive pressure has higher productivity. Dropping observations depending on the used proxy implies sample selection; this can be observed from magnitude differences among coefficients (EP-all versus others).

Panel B in Table 5 presents the estimates for the EOE industry. The lowest value for the non-technical labor coefficient (0.442) is given by LP estimator; for technical labor (0.240) by the LP estimator; and for capital (0.110) by the B-2 and B-3 estimators. The largest value for the non-technical labor coefficient (0.545) is given by the OP estimator; for technical labor (0.307) by the OLS; and for capital (0.241) by the ACF-i estimator. There is an interesting story in the Buettner estimates, where labor has dynamic implication and productivity is recovered from investment. Controlling for R&D in the productivity process (B-2 and B-3) increases the coefficient of technical labor and decrease the coefficients of capital, i.e., an increase in R&D spending will increase the coefficient of technical labor to have dynamic implications, the labor coefficients decreases from 0.545 (OP) to 0.493 (ACF-i) and 0.469 (EP-all). The labor estimates in EP-all are close to the ACF-i estimates and the capital coefficient (0.204) is close to the OP estimate (0.209).

The value-added-generating function estimates for the MTE industry are presented in Panel C. The lowest value for the non-technical labor coefficient (0.389) is given by the EP-i estimator; for technical labor (0.165) by the LP estimator; and for capital (0.110) by the B-2 and B-3 estimators. The largest value for the non-technical labor coefficient (0.714) is given by the OLS estimator; for technical labor (0.259) by the B-2 and B-3; and for capital (0.170) by the ACF-i estimator.¹⁸ If we compare ACF-i and EP-i estimates, by endogenizing productivity in the EP-i estimator, the labor coefficients decrease and the capital coefficient increases (0.133 in ACF-i and 0.170 in EP-i).

Summarizing, I find that selection plays an important role. Allowing labor to have dynamic implications is important when endogenizing the productivity process. Over 75 percent of the observations are dropped when only data with positive R&D spending is used (the B-2 and B-3 estimators). Using the B-1 estimator, where capital is used as a proxy for the choice distribution of productivity, does not improve the estimates for capital. In addition to R&D spending, other factors, which are not captured by the model, affect the distribution of productivity. The estimated capital coefficient drops drastically to unreasonable levels when lagged positive R&D spending is introduced to control for expected productivity in the Buettner (2004) specification. On the one hand, this might be due to an endogeneity problem with respect to R&D, i.e., if R&D data is used in the construction of the value-added measure.

The present paper estimates various specifications with the competitive pressure variables mentioned earlier; however, it presents only partial results. Accounting for competitive pressure gives better estimates for capital, i.e., the capital coefficient increases and the labor coefficients move in the direction suggested by theory and previous empirical findings. Presence in foreign markets exposes firms to international competition. Facing international competition makes them invest in the latest technologies. Hence , the observed increase in the capital coefficient is expected.

Summary statistics: firms' productivity level and growth. Giving the estimated coefficients of the value-added generating function, the paper analyzes the summary statistics for both productivity level and growth in order to point out the importance of each estimator. Table 6 shows summary statistics for estimated productivity levels and growth distributions at the firm level when investment is used as a proxy for productivity. The EP-i and EP-all estimators provide the largest productivity levels, i.e, accounting for R&D spending and competitive pressure

 $^{^{18}\}mathrm{The}$ very large value of capital (0.694) from the B-1 estimator indicates an identification problem.

shifts productivity distribution to the right. The ACF-i estimator provides the smallest interquantile range (0.068) among all estimators in the MME industry, EP-i in the EOE industry, and OP and EP-all in the MTE industry (0.064).¹⁹ All estimators indicate a productivity growth of the 75th percentile of firm around 13 percent in the MME industry; 15 percent in the EOE; and around 13 percent in the MTE industry. There is a negative productivity growth for the 25th percentile firms. The paper finds mixed results for median productivity growth: only ACF-i and EP-all indicate a positive growth in the MME industry, and only EP-i and EP-all in the EOE industry. All estimators show a positive median firm productivity growth in the MTE industry. On the one hand, those results should be interpreted with care since all statistics are calculated for the whole period from 1996 to 2002 and are influenced by the 2001 downturn. On the other hand, the findings emphasize that allowing for endogeneity in the productivity process has important implications for the firms located between the 25th and 75th percentiles, i.e., it corrects possible underevaluation of productivity growth for an exogenous productivity process.

Though early literature on R&D and productivity studied the average effect of R&D on productivity, my approach treats R&D subject to stochastic accumulation. This allows estimation of the entire conditional distribution of productivity realizations, and gives a more complete picture of the effect of R&D investment on productivity and links productivity with competition.

Effect of R&D spending on productivity growth. Using the estimated productivity, the study investigates the impact of R&D spending per value-added on the empirical distribution of productivity growth. Table 7 presents OLS and percentile regressions of productivity growth, defined as $\omega_{jt} - \omega_{jt-1}$, on R&D intensity (R&D spending/Value added) and competitive pressure (number of firms with fewer than 100 employees, change in industry concentration [C4], median/mean R&D spending at the five-digits industry level, and foreign demand):

(14)
$$\Delta \hat{\omega}_{jt} = \mu + r^P (R \& D / ValueAdded)_{jt-1} + CompetitivePressure_{t-1} \beta + \epsilon_{jt},$$

where r^P is the private rate of return to R&D and the shocks ϵ_{jt} are i.i.d. The OLS regression estimates the mean effect of R&D intensity on productivity growth, while the quantile regression estimates the effect of the conditional distribution on

 $^{^{19}\}mathrm{Interquantile}$ range is defined as the difference between the 75th percentile and the 25th percentile over the median.

different quantiles. The quantile regressions' reported standard errors are bootstrapped. I have to distinguish between the private return and the social return to R&D. In my case, I estimate the private return to R&D using firms' own shares as explanatory variables. The social return to R&D captures inter-firms technology spillovers by focusing on the industry level and alleviates measurement problems. The regressions are run on all firms since the productivity is constructed from the EP-all estimator, i.e, there is no need to control for censoring of the distribution through exit or through negative investments.

My findings, in Table 7, indicate a median rate of return to R&D of around 20 percent in the MME industry; of around 10 percent in the EOE industry; and about 21 percent in the MTE industry. I find 5 (the EOE industry) and 10 percentage point (the MME and MTE industries) higher return rates - for firms with productivity growth such that 75 percent of all productivity growth in the 75th percentile. The number of small firms has a positive impact on productivity growth in the MME and EOE industries. The magnitude of the effect on percentiles is the same for the MME industry and somewhat larger in the tails for the EOE industry. A positive change concentration has a negative impact on firms' productivity growth in the MME and MTE industries and the effect is larger in the higher percentiles.²⁰ The paper finds a positive effect of median/mean R&D spending on firms' productivity growth only at the median in the MME industry and at the mean in the EOE industry.²¹ In addition, an increase in foreign demand has a positive impact on productivity growth only for firms in the higher percentiles (the MME and MTE industries). My results are in line with previous private rate of return findings based on U.S. data (Hall, 1995) and they are robust to the method used to estimate productivity (EP-i versus EP-all).²²

Optimal R&D investment. Do manufacturing industries engage too much or too little in R&D? - The paper provides an estimate of how much private investment in research differs from optimal investment. High rates of return to R&D would suggest substantial underinvestment. Table 7 suggests that not accounting for competitive pressure might lead to an overestimated private rate of return to R&D (underinvestment), e.g., by about 4 percentage points in the MTE industry (median regression). This is important since the private rates of return

 $^{^{20}\}mathrm{The}$ results remain valid when concentration level is used.

 $^{^{21}}$ Due to the numerical problems in estimation, mean of R&D spending is used in the estimation of productivity for the EOE and MTE industries.

 $^{^{22}\}mathrm{A}$ detailed appendix that includes the estimates using different productivity methods is available from the author upon request.

are already overestimated due to the unobserved variable correlated over time (or other unobserved demand factors at the industry level). Jones and Williams (1998) emphasize that r^P represents an underestimate of the true rate of return to R&D with a maximum down-bias equal to the rate of output growth. Their argument is based on the following assumption: we allocate one unit of output from consumption to R&D today and then consume the proceeds tomorrow, i.e., we reduce the R&D tomorrow to have the subsequent stock of ideas unchanged. They define the true rate of return to R&D as the gain in consumption associated with this variation, and the optimal amount of research as the condition where the rate of return is equal to the real interest rate, r. Using a growth model, Jones and Williams (1998) show that actual rate of investment in R&D by the industry, s^{actual} , satisfies the equation $r^P = \lambda g_{\omega}/s^{actual}$, where g_{ω} is the productivity growth and λ is a parameter in the production function for new ideas and the presence of $0 < \lambda \leq 1$ may reflect duplication of effort in research process - the social marginal product of R&D may be less than the private marginal product (Jones and Williams, 1998 provide a fruitful discussion). The optimal rate of investment in R&D along a balanced growth path is $s^{optimal} = \lambda g_{\omega}(r - (1 - \lambda)g_{output})$, where g_{output} is the output growth. Therefore, the ratio of optimal investment to actual investment in R&D is

(15)
$$\frac{s^{optimal}}{s^{actual}} = \hat{r}^P / (r - (1 - \lambda)g_{output}).$$

Having the estimate of \hat{r}^P , we can compute a *lower bound* on this ratio. The denominator is no greater than the real rate of return for the economy. The yearly average real return on the stock market in Sweden was around 7.6 percent in 2000. Table 8 shows conservative estimates of the ratio $s^{optimal}/s^{actual}$ using the estimated private rates of return from the OLS and percentile regressions (Table 7). With an average rate of return of 7.6 percent, the figures indicate a ratio of about 2.5 for the MME industry, of 1.3 for the EOE industry, and of 4 for the MTE industry (using median estimates). If we double the private rates of return to 15 percent, the ratios are about 1.3 for the MME, 0.7 for the EOE industry (i.e., over-investment), and 1.4 for the MTE industry. Hence, the optimal share of resources to invest in R&D is estimated to be 2-4 times larger than the actual amount invested in the MME and MTE industries. The EOE industry is close to the optimal rate of investment. It is important to stress that those ratios are computed with the actual rates of return estimated for median firms,

i.e., they have a productivity growth that is higher than 50 percent of all firms in the industry. Using the actual rate of return to R&D estimated for firms in higher productivity growth percentiles, the conservative estimates suggest that the optimal R&D spending is at least 2-4 times the actual spending in the MME and MTE industries, and 1-2 times in the EOE industry.

Productivity decomposition. To check the importance of productivity gains stemming from the reshuffling of resources from the less to the more efficient firms, I compute aggregate industry productivity measures for each year. The aggregate industry productivity, Ω_t , is a weighted average of firms' individual productivities, ω_{jt} , with an individual firm's market share, s_{jt} . Following Foster et al. (2001), the change in industry productivity from year t_0 to year t_1 can be written as

(16)
$$\Delta\Omega_{t_0,t_1} = \sum_{j \in C_{t_0,t_1}} s_{jt_0} \Delta\omega_{jt_0,t_1} + \sum_{j \in C_{t_0,t_1}} \Delta s_{jt_0,t_1} (\omega_{jt_0} - \Omega_{t_0}) + \sum_{j \in C_{t_0,t_1}} \Delta s_{jt_0,t_1} \Delta\omega_{jt_0,t_1} + \sum_{j \in E_{t_0,t_1}} s_{jt_1} (\omega_{jt_1} - \Omega_{t_0}) - \sum_{j \in X_{mt_0,t_1}} s_{jt_0} (\omega_{jt_0} - \Omega_{t_0})$$

where Δ is the difference operator $(\Delta\Omega_{t_0,t_1} = \Omega_{t_1} - \Omega_{t_0})$; C_{t_0,t_1} is the set of continuing firms, i.e., operating both in t_0 and t_1 ; E_{t_0,t_1} is the set of entering firms, i.e., that operated in t_1 but not in t_0 ; and X_{t_0,t_1} is the set of exiting firms, i.e., that operated in t_0 but not in t_1 . The decomposition (16) consists of five terms. The first term (*Within*) is the increase in productivity when the continuing firms increase their productivity at initial market share. The second term (*Between*) is the increase in productivity when continuing firms with above-average productivity expand their market shares relative to firms with below-average productivity. The third term (*Cross*) captures the increase in productivity when continuing firms increase their market shares, while the fourth and fifth terms (*Entry* and *Exit*) are productivity increases due to entry and exit, respectively.²³ ²⁴

Table 9 shows the results of the productivity decomposition for the three in-

$$\Omega_t = \sum_j s_{jt} \omega_{jt} = \overline{\omega}_{jt} + \sum_j (s_{jt} - \overline{s}_t)(\omega_{jt} - \overline{\omega}_t),$$

²³Decomposition of productivity is based on entry and exit defined by organization number (FS-RAMS database).

²⁴Another decomposition is proposed by Olley and Pakes (1996), where the weighted aggregate measure Ω_t is decomposed into two parts: the unweighted aggregate productivity measure and the total covariance between a firm's share of the industry output and its productivity:

where the bar over a variable denotes a mean of all firms in a given year. Melitz and Polanec (2009) propose a dynamic version of OP decomposition and discuss possible bias contribution of surviving, entering, and exiting firms in widely-used decomposition methods.

dustries 1996-2000 and 1996-2002 using Foster et al. (2001). I consider the decomposition over these two periods to control for the possible effect of the bursting of the dot-com bubble on productivity growth. From 1996 to 2002, the aggregate productivity gains range from around 8 percent in the MME industry to around 22 percent in the EOE and MTE industries. The aggregate productivity growth from 1996 to 2000 is around 4 percent in the MME industry, around 59 percent in the EOE industry, and around 47 percent in the MTE industry. These results might emphasize a possible negative impact of the bursting of the dot-com bubble on productivity in the EOE industry and the MTE industries in 2001-2002. During this period, productivity growth was reduced to half in the EOE and MTE industries. Almost all productivity growth in the MME industry comes from firms that increased both productivity and market share. Net entry had a contribution of around 4 percent, mostly driven by exit (EP-all). The positive contribution of net entry compensates the negative *Within* and *Between* terms. The productivity growth in EOE was driven by continuing firms that had increased both productivity and market shares (Cross) and by continuing firms that had increased their productivity at their initial market shares (*Within*). The latter ones have a positive contribution only in 1996-2000. While the entrants contributed around 8 percent to the productivity growth, their contribution was substantial after 2000. In the MTE industry, the productivity growth from 1996 to 2002 was by the continuing firms that increased their productivity (Within and Cross). However, both contribution channels to productivity growth shrunk proportionally due to the decrease in productivity growth after 2000.

6 Discussion and conclusions

This paper proposes a dynamic structural model to estimate productivity in intensive R&D industries where competitive pressure is a key factor for investment. The model, an extension of the two-step structural technique suggested by Olley and Pakes (1996), endogenizes productivity. In an industry where competitive pressure affects both firms' R&D spending and productivity, the true underlying model of firm dynamics should explicitly account for these factors. If this is not the case, then it is unclear whether the Olley and Pakes (1996) or the Buettner (2004) approach can be applied.

This paper explores how R&D spending and competitive pressure influence the

stochastic evolution of productivity in the Swedish R&D intensive manufacturing industries: machinery and equipment, electrical and optical equipment, and transport equipment. The paper also uses different semiparametric estimators derived from the OP framework to measure productivity in the three Swedish manufacturing industries. Not accounting for competitive pressure when productivity evolves as an endogenous process results in an underestimation of productivity. This paper shows in a theoretical framework how competitive pressure and R&D spending affect firm dynamics productivity. Productivity is expressed as a function of capital, investment, and competitive pressure. The endogenous productivity choice model justifies the retention of observations with non-positive investment when competitive pressure is included.

The paper also provides an analysis of rates of return to R&D on different parts of the productivity growth distribution. The results show that by analyzing the average rates of return to R&D, the researcher might obtain upper bias estimates, which implies an underestimation of the actual investment for median firms. Furthermore, those rates are also overestimated if the researcher fails to control for competitive pressure in productivity growth regression. Using Swedish data from 1996-2002, I find evidence that R&D spending enhances performance in Swedish manufacturing industries - but the overall effect on R&D depends of actual firm productivity and market conditions (domestic and foreign). My results indicate that the optimal investment in R&D should be 2 to 4 times the actual investment (the machinery and equipment industry, and the transport and equipment industry). The actual R&D investment in the electrical and optical equipment industry is closer to the optimum for median firms, but not for firms in the upper part of the productivity growth distribution (75th percentile).

References

- ACKERBERG, D., L. BENKARD, S. BERRY, AND A. PAKES (2007): "Econometric Tools for Analyzing Market Outcomes," *Handbook of Econometrics*, 6, 4171– 4276.
- ACKERBERG, D., K. CAVES, AND G. FRASER (2006): "Structural Identification of Production Functions," Mimeo, UCLA.
- AGHION, P., N. BLOOM, R. BLUNDELL, R. GRIFFITH, AND P. HOWITT (2005): "Competition and Innovation: An Inverted U Relationship," *Quarterly Journal of Economics*, 120(2), 701–728.
- AGHION, P., R. BLUNDELL, R. GRIFFITH, P. HOWITT, AND S. PRANTL (2009):
 "The Effects of Entry on Incumbent Innovation and Productivity," *Review of Economics and Statistics*, 91(1), 20–32.
- ATHEY, S. (2000): "Characterizing Properties of Stochastic Objective Functions," mimeo, MIT and NBER.
- ATKESON, A., AND A. BURSTEIN (2007): "Innovation, Firm Dynamics, and International Trade," Discussion Paper 13326, NBER Working paper.
- Aw, B. Y., S. CHUNG, AND M. J. ROBERTS (2003): "Productivity, Output, And Failure: A Comparison Of Taiwanese And Korean Manufacturers," *Economic Journal*, 113(491), 485–510.
- Aw, B. Y., M. J. ROBERTS, AND D. Y. XU (2008): "R&D Investments, and the Evolution of Firm Productivity," American Economic Review, Papers and Proceedings, 98(2), 451–456.
 - (2009): "R&D Investment, Exporting, and Productivity Dynamics," Mimeo, The Pennsylvania State University.
- BIESEBROECK, J. V. (2007): "Robustness of Productivity Estimates," The Journal of Industrial Economics, LV, 529–569.
- BLOMSTRÖM, M., AND A. KOKKO (1994): chap. Home-country effects of Foreign-Direct Investment: Sweden', in Globerman, S. (ed.), Canadian-Based Multinationals, The Industry canada Research Series, 4., Calgary: University of Calgary Press.

- BLUNDELL, R., AND S. BOND (2000): "GMM Estimation with Persistent Panel Data: An Application to Production Functions," *Econometric Reviews*, 31(3), 549–569.
- BLUNDELL, R., R. GRIFFITH, AND J. VAN REENEN (1999): "Market Share, Market Value, and Innovation in a Panel of British Manufacturing Firms," *Review of Economic Studies*, 66(3), 529–554.
- BOONE, J. (2000): "Competitive Pressure: The Effects on Investments in Product and Process Innovation," *RAND Journal of Economics*, 31(3), 549–569.
- BUETTNER, T. (2004): "R&D and the Dynamics of Productivity," Mimeo, London School of Economics.
- COSTANTINI, J. A., AND M. J. MELITZ (2007): "The Dynamics of Firm-Level Adjustment to Trade Liberalization," Discussion paper, Mimeo, Princeton University.
- DE LOECKER, J. (2009): "Product Differentiation, Multi-Product Firms and Structural Estimation of Productivity," Mimeo, Princeton University.
- DORASZELSKI, U., AND J. JAUMANDREU (2009): "R&D and Productivity: Estimating Production Functions when Productivity is Endogenous," Mimeo, Harvard University.
- ERICSON, R., AND A. PAKES (1995): "Markov-Perfect Industry Dynamics: A Framework for Empirical Work," *Review of Economic Studies*, 62, 53–83.
- EU (2009): "Monitoring Industrial Research: The 2009 EU Industrial R&D Investment Scoreboard," Discussion paper, European Commission.
- FORS, G. (1997): "Utilization of R&D Results in the Home and Foreign Plants of Multinationals," *The Journal of Industrial Economics*, 45(3), 341–358.
- FOSTER, L., J. HALTIWANGER, AND C. KRIZAN (2001): "The Link Between Aggregate and Micro Productivity growth: Evidence from Retail Trade," NBER Working Paper 13783.
- FOSTER, L., J. HALTIWANGER, AND C. SYVERSON (2008): "Reallocation, Firm Turnover, and Efficiency: Selection on Productivity or Profitability?," *American Economic Review*, 98(1), 394–425.

- GEROSKI, P. (1990): "Innovation, Technological Opportunity, and Market Structure," Oxford Economic Papers, 42(3), 586–602.
- GREENSTREET, D. (2005): "Exploiting Sequential Learning to Estimate Establishment Level Productivity Dynamics and Decision Rules," Mimeo, University of Michigan.
- GRILICHES, Z. (1998): *R&D and Productivity*. The University of Chicago Press, Chicago.
- HALL, B. H. (1995): "The Private and Social Returns to Research and Development: What Have We Learned?," mimeo, University of California at Berkley.
- HOPENHAYN, H. A. (1992): "Entry, Exit and Firm Dynamics in Long Run Equilibrium," *Econometrica*, 60(5), 1127–1160.
- HULTEN, C., AND F. WYKOFF (1981): chap. The measurement of Economic Depreciation, in C. Hulten (ed.), *Depreciation, Inflation, and Taxation of Income* from Capital, Urban Institute Press, Washington, D.C.
- JONES, C., AND J. WILLIAMS (1998): "Measuring the Social Return to R&D," The Quarterly Journal of Economics, pp. 1119–1135.
- JOVANOVIC, B. (1982): "Selection and the Evolution of Industry," *Econometrica*, 50(5), 649–670.
- KLETTE, T., AND Z. GRILICHES (1996): "The Inconsistency of Common Scale Estimators when Output Prices are Unobserved and Endogenous," *Journal of Applied Econometrics*, 11(4), 343–361.
- LEVINSOHN, J., AND A. PETRIN (2003): "Estimating Production Functions Using Inputs to Control for Unobservables," *Review of Economic Studies*, 70(2), 317–341.
- MACDONALD, J. (1994): "Does Import Competition Force Efficient Production?," Review of Economics and Statistics, 76(4), 721–727.
- MAICAN, F., AND M. ORTH (2009): "Productivity Dynamics and the role of "Big-Box" Entrants in retailing," Working Paper no. 328, Department of Economics, University of Gothenburg.

- MARSCHAK, J., AND W. H. ANDREW (1944): "Random Simultaneous Equations and the Theory of Production," *Econometrica*, 12(3-4), 143–205.
- MELITZ, M. (2000): "Estimating Firm Level Productivity in Differentiated Product Industries," Mimeo, Harvard University.

(2003): "The Impact of Trade on Intra-Industry Reallocations and Aggregate Industry Productivity," *Econometrica*, 71(6), 1695–1725.

- MELITZ, M., AND S. POLANEC (2009): "Dynamic Olley-Pakes Decomposition with Entry and Exit," Mimeo, Princeton University.
- MUENDLER, M. (2005): "Estimating Production Functions when Productivity Change is Endogenous," mimeo, University of California, San Diego and CESifo.
- NICKELL, S. J. (1996): "Competition and Corporate Performance," Journal of Political Economy, 104(4), 724–746.
- OLLEY, S., AND A. PAKES (1996): "The Dynamics of Productivity in the Telecommunications Equipment Industry," *Econometrica*, Vol. 64(No. 6), 1263– 1297.
- PAKES, A. (1994): chap. The Estimation of Dynamic Structural Models: Problems and Prospects Part II. Mixed Continuous-Discrete Control Models and Market Interactions, Laffont, J. J. and Sims, C. eds, Advances in Econometrics: Proceedings of the 6th World Congress of the Econometric Society, Chapter 5.
- PORTER, M. (1990): The Competitive Advantage of Nations. Macmillan Press.
- RAITH, M. (2003): "Competition, Risk, and Managerial Incentives," American Economic Review, 93, 1425–1436.
- ROBINSON, P. (1988): "Root-N-Consistent Semiparametric Regression," *Econo*metrica, 56(4), 931–954.
- SCHMIDT, K. (1997): "Managerial Incentives and Product Market Competition," *Review of Economic Studies*, 64, 191–213.
- SIVADASAN, J. (2004): "Barriers to Entry and Productivity: Micro-Evidence from Indian Manufacturing Sector Reforms," mimeo, University of Chicago.

- SMITH, J. E., AND K. J. MCCARDLE (2002): "Structural Properties of Stochastic Dynamic Programs," Operations Research, 50(5), 796–809.
- SVENSSON, R. (2008): "Growth through research and development," Discussion Paper 19, Vinnova.
- SYVERSON, C. (2004a): "Market Structure and Productivity: A Concrete Example," Journal of Political Economy, 112(2), 1181–1222.
- (2004b): "Product Substitutability and Productivity Dispersion," *Review* of *Economics and Statistics*, 86(2), 534–550.
- VIVES, X. (2004): "Innovation and Competitive Pressure," mimeo, INSEAD and Centre for Economic Policy Research.
- WOOLDRIDGE, J. M. (2005): "On Estimating Firm-Level Production Functions using Proxy Variables to Control for Unobservables," mimeo, Michigan State University.

A. Ma	chinery a	and equipment	industry			
Year	Firms	Sales	Value	Employment	Technical	R&D
			Added		Employment	Spending
1996	1053	136, 343, 571	44,739,164	93,847	14,089	5,047,750
1997	1083	133,722,006	45,996,676	92,422	13,844	4,517,149
1998	1093	$138,\!645,\!778$	45,965,565	92,880	14,203	4,309,608
1999	1118	141,833,765	45,870,843	92,752	13,918	4,907,777
2000	1101	147,492,842	46,241,629	90,805	14,763	5,238,998
2001	1080	149,811,427	46,423,050	88,544	15,271	7,793,836
2002	1052	$153,\!673,\!625$	47,627,956	87,741	15,967	4,589,227
B. Ele	ctrical ar	nd optical equip	oment industr	у		
1996	741	157,691,151	40,351,568	80,231	21,447	18,252,000
1997	798	176,798,204	52,169,050	86,082	22,038	22,658,616
1998	825	200,601,253	54,560,220	88,670	22,866	28,076,795
1999	827	232,935,044	56,349,046	91,548	22,213	33,557,319
2000	843	290,674,021	52,103,204	100,608	28,383	42,609,466
2001	843	233,316,414	24,521,650	112,650	28,925	41,988,748
2002	785	194,929,847	30,347,840	86,156	28,292	34,700,542
C. Tra	ansport e	quipment indus	stry			
1996	328	179,072,921	36,994,328	85,445	13,025	5,362,125
1997	346	200,714,936	41,600,851	86,075	13,133	11,979,013
1998	345	221,813,950	50,986,649	89,018	14,216	4,511,031
1999	359	$243,\!678,\!772$	60,958,372	90,780	14,745	12,148,561
2000	368	265,993,715	62,874,146	93,229	15,643	12,975,308
2001	373	216,054,572	53,704,239	93,515	16,109	16,930,037
2002	369	$215,\!549,\!473$	46,973,965	91,474	17,205	24,164,467

 Table 1: Characteristics of the data

NOTE: Firms have at least one technical employee (at least three years of undergraduate school) or made at least one R&D investment during 1996-2002. Sales, value-added, and R&D spending are measured in thousand 1996 SEK.

A. Mach	inery and e	quipment industry: e	ntrants active in	2001.				
Year of	Number	Share of Number	Share of 2001	Share of 2001	Share of 2001 Technical	Share of 2001		
Entry		Active in 2001(%)	Sales(%)	Employment(%)	Employment(%)	R&D(%)		
1973	82	7.59	31.50	30.51	30.86	63.44		
1983	19	1.76	6.69	5.01	5.55	7.82		
1993	51	4.72	4.59	4.66	4.75	2.56		
1996	396	36.67	6.30	8.18	6.68	0.87		
1997	34	3.15	1.42	2.21	1.26	1.32		
1998	32	2.96	1.07	1.38	1.00	0.37		
1999	27	2.50	1.67	1.64	1.36	0.67		
B. Electrical and optical equipment industry: entrants active in 2001.								
1973	33	3.91	7.25	10.12	8.20	1.84		
1983	13	1.54	32.72	12.94	22.44	60.12		
1993	41	4.86	0.73	1.27	0.74	0.00		
1996	328	38.91	5.38	7.90	6.67	0.78		
1997	45	5.34	9.63	20.69	7.08	1.35		
1998	23	2.73	2.41	2.23	1.18	0.15		
1999	30	3.56	0.94	1.64	0.81	0.02		
C. Trans	port equipr	nent industry: entran	ts active in 2001.					
1973	41	10.99	13.70	19.73	20.57	5.42		
1983	7	1.88	10.75	7.94	7.04	14.95		
1993	14	3.75	0.68	1.11	0.55	0.01		
1996	98	26.27	4.44	6.71	3.46	0.74		
1997	12	3.22	0.95	1.22	2.17	0.09		
1998	11	2.95	0.25	0.58	0.27	0.00		
1999	17	4.56	1.14	1.38	0.58	0.02		

Table 2: Entrants active in 2001

NOTE: The sample contains firms that had at least one technical employee or made at least one R&D investment during 1996-2002.

Table 3: Incumbents exiting by 2	00)]	1
-----------------------------------------	----	----	---

A. Machin	A. Machinery and equipment industry: incumbents exiting by 2001.									
Activ in	Number	Share of Number	Share of Sales	Share of	Share of Technical	Share of				
	Active in Base in Base Employment in Employment in R									
		Year(%)	$\operatorname{Year}(\%)$	Base Year(%)	Base Year(%)	Base Year(%)				
1997	311	28.72	28.74	17.33	19.62	19.50				
1998	273	24.98	24.99	15.90	17.53	18.43				
1999	213	19.05	19.06	11.87	12.39	12.35				
2000	131	11.90	11.91	7.52	7.20	7.20				
B. Electrical and optical equipment industry: incumbents exiting by 2001.										
1997	248	31.08	31.09	17.64	23.10	19.48				
1998	229	27.76	27.76	16.56	21.44	18.74				
1999	165	19.95	19.95	13.41	17.15	12.82				
2000	100	11.86	11.86	8.33	10.14	10.48				
C. Transp	oort equipm	ent industry: incum	bents exiting by 2	2001.						
1997	97	28.03	28.06	11.33	18.85	12.92				
1998	81	23.48	23.50	10.01	16.32	14.81				
1999	62	17.27	17.28	9.53	15.32	14.43				
2000	32	8.70	8.71	0.82	2.14	1.06				

NOTE: The sample contains firms that had at least one technical employee or made at least one R&D investment during 1996-2002.

A. Ma	A. Machinery and equipment industry								
Year	Average R&D-to-sales	Median R&D-to-sales	Median R&D-to-sales						
	Ratio by year(%) Ratio for firms with sales Ratio for firms with $(\%)$								
		below median sales($\%$)	above median sales(%)						
1996	3.07	1.26	2.74						
1997	3.49	2.01	2.51						
1998	3.41	2.15	2.68						
1999	3.66	2.17	2.72						
2000	3.60	2.16	2.81						
2001	4.19	2.57	2.54						
2002	4.89	2.41	3.36						
B. Electrical and optical equipment industry									
1996	6.83	3.46	6.24						
1997	7.69	4.36	5.82						
1998	7.19	3.05	7.17						
1999	8.73	4.01	7.92						
2000	18.35	3.73	6.40						
2001	58.29	5.77	6.82						
2002	14.57	5.44	8.01						
C. Tra	ansport equipment industr	сy							
1996	2.74	0.72	1.74						
1997	3.37	1.00	3.93						
1998	3.45	1.32	2.95						
1999	3.73	1.32	3.31						
2000	4.81	2.74	2.90						
2001	5.37	1.83	5.02						
2002	57.26	1.88	5.42						

 Table 4: Scale effects: R&D-to-sales ratio during 1996-2002

NOTE: The sample contains firms with a positive R&D investment.

A. Machinery and equi	pment ind	ustry								
Estimation						Buettne	er (2004) p	rocedure	Endoger	ious productivity
procedure	SIO	OP	LP-m	ACF-i	ACF-m	B-1	B-2	B-3	EP-i	EP-all
	(1)	(2)	(3)	(4)	(5)	(9)	(2)	(8)	(6)	(10)
Non-technical labor	0.693	0.678	0.592	0.675	0.680	0.649	0.626	0.626	0.410	0.498
Std. error	(0.008)	(0.013)	(0.035)	(0.024)	(0.016)	(0.012)	(0.032)	(0.056)	(0.010)	(0.005)
Technical labor	0.265	0.265	0.194	0.247	0.225	0.239	0.233	0.231	0.206	0.241
Std. error	(0.007)	(0.007)	(0.011)	(0.015)	(0.022)	(0.012)	(0.029)	(0.018)	(0.00)	(0.007)
Capital	0.099	0.191	0.179	0.145	0.139	0.072	0.051	0.025	0.214	0.179
Std. error	(0.005)	(0.010)	(0.036)	(0.045)	(0.032)	(0.032)	(0.029)	(0.008)	(0.022)	(0.029)
R&D spending	No	No	No	No	No	No	\mathbf{Yes}	Yes	Yes	Yes
Competitive pressure	No	No	No	No	No	No	No	No	Yes	Yes
# Obs. stage I	7,393	6,739	7,018	6,739	7,018	5,915	1,075	1,075	6,739	7,393
# Obs. stage II		4,795	5,350	4,795	5,350	4,361	774	774	4,795	5,915
B. Electrical and optica	al equipme	int industr	y							
Non-technical labor	0.535	0.545	0.442	0.493	0.445	0.544	0.531	0.531	0.528	0.469
Std. error	(600.0)	(0.010)	(0.00)	(0.010)	(0.010)	(0.012)	(0.011)	(0.010)	(0.010)	(0.006)
Technical labor	0.307	0.296	0.240	0.283	0.251	0.263	0.287	0.287	0.288	0.280
Std. error	(0.008)	(0.008)	(0.007)	(0.008)	(0.007)	(0.010)	(600.0)	(0.008)	(0.008)	(0.006)
Capital	0.184	0.209	0.183	0.241	0.166	0.134	0.125	0.110	0.199	0.204
Std. error	(0.007)	(0.007)	(0.007)	(0.008)	(0.007)	(0.008)	(0.007)	(0.00)	(0.007)	(0.005)
R&D spending	No	No	No	No	No	No	Yes	\mathbf{Yes}	Yes	Yes
Competitive pressure	No	No	No	No	No	No	No	No	Yes	Yes
# Obs. stage I	5,344	4,880	5,040	4,880	5,040	4,880	720	720	4,880	5,344
# Obs. stage II	,	3,437	3,780	3,437	3,780	3,437	502	502	3,437	4,053
C. Transport equipmer	at industry									
Non-technical labor	0.714	0.713	0.684	0.447	0.720	0.676	0.639	0.639	0.389	0.679
Std. error	(0.014)	(0.015)	(0.014)	(0.016)	(0.014)	(0.070)	(0.066)	(0.049)	(0.015)	(0.011)
Technical labor	0.193	0.192	0.165	0.192	0.180	0.173	0.258	0.259	0.190	0.205
Std. error	(0.012)	(0.012)	(0.011)	(0.012)	(0.011)	(0.010)	(0.009)	(0.008)	(0.00)	(0.012)
Capital	0.136	0.156	0.142	0.133	0.115	0.694	0.098	0.098	0.170	0.137
Std. error	(0.009)	(0.009)	(0.008)	(600.0)	(0.009)	(0.046)	(0.064)	(0.057)	(0.013)	(0.010)
R&D spending	No	No	No	No	No	No	Yes	Yes	Yes	Yes
Competitive pressure	No	No	No	No	No	No	No	No	Yes	Yes
# Obs. stage I	2,409	2,189	2,214	2,189	2,214	2,189	389	389	2,189	2,409
# Obs. stage II		1,543	1,656	1,543	1,656	1,543	275	275	1,543	1,826
NOTE: The dependent	variable i	s the log o	f value adde	ed. The mc	dels are as	follows: O	LS - ordina	rry least square	, OP - the Oll	ey and Pakes
et al. (2006) method us	sing investi	ment as ni	oxv for pro	o) memou admetivity:	ACF-m - th	e Ackerhei	uus as proz "ø et al. (2	one) method u	/ILY; AUF-I - L sing intermedi	ne Ackerberg ate innuts as
proxy for productivity;	B-1 - the	Buettner	(2004) met	thod that c	aptures the	effect of I	3&D via c	apital and cont	trol for selection	on; B-2 - the
Buettner (2004) metho	d that cap	otures the	effect of R.	&D, but th	nat does not	t control fo	or selection	i, B-3 - the Bu	lettner (2004)	method that
captures the effect of F	t&D and c	ontrols for	selection;]	EP-i - uses	positive inv	estment as	proxy for	productivity, c	aptures the ef	fects of R&D
and competitive pressu	re on prod:	uctivity pr	ocess; EP-a	ll - uses all	data, captui	res the effe	cts of R&L) and competiti	ve pressure on	productivity
process. The following	measures	are include	ed to accou	int for com	petitive pres	ssure: nun	uber of firm	is with fewer tl	han 100 emplo	yees, median
R&D spending at the	sub-indust	ry level (f	ive digits),	change in	concentratic	on measure	e (c4), fore	ign demand -	total sales to	other foreign
firms. All standard err	ors for sem	niparametr	ic methods	are bootsr:	apped using	50 replica	tions.			

parameters
function
generating
of value-added
Estimates o
Table 5:

A: Machinery and equipment industry									
		Pr	oductivity	level			Productiv	vity growth	L
	Q25	Q50	Mean	Q75	IQM	Q25	Q50	Mean	Q75
OP	5.104	5.300	5.315	5.517	0.078	-0.137	-0.005	-0.012	0.124
ACF-i	5.542	5.730	5.738	5.936	0.068	-0.129	0.001	-0.004	0.128
EP-i	5.831	6.099	6.098	6.377	0.089	-0.132	-0.001	-0.008	0.128
EP-all	5.809	6.047	6.039	6.288	0.079	-0.135	0.002	-0.002	0.139
B: Electrical and optical equipment industry									
OP	5.356	5.571	5.568	5.802	0.080	-0.157	-0.001	-0.006	0.147
ACF-i	5.285	5.506	5.497	5.740	0.082	-0.160	-0.003	-0.008	0.146
EP-i	5.501	5.716	5.711	5.944	0.077	-0.155	0.001	-0.003	0.150
EP-all	5.640	5.867	5.864	6.121	0.081	-0.156	0.004	-0.004	0.157
C: Transport	equipment in	dustry							
OP	5.350	5.520	5.525	5.706	0.064	-0.118	0.006	0.011	0.130
ACF-i	6.271	6.592	6.584	6.909	0.096	-0.106	0.020	0.027	0.140
EP-i	6.276	6.628	6.622	6.968	0.104	-0.111	0.021	0.026	0.140
EP-all	5.608	5.792	5.784	5.984	0.064	-0.122	0.009	0.012	0.131

 Table 6: Summary statistics: productivity level and growth

In an 5.005 5.132 5.754 5.354 5.009 5.120 5.12 5.104 5.131 NOTE: Productivity levels are in logs. Productivity growth is defined as $log(\omega_{jt}) - log(\omega_{jt-1})$. IQM is standardized interquantile range, i.e., the difference between the quantile 75 and the quantile 25 over the median.

Estimation Quantile Quantile	
Listilitation Qualitie Qualitie	
procedure OLS 0.25 0.50 0.75 OLS 0.25 0.50	0.75
Intercept -0.009 -0.138 -0.002 -0.135 -0.930 -0.804 -0.748	-0.672
Std. error (0.004) (0.004) (0.003) (0.004) (0.255) (0.236) (0.177)	(0.208)
R&D intensity _{t-1} 0.312 0.173 0.217 0.372 0.314 0.140 0.196	0.369
Std. error (0.031) (0.105) (0.101) (0.138) (0.031) (0.122) (0.107)	(0.134)
No. of small firms $(<100)_{t-1}$ 0.0009 0.0007 0.0008	0.0008
Std. error (0.0002) (0.0002) (0.0001)	(0.0002)
Change in $C4_{t-1}$ -0.077 -0.076	-0.129
Std. error (0.040) (0.034) (0.027)	(0.028)
Median R&D spending_{t-1} 0.0001 0.000 0.0001	0.0001
Std. error (0.0001) (0.0001) (1e-6)	(0.001)
Foreign demand _{$t-1$} $3.335e-9$ 0.000 0.000	1e-8
Std. error (1.945e-9) (0.000) (0.0001)	(1e-9)
#Obs. 5,692 5,692 5,692 5,692 5,692 5,692 5,692	5,692
B. Electrical and optical equipment industry	
Intercept -0.008 -0.156 0.001 0.154 -0.116 -0.183 -0.231	-0.167
Std. error (0.007) (0.006) (0.004) (0.005) (0.185) (0.142) (0.107)	(0.133)
R&D intensity _{t-1} $0.086 -0.025 0.102 0.159 0.087 0.140 0.102$	0.159
Std. error (0.015) (0.155) (0.031) (0.050) (0.015) (0.161) (0.020)	(0.081)
No. of small firms $(< 100)_{t-1}$ 0.0002 5e-5 0.0003	0.0005
Std. error (0.0002) (0.0001)	(0.0002)
Change in $C4_{t-1}$ -0.005 0.816 0.018	-0.017
Std. error (0.060) (0.053) (0.037)	(0.047)
Mean R&D spending_{t=1} $5.254e-7 0.000 0.000$	0.000
Std. error (2.281e-7) (0.0001) (0.001)	(0.0001)
Foreign demand _{$t-1 -7.942e-9$ 0.000 0.000}	0.000
Std. error $(2.927e-9)$ (0.0001) (0.0001)	(0.0001)
#Obs. 4,053 4,053 4,053 4,053 4,053 4,053 4,053	4.053
C. Transport equipment industry	
Intercept 0.004 -0.124 0.003 0.125 0.156 -0.079 0.223	0.501
Std. error (0.008) (0.007) (0.005) (0.007) (0.284) (0.258) (0.200)	(0.226)
B&D intensity, 1 0.259 0.150 0.241 0.333 0.261 0.148 0.208	0.313
Std error (0.041) (0.011) (0.135) (0.074) (0.023) (0.023) (0.140)	(0.040)
No. of small firms (< 100), $-0.0005 - 0.0001 - 0.0008$	-0.001
Std. error (0.001) (0.0007)	(0.0008)
Change in $C4_{t-1}$ -0.211 -0.160 -0.118	-0.172
Std error (0.081) (0.074) (0.052)	(0.059)
Mean R&D spending,	0.000
Std. error $(1.485e^{-7})$ (0.0001) (0.001)	(0.0001)
Foreign demand, 3.214e-10 0.000 0.000	1.e-9
Std. error $(1.068e-9)$ (0.0001) (0.0001)	(1e-10)
#Obs. 1,826 1,826 1,826 1,826 1,826 1,826 1,826	1.826

Table 7: Quantile regressions on the conditional distribution of productivity growth

NOTE: The dependent variable is productivity growth. All standard errors are bootsrapped using 50 replications in quantile regressions.

Table 8: The ratio of optimal investment to actual investment in R&D

			Quantile	
	OLS	0.25	0.50	0.75
Machinery and equipment	2.09, 4.13	0.93, 1.84	1.30, 2.58	2.46, 4.85
Electrical and optical equipment	0.58, 1.14	0.93, 1.84	0.68, 1.34	1.06, 2.09
Transport equipment	1.74, 3.43	0.98, 1.94	1.38, 2.73	2.08, 4.11

NOTE: The figures -,- give the minimum and the maximum for the ratio of optimal investment to actual investment in R&D. The ratio, $s^{optimal}/s^{actual}$ is approximated by the ratio between the rate of return to R&D and the average real return on the stock market (7.6% is considered here). The minimum ratio is obtained when the average real return on the stock market doubles (15%).

					Percentage o	of growth fro	m	
Period	Productivity	Overall	Within	Between	Cross	Entry	Exit	Net Entry
	measure	industry	firms	firms	firms			
		growth	(1)	(2)	(3)	(4)	(5)	(4) - (5)
A. Machine	erv and equipme	nt industry						
1996-2000	Labor	11.27	2.80	-0.50	6.28	-0.18	-2.33	2.15
1996-2000	TFP-OP	7.10	-3.30	1.42	5.77	3.19	-0.01	3.20
1996-2000	TFP-ACF-i	6.60	-2.15	0.17	6.27	1.48	-0.83	2.31
1996-2000	TFP-EP-i	4.19	-2.84	-1.42	7.27	-1.18	-2.07	0.88
1996-2000	TFP-EP-all	4.19	-2.84	-1.42	7.27	-1.19	-2.07	0.88
1996-2002	Labor	14.51	3.86	-0.92	6.38	2.23	-2.96	5.18
1996-2002	TFP-OP	8.87	-2.38	0.55	5.25	6.89	1.47	5.42
1996-2002	TFP-ACF-i	8.58	-1.19	-0.39	5.84	4.89	0.57	4.32
1996-2002	TFP-EP-i	7.00	-2.47	-2.26	7.23	1.16	-3.33	4.49
1996-2002	TFP-EP-all	7.00	-2.47	-2.26	7.23	1.16	-3.33	4.49
B. Electric	al and optical eq	uipment industry						
1996-2000	Labor	60.91	30.41	8.10	23.29	-0.48	0.39	-0.88
1996-2000	TFP-OP	57.61	23.40	-0.69	27.44	3.28	-4.17	7.46
1996-2000	TFP-ACF-i	57.41	22.51	0.69	27.43	2.82	-3.95	6.78
1996-2000	TFP-EP-i	59.18	23.61	1.09	27.78	2.65	-3.98	6.63
1996-2000	TFP-EP-all	59.61	26.39	7.33	24.11	1.58	-0.20	1.78
1996-2002	Labor	19.86	2.24	-7.94	12.61	8.42	-4.53	12.96
1996-2002	TFP-OP	21.41	-1.12	0.52	9.91	9.20	-2.92	12.12
1996-2002	TFP-ACF-i	17.68	-1.42	-0.99	10.17	8.38	-1.54	9.92
1996-2002	TFP-EP-i	16.79	-0.92	-1.59	10.60	7.56	-1.14	8.71
1996-2002	TFP-EP-all	21.55	10.75	-6.71	5.19	8.25	-4.06	12.31
C. Transpo	ort equipment inc	lustry						
1996-2000	Labor	49.29	36.91	-1.57	13.91	1.22	1.19	0.03
1996-2000	TFP-OP	47.26	27.38	-6.40	23.00	7.12	-3.84	3.28
1996-2000	TFP-ACF-i	51.68	28.29	2.90	24.44	-0.03	4.32	-4.35
1996-2000	TFP-EP-i	52.27	28.06	4.06	24.91	-0.47	4.29	-4.76
1996-2000	TFP-EP-all	47.48	35.50	-3.91	13.43	4.73	2.28	2.45
1996-2002	Labor	25.00	14.35	-1.97	6.31	6.89	0.59	6.31
1996-2002	TFP-OP	25.42	13.02	-0.87	5.73	11.28	3.78	7.54
1996-2002	TFP-ACF-i	22.54	13.05	-1.73	7.43	0.16	-3.63	3.79
1996-2002	TFP-EP-i	21.81	12.78	-1.72	7.54	-1.35	-4.57	3.22
1996-2002	TFP-EP-all	22.66	11.87	-1.27	5.65	8.93	2.52	6.41

Table 9: Decomposition of productivity growth, 1996 to 2000 and 1996 to 2002 (percent)

NOTE: The Foster, Haltiwanger, and Krizan's (2001) decomposition is used (Section 5). Labor productivity is defined as log of value added per employee. The shares of value added at the industry level are used as weights in the decomposition.

Appendix A: Data sources. I here describe the variables used. Value added is total shipments, adjusted for changes in inventories, minus the cost of materials. Real value added is constructed by deflating value added by a five-digit industry output deflater. The deflectors are taken from Statistics Sweden. The technical labor variable is the total number of employees with at least 3 years of technical school education. The non-technical labor defines the remaining employees. Data on the research and development variable stems from FS and covers all firms with at least one employee who works at least half-time in R&D activities. The FS is updated annually and it is compulsory for firms to reply. Firms must give an exact figure for R&D spending or answer in an interval scale. I deflated the R&D spending, sales, and investment by the consumer price index(CPI) from IMF-CDROM 2005. The capital measure is constructed using a perpetual inventory method, $k_{t+1}(1-\delta)k_t + i_t$. Since the capital data distinguishes between buildings and equipment, all calculations of the capital stock are done separately for buildings and equipment. As suggested by Hulten and Wykoff (1981) buildings are depreciated at a rate of 0.361 and equipment at 0.1179.

In order to construct capital series using the perpetual inventory method, I need an initial capital stock. Some of the firms are in FS since 1973. I set the initial capital stock to the first occurrence in FS. I define entry when the year of entry in FS is the same as the year of first data collection. FS contains all firms in different industries after 1996.

Appendix B. Properties of the value function. The Bellman equation can be rewritten in terms of the expected value of profits in the following period and the continuation thereafter

$$V(\omega, k, \theta) = \max\{\phi, \pi(\omega, k, \theta) - c(\tilde{i}(\omega, k, \theta), k) - z(\tilde{\psi}(\omega, k, \theta), \omega) \\ +\beta \int \chi(\omega', k', \theta') \left[\pi(\omega', k', \theta') - c(\tilde{i}(\omega', k', \theta'), (1-\delta)k + \tilde{i}(\omega, k, \theta)) \right] \\ (17) \quad -z(\tilde{\psi}(\omega', (1-\delta)k + \tilde{i}(\omega, k, \theta), \theta'), \omega', \theta') P(d\omega'|\tilde{\psi}(\omega, k, \theta), \tilde{i}(\omega, k, \theta), \theta) \\ +\beta \phi \int [1 - \chi(\omega', k', \theta')] + \beta^2 \int \chi(\omega'', k'', \theta'') V(\omega'', k'', \theta'') \\ P(d\omega') [\tilde{\psi}(\omega', k', \theta'), \tilde{i}(\omega', k', \theta'), \theta') P(d\omega') [\tilde{\psi}(\omega, k, \theta), \tilde{i}(\omega, k, \theta), \theta).$$

We want to find a set of alternative programs that leave the last term in this expression unchanged. The distribution of ω " conditional on ω and each alternative policy is the same as the distribution of ω " conditional on ω and optimal policy. We select the optimal policy such that

(18)
$$\begin{aligned} &\int_{\omega'} P(\omega^{"} > \tilde{\omega} | \psi^{"}, i(\omega', k', \theta'), \theta') P(d\omega' | \psi', i(\omega, k, \theta), \theta) = \\ &\int_{\omega'} P(\omega^{"} > \tilde{\omega} | \psi^{"} + \Delta(\psi^{"}, \omega', \theta', \epsilon), i(\omega', k', \theta') + \Delta(\psi^{"}, \omega', \theta', \epsilon), \theta') + \\ &P(d\omega' | \psi' - \epsilon, i(\omega, k, \theta) - \epsilon, \theta), \end{aligned}$$

where ϵ and $\Delta(\cdot)$ are chosen such that $\Delta(\cdot, \epsilon) = 0$ at $\epsilon = 0$. The optimal policy produces a distribution of ω " conditional on ψ' as a convolution of $P(\cdot|\psi', i(\omega, k, \theta), \theta)$ and $P(\cdot|\psi'', i(\omega', k', \theta'), \theta')$. This gives the same convoluted distribution by perturbing i and ψ' by ϵ and i' and ψ'' by $\Delta(\psi'', \omega', \theta', \epsilon)$.

Lemma 1 The value function $V(\omega, k, \theta)$ is bounded, non-decreasing in ω and k, supermodular in (ω, θ) and (ω, k) , and unique.

Proof: The proof is a consequence of the Proposition 5 in Smith and McCardle (2002). I reformulate Smith and McCardle (2002)'s proposition in Proposition 1. All the properties in Lemma 1 are closed convex cone properties.

Definition 1 P is a closed convex cone property (CCC) if the set of functions satisfying P forms a closed convex cone in the topology of pointwise convergence.

Proposition 1 (Smith and McCardle, 2002) Let U be a set of functions on $\Omega \times K \times \Theta$ satisfying a CCC property P, and let P^* be a joint extension of P on $\Psi \times \mathbb{R}_+ \times \Theta$. If, for all t, (a) the net profit functions $r_t(\psi', i, \omega, k, \theta)$ satisfy P^* and (b) the transitions $\tilde{\psi}$ and \tilde{i} satisfy P^* (\succeq_U), then each V_t satisfies P and $\lim_{t\to\infty} V_t$, if it exists, also satisfies P.

The properties P and P^* are the following: $P - V(\omega, k, \theta)$ is bounded, increasing in ω and k, and supermodular in (ω, k) and (ω, θ) ; and P^* - for each $\tilde{\psi}(\omega, k, \theta)$ and $\tilde{i}(\omega, k, \theta), r(\psi', i, \omega, k, \theta)$ is bounded, nondecreasing in ω and k, and supermodular in (ω, k) and (ω, θ) .

The net profit function is bounded above because the profit function is bounded above. In addition, cost and R&D functions are nonnegative. The expected net present value of the future one period return is bounded above due to the fact that $\beta < 1$. In addition, ϕ puts a lower bound on the value function so that $V_t(\cdot)$ is bounded.

The net profit is a non-decreasing function in (ω, k) , and is supermodular (Athey, 2000); this combination of properties is the *P* that we want to show that the value function $V(\cdot)$ satisfies. Each of these properties is a *single-point property*,

and so is P. The joint extension P^* of P requires that P holds for each action $(\tilde{\psi}, \tilde{i})$. The net profit function satisfies P^* for each choice of action and therefore satisfies P^* . From Lemma 2 and Lemma 3, we have that the transitions $\tilde{\psi}$ and \tilde{i} satisfy P^* (\succeq_U). Thus, each $V_t(\cdot)$ satisfies P and so does $\lim_{t\to\infty} V_t(\cdot)$.

Lemma 2 The optimal physical investment choice conditional on (ψ', k, θ)

$$\tilde{i}(\psi',k,\theta) = \arg \sup_{i} \left[-c(i,k) + \beta \int V(\omega',k',\theta') P(d\omega'|\psi',\theta) \right]$$

is non-decreasing in ψ' , k, and θ .

Proof: The value function $V(\omega', k', \theta')$ is supermodular in (ω', k') and (ω', θ') . The integral $\int V(\omega', k', \theta') P(d\omega'|\psi', \theta)$ is supermodular in (ψ', θ) because $P(d\omega'|\psi', \theta)$ is stochastically non-decreasing in ψ' and θ (Athey, 2000). This implies that the optimal investment choice $\tilde{i}(\psi', k, \theta)$ is non-decreasing in capital k, non-decreasing in ψ' and θ . The function -c(i, k) is supermodular implying that the objective function is supermodular. ■

Lemma 3 The policy function for the choice of distribution

$$\tilde{\psi}(\omega, k, \theta) = \arg \sup_{\psi'} \left[\pi(\omega, k, \theta) - c(\tilde{i}(\psi', k, \theta), k) - z(\psi', \omega) + \beta \int V(\omega', k', \theta') P(d\omega'|\psi', \theta) \right]$$

is non-decreasing in ω and strictly non-decreasing in ω on the sets

$$\left\{(\omega,k,\theta)|z(\tilde{\psi}'(\omega,k,\theta),\omega)>0\right\}\cup\left\{(\omega,k,\theta)|\pi(\omega,k,\theta)\quad\text{is supermodular in}\quad(\omega,\theta)\right\}.$$

Proof: The objective function, $r(\cdot)$ is supermodular in (ψ', ω) and (ω, θ) . It is a sum of supermodular functions (by assumption, the R&D spending $-z(\psi', \omega)$ is supermodular, and so is the profit function). This implies that the objective function is non-decreasing in ω .

To prove strict monotonicity, I use an Euler equation $F(\omega, k, \psi', \theta) = 0$ for a perturbation of the optimal $\tilde{\psi}(\cdot)$ between periods t and t + 1 (Pakes, 1994). We want to see what the implications of an increasing in productivity are on the Euler equation. The Euler equation has to remain satisfied for an increasing in productivity. The choice of distribution $\tilde{\psi}(\cdot)$ given competitive pressure θ affects the stochastic evolution of the future productivity ω' . The future productivity ω' affects the future pressure θ' . We construct an alternative program that leaves the joint distribution of the state variables from period t + 2 and onwards unchanged (conditional on the state in t). If ψ' denotes the choice distribution under the
optimal program, let us consider the perturbation $\psi^* = \psi' - \epsilon$. The next period productivity has the distribution $P(d\omega'|\psi' - \epsilon, \theta)$ under this perturbation. Let us define

$$\begin{split} \omega^* &= P^{-1} \left(P(d\omega' | \psi', \theta) | \psi' - \epsilon, \theta \right) = g(\omega', \psi', \theta, \epsilon), \\ \Delta(\omega', \psi', \theta, \epsilon) &= \omega' - \omega^* = \omega' - g(\omega', \psi', \theta, \epsilon), \\ \Gamma(\omega', \psi', \theta, \epsilon) &= \theta' - \theta^*, \end{split}$$

where $\Delta(\cdot, \epsilon) = 0$ and $\Gamma(\cdot, \epsilon) = 0$ at $\epsilon = 0$; $\Delta(\cdot)$ and $\Gamma(\cdot)$ are differentiable as P and P^{-1} are differentiable. The difference in period t between the value function of the original and alternative program is

$$\begin{split} V(\omega,k,\theta) - V(\omega,k,\theta,\epsilon) &= -z(\psi',\omega,\theta) + z(\psi'-\epsilon,\omega,\theta) \\ +\beta \int \chi(\omega',k',\theta') \left[\pi(\omega',k',\theta') - \pi[\omega'-\Delta(\omega',\psi',\theta,\epsilon),k',\theta'-\Gamma(\omega',\psi',\theta,\epsilon)] \right] \\ -z(\psi'',\omega') + z(\psi'',\omega'-\Delta(\omega',\psi',\theta,\epsilon)) \right] P(d\omega'|\psi',\theta). \end{split}$$

This expression must be non-negative in a neighborhood of the $\epsilon = 0$ since the original program is optimal. Its differentiable in ϵ , in a neighborhood of the $\epsilon = 0$, must be zero, which implies the Euler equation

$$\begin{split} F(\omega,k,\theta,\psi') &= -\frac{\partial z(\psi',\omega,\theta)}{\partial \psi'} + \beta \int \chi(\omega',k',\theta') \left[\frac{\partial \pi(\omega',k',\theta')}{\partial \omega'} - \frac{\partial z(\psi^{n},\omega')}{\partial \omega'} \right] \frac{\partial \Delta(\omega',\psi',\theta,\epsilon))}{\partial \epsilon} \\ P(d\omega'|\psi',\theta) + \beta \int \chi(\omega',k',\theta') \left[\frac{\partial \pi(\omega',k',\theta')}{\partial \theta'} - \frac{\partial z(\psi^{n},\omega')}{\partial \theta'} \right] \frac{\partial \Gamma(\omega',\psi',\theta,\epsilon))}{\partial \epsilon} \\ P(d\omega'|\psi',\theta) &= 0, \end{split}$$

for each (k, θ, ψ') . $F(\omega, k, \theta, \psi')$ is a continuous, strictly increasing function of ω for every (k, θ, ψ') . For a fixed k, an increase in ω has to trigger change in ψ' and θ for $F(\omega, k, \theta, \psi') = 0$ to remain satisfied. My setting accounts for the effect of competitive pressure on the firm's profit when we have an increase in productivity. Thus, the choice distribution $\tilde{\psi}(\omega, k, \theta)$ is non-decreasing in ω on the set $\{(\omega, k, \theta) | z(\tilde{\psi}(\omega, k, \theta), \omega) > 0\}$ and where the profit function in supermodular in (ω, θ) .

Theorem 1 The policy function for the **investment choice** $\tilde{i}(\omega, k, \theta) = \tilde{i}(\tilde{\psi}(\omega, k, \theta), k, \theta)$ is non-decreasing in ω and strictly non-decreasing in ω on the sets

$$\begin{split} & \Big\{ (\omega,k,\theta) | \tilde{i}(\omega,k,\theta) > 0 \wedge z(\tilde{\psi}(\omega,k,\theta),\omega) > 0 \Big\} \cup \\ & \{ (\omega,k,\theta) | \pi(\omega,k,\theta) \quad \text{is supermodular in} \quad (\omega,\theta) \} \end{split}$$

Proof: Lemma 2 and 3 give us that the investment choice $\tilde{i}(\psi', k, \theta)$ is nondecreasing in ψ' , which is non-decreasing in ω and θ . This implies that the optimal investment choice $\tilde{i}(\omega, k, \theta)$ is non-decreasing in ω and θ . Let us consider the following alternative programme: $i^*(\omega, k, \theta) = \tilde{i}(\omega, k, \theta) - \epsilon$, $\theta'^* = \theta' - \Gamma(\epsilon)$ (actual investment affects future productivity that affects competitive pressure), $i^*(\omega', k', \theta') =$ $\tilde{i}(\omega', k', \theta' - \Delta(\epsilon))$, $\psi^* = \tilde{\psi}(\omega, k + \epsilon, \theta)$, and $\chi^*(\omega, k, \theta) = \chi^*(\omega, k + \epsilon, \theta)$. The difference in period t between the value function of the original and alternative programme is

$$\begin{split} V(\omega,k,\theta) - V(\omega,k,\theta,\epsilon) &= -c(i,k) + c(i-\epsilon,k) \\ +\beta \int \chi(\omega',k',\theta') \left[\pi(\omega',k',\theta') - \pi(\omega',k'-\epsilon,\theta'-\Delta(\epsilon)) \right] \\ -c(\tilde{i}(\omega',k',\theta'),k') + c(\tilde{i}(\omega',k',\theta'-\Delta(\epsilon)),k'-\epsilon) \right] P(d\omega'|\psi',\theta) \end{split}$$

This expression must be non-negative in a neighborhood of the $\epsilon = 0$ because the original programme is optimal. Its differentiable in ϵ , in a neighborhood of $\epsilon = 0$, must be zero at $\epsilon = 0$, which implies the Euler equation

$$F(\omega, k, \theta, i) = -\frac{\partial c(i,k)}{\partial i} + \beta \int \chi(\omega', k', \theta') \left[\left(\frac{\partial \pi(\omega', k', \theta')}{\partial i} - \frac{\partial c(\tilde{i}(\omega', k', \theta'), k')}{\partial i} \right) \frac{\partial \Delta(\epsilon)}{\partial \epsilon} \right] \\ \left(\frac{\partial \pi(\omega', k', \theta')}{\partial \theta'} - \frac{\partial c(\tilde{i}(\omega', k', \theta'), k')}{\partial \theta'} \right) \frac{\partial \Gamma(\epsilon)}{\partial \epsilon} + P(d\omega' | \psi', \theta) = 0$$

for each (θ, k, i) . $F(\omega, k, \theta, i)$ is a continuous, strictly increasing function of ω for every (k, θ, i) . For a fixed k, an increase in ω has to trigger change in i and θ for $F(\omega, k, \theta, i) = 0$ to remain satisfied.

Paper III

Industry Dynamics and Format Repositioning in Retail^{*}

Florin G. Maican[†]

Draft: April 30, 2010

Abstract

In differentiated product markets, when firms are affected by demand shocks, they may react by repositioning their products, which in turn affects market structure. This paper proposes a dynamic oligopoly model to estimate the costs of repositioning store formats together with sunk costs of entry and sell-off values of exit in the retail industry. The model gives important information about driving forces behind format changes and how such repositioning can be linked to entry and exit. Using data from Sweden, the results indicate that both repositioning and entry costs increase with market size, and their growth decreases when moving to larger markets. Small markets have higher sell-off values than repositioning costs, but large entry costs. The difference between higher entry and lower repositioning costs explains why the number of observed repositionings is higher than the number of entrants. Since entry is regulated in most of OECD countries, repositioning costs and their link to competition have important implications for competition policy.

Keywords: imperfect competition; dynamic oligopoly; dynamic estimation; industry dynamics; repositioning; retail. *JEL Classification*: L1, L13, L81

^{*}I would like to thank Jaap Abbring, Dan Ackerberg, Victor Aguirregabiria, Eric Bartelsman, Amit Gandhi, Lapo Filistrucchi, Lennart Hjalmarsson, Matilda Orth, Catherine Schaumans, Che-Lin Su, Johan Stennek, and Andrew Sweeting for very useful comments and suggestions. In addition, I thank participants at EARIE (Toulouse), EARIE (Ljubliana), the 2008 Nordic Retail Conference (Stockholm), Tilburg University, and the University of Gothenburg Industrial Organization seminars. Financial support from the Swedish Competition Authority is gratefully acknowledged.

[†]Department of Economics, University of Gothenburg, Box 640, SE 405 30, Göteborg, Sweden, Phone +46-31-786 4866, Fax: +46-31-786 4154, *E-mail*: florin.maican@handels.gu.se

1 Introduction

There have been major structural changes in retail markets during the last decades, e.g., decreasing number of stores and the rise of the "big-box" format. Due to the increasing importance of information technology and distribution systems, large retail firms dominate, each operating a number of well-defined store formats, and continuously reconsidering store formats as well as possible entry of new stores, or exit. Recent investments in retail aim to increase product differentiation in store formats. However, each investment implies sunk costs. There are only a few studies that emphasize firms' strategies on format repositioning in local markets in response to strategies of rival firms (Sweeting, 2007; Gandhi et al., 2008). But, the retail industry has large scale-and-scope economies where format repositioning has key implications for competition.¹

The aim here is to provide a model that estimates and links respositioning costs with sunk costs of entry and sell-off (exit) values. If rival firms enter with new stores, the reaction of the firm may be to change the formats of affected stores or to shut down some stores. Shifts in costs of entry and repositioning can then lead to markets with few stores, with little product differentiation and low competition. A retail market increasingly dominated by a small number of stores is bad for both consumers and suppliers. In Europe, there are countries where the top five firms made up 70 percent or more of the grocery market in 2005: Germany (70 percent), France (70 percent), Austria (79 percent), Estonia (79 percent), Ireland (81 percent), Slovenia (82 percent), Sweden (82 percent), and Finland (90 percent). Since entry is regulated in most of OECD countries, sunk costs of entry, and repositioning-format costs, have important implications for policy analysis.²

This paper uses a fully dynamic oligopoly model to estimate the costs of format repositioning, sunk costs of entry, and sell-off values of exit in the Swedish

¹Focusing on food prices in the EU, the European Commission published a report in December 2008, recommending among other things that, "regulations that restrict entry of new companies into the market need to be scrutinized and removed when appropriate, while keeping in mind their environmental and social goals" (European Commission 2008:321).

²Pilat (1997) surveys entry regulations in OECD countries.

retail food industry.³ ⁴During the period 2001-2006 the number of stores that changed format was substantially larger than the number of stores that entered. For example, over 80 percent more stores changed format than entered in 2006. Four important firms dominate the Swedish retail food market. The focus here is on the individual store, but the paper also accounts for firm's strategies.⁵ This is important, since firms try to change the formats of their stores in response to local competition. The format of the store is chosen to maximize the store's expected future profits.⁶ But these choices may affect the future format choices of other stores in the market.

In each period, a discrete choice demand model, that accounts foe spatial differentiation, is estimated to recover unobserved store quality. Store quality is defined as the mean of unobserved store characteristics across consumers. It is unobserved by the researcher but known to the store. Section 3.1 provides a detailed discussion about what I measure by quality. By changing format, stores try to increase the their store quality (quality effect), but this changing cannot be done without any cost. First, though stores try to keep their old consumers while hoping to also gain new ones, there may be a fall in sales in the short-run until the customers adjust to the new format. But second, there may also be sunk costs of investment associated with format repositioning. Important factors for format repositioning that are observable and differ across markets, e.g., local demographic characteristics allow to estimate mean repositioning costs the for observed changes.

Returns from format repositioning are realized over future periods and, therefore, a dynamic model is best for estimating repositioning costs and benefits. The dynamic approach used here is based on the two-step procedure proposed by Bajari et al. (2007) (BBL). In the first step, I estimate price adjusted quality of each

³Maican (2008) uses a similar approach to estimate the sunk costs of store-type repositioning, i.e., switching from a grocery store to a convenience store. But due to the small number of observed changes in store type, an accurate estimation of sunk costs is hard to obtain. The present paper extends Maican (2008) by introducing spatial competition and uses store concept instead of store type to define store-format. This is important, since store type is more related to size, while store concept is related to the firm's business model. A firm can change store concept to adjust to market competition. A more detailed discussion about the store-format definition is given in Section 2.

⁴There are studies that estimate costs paid by individuals/households when moving between different cities offering different market opportunities (Kennan and Walker, 2006; Bayer and Falko, 2006; and Gemici, 2007).

⁵While the dynamic setting at firm level could be problematic, since techniques for estimating dynamic games with incomplete information assume stationarity, the growth of the retail industry may be a non-stationary process, since some firms never exit.

⁶Since the store format is based on the business model given by the firm, the model presented at the store level surprises also the firm effect.

store from a random-coefficients demand model (Berry et al., 1995; Nevo, 2001; Davis, 2006). Using assumptions on the timing of innovations in store-quality relative to store-format choices, this estimation allows for endogenous format choices. These timing assumptions are used in the production-function estimation literature (Olley and Pakes, 1996; Blundell and Bond, 2000; Levinsohn and Petrin, 2003; and Ackerberg et al., 2006). The estimated quality is then used to estimate the sales-generating function. In addition, the paper also estimates entry, exit, and format-attractiveness policies. In the second step, using an inequality estimator (Pakes et al., 2007b), the paper recovers the sunk costs of entry, sell-off (exit) value, and format-repositioning costs. An advantage of using inequality estimator is that it is robust to simulation errors in the value function. This estimation approach is also used by Ho (2007) and Ishii (2005) when estimating models in a static setting, and by Holmes (2008) in a dynamic study of Wal-Mart's store locations.

Recent literature on estimation of dynamic games with Markov perfect equilibria has developed alternative extensions to the Hotz and Miller (1993) and Hotz et al. (1994) approaches (Aguirregabiria and Mira, 2007; Bajari et al., 2007; Pakes et al., 2007a; and Pesendorfer and Schmidt-Dengler, 2003). Several recent papers have estimated dynamic oligopoly games using industry data. First, there are studies of entry and exit in homogenous product markets: Ryan (2009) analyzes the cement industry and Collard-Wexler (2006) studies the ready-mix concrete industry. Second, there are papers that allow for vertical product differentiation by using logit-demand models: Beresteanu and Ellickson (2006) and Macieira (2006) analyze the supermarket and supercomputer industries, respectively. Third, there are few studies of both horizontal and vertical product differentiation: Analyzing the U.S. radio industry, Sweeting (2007) uses a random coefficients demand model to measure the costs of product repositioning.

This paper is closely related to Sweeting (2007) and Ryan (2009). Both those papers use the BBL dynamic framework to estimate sunk costs. Sweeting (2007) does not model entry and exit. This paper explicitly model entry and exit since there is a close connection between entry, exit, and format repositioning in retail markets. To my knowledge, this is the first paper that models entry, exit as well as repositioning at the same time. The sunk costs of entry are backed out as in Ryan (2009). However, an important difference is that I also account for spatial competition. Location is a key factor for a store, and consumers have preferences over both geographic and store characteristics. Finally, this paper is first, to my knowledge, to estimate repositioning costs in the retail industry.

Another alternative is to model at the firm level (Aguirregabiria and Vicentini, 2006; Jia, 2008). Using a static setting, Jia (2008) provides an empirical model for measuring the impact of chain stores on other discount retailers and quantifying the scale-economies within a chain. Her model allows for flexible competition patterns among all players. However, it has some limitations, such as that it cannot be applied to oligopoly games with three or more chains. In this case the strategy of the firm is modeled choosing the number, format, and location of its stores. The researcher then explicitly models firm behaviour, but might miss important information at store level. The researcher assumes that stores in the same format are identical, the only difference between two stores in the same format being their location. By aggregating stores, the researcher loses important information, such that why a firm changed the format of one store but kept others in the same format, if all were in similar locations. In addition, the researcher can then only estimate the distribution of store quality. In my case, recovering the quality of individual store is important, since store quality influences demand, and therefore is an important decision-factor for the firm.

This paper uses detailed data on all retail food stores in Sweden during 2001 to 2006. This is the first model of the store format repositioning. Given the complexity of this industry (multiple ownership, spatial differentiation, and regulation), the estimated parameters are preliminary, i.e. they cannot be used as a direct guide for policies. This paper provides initial estimates of these costs while future research should consider robustness tests and consider the implications of the results. I assume that the estimated repositioning costs are the same for each store within each group market. The results indicate that the costs of format repositioning increase with market size. For three groups by market-size, the ratio between median sales and average repositioning costs is over 20. Format repositioning seems most profitable, however, in medium-sized markets, with a population between 20,000 and 60,000. I find higher entry costs per median sales in medium than in large markets. Sell-off values on exit are about twice as high in the large markets as in the small ones.

The findings show that stores are less likely to exit if they have high quality, if they are located in large markets, or if the firm operates many stores in the same format. Entry is more likely in large markets and if rivals have high quality, i.e., if there is room for product differentiation. Stores with high quality are more likely to be in large formats, and old stores are less likely to reposition themselves. Furthermore, distance is found to be a key factor when consumers choose a store. Store's quality is more persistent for non-repositioning stores.

The next section gives a brief overview of the Swedish retail food industry and relevant recent events, and also discusses the data sources and introduce the variables. Section 3 then presents the theoretical model, while Section 4 discusses the results whereas Section 5 summarises and draw conclusions.

2 Overview of the Swedish Retail Food Industry

Annual retail sales in Sweden in 2004 were around SEK 400 billion, one-third of private consumption, of which 52 percent was grocery sales and 48 percent is nonfood. Four large firms dominate Swedish food retail: ICA, Coop, Axfood, and Bergendahls together had more than 90 percent of all food retail sales in 2004. ICA, the largest firm (with 44 percent), consists of independently owned stores but with a fairly high degree of centralized decision-making. Axfood is a mix of franchisees and fully-owned stores.⁷ Coop, on the other hand, consists of centralized cooperatives where decisions are made at the cooperative level (national or local). Axfood and Coop together have market shares slightly over 20 percent each. Bergendahls (4 percent), operates mainly in the south/south-western parts of the country. In addition, international firms with well-defined discount formats (Netto and Lidl) entered the Swedish market in 2002 and 2003, respectively. So far, they have fairly modest market shares. Finally, independent stores have about 8 percent market share.

Data. The data-set comes from Delfi Marknadsparter AB (DELFI); details of its sources are given in Appendix A. The unit of observation is stores defined by physical location. The data contains yearly information on all retail food stores in the Swedish market during 2001 to 2006, including format, age, owner/firm, sales, sales space, and location. Store format depends on firm/owner, sales space (size), parking, product assortment, etc. Since retail food demand is a function of the market's population but varies across income levels, I connected demographic information from Statistics Sweden(SCB), such as population by age-groups and average income, to the store data from DELFI.

Store format. The retail food industry consists of firms that operate stores in

 $^{^7\}mathrm{Axel}$ Johnson and the D-group merged at the end of the 1990s, to create Axfood, again with fairly centralized decision-making and uniformly-designed stores.

different sizes, where each store has a well defined business concept. The name of a store is usually affiliated with the owner/firm and its self-defined store-concept (such as "very large" or "near you"). The main purpose of the paper is to measure average repositioning costs from one concept to another. However, since the number of store concepts is large (over 30), to reduce the space-dimensionality I group them into 18 *formats* (Table 1), each format containing one or more close store concepts of the same firm. The most important store concepts of one firm are kept in one format, however.

I distinguish cases where, the firm decided to replace one format with another for all its stores in that format. These aggregate format changes, decided at the firm level, are not considered as format repositioning for purposes of this study. On the other hand, a store can change format within or across firms, limited to some extend by sales space available. For example, a hypermarket is not likely to switch to a convenience store (more below). Because of the restricted entry (regulations), the reserve is also unlikely.⁸

To define the possible repositioning alternatives, I re-group the stores in four size groups based on sales space: very large (i.e., hypermarkets), large (i.e., supermarkets), medium (i.e., convenience stores), and small. The four main firms have store concepts in all these four size-groups. I only allow stores to change to other formats within the current size-group or in the next larger or smaller size-group.

Table 1 presents summary statistics for all formats grouped by firm. The largest is *ICA Maxi* (3,3378 m^2) followed by *Coop Large*, *ICA Kvantum*, and *Axfood Willys*, while the smallest is *Others*, i.e., gas station stores, small corner stores and the hard discounters Netto and Lidl. *ICA Maxi* has the highest average sales per square meter (SEK 88,000/ m^2), followed by *Begendahls Vi* (SEK 81,000/ m^2), while *Axfood Handlarń* has the lowest (SEK 36,000/ m^2). *Axfood Vivo* and *ICA Rimi* disappeared from 2001 to 2006, while *Begendahls Vi* and *Coop Nära* appeared.

Figure 1 shows how the numbers of stores by format evolve during the study period. ICA increased the number of *ICA Maxi* hypermarkets, but reduced the number of *ICA Nära* small stores. Axfood increased the number of *Willys* supermarket, and of *Tempo* and *Handlarń* small stores, but reduced the number of *Hemköp* supermarket. During the study period, Coop tries to redefine its store

⁸The Swedish Plan and Building Act (PBA) authorizes the 290 municipalities to decide over applications for new entrants, while inter-municipality cases are handled by the 21 county administrative boards. Several reports have stressed the need to better analyze how regulation affects market outcomes (Pilat 1997; Swedish Competition Authority 2001:4, 2004:2, 2008:5).

formats towards well defined formats such as *Nära* and *Forum* (in different sizes). Finally, besides starting *Vi*, Bergendahls expanded the number of its *Other* stores. **Market definition.** Food products fulfill basic needs and consumers typically travel relatively short distances when buying food (except if prices are sufficiently low). Consequently, nearness to work and home are key aspects for consumers when choosing a store, though the distance likely increases with store-size.⁹

Local markets must be isolated geographic units, such that stores competitively interact only with other stores in the same one. *Postal areas* (in total 1534) are not large enough for large stores, which leaves the 88 *local labor markets* defined by the 290 *municipalities*. The *local labor markets* take commuting patterns into account, which are important for largest stores. *Municipalities* are more appropriate for, while matching the local-government decisions. Therefore, I use municipalities as local markets.

Descriptives. Table 2 presents store characteristics during the study period. The total number of stores decreases from 6,524 in 2001 to 5,953 in 2006 (about 9%), which confirms the trend towards fewer retail stores discussed above. In all years, the number of exits exceed the number of entrants. The number of format repositionings vary between 589 (in 2005) and 243 (in 2006). Average annual sales increases around 25 percent during the period, but only 14 percent for industry as a whole, implies more larger stores at the end of the period. Median population at the local-market (municipality) level increased almost 6 percent, while the median number of families increases by almost 5 percent. The median sales of repositioning stores are lower than the average sales for the full sample, indicating that repositioning is most common among small stores.

The next step is to analyze the difference between repositioning markets (where at least one repositioning occurred during the study period) and non-repositioning markets. There are 52 markets where I observe repositioning every year (2001-2006); 41 with at least one during 4 years; 71 during 3 years; 63 during 2 years; 51 during only 1 year; and 12 markets where there are no format repositionings during the whole period.

Table 3 presents median characteristics for markets with and without repositioning. Repositioning markets have about twice the median population and twice the median number of stores throughout period. Repositioning markets are less

⁹According to surveys made by the Swedish Institute for Transport and Communication Analysis, the average travel distance for trips with the main purpose of buying retail food products was 9.83 kilometers (1995-2002).

concentrated, with C_4 only 0.19 in 2006 compared to 0.39 in non-repositioning markets. They also have more entrants and exits. There is persistent high correlation between the numbers of repositionings and of entries, increasing over time (0.63 in 2002 compared to 0.88 in 2006), as does the correlation between the numbers of repositionings and exits (0.50 in 2002 compared to 0.66 in 2005). These high correlations are why it is important to analyze entry, exit, and repositioning at the same time as done here.

Another important question is what characterizes markets where each format is present. Table 4 shows median characteristics of repositioning and nonrepositioning markets by format. The formats present in most municipalities are *ICA Supermarket* (230 markets), *ICA Nära* (247 markets), and *Coop Medium* (219 markets). For all formats, markets with repositioning have larger median populations than markets without. Median market-share in repositioning markets is lower than in non-repositioning markets for all formats except *Coop OBS* and *Vi.* All formats expect *Rimi*, *OBS et al.*, *Axfood Others*, *Bergendahls Others*, and *Others* have median sales per square meter.

The total numbers of format repositionings from 2001 to 2006 for each firm are presented in Tables 5-8. For example, there are 8 repositionings from *ICA Kvantum* to *ICA Maxi*, and 2 repositionings from *ICA Maxi* to *ICA Kvantum* during the period (Table 5). The highest number of repositionings are from *ICA Nära* to *Others* (220). As mentioned earlier, the format *Rimi* disappeared with most of the stores switching to *ICA Supermarket* (100) and *ICA Nära* (30). There are also many repositionings from *ICA Nära* to Axfood's formats *Handlarń* and *Tempo*.

Table 6 presents repositionings from Axfood, including many from Vivo to Hemköp et al. and to Vi, as well as many from Hemköp et al. to Willys et al. and to Tempo, and from Axfood Others to Handlarń and to Others. The number of repositionings from Bergendahls' formats is small and mostly to other firms (Table 7). Finally, most Coop repositionings were from Coop Medium to Coop Nära or Coop Large, and to other firms' formats such as Handlarń and Bergendahs Others (Table 8).

3 The Modeling Approach

Local markets in the retail industry are characterized by simultaneous entry, exit, and format repositioning. My model, built on the work of Ericson and Pakes (1995), provides a theoretical framework of industry dynamics to account for these features.

All economically important characteristics of stores are included in a vector of commonly-observed state variables. Stores receive state-dependent revenues from selling products and services in each period. Store format, local demand, and competition influence the evolution of the state-vector. Equilibrium is attained when stores follow strategies that maximize the discounted present value of their expected stream of revenues given the expected strategies of their competitors. An important assumption is that stores are maximizing their individual payoffs even if they have the same owner/firm. However, since firms try to avoid cannibalization, and also benefit from economies of scope, common ownership may affect store formats. This effect of common ownership is not modeled explicitly. However I control, for the effect of common ownership on store revenues and policies (as discussed in the Introduction).

Timing. There is an infinite sequence of periods, years in this case. The timing of the simultaneous game is as follows:

- 1. Incumbent stores observe their current store-quality, formats, and local market demographics.
- 2. Each potential entrant receives a draw from the distribution of entry costs, and then makes decisions.
- 3. Consumers choose to buy from a store based on its quality. Consumers and local market demographics generate the store's revenues. There is also a fixed cost for incumbent stores.
- 4. Each store receives a private shock v to its payoffs from choosing a specific format for the following year. The private shocks are assumed i.i.d over stores, formats, and years. After observing its private shock, the store decides its format for the year after that.
- 5. Stores choose their formats. State space variables quality and local market demographics evolve according to stochastic processes described below (section 3.3).

6. Incumbent stores that exit the market receive their sell-off values. Stores that enter pay an entry fee.

Stores that exit of course sell products and services in the year before leaving the market. Furthermore, stores change formats (or not) without knowing the decisions of their competitors. At the beginning of each year, they observe quality and the entry, exit, and repositioning decisions of their rivals in the previous period. Since private shocks are i.i.d., stores do not update their expectations of their rivals future behavior after observing their actions.

Equilibrium. The actions that a store takes in a given period (exit or repositioning) affect current profits and the state variables, and, therefore, future strategic interactions. In this way, the model captures dynamic competition via entry, exit, and repositioning decisions. There are N_m stores in market m, denoted $j = 1, \dots, N_m$, that make decisions at times $t = 1, 2, \dots, \infty$. Store characteristics at t are summarized by the vector of state variables, quality $\boldsymbol{\omega}_t \in \mathbb{R}^{N_m}$. Given states $\boldsymbol{\omega}_t$, the stores choose entry, exit, and repositioning simultaneously. Each store j receives a private shock v_{jt} , drawn independently across stores and over time from a distribution $Q_j(\cdot|\boldsymbol{\omega}_t)$. Differences in each store's productivity might be one explanation for the existence of private shocks. I denote $a_{jt} \in A_j$ the action of store j and $\boldsymbol{a}_t = (a_{1t}, \dots, a_{Nt}) \in \boldsymbol{A}$ the vector of actions at time t. The vector of private shocks is $\boldsymbol{v}_t = (v_{1t}, \dots, v_{Nt})$.

The profits of store j at time t, $\pi_j(\boldsymbol{a}_t, \boldsymbol{\omega}_t, v_{jt})$, depend on its quality (state) ω_{jt} , the actions of all the stores in the market \boldsymbol{a}_t , and the store's private shock v_{jt} . Profits are net of fixed and sunk costs at time t, such as entry costs, and repositioning costs, as well as sell-off value. In addition, all stores are assumed to have a common discount factor $0 < \beta < 1$. Conditional on current quality $\boldsymbol{\omega}_t$, the expected future profit of store j, evaluated prior to realization of the private shock, is

$$\mathbb{E}\left[\sum_{\tau=t}^{\infty}\beta^{\tau-t}\pi_{j}(\boldsymbol{a}_{\tau},\boldsymbol{\omega}_{\tau},\boldsymbol{\upsilon}_{j\tau})|\boldsymbol{\omega}_{t}\right]$$

Finally, to define the transition between states, I assume that quality evolves as an AR(1) process, where the speed of quality-adjustment is estimated from a static-demand model. To be more precise, quality at t + 1, $\boldsymbol{\omega}_{t+1}$, is drawn from a probability distribution $P(\boldsymbol{\omega}_{t+1}|\boldsymbol{a}_t, \boldsymbol{\omega}_t)$. This implies that entry, exit, or changing format might affect future competition. I focus on pure strategy Markov perfect equilibria (MPE). As in Bajari et al. (2007), I assume that there is at least one MPE (Doraszelski and Satterthwaite, 2010) for details on existence and uniqueness). The existence of which implies that each store's behaviour depends on its current quality and its current private shock. If Ω is the quality space, a profile of vector strategies is $\sigma = (\sigma_1, \dots, \sigma_N)$, where $\sigma : \Omega \times \Upsilon_1 \times \dots \times \Upsilon_N \to A, \Upsilon_j$ is the space for the private shock v_j and A is the space of actions. Assuming Markov behaviour implies that store j's expected profit, given state $\boldsymbol{\omega}$, can be written

$$V_{j}(\boldsymbol{\omega}|\sigma) = \mathbb{E}_{\boldsymbol{v}}\left[\pi_{j}(\sigma(\boldsymbol{\omega},\boldsymbol{v}),\boldsymbol{\omega},\boldsymbol{v}_{j}) + \beta \int V_{j}(\boldsymbol{\omega}|\sigma)dP(\boldsymbol{\omega}'|\sigma(\boldsymbol{\omega},\boldsymbol{v}),\boldsymbol{\omega})|\boldsymbol{\omega}\right].$$

A strategy profile σ^* is a Markov perfect equilibrium given opponent profile σ_{-j} if each store j prefers strategy σ_j to all Markov strategies σ'_j , so that

(1)
$$V_j(\boldsymbol{\omega}|\sigma_j^*,\sigma_{-j}^*) \ge V_j(\boldsymbol{\omega}|\sigma_j',\sigma_{-j}^*)$$

for all $j, \boldsymbol{\omega}$, and σ'_{j} .

4 Estimation

Store demand. The utility of consumer *i* from buying from store *j* in market *m* in period *t* is a function of observed and unobserved vector of store characteristics $(\mathbf{x}_{jmt}, \omega_{jmt})$, a vector of observed consumer characteristics \mathbf{z}_{imt} , and a vector of unobserved consumer characteristics $\boldsymbol{\nu}_{imt}$. Unobserved store quality is difficult to quantify but is a determinant of demand. Stores may have the same format but differ in consumers' perceptions of their display, variety offered, advertising, and service, all elements of quality.

Each store's market-share also depends on its format, as well as other store characteristics such as owner, age, sales space, local demand, and competition. Since quality is not directly observed in the data, it is backed out through estimation of the demand model.

The utility of consumer *i* from buying from store *j* is then given by the scalar $u_{ijmt} = u(\mathbf{z}_{imt}, \mathbf{x}_{jmt}, \omega_{jmt}; \boldsymbol{\theta})$, where $\boldsymbol{\theta}$ is a vector of parameters to be estimated. Consumers with different characteristics $(\mathbf{z}_{imt}, \boldsymbol{\nu}_{imt})$ make different store choices. By integrating out the choice function over the distribution of \mathbf{z}_{imt} and $\boldsymbol{\nu}_{imt}$, the

aggregate demand of the store can be obtained. It is assumed that ν_{imt} follows a normal distribution, so that its parameters can be estimated. The mean and standard deviation of ν_{imt} then appear in the utility function as part of the vector θ . Consumer *i* chooses store *j* if and only if

(2)
$$u(\mathbf{z}_{imt}, \boldsymbol{\nu}_{imt}, \mathbf{x}_{jmt}, \omega_{jmt}; \boldsymbol{\theta}) \ge u(\mathbf{z}_{imt}, \boldsymbol{\nu}_{imt}, \mathbf{x}_{rmt}, \omega_{rmt}; \boldsymbol{\theta}) \text{ for } j \neq r$$

where the alternatives $r = 1, \dots, N_m$ represent the competing stores in the market m.

An outside alternative is the option of buying from stores with different formats. I set stores that never changed format during the study period as outside alternative for each local market. In my case the outside alternative is *Others* (Table 1). The presence of this outside alternative allows us to model changes in total sales as a function of store characteristics. Consider

$$B_{jmt} = \{ \boldsymbol{\nu} | u(\mathbf{z}_{imt}, \boldsymbol{\nu}_{imt}, \mathbf{x}_{jmt}, \omega_{jmt}; \boldsymbol{\theta}) \ge u(\mathbf{z}_{imt}, \boldsymbol{\nu}_{imt}, \mathbf{x}_{rmt}, \omega_{rmt}; \boldsymbol{\theta}), \quad j = 0, 1, \cdots, N_m \},\$$

where B_{jmt} is the set of values for $\boldsymbol{\nu}_{imt}$ that induces the choice of store j rather than store r. Assuming that the $F_0(\boldsymbol{\nu})$ provides the density of $\boldsymbol{\nu}$ in the population of interest, the market-share of store j as a function of the characteristics of all stores in the market is

(3)
$$\mathbf{s}_{jmt}(\mathbf{x}_{jmt},\omega_{jmt};\boldsymbol{\theta}) = \int_{\boldsymbol{\nu}\in B_{jmt}} F_0(d\boldsymbol{\nu}).$$

Let $\mathbf{s}(\cdot)$ be the N_m -element vector of functions whose j^{th} component is given by (3)

$$\mathbf{s}(\mathbf{x},\boldsymbol{\omega};\boldsymbol{\theta}) = [s_1(\mathbf{x}_1,\omega_1;\boldsymbol{\theta}),\cdots,s_{N_m}(\mathbf{x}_{N_m},\omega_{N_m};\boldsymbol{\theta})]'.$$

Then, if *pop* is the number of consumers in market m, the vector of demand for the N_m stores is $pop \times \mathbf{s}(\mathbf{x}, \boldsymbol{\omega}; \boldsymbol{\theta})$. In the empirical application I set *pop* equal to the population in each market.

In the demand model, I allow for interaction between individual and store characteristics. Following Berry et al. (1995), I also allow each individual to have different preferences for some observed store characteristics. The randomcoefficients model generated is then

(4)
$$u_{ijmt} = \mathbf{D}_{jmt}\boldsymbol{\theta}_1 + \mathbf{x}_{jmt}\boldsymbol{\theta}_1 + \mu_{ijmt} + \omega_{jmt} + \epsilon_{ijt}$$

where u_{ijmt} is the utility of consumer *i* from buying from store *j* in market *m* in period *t*; \mathbf{D}_{jmt} is a row vector that contains 0 and 1, where 1 indicates the format of the store; the term μ_{ijmt} captures the interaction between the store's format and consumer characteristics; and \mathbf{x}_{jmt} are store characteristics *j* i.e. age and distance to center of zip code; ω_{jmt} is quality of store *j*; and ϵ_{ijt} represents unobserved sources of variation that are independent across consumers, given the store, and across stores, given the consumer. The term μ_{ijmt} contains two components: (i) interaction between observed consumer characteristics (\mathbf{z}_{imt}) and store format (\mathbf{D}_{jmt}), and (ii) interaction between unobserved consumer characteristics (the $\boldsymbol{\nu}_{imt}$) and store format, or

$$\mu_{ijmt} = \mathbf{z}_{imt} \mathbf{D}_{jmt} \boldsymbol{\theta}_2 + \boldsymbol{\nu}_{imt} \mathbf{D}_{jmt} \boldsymbol{\Psi}, \qquad \boldsymbol{\nu}_{imt} \sim N(0, \boldsymbol{\psi}^2),$$

where $\boldsymbol{\nu}_{imt}$ is assumed to have multivariate normal distribution.

In the demand specification, I also control for spatial differentiation between stores. Store location is one of the most important factors that generate sales. Consumers tend to shop closer to their home and work, though of course the choice of store also depends on store format, prices, and product assortment, etc. Location is defined by three digit postal-codes, with the assumption that consumers are located in the center of each one. One of the reasons for using postal-codes is that competition is more intense in smaller areas. With the geographical coordinates of each store and its postal-code, I compute the distance between the store and the center of the postal-code area using the Haversine formula.¹⁰

An alternative measure would be the distance to that store of the same size or format that is closest to the center of the postal-code area. First, the minimum distance for each size-groups (4 different) in the postal-code area is computed. Then, relative distance is computed as the difference between each store's distance and the minimum distance. To be more precise, I give an example. Assume

$$d_{A,B} = 2R \arcsin\left[\min\left\{\left((\sin(0.5(lat_B - lat_A)))^2 + \cos(lat_A)\cos(lat_B)(\sin(0.5(lon_B - lon_A)))^2\right)^{0.5}, 1\right\}\right]$$

 $^{^{10}{\}rm The}$ Haversine formula is based on latitude and longitude measures with R the radius of the earth, the distance between two points A and B is given by

a postal area where three stores operate with the following distances to the center: ICA Nära (300m), Coop Nära (600m), and Tempo (500m). The relative distances are then: ICA Nära (0m), Coop Nära (300m), and Tempo (200m). Consequently, ICA Nära has an advantage over Coop Nära (300m) of being closer to consumers. Choosing size-groups instead of formats to define relative distance allows consumers to shop from other firms that have a store of the same size. The advantage of relative distance is that it models consumer preference for size-groups, but it restricts consumer choices. In the empirical part I use distance to the center of the postal area to account for spatial differentiation.

Demand estimation. The main objective of the demand estimation is to obtain a measure of the store's quality, ω_{jmt} . To back out the quality, I use the same method used in the production function literature to back out the unobserved productivity. I assume that a store's quality evolves as an AR(1) process, and that all customers value it in the same way. The evolution of store quality is estimated using a standard approach for random-coefficient demand (Berry et al., 1995, Nevo, 2000, and Ackerberg et al., 2008). A drawback of this approach is that it assumes that observed product characteristics is exogenous. In my case, ω_{jmt} measures price adjusted quality because the store prices are unobserved. It is a difficult task to construct an informative and consistent price index at the store level, e.g., different stores offer different product varieties. One strategy is to specify a price equilibrium equation and introduce it in the demand specification. However, this might be problematic due to the existence of multiple equilibria.

Innovations in the quality of format repositioning and non-repositioning stores, ξ_{jmt}^{re} and ξ_{jmt}^{nre} , are unknown when stores make decisions for the following year, so that

- (5) Repositioning: $\omega_{jmt} = \rho_1^{re} \omega_{jmt-1} + \rho_0^{re} + \xi_{jmt}^{re}$
- (6) Non-repositioning: $\omega_{jmt} = \rho_1^{nre} \omega_{jmt-1} + \rho_0^{nre} + \xi_{jmt}^{nre}$

where $\xi_{jmt}^{re} \in N(0, \eta^{re})$ and $\xi_{jmt}^{nre} \in N(0, \eta^{nre})$. Moreover, innovations ξ_{imt}^{re} and ξ_{imt}^{nre} are assumed to evolve independently across markets.

I use these innovations to form a set of moment conditions used in the estimation (Berry et al., 1995; Sweeting, 2007). The mean utility provided by store j in market m at time t is then

(7)
$$\delta_{jmt} = \mathbf{D}_{jmt} \overline{\boldsymbol{\theta}}_1 + \mathbf{x}_j \boldsymbol{\theta}_1 + \omega_{jmt} = \mathbf{y}_{jmt} \widetilde{\boldsymbol{\theta}}_1 + \omega_{jmt},$$

where $\tilde{\boldsymbol{\theta}}_1 = (\overline{\boldsymbol{\theta}}_1, \boldsymbol{\theta}_1)$. In the first step of the estimation, I obtain an estimate of quality $\omega_{jmt}(\cdot)$ as a function of $\boldsymbol{\theta}$. In the second step, an identification assumption that $\boldsymbol{\theta} = \boldsymbol{\theta}_0$ is needed, where $\boldsymbol{\theta}_0$ is the true value of $\boldsymbol{\theta}$. In the third step, I use method of moments to find $\boldsymbol{\theta}$.

An approximation of the market-shares conditional on (δ, θ) is given by

(8)
$$s_{jmt}(\boldsymbol{\theta}, \boldsymbol{\delta}) = \int \frac{\exp[\delta_{jmt} + \mu_{ijmt}]}{1 + \sum_{r} \exp[\delta_{rmt} + \mu_{irmt}]} f(\boldsymbol{\nu}) d(\boldsymbol{\nu})$$

Pakes's (1986) simulation method is used to find this approximation, $s_{jmt}(\theta, \delta, P^{ns})$, where P^{ns} is the empirical distribution of the simulation draws.¹¹ For a given θ and the set of simulation draws for ν , the unique values of δ that predict the observed market-shares are found using contraction mapping (Berry et al., 1995). Equation (7) implies that

(9)
$$\omega_{jmt}(\boldsymbol{\theta}, P^{ns}) = \delta_{jmt} - \mathbf{D}_{jmt}\overline{\boldsymbol{\theta}}_1 - \mathbf{x}_j \boldsymbol{\theta}_1 = \delta_{jmt} - \mathbf{y}_{jmt}\tilde{\boldsymbol{\theta}}_1$$

i.e., that quality $\omega_j(\cdot)$ is a function of parameters, the data, and simulation draws. **Identification.** An endogeneity problem arises because unobserved quality ω_{jmt} might be correlated with store-format. For each store, innovation in quality ξ_{jmt} is given by

(10)
$$\xi_{jmt} = \omega_{jmt} - \rho_1 \omega_{jmt-1} - \rho_0 (\delta_{jmt} - \rho_1 \delta_{jmt-1}) - (y_{jmt} - y_{jmt-1}) \hat{\theta}_1 - \rho_0$$

These innovations are uncorrelated with store-format at t and t - 1, which allows us to form the following moment-conditions:

(11)
$$E[w_{jmt}\xi_{jmt}(\boldsymbol{\theta})] = 0,$$

where $\boldsymbol{\theta}$ are all parameters of the demand-system and quality-transitions, and w_{jmt} is a set of instruments. y_{jmt} , y_{jmt-1} , and δ_{jmt-1} together with log of population plus competition variables interacted with store format, are used as instruments to estimate $\boldsymbol{\theta}$ by minimizing $\| G_{N,ns}(\boldsymbol{\theta}) \|$, where

(12)
$$G_{N,ns}(\boldsymbol{\theta}) = \sum_{j} \xi_{jmt}(\boldsymbol{\theta}, P^{ns}) \times w_j.$$

 $^{^{11}}$ When using simulation to compute the integral, a simulation error is introduced, the variance of which error decreases with the number of simulations used. Pakes (1996) and Berry et al. (1995) discuss this in detail.

In addition, in order to identify ρ_1 , I also include the log of market-share at t-1 interacted with an indicator of whether if the store changes format or not.

Payoffs, entry, and exit. A store's full period payoff function depends on whether it is an entrant, a continuing incumbent, a repositioning incumbent, or exits. Store payoff depends on store format, repositioning costs, and fixed costs.

The payoff for an incumbent store j in market m in period t is

(13)

$$\pi_{jmt}(\omega_{jmt}, \boldsymbol{\omega}_{-jmt}; \boldsymbol{\alpha}, \boldsymbol{\gamma}) = r(\boldsymbol{\omega}; \boldsymbol{\alpha}) - \gamma_1 I(f_{jmt+1} \neq f_{jmt}) - \gamma_2 I(f_{jmt} \neq 0) + \gamma_3 \upsilon_{jmt}$$

where $r(\cdot)$ is the revenue function, γ_1 are the sunk costs of repositioning format, and γ_2 are fixed costs paid every period the store operates. The coefficient γ_3 is scale of the i.i.d payoff shocks v_{jmt} , which are assumed to be drawn from a Type I extreme-value distribution. The revenues of store j in market m in period t are:

(14)
$$r_{jmt}(\omega_{jmt}, \boldsymbol{\omega}_{-jmt}) = \boldsymbol{\alpha}_{mt}(1 + K_{jmt}\boldsymbol{\alpha}^K)(1 + H\boldsymbol{\alpha}^H) + \epsilon_{jmt}^r$$

where α_{mt} are the set of year-market fixed-effects dummies. While additional store characteristics are collected in K, H contains variables that measure the competition from other formats in the local market, including estimated storequality of competitors.

Incumbent stores that choose to exit have the payoff

(15)
$$\pi_{jmt}(\omega_{jmt}, \boldsymbol{\omega}_{-jmt}) = \tilde{r}_{jmt} + \gamma_4$$

where γ_4 is the sell-off value associated with closing down the store and exiting the market. Finally, the payoff for an entrant is a simple function of the fixed cost of entry $(entry_f)$:

$$\pi_{jmt}(\boldsymbol{\omega}_{-jmt}) = -sunk_{jmt},$$

where $sunk_{jmt}$ is the sunk cost of entry.

Evolution of state space. The probability of moving to another state is given by the combinations of all paths that lead to that state. To obtain a new state, an incumbent has two options: (i) it stays in the market and moves to a new state; or (ii) it exits and is replaced by a new entrant in the new state. For any change in the state vector, I have to account for the entry, exit, and repositioning decisions of incumbents and potential entrants. I model entry is a restricted way because in my data entry is based on the address. Therefore, in the forward simulations, potential entrants can enter only in locations where stores exit. The probability of entry and exit can be written in terms of optimal entry and exit strategies:

(16)
$$Pr(entry|\omega_j) = \int \Theta(\omega_j, sunk_j) dG(sunk_j)$$

(17)
$$Pr(exit|\omega_j) = \Phi(\omega_j).$$

There are a few further assumptions required. First, because in many cases, entry and exit strategies take the form of simple cutoff rules in dynamic oligopoly models (Beresteanu and Ellickson, 2006), I assume that both conditional probabilities (16) and (17) can be approximated using probit models. Second, from the demandestimation assumption we have that quality evolves stochastically according to the AR(1) processes described by (5) and (6). Third, since each market m is defined by its characteristics, e.g., the total number of formats and population groups, I assume that the growth rates for population groups evolve according to the following AR(1) processes

(18)
$$pop_{mt}^g = \delta_{1,g}^{pop} pop_{mt-1}^g + \delta_{0,g}^{pop} + \upsilon_{mt}^{pop},$$

where g is one of population groups and $v_{mt}^{pop} \sim N(0, \eta^{pop})$.

Value functions. Given the policy functions and the evolution of the state space, the value functions for incumbents and entrants can be computed, giving the expected discounted present value of the store of a given quality. The value function has two components: the per period payoff function (profit of incumbents, or sunk costs of entry for entrants), and the expected value of next period. Stores use the value function to choose their optimal format, entry, or exit. When a store considers changing format, it compares the marginal benefit of having a new quality against the cost of achieving it. Similarly sell-off value at exit is compared with its continuation value.

A potential entrant compares its draw from the distribution of sunk entry-costs against its expected value if it enters. Since private shocks on profits are assumed to be i.i.d, I integrate out all private information in the store's per-period payoff function when computing these value functions. In other words, stores choose their optimal strategy on given the ex-ante value function of next period's potential store qualities. I assume that the potential entrant only lives one year, so that there is no reason to solve for an optimal stopping rule. If it has a higher sunk-costs draw in the current period, it may postpone entering until it receives a more favorable draw. Given current quality and its sunk-costs of entry draw, $sunk_j$, the value function of the potential entrant that decides to enter is

(19)
$$V_{j}^{e}(\boldsymbol{\omega}, sunk_{j}) = max_{f_{j}^{e}} \left\{ -sunk_{j} + \beta \int V_{j}(\boldsymbol{\omega}') dP(\boldsymbol{\omega}'|\boldsymbol{\omega}, \boldsymbol{v}) \right\}.$$

This value function includes the optimal choice of store-format. Since the potential entrant is forward-looking and rational, its expected value of entering accounts for the chosen formats of other stores and their entry and exit decisions. The choice of store-format does not depend on $sunk_j$, i.e., stores choose format f_j^e conditional on entering. In other words, for a given quality there is a draw from the distribution of sunk-costs of entry such that a store would be indifferent between entering or not,

(20)
$$\overline{sunk_j} = \beta \int V_j(\boldsymbol{\omega}') dP(\boldsymbol{\omega}'|\boldsymbol{\omega}, \boldsymbol{v}).$$

The entry function is denoted $\Theta(\omega_j, sunk_j)$. In equilibrium, a store enters the market if it receives a sunk-costs of entry draw less than this value.

If a store decides to leave the market, it obtains profit $\pi_j(\boldsymbol{\omega})$ and sell-off value $scrap_j$. On the other hand, if it stays in the market it receives the following payoff, which depends on the cost of repositioning if it changes format

(21)

$$V_{j}^{stay}(\boldsymbol{\omega}) = max_{f_{j}} \left\{ -\gamma_{1}I(f_{jt+1} \neq f_{jt}) - \gamma_{2}I(f_{j} \neq 0) + \beta \int V_{j}(\boldsymbol{\omega}')dP(\boldsymbol{\omega}'|\boldsymbol{\omega}, \boldsymbol{v}) \right\}.$$

Summarizing, then the value function of an incumbent is a combination of its expected payoffs if it stays in the market and if it exits,

(22)
$$V_j(\boldsymbol{\omega}) = \int V_j(\boldsymbol{\omega}') dP(\boldsymbol{\omega}'|\boldsymbol{\omega}, \boldsymbol{v}) + (1 - \Phi(\omega_j)) V_j^{stay}(\boldsymbol{\omega}) + \Phi(\omega_j) \gamma_4.$$

Distribution of sunk entry-costs. The estimated policy functions describe how a store will behave at each point. In addition, given the primitives of the model, which quantify the benefits and costs of those actions, it is possible to find the distribution of sunk entry-costs (Bajari et al., 2007; Ryan, 2009). Knowledge about

store behaviour whether it enters, exits, or repositions, and the revenues associated with those behaviours, allows computation of the expected value of entry. If that value is positive yet the store did not enter, the store must have received a large entry-costs drawn that made it unprofitable to enter. The distribution of sunk entry-costs can be recovered by matching its cumulative distribution to the predicted probability of entry. A store enters when the value of doing so, $EV^e(\boldsymbol{\omega})$, is larger than $sunk_j$. By simulating many forward paths of possible outcomes given that the firm entered, and averaging over those paths, I obtain the expected value of entry, which I then match against observed rates of entry at different quality states. Therefore, the probability that a store enters is given by

(23)
$$Pr(sunk_j \leq EV^e(\boldsymbol{\omega})) = F^e(EV^e(\boldsymbol{\omega}); \mu_F, \sigma_F^2),$$

where $F^{e}(\cdot)$ is the cumulative distribution of sunk entry-costs. The entry probability estimated, by probit, gives us $Pr(entry|\omega)$. If ns is the number of quality states from which I simulate EV^{e} , I recover the parameters of the distribution by market-size from the following optimization problem:

(24)
$$\min_{\mu_F,\sigma_F} \frac{1}{ns} \sum_{k}^{ns} [Pr(entry|\boldsymbol{\omega}) - F^e(EV^e(\boldsymbol{\omega}))^2; \mu_F, \sigma_F^2).$$

Estimation of sunk repositioning costs. The next step is estimation of average sunk repositioning costs (γ_1), fixed costs of operation (γ_2), the scaling parameter for heterogeneity of sunk costs (γ_3), and the scrap value (γ_4). The estimation is done using Pakes et al. (2007b)'s (PPHI) moment-inequality estimator. The inequalities are formed from the condition that the payoffs obtained using stores' actual strategies must have been higher than those from any alternative strategy (equation 1): $V_j(\boldsymbol{\omega}|\sigma_j^*, \sigma_{-j}^*) - V_j(\boldsymbol{\omega}|\sigma_j', \sigma_{-j}^*) \geq 0$. By construction, the value function

(25)
$$V_{j}(\boldsymbol{\omega}|\sigma_{j}^{*},\sigma_{-j}^{*}) = \mathbb{E}_{\sigma_{j}^{*},\sigma_{-j}^{*}} \sum_{t=0}^{\infty} \beta^{t} r(\cdot;\hat{\boldsymbol{\alpha}}) - \gamma_{1} \mathbb{E}_{\sigma_{j}^{*},\sigma_{-j}^{*}} \sum_{t=0}^{\infty} \beta^{t} I(f_{jt} \neq f_{jt+1}) - \gamma_{2} \mathbb{E}_{\sigma_{j}^{*},\sigma_{-j}^{*}} \sum_{t=0}^{\infty} \beta^{t} I(f_{jt} \neq 0) + \gamma_{3} \mathbb{E}_{\sigma_{j}^{*},\sigma_{-j}^{*}} \sum_{t=0}^{\infty} \beta^{t} \upsilon_{jt}(f_{jt+1}), + \gamma_{4} \mathbb{E}_{\sigma_{j}^{*},\sigma_{-j}^{*}} \sum_{t=0}^{\infty} \beta^{t} I(\chi_{jt+1=1}),$$

where $\mathbb{E}_{\sigma_j^*,\sigma_{-j}^*}$ is the expectation operator over future states conditional on strategies, is linear in γ . Having an estimator for the value function, I use the PPHI estimator, which allows for simulation error in the estimated value function.¹² While the simulation error can also be reduced by increasing the number of forward simulations, this would be expensive here and, therefore, it is more efficient to use the PPHI estimator.¹³

Since we are not able to measure profits $\pi(\cdot)$ exactly, we can calculate an approximation, denoted $\tilde{\pi}(\cdot; \boldsymbol{\gamma})$, which is known up to the parameter-vector $\boldsymbol{\gamma}$. This approximation has the arguments: strategies σ_j and σ_{-j} ; the observed vector of determinants of profits, \boldsymbol{y} ; and the parameter vector $\boldsymbol{\gamma}$. An approximation to the difference in profits that the store would have earned if it had chosen σ'_j instead of σ_j is denoted $\Delta \tilde{\pi}(\sigma_j, \sigma'_j, \cdot)$. The change in true profits can be written as

(26)
$$\Delta \pi(\sigma_j, \sigma'_j, \boldsymbol{\sigma}^*_{-j}, \boldsymbol{y}, \boldsymbol{\gamma}) = \Delta \tilde{\pi}(\sigma_j, \sigma'_j, \boldsymbol{\sigma}^*_{-j}, \boldsymbol{y}; \boldsymbol{\gamma}) + \nu_{1,j,\sigma_j,\sigma'_j} + \nu_{2,j,\sigma_j,\sigma'_j},$$

where ν_1 and ν_2 are unobserved determinants of true profits, differing in what the store knows about them. The store knows ν_2 before it chooses its strategy for the next period, so ν_2 is part of its information set \mathcal{I}_j . I assume that the store-decision does not depend on ν_1 , so $\mathbb{E}[\nu_{1,j,\sigma_j,\sigma'_j}|\mathcal{I}_j] = 0$ by construction and profits are observable up to the parameter-vector $\boldsymbol{\gamma}$ plus an error which is mean-conditional on the store's information set (ν_1) , i.e., I assume that $\nu_{2,j,\sigma_j,\sigma'_j}$ is identically zero for all σ_j and σ'_j .

The PPHI estimator requires that expected profits using the actual strategy σ_j^* be higher than under alternative σ_j'

(27) $\mathbb{E}[\Delta \pi(\sigma_{j}^{*}, \sigma_{j}^{'}, \boldsymbol{\sigma}_{-j}^{*}, \boldsymbol{y})|\mathfrak{I}_{j}] \geq 0.$

But assumption $\mathbb{E}[\nu_{1,j,\sigma_i,\sigma'_i}|\mathcal{I}_j] = 0$, yields the inequality in approximated profits

(28)
$$\mathbb{E}[\Delta \tilde{\pi}(\sigma_{j}^{*}, \sigma_{j}^{'}, \boldsymbol{\sigma}_{-j}^{*}, \boldsymbol{y}, \boldsymbol{\gamma}) | \mathbf{w}_{j}] \geq 0,$$

where $\mathbf{w}_j \in \mathcal{I}_j$. Taking sample averages across observations yields the following moment-inequalities

(29)
$$\frac{1}{N}\sum_{j=1}^{N} \left[\Delta \pi(\sigma_{j}^{*}, \sigma_{j}^{'}, \boldsymbol{\sigma}_{-j}^{*}, \boldsymbol{y}, \boldsymbol{\gamma}) \otimes h(\mathbf{w}_{j}) \right] \geq 0,$$

 $^{^{12}}$ Sweeting (2007) and Holmes (2008) also use the PPHI estimator in their dynamic framework.

 $^{^{13}{\}rm Simulation}$ error appears since store-quality and format-decisions, as well as market demographics, can evolve in so many ways.

where $h(\mathbf{w}_j)$ is a set of instrument functions. The number of moment-inequalities can be increased by increasing the number of alternative strategies or by expanding the number of instruments. The moment-inequalities here are

$$\begin{array}{l} \frac{1}{N}\sum_{j=1}^{N} \left[& (T_{1,\sigma_{j}^{*},\sigma_{-j}^{*}} - T_{1,\sigma_{j}^{\prime},\sigma_{-j}^{*}}) - \gamma_{1}(T_{2,\sigma_{j}^{*},\sigma_{-j}^{*}} - T_{2,\sigma_{j}^{\prime},\sigma_{-j}^{*}}) \\ (30) & -\gamma_{2}(T_{3,\sigma_{j}^{*},\sigma_{-j}^{*}} - T_{3,\sigma_{j}^{\prime},\sigma_{-j}^{*}}) + \gamma_{3}(T_{4,\sigma_{j}^{*},\sigma_{-j}^{*}} - T_{4,\sigma_{j}^{\prime},\sigma_{-j}^{*}}) \\ & +\gamma_{4}(T_{5,\sigma_{j}^{*},\sigma_{-j}^{*}} - T_{5,\sigma_{j}^{\prime},\sigma_{-j}^{*}}) \right] \geq 0, \end{array}$$

for five alternatives of σ'_j , where T_k is the sample average of the term k in equation (25). Using inequality (30), the lower bound for the sunk cost of repositioning from the following inequality:

$$(31) \begin{array}{ccc} \gamma_{1} \geq & \frac{T_{1,\sigma_{j}^{*},\sigma_{-j}^{*}} - T_{1,\sigma_{j}^{'},\sigma_{-j}^{*}}}{T_{2,\sigma_{j}^{*},\sigma_{-j}^{*}} - T_{2,\sigma_{j}^{'},\sigma_{-j}^{*}}} - \gamma_{2} \frac{T_{3,\sigma_{j}^{*},\sigma_{-j}^{*}} - T_{3,\sigma_{j}^{'},\sigma_{-j}^{*}}}{T_{2,\sigma_{j}^{*},\sigma_{-j}^{*}} - T_{2,\sigma_{j}^{'},\sigma_{-j}^{*}}} + \gamma_{3} \frac{T_{4,\sigma_{j}^{*},\sigma_{-j}^{*}} - T_{4,\sigma_{j}^{'},\sigma_{-j}^{*}}}{T_{2,\sigma_{j}^{*},\sigma_{-j}^{*}} - T_{2,\sigma_{j}^{'},\sigma_{-j}^{*}}} \\ + \gamma_{4} \frac{T_{5,\sigma_{j}^{*},\sigma_{-j}^{*}} - T_{2,\sigma_{j}^{'},\sigma_{-j}^{*}}}{T_{2,\sigma_{j}^{*},\sigma_{-j}^{*}} - T_{2,\sigma_{j}^{'},\sigma_{-j}^{*}}} \end{array}$$

(if $T_{2,\sigma_j^*,\sigma_{-j}^*} < T_{2,\sigma_j',\sigma_{-j}^*}$). By increasing the number of format repositioning both the revenues and cost of repositioning increase. Thus, this policy that is not preferred by stores gives us the lower bound for the repositioning cost. The upper bound is obtained using a strategy that decreases the number of repositionings, which implies a decrease in revenues:

$$(32) \begin{array}{c} \gamma_{1} \leq & \frac{T_{1,\sigma_{j}^{*},\sigma_{-j}^{*}} - T_{1,\sigma_{j}^{*},\sigma_{-j}^{*}}}{T_{2,\sigma_{j}^{*},\sigma_{-j}^{*}} - T_{2,\sigma_{j}^{*},\sigma_{-j}^{*}}} - \gamma_{2} \frac{T_{3,\sigma_{j}^{*},\sigma_{-j}^{*}} - T_{3,\sigma_{j}^{*},\sigma_{-j}^{*}}}{T_{2,\sigma_{j}^{*},\sigma_{-j}^{*}} - T_{2,\sigma_{j}^{*},\sigma_{-j}^{*}}} + \gamma_{3} \frac{T_{4,\sigma_{j}^{*},\sigma_{-j}^{*}} - T_{4,\sigma_{j}^{*},\sigma_{-j}^{*}}}{T_{2,\sigma_{j}^{*},\sigma_{-j}^{*}} - T_{2,\sigma_{j}^{*},\sigma_{-j}^{*}}} \\ + \gamma_{4} \frac{T_{5,\sigma_{j}^{*},\sigma_{-j}^{*}} - T_{5,\sigma_{j}^{*},\sigma_{-j}^{*}}}{T_{2,\sigma_{j}^{*},\sigma_{-j}^{*}} - T_{2,\sigma_{j}^{*},\sigma_{-j}^{*}}} \end{array}$$

(if $T_{2,\sigma_{j}^{*},\sigma_{-j}^{*}} > T_{2,\sigma_{j}^{'},\sigma_{-j}^{*}}$).

The lower bound for the fixed cost is zero. The upper bound of the fixed cost is obtained implementing a strategy that reduces the number of format repositioning:

$$(33) \begin{array}{ccc} \gamma_{2} \leq & \frac{T_{1,\sigma_{j}^{*},\sigma_{-j}^{*}} - T_{1,\sigma_{j}^{\prime},\sigma_{-j}^{*}}}{T_{3,\sigma_{j}^{*},\sigma_{-j}^{*}} - T_{3,\sigma_{j}^{\prime},\sigma_{-j}^{*}}} - \gamma_{1} \frac{T_{2,\sigma_{j}^{*},\sigma_{-j}^{*}} - T_{2,\sigma_{j}^{\prime},\sigma_{-j}^{*}}}{T_{3,\sigma_{j}^{*},\sigma_{-j}^{*}} - T_{3,\sigma_{j}^{\prime},\sigma_{-j}^{*}}} + \gamma_{3} \frac{T_{4,\sigma_{j}^{*},\sigma_{-j}^{*}} - T_{4,\sigma_{j}^{\prime},\sigma_{-j}^{*}}}{T_{3,\sigma_{j}^{*},\sigma_{-j}^{*}} - T_{3,\sigma_{j}^{\prime},\sigma_{-j}^{*}}} \\ + \gamma_{4} \frac{T_{5,\sigma_{j}^{*},\sigma_{-j}^{*}} - T_{5,\sigma_{j}^{\prime},\sigma_{-j}^{*}}}{T_{3,\sigma_{j}^{*},\sigma_{-j}^{*}} - T_{3,\sigma_{j}^{\prime},\sigma_{-j}^{*}}} \end{array}$$

(if $T_{3,\sigma_j^*,\sigma_{-j}^*} > T_{3,\sigma_j',\sigma_{-j}^*}$). In the empirical section, this strategy is implemented by reducing by 0.05 the probability to repositioning in every state. Another alterna-

tive strategy is not allowing repositioning.

A store that changes the format receives a large draw of ν . Large repositioning costs can explain why some stores never change their format even if they receive favorable draws of ν . The upper bound of the scale parameter γ_3 is obtained using an alternative strategy that increases ν , reduce the revenues, and makes repositioning a constant strategy, i.e., format choices are made equal. I implement this strategy by equalizing the choice probabilities for other formats and leaving unchanged the choice of the current format (Sweeting, 2007).

$$(34) \begin{array}{ccc} \gamma_{3} \leq & \frac{T_{1,\sigma_{j}^{*},\sigma_{-j}^{*}} - T_{1,\sigma_{j}^{'},\sigma_{-j}^{*}}}{T_{4,\sigma_{j}^{*},\sigma_{-j}^{*}} - T_{4,\sigma_{j}^{'},\sigma_{-j}^{*}}} + \gamma_{1} \frac{T_{2,\sigma_{j}^{*},\sigma_{-j}^{*}} - T_{2,\sigma_{j}^{'},\sigma_{-j}^{*}}}{T_{4,\sigma_{j}^{*},\sigma_{-j}^{*}} - T_{4,\sigma_{j}^{'},\sigma_{-j}^{*}}} + \gamma_{2} \frac{T_{3,\sigma_{j}^{*},\sigma_{-j}^{*}} - T_{3,\sigma_{j}^{'},\sigma_{-j}^{*}}}{T_{4,\sigma_{j}^{*},\sigma_{-j}^{*}} - T_{4,\sigma_{j}^{'},\sigma_{-j}^{*}}} \\ + \gamma_{4} \frac{T_{5,\sigma_{j}^{*},\sigma_{-j}^{*}} - T_{5,\sigma_{j}^{'},\sigma_{-j}^{*}}}{T_{4,\sigma_{j}^{*},\sigma_{-j}^{*}}} \end{array}$$

(if $T_{4,\sigma_j^*,\sigma_{-j}^*} < T_{4,\sigma_j',\sigma_{-j}^*}$). The drawback of this strategy is that the store can also choose to exit.

The next step is to propose an alternative strategy to estimate the upper bound of the sell-off value (scrap value), γ_4 . This bound is obtained by increasing the likelihood to exit. In the empirical implementation, I increase exit probability by 0.05.

$$(35) \begin{array}{rcl} \gamma_{4} \leq & -\frac{T_{1,\sigma_{j}^{*},\sigma_{-j}^{*}} - T_{1,\sigma_{j}^{'},\sigma_{-j}^{*}}}{T_{5,\sigma_{j}^{*},\sigma_{-j}^{*}} - T_{5,\sigma_{j}^{'},\sigma_{-j}^{*}}} + \gamma_{1} \frac{T_{2,\sigma_{j}^{*},\sigma_{-j}^{*}} - T_{2,\sigma_{j}^{'},\sigma_{-j}^{*}}}{T_{5,\sigma_{j}^{*},\sigma_{-j}^{*}} - T_{5,\sigma_{j}^{'},\sigma_{-j}^{*}}} + \gamma_{2} \frac{T_{3,\sigma_{j}^{*},\sigma_{-j}^{*}} - T_{3,\sigma_{j}^{'},\sigma_{-j}^{*}}}{T_{5,\sigma_{j}^{*},\sigma_{-j}^{*}} - T_{5,\sigma_{j}^{'},\sigma_{-j}^{*}}} \\ - \gamma_{3} \frac{T_{4,\sigma_{j}^{*},\sigma_{-j}^{*}} - T_{5,\sigma_{j}^{'},\sigma_{-j}^{*}}}{T_{5,\sigma_{j}^{*},\sigma_{-j}^{*}} - T_{5,\sigma_{j}^{'},\sigma_{-j}^{*}}} \end{array}$$

(if $T_{5,\sigma_j^*,\sigma_{-j}^*} < T_{5,\sigma_j',\sigma_{-j}^*}$). Due to heterogeneity across Swedish local markets, I estimate the cost-parameters for a groups of markets.¹⁴

5 Results

This section presents the results from: demand estimation; revenue generating function estimation; entry, exit, and repositioning policies; and repositioning and entry costs.

Demand. Table 9 shows the estimates from the demand model presented in Section 3.1. The demographic format-taste parameters show to some extent expected

¹⁴Appendix C presents an alternative estimator for repositioning costs.

patterns. Stores in markets with a large proportion of children (population 0-14) tend to have large and very large formats (Table 9). *ICA Supermarket* is the most preferred format in these markets. The very large formats *ICA Maxi* (1.03) and *Coop Large* (0.88) are are preferred in these markets. *ICA Kvantum*, also a large format, is found to be less preferred in markets with a high proportion of children, however. Perhaps the effect of demographics is captured by other demographics. Axfood's large format *Willys et al.*, which is promoted as "Sweden's cheapest bag of groceries", is the one preferred in markets with the most kids (1.25), however. Young adults (15-34) prefer small formats such as *ICA Nära* (0.69), *Tempo* (0.80), and *Coop Nära* (1.99). *ICA Maxi* (0.68) is the most preferred very large format in markets with large share of young adults. The medium format *ICA Supermarket* (0.82), and small formats such as *Tempo* (1.98) are preferred in markets with large share of older people (over 65). As with young adults, *ICA Maxi* is the most preferred in markets with large share of older people.

As discussed earlier, distance to the center of postal-code (three digit postalcode area) is used to control for spatial differentiation in the demand estimation, allowing us to explore the benefits of a store being closer to consumers. The estimated coefficients of distance represents the travel costs. *ICA Supermarket, Vi*, and *Coop Nära* are the formats that are close to the consumers. However, they are found to have relatively large travel cost. Consumers are less likely to choose these formats based on distance, i.e., there are other factors that affect consumers choices.

The age of the store is positive (0.65) and significant at traditional levels. Perhaps indicating that older stores might have better location. Sales space is also positive and significant, because large stores usually provide a wider range of products and lower prices.

The standard deviations of the random components format-tests are small and mostly are insignificantly different from zero. Demographics, therefore, seems to capture most of the systematic differences in tastes for formats within markets. Given the small standard deviation of the random coefficient on *Others*, there is an acceptable degree of substitution with the outside good.

As explained earlier, unobserved is backed out quality from the demand estimation. The AR(1) parameters for repositioning and non-repositioning are both less than one. This implies that the processes that describe quality are stationary, through quality is more persistent for the non-repositioning stores ($\rho_{nre} = 0.62$), as one might expect. Repositioning stores have higher quality than non-repositioning stores in the upper tail of the distribution, but lower in the lower tail. The median quality of a repositioning store is larger that of a non-repositioning store, however.

Sales generating function estimation. Table 10 shows estimates of four salesgenerating function specifications, each including market demographics and store characteristics such as age and quality. Including quality and quality squared allows control for its effect on sales.

Model 2 adds repositioning, while Model 3 also controls for format competition. Model 4 also controls for the effect of current on future sales (non-linear sales-effects) by including a dummy that specifies whether the store's marketshare is less than the market-average. Increasing in sales due to an increase in store quality decreased with quality in all four models.

Increasing the proportion of kids has a positive impact on sales, while the effect of young adults is negative, perhaps because a higher proportion of this group might imply a lower income-level relative to the control group (35-65). While the store's age has a positive effect on sales (Model 4), the distance has a negative impact. As might be expected, both the number of stores in the same format and the number of stores owned by other firms have negative effects on sales. The direct effect of repositioning is positive, but not significant, which is not surprising since it might take time for consumers to adjust to the new format. Low previous market-share has a negative impact on sales.

To estimate sunk costs of entry and repositioning, I use Model 4 since it provides the highest correlation between observed and predicted sales for both repositioning and non-repositioning stores.

Policy functions. Table 11 reports the multinomial logit estimates of incumbent stores' format strategies. The first part of the table reports the marketdemographic variables that affect the prevalence of each format, i.e., the proportion of the population in by age group and changes in those groups. In markets with a large proportion of children, stores are more likely to choose large formats such as *ICA Maxi* (345), *ICA Kvantum* (334), *Coop Large* (336), and *Bergendahls Others* (341). In markets with a large proportion of young adults or with increasing proportion of young adults, large formats such as *ICA Maxi* (72), *Coop Large* (49), and *Willys et al.* (16) are more likely. In markets with a large proportion of adults, *Coop Nära* is the most preferred format.

Stores with high quality are more likely to be in large formats (*ICA Maxi, Coop Large*). Axfood's small formats, *Tempo* and *Handlarń*, are also associated to have

high quality (Axfood increased the number of stores in these formats during the period).

In markets where rivals have high quality, ICA focuses most on *ICA Kvantum* (0.098), *ICA Supermarket* (0.018), and *ICA Nära* (0.020); Axfood on *Handlarń* (0.039) and Coop on *COOP Medium* (0.026). Thus, three of ICA's formats are more likely in markets where rivals have high quality (product differentiation). All formats are less likely the more other stores there were the same format. The positive parameters on other formats show that stores try to differentiate in format.

The right part of Table 11 reports the coefficients on age, distance from other stores in rival format, and market-share. I allow the coefficients to differ depending on whether a store remains in the same format or changes format. Old stores in large formats are less likely to change format. Old stores that change are less likely to be in one of ICA's formats. On the other hand, old stores that change are more likely to become *Coop Nära*, *Axfood Others*, or *Hemköp et al.*. Stores far from others that changed format are more likely to become *Bergendahls Others*, *Handlarń*, or *OBS et al.*.

Table 12 reports estimates of entry and exit policies, aggregated for all formats, but accounting for market and format fixed-effects. Stores with high quality are less likely to exit. Exit is also less likely in large markets, and if the firm has many stores of the same format in the same market, which might indicate economies of density (Holmes 2008). On the other hand, stores are more likely to exit if they are old or if rivals have high quality.

Entry is more likely if rivals have high quality, and in markets with large population. Entry is less likely if the firm already operates many stores in that format in the same market, or if there are many other-firm rivals.

Repositioning costs and sell-off values. Using estimated quality, sales estimates, and policy functions, I can recover the cost parameters. Table 13 presents repositioning-cost estimates (identified sets) for markets with population below 20,000, population 20,000-60,000, and over 60,000, as well as median sales of repositioning stores, and the median number of repositioning stores per year during the period.

The confidence intervals for the estimated parameters are simulated (see Pakes et al., 2007b). To apply bootstrap is very time consuming because of the forward simulations. In addition to PPHI, there are different approaches for inferences with moment inequalities (Imbens and Manski, 2004; Chernozhukov et al., 2007; and Andrews and Soares, 2010). In this case, the confidence intervals are the extreme

points of identified set. PPHI suggests a simulation method to construct the "inner" and "outer" confidence intervals. These intervals are asymptotically the true confidence intervals for the estimated bounds. I only report the "outer" threshold, i.e., the conservative values. The "outer" confidence intervals are obtained using 100 simulations.¹⁵ The small number of simulations is due to computation burden, i.e., 100 programming problems have to be solved.¹⁶

Average repositioning costs depend on the size of the market, being about 63 percent higher in markets with population 20,000-60,000 than in smaller markets, and another 16 percent higher in the largest.¹⁷ Median sales are more than 20 times higher than average repositioning costs, especially high for markets of 20,000-60,000. By far most repositioning are in large markets, though repositioning seems to be most profitable in middle size-group ones. Sell-off values on exit are twice as high in the large markets as in the small ones. Due to store heterogeneity, there is a large spread in sell-off values in large markets.

For robustness, Table 15 (Appendix C) shows the point estimates of the repositioning and sell-off values using the minimum distance estimator. The values of estimated parameters belong to the identified sets estimated using PPHI.

Entry costs. Table 14 shows estimated sunk costs of entry, again for markets grouped by size. It is far more expensive to enter a market than to reposition. This might explain why the number of repositioning is larger than the number of entrants. Mean entry costs increases substantially with market size. Entry costs are more than twice median sales in small markets, over four times median sales in medium markets, and a bit less in large markets. Thus, there is higher entry costs per median sales in medium than in large markets. The large estimation of entry costs in large markets may be due to big entrants that shift the mean.

 $^{^{15}\}mathrm{In}$ Pakes et al. (2007b), Section 3.1.2 presents the estimation details of the confidence intervals.

¹⁶IPOPT optimizer is used (https://projects.coin-or.org/Ipopt). Ipopt is an optimizer for large scale nonlinear problems. In case that absolute norm is used, it is computationally efficient to used linear programming solvers. For example, the GLPK (http://www.gnu.org/software/glpk/), which is a package for solving large-scale linear programming.

¹⁷To get a conservative estimate, I used the upper bound to measure the changes in costs.

6 Conclusions

While there have been important contributions in modelling both static and dynamic consumer demand for differentiated products, there have been few attempts at modelling their supply. The fact the products are differentiated means that shocks might cause firms to change their product assortment.

A dynamic oligopoly model is estimated in Swedish retail food industry to measure the costs associated with repositioning by changing store formats, which also often includes major changes in product assortment. The estimation gives important information about driving forces behind repositioning, associated costs, and how it can be linked to entry and exit.

There are high correlations between entry, exit, and repositioning. More generally, the paper provides a framework for studying repositioning in any industry where entry and exit are important. Understanding the potential role of repositioning in the trade-off between repositioning, entry, and exit is one of the aims of this paper. Since entry is regulated in most OECD countries, knowledge about repositioning and entry costs has important implications for policy.

In the Swedish retail food industry, repositioning costs increase with the size of the market, though at a decreasing rate. Stores are less likely to exit if they have high quality, are in large markets, or if there are many same-format stores in the same market. Entry is more likely in large markets or if rivals have high quality, so that there is room for product differentiation. Stores with high quality are more likely to be in large formats. Old stores are less likely to reposition. Distance is an important factor for consumers choosing a store. Finally, store's quality is more persistent for non-repositioning stores.

In future work, cost estimates of the four different size groups (see Section 2) of stores will provide valuable information about repositioning. Having the estimated structural parameters, different policy experiments can be implemented. My interest is to evaluate how the cost of repositioning could be affected by lowering the cost of entry, and how sunk cost of entry could be affected by lowering the cost of repositioning. I modelled multi-product ownership, i.e., the fact that each firm operates many stores, in a limited way. Since I only control for owners in stores' policies, I did not explicitly model the dynamics of their product selection. Selection of products in multi-product firms is an important topic for future research. Understanding it will give us more information about market power, business cannibalization, and economies of scope.

References

- ACKERBERG, D., L. BENKARD, S. BERRY, AND A. PAKES (2008): Handbook of Econometrics, vol. 6, chap. Econometric Tools for Analyzing Market Outcomes, pp. 4171–4276. Elsevier.
- ACKERBERG, D., K. CAVES, AND G. FRASER (2006): "Structural Identification of Production Functions," Mimeo, UCLA.
- AGUIRREGABIRIA, AND VICENTINI (2006): "Dynamic Spatial Competition between Multi-Store Firms," Mimeo, University of Toronto.
- AGUIRREGABIRIA, V., AND P. MIRA (2007): "Sequential Estimation of Dynamic Discrete Games," *Econometrica*, 75(1), 1–53.
- ANDREWS, D. W. K., AND G. SOARES (2010): "Inference for Parameters Defined by Moment Inequalities Using Generalized Moment Selection," *Econometrica*, 78, 119–157.
- BAJARI, P., L. BENKARD, AND J. LEVIN (2007): "Estimating Dynamic Models of Imperfect Competition," *Econometrica*, 75(5), 1331–1370.
- BAYER, C., AND J. FALKO (2006): "On the Dynamics of Interstate Migration: Migration Costs and Self-Selection," mimeo, University of Dortmund.
- BERESTEANU, A., AND P. ELLICKSON (2006): "The Dynamics of Retail Oligopoly," Mimeo, Duke University.
- BERRY, S., J. LEVINSOHN, AND A. PAKES (1995): "Automobile Prices in Market Equilibrium," *Econometrica*, 63(4), 841–890.
- BLUNDELL, R., AND S. BOND (2000): "GMM Estimation with Persistent Panel Data: An Application to Production Functions," *Econometric Reviews*, 31(3), 549–569.
- CHERNOZHUKOV, V., AND H. HONG (2003): "A MCMC Approach to Classical Estimation," *Journal of Econometrics*, 115(2), 293–346.
- CHERNOZHUKOV, V., H. HONG, AND E. TAMER (2007): "Estimation and Confidence Regions for Parameter Sets in Econometric Models," *Econometrica*, 75, 1243–1284.
- COLLARD-WEXLER, A. (2006): "Demand Fluctuations and Plant Turnover in the Ready-Mix Concrete Industry," (Mimeo, New York University).
- DAVIS, P. (2006): "Spatial Competition in Retail Markets: Movie Theaters," RAND Journal of Economics, 37(4), 964–982.
- DORASZELSKI, U., AND M. SATTERTHWAITE (2010): "Computable Markov-Perfect Industry Dynamics," *Rand Journal of Economics*, 41(2), 215–243.

- ERICSON, R., AND A. PAKES (1995): "Markov-Perfect Industry Dynamics: A Framework for Empirical Work," *Review of Economic Studies*, 62, 53–83.
- EUROPEAN COMMISION (2008:321): "Tackling the Chalenge of Rising Food Prices Directions for EU action," (COM).
- GANDHI, A., L. FROEB, S. TSCHANTZ, AND G. WERDEN (2008): "Post-Merger Product Repositioning," *The Journal of Industrial Economics*, LVI(1), 49–67.
- GEMICI, A. (2007): "Family Migration and Labor Market Outcomes," mimeo, University of Pennsylvania.
- Ho, K. (2007): "Insurer-Provider Networks in the Medical Care Market," Discussion paper, mimeo, Columbia University.
- HOLMES, T. (2008): "The Diffusion of Wal-Mart and Economies of Density," NBER Working Paper 13783.
- HOTZ, V., AND R. MILLER (1993): "Conditional Choice Probabilities and the Estimation of Dynamic Models," *Review of Economic Studies*, 60(3), 497–529.
- HOTZ, V., R. MILLER, S. SANDERS, AND J. SMITH (1994): "A Simulation Estimator for Dynamic Models of Discrete Choice," *Review of Economic Studies*, 61(2), 265–289.
- IMBENS, G. W., AND C. F. MANSKI (2004): "Confidence Intervals for Partially Identified Parameters," *Econometrica*, 72, 1845–1857.
- ISHII, J. (2005): "Compatibility, Competition and Investment in Network Industries: ATM Networks in the Banking Industry," Discussion paper, mimeo, Stanfort University.
- JIA, P. (2008): "What Happens when Wal-Mart comes to Town: An Empirical Analysis of the Discount Retailing Industry," *Econometrica*, 76(6), 1263–1316.
- KENNAN, J., AND J. R. WALKER (2006): "The Effect of Expected Income on Individual Migration Decisions," mimeo, University of Wisconsin-Madison.
- LEVINSOHN, J., AND A. PETRIN (2003): "Estimating Production Functions Using Inputs to Control for Unobservables," *Review of Economic Studies*, 70(2), 317–341.
- MACIEIRA, J. (2006): "Measuring the Effects of Innovation Subsidies: The Case of Supercomputers," Discussion paper, mimeo, Northwestern University.
- MAICAN, F. (2008): "Dynamic Store Adjustment: Switching Store Formats in Retail," mimeo, Department of Economics, Göteborg University.

- NEVO, A. (2000): "A Practitioner's Guide to Estimation of Random-Coefficients Logit Models of Demand," *Journal of Economics and Management Strategy*, 9(4), 513–548.
 - (2001): "Mergers with Differentiated Products: The Case of the Ready-to-Eat Cereal Industry," *Econometrica*, 69(2), 307–342.
- OLLEY, S., AND A. PAKES (1996): "The Dynamics of Productivity in the Telecommunications Equipment Industry," *Econometrica*, Vol. 64(No. 6), 1263– 1297.
- PAKES, A., M. OSTROVSKY, AND S. BERRY (2007): "Simple Estimators for the Parameters of Discrete Dynamic Games, with Entry/Exit Examples," *RAND Journal of Economics*, 38(2), 373–399.
- PAKES, A., J. PORTER, J. HO, AND J. ISHII (2007): "Moment Inequalities and their Application," Discussion paper, mimeo, Harvard University.
- PESENDORFER, M., AND SCHMIDT-DENGLER (2003): "Identification and Estimation of Dynamic Games," Mimeo, London School of Economics.
- PILAT, D. (1997): "Regulation and Performance in the Distribution Sector," OECD Working Papers 180.
- RYAN, S. (2009): "The Costs of Environmental Regulation in a Concentrated Industry," Mimeo, MIT.
- SWEDISH COMPETITION AUTHORITY (2001:4): "Kan Kommunerna Pressa Matpriserna? (Can the Municipalities put Pressure on Prices?)," Technical Report 4, Stockholm.

(2004:2): "Konsumenterna, Matpriserna och Konkurrensen (Consumers, Retail Food Prices, and Competition)," Technical Report 2, Stockholm.

(2008:5): "Action for Better Competition (Åtgärder För Bättre Konkurrens)," Technical Report 5, Stockholm.

SWEETING, A. (2007): "Dynamic Product Repositioning in Differentiated Product Markets: The Case of Format Switching in the Commercial Radio Industry," Discussion Paper 13522, NBER.
Format	Firm	Store concepts into Sa	ales space	Average sales			Number o	f stores		
		the defined format	m^2	per m^2 (SEK)	2001	2002	2003	2004	2005	2006
1	ICA	ICA Maxi	3,378	88	31	32	34	35	41	49
2	ICA	ICA Kvantum	2,205	75	119	121	124	126	127	122
3	ICA	ICA Supermarket	875	99	450	437	534	524	496	484
4	ICA	ICA Nära	263	52	1,226	1,134	1,152	1,044	937	930
5	Axfood	Willys et al.	1,559	69	56	26	111	135	146	150
9	Axfood	Vivo	745	02	253	121	91	88	0	0
7	Axfood	Hemköp et al.	1,214	56	180	269	251	224	174	161
×	Axfood	Handlarn	167	36	117	172	187	208	217	221
6	Axfood	Tempo	286	43	89	103	103	111	147	151
10	ICA	Rimi	801	63	134	135	0	0	0	0
11	Axfood	Axfood Others	281	61	281	172	162	129	158	157
12	COOP	Coop Nära	302	68	0	0	0	0	97	114
13	COOP	Coop Large	2,349	59	63	09	61	88	66	112
14	COOP	Coop Medium	535	54	801	815	786	725	576	508
15	COOP	OBS et al.	1,382	65	122	61	60	56	40	25
16	Bergendahls	Vi	867	81	0	0	0	0	86	98
17	Bergendahls	Bergendahls Others	1,290	47	62	73	80	83	92	108
18	Others	Others	134	45	2,540	2,447	$2,440\ 2$	638	2,679	2,563
NOTES:	Data from DEI	LFI. Average sales per square meter are in 20	001 thousar	d SEK (1 EUR-	=9 SEK,	1 USD=	= 8 SEK). I	Each for	mat cont	ains
one or m	ore store concer	ots. ICA Nära includes also ICA Express stor	res: Willius	et al. includes W	illys and	Willvs	Hemma: H	emkön ei	al incl	ndes

 Table 1: Store formats by firm

 Format
 Firm
 Store concepts

Exer, Smores corrections for the and Billialis, *Axfood Others* (format number an includes Jour Livs, Matraira, Axfood Nariys Raft Pris, Exer, Spar, Matex, Axfood Storifvs, and Billialis, *Axfood Others* (format number 11) includes Jour Livs, Matraira, Axfood Nariys Raft Pris, Axfood Lågpris, and Östenssons; *Coop Large* (format number 13) includes COOP Forum level 2 and 3, B & W, Gröna Konsum, COOP Extra, and Konsum Extra; *Coop Medium* includes stores Coop Konsum level 1 and 4; *OBS et al.* includes OBS, Robin Hood, Donns, Prix, and Fakta, *Bergendulls Others* (format number 17) includes Favör, Matöppet, Prixstra, City Gross, and Ekohallen, Eko, and AG-Favör; *Others* includes Statiol, OKQ8, Nära Dej, Matbulken, Fri Mat, "Eleven, Presshyrån Rivs, Shell, Preem, Bilisten, Lidl, Netto, Norsk Hydro, Din-X, Uno-X, Pump, Samuelsons Servicehandeh, Pressbyrån Franchise, Q Star, and yet others. NC



Figure 1: Evolution of the number of stores by format (see Table 1) during 2001-2006.

2001 - 2006	
characteristics,	
Store	
ä	
Table	

Year	# of	# of Repos.	# of	# of	Average	Total	Format	repositioni	ing sales	Median	Median
	Stores	Format	Entries	Exits	Sales	Sales				pop.	families
							25%	50%	75%		
2001	6,524		1	339	23,594	153,927,750				38,706	19,852
2002	6,228	481	94	167	25,368	157,993,250	5,500	17,500	45,000	38,706	19,852
2003	6,176	321	134	239	27,080	167, 247, 500	12,500	35,000	55,000	38,706	19,852
2004	6,214	300	311	261	27,386	170, 178, 020	3,500	9,000	35,000	39,145	20,410
2005	6,112	589	171	293	28,242	172, 613, 250	4,500	12,500	27,500	39,477	20,554
2006	5,953	243	134		29,463	175, 395, 290	4,500	9,000	31,250	40,873	20,770
NOTE	S: Sales are	reported in thou	usands of 2	2001 SE	K. Populat	ion and number	rs of familie	es are at m	inicipality	level (290).	

	$\Im or(R, X)$.50	.53	.80	.66		re all	(X)	
	$(R,E) \in$		0	0	0	0	1	arkets aı	or exit	
	Cor		0.63	0.72	0.78	0.76	0.88	ning ma	and (R)	
	C_4	Non- Benos	0.365	0.370	0.358	0.450	0.394	-Repositio	ries (E) , i	
arkets		Repos.	0.236	0.168	0.165	0.221	0.186	year. Non	 and ent 	
ioning m	xits (X)	Non- Repos	1.0	1.0	2.0	1.0	'	during the	itionings (1	
n-reposit	# of E	Repos.	3.5	3.5	4.5	4.0	,	g occurred	rs of repos	
and no	tries (E)	Non- Benos	1.0	1.0	2.0	1.0	2.5	positionin	the numbe	
sitioning	# of Er	Repos.	2.5	3.0	5.0	3.5	2.5	least one r	s between	
s of repc	Stores	Non- Benos	15.5	17.0	18.5	13.0	17.0	e where at	correlation	urred.
cteristic	; jo #	Repos.	33.5	32.5	32.0	31.0	31.0	ts are those	R, X) are	ioning occı
an chara	ation	Non- Benos	10,549	11,014	12,615	11,270	11,804	<i>ving</i> marke	and Cor(ere reposit
3: Medi	Popul	Repos.	21,450	26,355	22,319	23,104	27,120	Reposition	Cor(R, E)	narkets wh
Table	Year		2002	2003	2004	2005	2006	NOTE:	others.	across n

2004 200 34 39	5 2006	Renos	Mon	¢	Mon	¢	
34 39		Techor.	-TION	Repos.	-TION	Repos.	Non-
34 39			Repos.		Repos.		Repos.
	45	69,741	45,266	0.233	0.284	89.10	83.05
102 104	66	37,057	26,270	0.193	0.290	72.29	68.77
233 233	230	23,922	13, 324	0.056	0.145	61.87	61.51
258 248	247	24,935	13,725	0.014	0.024	46.79	43.75
89 93	66	49,863	30,532	0.086	0.136	59.21	56.94
20 0	0	54, 357	32,269	0.023	0.036	76.24	53.25
125 108	100	33,234	20,116	0.051	0.131	53.57	52.76
124 128	126	27, 120	15,453	0.006	0.012	30.77	30.22
68 81	90	36,690	21,715	0.011	0.021	40.12	39.56
0 0	0	37,448	30, 229	0.047	0.064	57.41	61.11
73 84	82	37,052	20,017	0.003	0.013	40.34	42.86
0 41	52	53,302	24,953	0.013	0.021	64.10	53.52
70 79	89	49,918	25,764	0.122	0.147	60.24	47.60
244 232	219	23,947	12,820	0.026	0.056	49.64	46.36
45 35	24	26,300	26,131	0.151	0.149	54.69	55.63
0 21	сс С	56,467	29,475	0.023	0.022	70.00	50.00
47 52	55	33,494	19,868	0.020	0.055	41.20	45.00
287 289	287	22, 328	13,000	0.006	0.011	32.25	31.34
positioning	occurred	during tl	ie year.	Non-Rep	osition in	g market	s are all
244 232 45 35 0 21 47 52 287 289 positioning	219 24 33 55 287 occurred	23,94 26,30 56,46 33,49 22,32 during	C 0 7 4 8	7 12,820 0 26,131 7 29,475 4 19,868 8 13,000 8 13,000 5 the year.	$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	7 12,820 0.026 0.056 49.64 0 26,131 0.151 0.149 54.69 7 29,475 0.023 0.022 70.00 4 19,868 0.020 0.055 41.20 8 13,000 0.006 0.011 32.25 11 extreme the the the the the the the the the th

 Table 4: Median characteristics of markets where formats are present, 2001-2006

 Emeration (arrange)

		fr	om ICA		
New Format (see Table 1)	Maxi	Kvantum	Supermarket	Nära	Rimi
ICA Maxi	-	8	0	0	0
ICA Kvantum	2	-	7	0	2
ICA Supermarket	0	8	-	7	100
ICA Nära	0	0	48	-	30
Willys et al.	0	0	0	2	0
Vivo	0	0	1	0	1
Handlarń	0	0	0	31	0
Tempo	0	0	0	21	0
Rimi	0	1	5	2	-
Axfood others	0	0	0	9	0
Vi	0	0	0	2	0
Bergendahls others	0	0	1	16	0
Others	0	2	1	220	1

Table 5: Format repositioning from ICA, 2001-2006

NOTE: The figures represent the number of stores that switched format (see Table 1).

Table 6:	Format	$\operatorname{repositioning}$	from	Axfood,	2001-2006

New Format (see Table 1)	Willys et al.	Vivo	Hemköp	Handlarń	Tempo	AxFood
			et al.			Others
ICA Kvantum	0	2	0	0	0	0
ICA Supermarket	0	0	1	0	0	0
ICA Nära	0	1	4	0	1	1
Willys et al.	-	9	60	0	1	7
Vivo	0	-	0	0	1	1
Hemköp et al.	0	113	-	0	0	7
Handlarń	0	3	0	-	12	68
Tempo	1	19	45	3	-	14
Rimi	0	1	0	0	0	0
Axfood Others	0	5	4	8	4	-
COOP Medium	0	0	2	0	0	0
Vi	0	78	1	0	2	0
Bergendahls Others	0	4	3	0	6	4
Others	3	20	8	40	8	93

NOTES: The figures represent the number of stores that switched format (see Table 1).

Vi Others		0 1	1 19	
COOP	Medium	0	1	cepts.
Axfood	Others	0	2	e store con
ndlarń Tempo		2	1	tains one or mor
Hemköp Ha	et al.	3 1	0 0	One format con
et				format.
Willys 21	al.	1	0	tch their
ICA Nära		0	1	stores that swi
ICA	Supermarket	0	1	ant the number of
Old Format (see Table 1)		Vi	Bergendahls Others	NOTES: The figures represe

 Table 7: Format repositioning from Bergendahls, 2001-2006

 Table 8: Format repositioning from Coop, 2001-2006

 Old Format (see Table 1)
 ICA Nära
 Willys et Vivo Hand

Old Format (see Table 1)	ICA Nära	Willys et	Vive) Handlarffer	npo Axf	ood Cool	p Coo	p Coop	OBS et al.	Vi	Bergendahls	Others
		du.			Oth	iers Nära	ı Lar	ge Medium			Others	
Coop Large	0	1	0	0 0	0	0	0	43	0	0	0	0
Coop Medium	1	1	ъ	24 14	1	115	39	0	3	ę	12	25
OBS et al.	0	0	0	0 0	0	0	53	38	0	0	0	1
Others	0	0	0	0 0	0	0	0	0	0	0	1	0
NOTES: The figures repres	ent the number	: of stores that	t swite	ched format.								

37

Format (see Table 1)	Std. Dev.	Distance	Pop.	Pop.	Pop.
			0-14	15 - 34	65-100
ICA Maxi	0.1336	-0.6704	1.0289	0.6838	0.6420
	(1.0002)	(0.0274)	(0.0012)	(0.0358)	(0.0321)
ICA Kvantum	0.0272	-0.7171	-0.9031	0.4654	0.2201
	(2.0435)	(0.0902)	(0.0631)	(0.0002)	(0.0106)
ICA Supermarket	0.1822	-1.6194	2.4134	0.1637	0.8203
	(4.0305)	(0.0335)	(0.2648)	(0.0312)	(0.4253)
ICA Nära	0.7535	-0.4094	-0.2264	0.6900	0.2715
	(2.1431)	(0.2003)	(0.1231)	(0.0243)	(0.0155)
Willys et al.	0.6057	-0.8821	1.2520	0.3079	0.2587
	(3.0246)	(0.0548)	(0.324)	(0.1260)	(0.0332)
Vivo	0.3944	-0.9255	0.9616	0.7370	0.1189
	(3.257)	(0.1005)	(0.1946)	(0.0004)	(0.0282)
Hemköp et al.	0.9085	-0.7188	0.4967	0.4152	0.4151
	(0.0303)	(0.0332)	(0.0340)	(0.2112)	(0.0553)
Handlarń	0.1837	-0.0559	0.0886	0.2558	0.3278
	(4.566)	(0.0012)	(0.0394)	(0.0122)	(0.0134)
Tempo	0.1504	-0.4019	0.2046	0.7964	1.9868
	(0.0201)	(0.0141)	(0.0244)	(0.0432)	(0.0043)
Rimi	0.1333	-0.2393	0.1031	0.7484	-0.9531
	(0.0223)	(0.0030)	(0.1506)	(0.0003)	(0.0245)
Axfood Others	0.9813	-0.5016	0.5548	0.7413	0.4577
	(1.0033)	(0.0302)	(0.0232)	(0.0191)	(0.0126)
Coop Nära	0.6035	-1.0515	0.5386	1.9743	0.1485
	(5.0130)	(0.0004)	(0.0930)	(0.0023)	(0.0165)
Coop Large	0.2385	-0.6393	0.8863	0.4163	0.1556
	(3.0201)	(0.0221)	(0.0562)	(0.0646)	(0.0427)
Coop Medium	0.8516	-0.1116	0.2041	0.4366	0.1325
	(10.2374)	(0.0047)	(0.1041)	(0.0406)	(0.0256)
OBS et al.	0.3486	-0.6849	0.4394	0.3244	0.3012
	(4.0504)	(0.0257)	(0.0452)	(0.0757)	(0.0432)
Vi	0.9622	-1.0666	0.5819	0.6543	0.3114
	(8.0250)	(0.0303)	(0.0940)	(0.0043)	(0.0165)
Bergendahls Others	0.2408	-0.9030	0.0820	0.2175	0.3407
	(9.0031)	(0.0221)	(0.0322)	(0.0046)	(0.0047)
Age	0.6561				
	(0.0135)				
Sales space	0.0828				
	(0.0409)				
Quality transition	0.1718				
Repositioning	(0.0042)				
Quality transition	0.6232				
Non-repositioning	(0.0023)				
Number of observations	18,519				
GMM objective	28647				
Sargan p-value	0.184				

Table 9: Demand model estimates

NOTES: Standard errors are in parentheses. Population is in logs at municipality level. Coefficients on market-format and year dummies are not reported.



Figure 2: Quality kernel density estimates, repositioning and non-repositioning stores

Table 10. Sales function estimates				
	Model 1	Model 2	Model 3	Model 4
Demographics				
Population (0-14 years)	0.0008	0.0007	0.0002	0.0009
	(0.0002)	(0.0002)	(0.0002)	(0.0002)
Population (15-34 years)	-0.0006	-0.0006	-0.0005	-0.0007
	(0.0001)	(0.0001)	(0.0001)	(0.0002)
Population (over 65 years)	0.0014	0.0014	0.0014	0.0017
	(0.0003)	(0.0003)	(0.0003)	(0.0004)
Store characteristics and competition	· /	. /	. /	
Age	-0.0209	-0.0202	-0.0115	0.0002
0	(0.0021)	(0.0021)	(0.0009)	(0.001)
Distance	-0.0282	-0.0282	-0.0165	-0.0011
	(0.0031)	(0.0031)	(0.0014)	(0.0006)
Store quality	0.0073	0.0074	0.0041	0.0004
	(0.0007)	(0,0007)	(0,0003)	(0.0001)
Store quality squared	-0.00005	-0.00005	-0.00003	-0.00001
Store quality squared	(0.00003)	(0.00001)	(0.000001)	(0.00001
Number of stores commonly owned in format	(0.00001)	(0.00001)	-0.0076	-0.0053
Number of stores commonly owned in format			(0.0007)	(0.0006)
Number of stores owned by other firms			0.0007)	0.0042
Number of stores owned by other firms			(0.0005)	(0.0043)
Format repositioning		0.0205	(0.0007)	(0.0005)
Format repositioning		(0.0205)	0.0594	(0.0047)
		(0.0069)	(0.0438)	(0.0023)
Revenue effects				0.01=4
Store market share less than market average				-0.2174
				(0.1399)
Number of the second is a second second	10 510	10 510	10 510	10 510
Number of observations	16,519	16,519	16,519	16,519
Sales for non-repositioning	00.017	00.017	00.017	00.017
actual mean	32,817	32,817	32,817	32,817
predicted mean	28,033	30,297	30,281	31,902
correlation of actual, predicted	0.10	0.42	0.42	0.45
Sales for repositioning	~~ ~ ~ ~ ~			
actual mean	35,120	35,120	35,120	35,120
predicted mean	30,118	33,428	33,715	34,717
correlation of actual, predicted	0.16	0.44	0.44	0.46

Table 10: Sales function estimates

 contension of actual, predicted
 0.10
 0.44
 0.44
 0.46

 NOTES: Estimation by non-linear least squares. Standard errors in parentheses. Specifications include market-year fixed effects. Sales are reported in thousands of 2001 SEK. Population is in logs at municipality level.
 Standard errors in parentheses.
 Specifications

		Table 11: Policy	parame	ter estim	nates: re	epositior	ning forn	nat strat	tegies									
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$	$ \begin{array}{c c c c c c c c c c c c c c c c c c c $	Format (see Table 1)	Pop. 0-14	Δ Pop. 0-14	Pop. 15-34	△ Pop. 15-34	Pop. 65-100	∆ Pop. 65-100	Own quality	Rivals quality	No. own Formats	No. rivals	$\frac{\text{Stay}}{\times A}$	Switch ge	Stay ×D	Switch ist.	Stay × Marke	Switch t share
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$	$ \begin{array}{llllllllllllllllllllllllllllllllllll$	ICA Maxi	345.30	-204.735	72.082	165.369	-271.604	430.359	0.100	0.003	-4.336	0.179	-0.358	-0.422	-0.670	-3.355	70.376	40.907
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	$ \begin{array}{llllllllllllllllllllllllllllllllllll$		(0.045)	(0.072)	(0.042)	(0.069)	(0.00)	(0.016)	(0.016)	(0.005)	(0.604)	(0.005)	(0.055)	(0.067)	(0.148)	(0.060)	(0.780)	(0.067)
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	$ \begin{array}{c c c c c c c c c c c c c c c c c c c $	ICA Kvantum	334.62	-135.473	-8.922	175.906	-248.186	363.985	-0.077	0.008	-1.334	0.167	0.269	0.225	-1.251	-14.282	70.164	42.078
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	$ \begin{array}{llllllllllllllllllllllllllllllllllll$		(0.028)	(0.040)	(0.023)	(0.052)	(0.008)	(0.013)	(0.010)	(0.003)	(0.057)	(0.004)	(0.035)	(0.039)	(0.228)	(0.007)	(0.539)	(0.156)
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$	ICA Supermarket	191.30	-86.721	-16.503	110.571	-219.378	251.676	0.006	0.018	-0.472	0.139	0.018	-0.054	-0.057	0.021	59.126	21.203
$ \begin{array}{c c c c c c c c c c c c c c c c c c c $	$ \begin{array}{llllllllllllllllllllllllllllllllllll$		(0.030)	(0.035)	(0.086)	(0.122)	(0.029)	(0.040)	(0.006)	(0.002)	(0.012)	(0.004)	(0.021)	(0.023)	(0.032)	(0.058)	(0.463)	(0.013)
Willy set al. 2023 [0.034] [0.035] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0.036] [0	Willys et al. 0.033 (0.133) (0.133) (0.133) (0.133) (0.133) (0.133) (0.133) (0.133) (0.133) (0.133) (0.133) (0.133) (0.133) (0.133) (0.133) (0.133) (0.133) (0.133) (0.133) (0.033) (0.033) (0.033) (0.033) (0.033) (0.033) (0.033) (0.033) (0.033) (0.033) (0.033) (0.033) (0.033) (0.033) (0.033) (0.033) (0.033) (0.033) (0.033) (0.033) (0.033) (0.033) (0.033) (0.033) (0.033) (0.033) (0.033) (0.033) (0.033) (0.033) (0.033) (0.033) (0.033) (0.033) (0.033) (0.033) (0.033) (0.033) (0.033) (0.033) (0.033) (0.033) (0.033) (0.033) (0.033) (0.033) (0.033) (0.033) (0.033) (0.033) (0.033) (0.033) (0.033) (0.033)	ICA Nära	188.20	-47.723	-11.604	85.374	-28.716	167.076	-0.022	0.020	-0.285	0.121	0.118	0.061	0.128	0.062	30.160	11.760
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	Willys at al	(10.034) (10.032)	(861.U)	(100.0) 16.858	(0.014) 150 010	(0.1.0) -163 264	(0.220) 326 505	(0.000)	(100.0)	(0.000) -0.660	0.155	(710-0)	(ornn)	(0.020) -0.344	(ccu.u) 0.449	(T.U34) 50 507	(0.070) 36.648
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	$ \begin{array}{llllllllllllllllllllllllllllllllllll$		(0.034)	(0.036)	(0.051)	(0.089)	(0.024)	(0.033)	(00.00)	(0.003)	(0.022)	(0.004)	(0.030)	(0.031)	(0.064)	(0.142)	(0.722)	(0.709)
$ \begin{array}{llllllllllllllllllllllllllllllllllll$	$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	Vivo	307.40	-137.283	-146.102	188.723	-310.258	235.689	-0.016	0.017	-0.344	0.126	0.089	0.044	-0.050	-0.345	57.083	16.709
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	$ \begin{array}{cccccccccccccccccccccccccccccccccccc$		(0.034)	(0.048)	(0.010)	(0.026)	(0.004)	(0.008)	(0.009)	(0.004)	(0.011)	(0.004)	(0.032)	(0.035)	(0.085)	(0.675)	(0.193)	(0.005)
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	Hemköp et al.	273.91	-94.412	9.804	112.722	-299.879	396.696	-0.028	0.014	-0.693	0.155	0.120	0.110	0.016	0.131	58.467	31.218
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	$ \begin{array}{c c c c c c c c c c c c c c c c c c c $		(0.027)	(0.036)	(0.020)	(0.035)	(0.00)	(0.014)	(0.007)	(0.002)	(0.020)	(0.004)	(0.026)	(0.027)	(0.047)	(0.061)	(0.626)	(0.166)
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	Handlarń	52.60	-36.425	-83.106	99.222	-276.976	249.792	0.030	0.037	-1.094	0.161	-0.072	-0.094	-0.145	-0.166	-106.767	-95.908
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$	$ \begin{array}{c c c c c c c c c c c c c c c c c c c $		(0.021)	(0.037)	(0.020)	(0.034)	(0.006)	(0.014)	(0.008)	(0.002)	(0.030)	(0.004)	(0.027)	(0.028)	(0.039)	(0.062)	(0.017)	(0.006)
$ \begin{array}{llllllllllllllllllllllllllllllllllll$	$ \begin{array}{llllllllllllllllllllllllllllllllllll$	Tempo	105.18	-1.073	-68.863	143.067	-124.473	237.636	0.037	0.024	-1.037	0.164	-0.097	-0.125	-0.269	-0.297	-43.008	-8.110
$ \begin{array}{llllllllllllllllllllllllllllllllllll$	$ \begin{array}{llllllllllllllllllllllllllllllllllll$		(0.024)	(0.041)	(0.015)	(0.028)	(0.005)	(0.013)	(0.009)	(0.003)	(0.032)	(0.004)	(0.031)	(0.031)	(0.053)	(0.076)	(0.024)	(0.022)
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	Axfood Others	139.22	-87.366	-204.179	164.845	-453.006	249.841	-0.300	0.036	-0.885	0.149	1.028	0.997	1.269	1.172	58.009	35.261
$ \begin{array}{cccccc} \mbox{Coop Nim} & 303.41 & 79.893 & -154.036 & 105.41 & -143.43 & -0.574 & -0.576 & 0.044 & -0.742 & 0.153 & -1.775 & -0.544 & -0.432 & -0.247 & -1.121.512 & -3.129 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.126 & -3.$	$ \begin{array}{c c c c c c c c c c c c c c c c c c c $		(0.019)	(0.037)	(0.013)	(0.027)	(0.007)	(0.017)	(0.008)	(0.002)	(0.025)	(0.004)	(0.029)	(0.029)	(0.043)	(0.120)	(0.164)	(0.030)
$ \begin{array}{c} Coop \mbox{ Large } & (0.046) & (0.083) & (0.011) & (0.083) & (0.011) & (0.083) & (0.014) & (0.058) & (0.0172) & (0.099) & (0.000) & (0.144) & (0.172) & (0.099) & (0.000) & (0.144) & (0.172) & (0.099) & (0.000) & (0.144) & (0.173) & (0.16) & (0.143) & (0.173) & (0.16) & (0.143) & (0.173) & (0.16) & (0.16) & (0.114) & (0.173) & (0.123) & (0.123) & (0.123) & (0.123) & (0.123) & (0.123) & (0.123) & (0.123) & (0.123) & (0.123) & (0.123) & (0.123) & (0.123) & (0.123) & (0.123) & (0.123) & (0.123) & (0.123) & (0.123) & (0.123) & (0.123) & (0.123) & (0.123) & (0.123) & (0.123) & (0.123) & (0.123) & (0.123) & (0.123) & (0.123) & (0.123) & (0.123) & (0.123) & (0.123) & (0.123) & (0.123) & (0.123) & (0.123) & (0.123) & (0.123) & (0.123) & (0.123) & (0.123) & (0.123) & (0.123) & (0.123) & (0.123) & (0.123) & (0.123) & (0.123) & (0.123) & (0.123) & (0.123) & (0.123) & (0.123) & (0.123) & (0.123) & (0.123) & (0.123) & (0.123) & (0.123) & (0.123) & (0.123) & (0.123) & (0.123) & (0.123) & (0.123) & (0.123) & (0.123) & (0.123) & (0.123) & (0.123) & (0.123) & (0.123) & (0.123) & (0.123) & (0.123) & (0.123) & (0.123) & (0.123) & (0.123) & (0.123) & (0.123) & (0.123) & (0.123) & (0.123) & (0.123) & (0.123) & (0.123) & (0.123) & (0.123) & (0.123) & (0.123) & (0.123) & (0.123) & (0.123) & (0.123) & (0.123) & (0.123) & (0.123) & (0.123) & (0.123) & (0.123) & (0.123) & (0.123) & (0.123) & (0.123) & (0.123) & (0.123) & (0.123) & (0.123) & (0.123) & (0.123) & (0.123) & (0.123) & (0.123) & (0.123) & (0.123) & (0.123) & (0.123) & (0.123) & (0.123) & (0.123) & (0.123) & (0.123) & (0.123) & (0.123) & (0.123) & (0.123) & (0.123) & (0.123) & (0.123) & (0.123) & (0.123) & (0.123) & (0.123) & (0.123) & (0.123) & (0.123) & (0.123) & (0.123) & (0.123) & (0.123) & (0.123) & (0.123) & (0.123) & (0.123) & (0.123) & (0.123) & (0.123) & (0.123) & (0.123) & (0.123) & (0.123) & (0.123) & (0.123) & (0.123) & (0.123) & (0.123) & (0.123) & (0.123) & (0.123) & (0.123) & (0.123) & (0.123) & (0.123) & (0.123) & (0.123) & (0.123) & (0.123) & (0.123) &$	$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	Coop Nära	303.41	79.893	-154.035	105.144	-143.469	-157.724	-0.056	0.004	-0.742	0.153	-1.715	0.204	-0.432	0.207	-112.152	-3.120
$ \begin{array}{c} Coop \ Large \\ Coop \ Large \\ Sa605 & -165,777 & 49.670 & -16.21 & -231.386 & 372.340 & 0.044 & 0.007 & -1.702 & 0.173 & -0.114 & -0.123 & -0.754 & -0.581 & -0.138 & -30.154 & -0.081 & 0.1088 & -0.1416 & -0.123 & -0.754 & -0.581 & -0.138 & -0.754 & -0.581 & -0.138 & -0.754 & -0.581 & -0.138 & -0.754 & -0.581 & -0.138 & -0.754 & -0.581 & -0.138 & -0.754 & -0.581 & -0.138 & -0.754 & -0.581 & -0.138 & -0.754 & -0.581 & -0.138 & -0.754 & -0.581 & -0.138 & -0.754 & -0.581 & -0.138 & -0.754 & -0.581 & -0.138 & -0.754 & -0.581 & -0.138 & -0.754 & -0.581 & -0.138 & -0.754 & -0.581 & -0.138 & -0.754 & -0.581 & -0.138 & -0.754 & -0.581 & -0.138 & -0.754 & -0.581 & -0.138 & -0.751 & -0.581 & -0.138 & -0.754 & -0.581 & -0.138 & -0.751 & -0.031 & -0.031 & -0.0321 & -0.031 & -0.0321 & -0.031 & -0.0321 & -0.0321 & -0.031 & -0.0321 & -0.031 & -0.0321 & -0.031 & -0.0321 & -0.031 & -0.0321 & -0.031 & -0.031 & -0.031 & -0.031 & -0.053 & -0.148 & -0.158 & -0.158 & -0.138 & -0.173 & -0.138 & -0.771 & -0.573 & -0.981 & -0.058 & -0.981 & -0.058 & -0.981 & -0.058 & -0.981 & -0.058 & -0.981 & -0.058 & -0.981 & -0.058 & -0.981 & -0.058 & -0.981 & -0.058 & -0.981 & -0.058 & -0.981 & -0.058 & -0.981 & -0.058 & -0.981 & -0.058 & -0.981 & -0.058 & -0.148 & -0.058 & -0.148 & -0.058 & -0.148 & -0.058 & -0.981 & -0.058 & -0.981 & -0.058 & -0.981 & -0.058 & -0.981 & -0.058 & -0.981 & -0.058 & -0.981 & -0.058 & -0.981 & -0.058 & -0.981 & -0.058 & -0.981 & -0.058 & -0.981 & -0.058 & -0.981 & -0.058 & -0.981 & -0.058 & -0.981 & -0.058 & -0.981 & -0.058 & -0.981 & -0.058 & -0.981 & -0.058 & -0.981 & -0.058 & -0.981 & -0.058 & -0.146 & -0.056 & -0.256 & -0.261 & -0.144 & -0.051 & -0.020 & -0.041 & -0.771 & -0.731 & -0.732 & -0.581 & -0.742 & -0.582 & -0.581 & -0.581 & -0.581 & -0.581 & -0.581 & -0.581 & -0.581 & -0.581 & -0.581 & -0.581 & -0.581 & -0.581 & -0.581 & -0.581 & -0.581 & -0.581 & -0.581 & -0.581 & -0.581 & -0.581 & -0.581 & -0.581 & -0.581 & -0.581 & -0.581 & -0.581 & -0.581 & -0.581 & -0.581 & -0.581 & -0.581 & -0.581 & -0.581 &$	$ \begin{array}{c} \mbox{Coop Large} & 336.05 & 116.477 & 49.670 & 116.421 & 231.398 & 373.40 & 0.044 & 0.007 & -1.72 & 0.114 \\ \mbox{Coop Medium} & 176.85 & -22.874 & -27.587 & 105.563 & 31.821 & 118.183 & -0.096 & 0.028 & 0.294 & 0.126 & 0.366 \\ \mbox{Coop Medium} & 176.85 & -22.874 & -27.587 & 105.563 & 31.821 & 118.183 & -0.096 & 0.028 & 0.294 & 0.126 & 0.366 \\ \mbox{OBS et al.} & 10.337 & (0.034) & (0.043) & (0.043) & (0.015) & (0.003) & (0.003) & (0.003) & (0.023) \\ \mbox{OBS et al.} & 127.11 & -94.328 & -13.533 & 17.730 & -11.15 & 0.1126 & (0.011) & (0.003) & (0.003) & (0.023) \\ \mbox{OBS et al.} & 127.11 & -94.328 & -13.281 & 116.79 & -11.15 & 0.0136 & (0.023) & (0.003) & (0.023) \\ \mbox{OBS et al.} & 127.11 & -94.328 & -10.381 & (0.014) & (0.043) & (0.013) & (0.033) & (0.033) & (0.033) & (0.033) \\ \mbox{Vi} & 42.37 & -20.972 & -152.186 & 160.746 & (0.114) & (0.013) & (0.033) & (0.033) & (0.033) & (0.033) & (0.033) & (0.033) & (0.033) & (0.033) & (0.033) & (0.033) & (0.033) & (0.033) & (0.033) & (0.033) & (0.033) & (0.033) & (0.033) & (0.033) & (0.033) & (0.033) & (0.033) & (0.033) & (0.033) & (0.033) & (0.033) & (0.033) & (0.033) & (0.033) & (0.033) & (0.033) & (0.033) & (0.033) & (0.033) & (0.033) & (0.033) & (0.033) & (0.033) & (0.033) & (0.033) & (0.033) & (0.033) & (0.033) & (0.033) & (0.033) & (0.033) & (0.033) & (0.033) & (0.033) & (0.033) & (0.033) & (0.033) & (0.033) & (0.033) & (0.033) & (0.033) & (0.033) & (0.033) & (0.033) & (0.033) & (0.033) & (0.033) & (0.033) & (0.033) & (0.033) & (0.033) & (0.033) & (0.033) & (0.033) & (0.033) & (0.033) & (0.033) & (0.033) & (0.033) & (0.033) & (0.033) & (0.033) & (0.033) & (0.033) & (0.033) & (0.033) & (0.033) & (0.033) & (0.033) & (0.033) & (0.033) & (0.033) & (0.033) & (0.033) & (0.033) & (0.033) & (0.033) & (0.033) & (0.033) & (0.033) & (0.033) & (0.033) & (0.033) & (0.033) & (0.033) & (0.033) & (0.033) & (0.033) & (0.033) & (0.033) & (0.033) & (0.033) & (0.033) & (0.033) & (0.033) & (0.033) & (0.033) & (0.033) & (0.033) & (0.033) & (0.033) & (0.033) & (0.033) $		(0.046)	(0.083)	(0.011)	(0.034)	(0.011)	(0.028)	(0.017)	(0.009)	(0.033)	(0.004)	(0.335)	(0.058)	(0.072)	(660.0)	(0.000)	(0.104)
$ \begin{array}{c} \mbox{Coop} \mbox{Medium} & (0.037) & (0.0137) & (0.0137) & (0.0137) & (0.0137) & (0.0137) & (0.0137) & (0.1137) & (0.1756) \\ \mbox{Coop} \mbox{Medium} & (0.034) & (0.034) & (0.034) & (0.037) & (0.0137) & (0.0137) & (0.0137) & (0.0137) & (0.0137) & (0.0137) & (0.0137) & (0.0137) & (0.0137) & (0.0137) & (0.0137) & (0.0137) & (0.0137) & (0.0137) & (0.0137) & (0.0137) & (0.0137) & (0.0137) & (0.0137) & (0.0137) & (0.0137) & (0.0137) & (0.0137) & (0.0137) & (0.0137) & (0.0137) & (0.0137) & (0.0137) & (0.0137) & (0.0137) & (0.0137) & (0.0137) & (0.0137) & (0.0137) & (0.0137) & (0.0137) & (0.0137) & (0.0137) & (0.0137) & (0.0137) & (0.0137) & (0.0137) & (0.0137) & (0.0137) & (0.0137) & (0.0137) & (0.0137) & (0.0137) & (0.0137) & (0.0137) & (0.0137) & (0.0137) & (0.0137) & (0.0137) & (0.0137) & (0.0137) & (0.0137) & (0.0137) & (0.0137) & (0.0137) & (0.0137) & (0.0137) & (0.0137) & (0.0137) & (0.0137) & (0.0137) & (0.0137) & (0.0137) & (0.0137) & (0.0177) & (0.0177) & (0.0177) & (0.0177) & (0.0177) & (0.0177) & (0.0177) & (0.0177) & (0.0177) & (0.0177) & (0.0177) & (0.0177) & (0.0177) & (0.0177) & (0.0177) & (0.0177) & (0.0177) & (0.0177) & (0.0177) & (0.0177) & (0.0177) & (0.0177) & (0.0177) & (0.0177) & (0.0177) & (0.0177) & (0.0177) & (0.0177) & (0.0177) & (0.0177) & (0.0177) & (0.0177) & (0.0177) & (0.0177) & (0.0177) & (0.0177) & (0.0177) & (0.0177) & (0.0177) & (0.0177) & (0.0177) & (0.0177) & (0.0177) & (0.0177) & (0.0177) & (0.0177) & (0.0177) & (0.0177) & (0.0177) & (0.0177) & (0.0177) & (0.0177) & (0.0177) & (0.0177) & (0.0177) & (0.0177) & (0.0177) & (0.0177) & (0.0177) & (0.0177) & (0.0177) & (0.0177) & (0.0177) & (0.0177) & (0.0177) & (0.0177) & (0.0177) & (0.0177) & (0.0177) & (0.0177) & (0.0177) & (0.0177) & (0.0177) & (0.0177) & (0.0177) & (0.0177) & (0.0177) & (0.0177) & (0.0177) & (0.0177) & (0.0177) & (0.0177) & (0.0177) & (0.0177) & (0.0177) & (0.0177) & (0.0177) & (0.0177) & (0.0177) & (0.0177) & (0.0177) & (0.0177) & (0.0177) & (0.0177) & (0.0177) & (0.0177) & (0.0177) & (0.0177) & (0.0177$	$ \begin{array}{c} \mbox{Coop} \mbox{Medium} & [0.037] & [0.041] & [0.070] & [0.012] & [0.026] & [0.011] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & [0.036] & $	Coop Large	336.05	-165.777	49.670	116.421	-231.398	372.340	0.044	0.007	-1.702	0.173	-0.114	-0.123	-0.754	-0.851	61.088	39.154
$ \begin{array}{c} \mbox{Coop Meduum} & 176385 & -22874 & -27757 & -2757 & -27587 & -27587 & -27587 & -0.2598 & -0.259 & -0.259 & -0.259 & -0.259 & -0.259 & -0.259 & -0.259 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -0.251 & -$	$ \begin{array}{c} \mbox{Coop Meduum} & 176.55 & -22.87 & 105.563 & 31.821 & 118.53 & -0.096 & 0.029 & 0.0294 & 0.1366 & 0.306 \\ \mbox{OBS et al.} & (0.034) & (0.084) & (0.303) & (0.515) & (0.142) & (0.142) & (0.006) & (0.003) & (0.004) & (0.023) \\ \mbox{OBS et al.} & (0.023) & (0.034) & (0.033) & (0.014) & (0.0136) & (0.1032) & (0.003) & (0.003) & (0.003) \\ \mbox{OI} & (0.023) & (0.033) & (0.014) & (0.048) & (0.015) & (0.032) & (0.069) & (0.069) & (0.069) & (0.023) \\ \mbox{OI} & (0.233) & (0.033) & (0.014) & (0.048) & (0.015) & (0.033) & (0.032) & (0.063) & (0.063) & (0.063) \\ \mbox{OI} & (0.233) & (0.023) & (0.044) & (0.044) & (0.134) & (0.135) & (0.032) & (0.063) & (0.063) & (0.053) \\ \mbox{OI} & (0.231) & (0.023) & (0.023) & (0.043) & (0.013) & (0.013) & (0.023) & (0.016) & (0.157) \\ \mbox{OI} & (0.133) & (0.232) & (0.023) & (0.013) & (0.013) & (0.023) & (0.016) & (0.157) \\ \mbox{OI} & (0.133) & (0.023) & (0.023) & (0.013) & (0.013) & (0.023) & (0.016) & (0.163) & (0.056) \\ \mbox{Bergendahls Others} & 34.7.7 & -160.366 & -26.770 & (0.015) & (0.013) & (0.021) & (0.047) & (0.006) & (0.575) \\ \mbox{Bergendahls Others} & 34.7.3 & -160.366 & -26.770 & (0.015) & (0.013) & (0.021) & (0.021) & (0.047) & (0.006) & (0.575) \\ \mbox{Number of observations} 18.519 & (0.055) & (0.056) & (0.015) & (0.021) & (0.012) & (0.004) & (0.041) & (0.041) \\ \mbox{Number of observations} 18.519 & (0.055) & (0.056) & (0.015) & (0.021) & (0.012) & (0.004) & (0.041) & (0.041) \\ \mbox{Log-Likelihood} & -70771.77 \\ \mbox{Log-Likelihood} & -70771.77 \\ \mbox{Log-Likelihood} & -70771.77 \\ \mbox{Log-Likelihood} & -70771 & 0.0021 & (0.015) & (0.015) & (0.012) & (0.014) & (0.041) & (0.041) & (0.041) \\ \mbox{Log-Likelihood} & -70771.77 \\ \mbox{Log-Likelihood} & -70711 & 0.0120 & (0.016) & (0.014) & (0.041) & $:	(0.037)	(0.044)	(0.041)	(0.070)	(0.019)	(0.026)	(0.011)	(0.003)	(0.094)	(0.004)	(0.037)	(0.037)	(0.145)	(0.277)	(0.734)	(0.765)
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	Coop Medium	176.85	-22.874	-27.587	105.563	31.821	118.183	-0.096	0.026	-0.294	0.126	0.366	0.312	0.431	-1.746	55.942	24.581
UBS et al. 127.11 -94.328 -1.3393 12.130 -91.739 -1.1179 0.1036 -1.11739 0.1037 0.1036 -1.11739 0.1037 0.1036 0.0037 0.0037 0.0137 0.0037 0.0037 0.0037 0.0037 0.0037 0.0037 0.0037 0.0037 0.0037 0.0037 0.0037 0.0037 0.0037 0.0037 0.0037 0.0037 0.0037 0.0037 0.0037 0.0037 0.0037 0.0039 0.0037 0.0039 0.0037 0.0039 0.0037 0.0039 0.0039 0.0039 0.0039 0.0039 0.0039 0.0037 0.0031 0.0037 0.0031 0.0031 0.0037 0.0039 0.0039 0.0039 0.0039 0.0031 0.0031 0.0031 0.0031 0.0031 0.0031 0.0031 0.0031 0.0031 0.0031 0.0031 0.0031 0.0031 0.0031 0.0031 0.0031 0.0031 0.0031 0.0031 0.0031 0.0031 0.0031 0.0031 0.0031 0.0031 0.0031 0.0031 0.0031 0.0031 0.0031 0.0031 0.0031 0.0031 0.0031 0.0031 0.0031 0.0031 0.0031 0.0031 0.0031 0.0031 0.0031 0.0031 0.0031 0.0031 0.0031 0.0031 0.0031 0.0031 0.0031 0.0031 0.0031 0.0031 0.0031 0.0031 0.0031 0.0031 0.0031 0.0031 0.0031 0.0031 0.0031 0.0031 0.0031 0.0031 0.0031 0.0031 0.0031 0.0031 0.0031 0.0031 0.0031 0.0031 0.0031 0.0031 0.0031 0.0031 0.0031 0.0031 0.0031 0.0031 0.0031 0.0031 0.0031 0.0031 0.0031 0.0031 0.0031 0.0031 0.0031 0.0031 0.0031 0.0031 0.0031 0.0031 0.0031 0.0031 0.0031 0.0031 0.0031 0.0031 0.0031 0.0031 0.0031 0.0031 0.0031 0.0031 0.0031 0.0031 0.0031 0.0031 0.0031 0.0031 0.0031 0.0031 0.0031 0.0031 0.0031 0.0031 0.0031 0.0031 0.0031 0.0031 0.0031 0.0031 0.0031 0.0031 0.0031 0.0031 0.0031 0.0031 0.0031 0.0031 0.0031 0.0031 0.0031 0.0031 0.0031 0.0031 0.0031 0.0031 0.0031 0.0031 0.0031 0.0031 0.0031 0.0031 0.0031 0.0031 0.0031 0.0031 0.0031 0.0031 0.0031 0.0031 0.0031 0.0031 0.0031 0.0031 0.0031 0.0031 0.0031 0.0031 0.0031 0.0031 0.0031	UBS et al. $12/\Lambda$ = 94.358 = 1.3.393 = 1.27.303 = 1.17.793 = 2.11.779 = 1.17.79 = 1.11.779 = 1.11.77 = 1.0.304 = 0.1.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.0.051 = 0.		(0.034)	(0.084)	(0.403)	(0.551)	(0.115)	(0.142)	(0.006)	(100.0)	(0.008)	(0.004)	(0.020)	(0.023)	(0.027)	0.931)	(0.651)	(0.018)
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	OBS et al.	11.721	-94.328	-13.093	127.303	-11/./29	6/0.112	-0.154	0.032	-1.040	0.148	0.003	0.484	0.573	0.742	170.00	38.113
Vi $1.2.57$ -20.972 -10.2186 10.0074 $0.10.29$ 10.0069 10.0074 $0.10.18$ 10.0024 0.0141 -3.788 -5.365 -0.999 1.336 -3.365 -0.999 1.336 -0.37 5.365 -0.999 1.336 0.031 0.051 0.001 0.001 0.0121 0.0124 0.0124 0.0124 0.0124 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0241 0.0041 0.0041 0.0041 0.0041 0.0041 0.0041 0.0041 0.0041 0.0041 0.0041 0.0041 0.0041 0.0041 0.0041 0.0041 0.0041 0.0041 0.0001	Vi $4.5.5$ $2.0.37$ $2.0.37$ 160.74 60.144 61.124 $1.145.8$ 0.223 -0.006 -0.425 0.141 -3.788 0.578 0.578 0.576 0.576 0.576 0.576 0.576 0.576 0.576 0.576 0.576 0.576 0.576 0.006 0.006 0.005 0.002 0.007 0.007 0.004 0.004 0.004 0.004 0.004 0.014 0.576 0.410 0.576 0.140 0.576 0.146 0.576 0.146 0.576 0.146 0.576 0.146 0.576 0.146 0.576 0.146 0.006 0.005 0.002 0.0697 0.142 0.376 0.440 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.004 0.0004		(0.023)	(0.053)	(0.014)	(0.048)	(0.016)	(0.036)	(0.015)	(0.003)	(0.069)	(0.005)	(0.051)	(0.055)	(0.087)	(0.100)	(0.845)	(0.087)
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	$ \begin{array}{llllllllllllllllllllllllllllllllllll$	Vi	42.37	-20.972	-152.186	160.744	61.124	-114.585	0.223	-0.006	-0.426	0.141	-3.788	-0.737	-8.365	-0.989	1.324	3.136
Bergendahls Others 341.73 -150.366 -26.270 151.754 51.350 218.054 -0.136 0.002 -0.697 0.142 0.480 0.479 0.531 0.599 66.461 28.589 Number of observations [0.046] (0.046] (0.055) (0.029) (0.056) (0.015) (0.021) (0.012) (0.004) (0.029) (0.041) (0.067) (0.086) (1.100) (0.121) Log-Likelihood -70771.77	Bergendahls Others 341.73 -150.366 -26.270 151.784 51.350 218.054 -0.136 0.002 -0.697 0.142 0.480 Number of observations [0.046] (0.055) (0.029) (0.056) (0.015) (0.021) (0.012) (0.004) (0.029) (0.004) (0.041) Log-Likelihood -70771.77		(0.051)	(0.082)	(0.029)	(0.047)	(0.006)	(0.018)	(0.024)	(0.011)	(0.047)	(0.006)	(0.575)	(0.081)	(0.065)	(0.148)	(0.000)	(0.121)
Number of observations 18,519 (0.055) (0.029) (0.056) (0.015) (0.021) (0.012) (0.004) (0.029) (0.004) (0.041) (0.067) (0.086) (1.100) (0.121) Log-Likelihood -70771.77	Number of observations [8,0.46] (0.055) (0.029) (0.056) (0.015) (0.021) (0.012) (0.004) (0.029) (0.004) (0.041) Log.Likelihood -70771.77	Bergendahls Others	341.73	-150.366	-26.270	151.784	51.350	218.054	-0.136	0.002	-0.697	0.142	0.480	0.479	0.531	0.599	60.461	26.389
Number of observations 18,519 Log-Likelihood -7077.1.77	Number of observations 18,519 Log-Likelihood -70771.77		(0.046)	(0.055)	(0.029)	(0.056)	(0.015)	(0.021)	(0.012)	(0.004)	(0.029)	(0.004)	(0.041)	(0.041)	(0.067)	(0.086)	(1.100)	(0.121)
Log-Likelihood -70771.77	Log-Likelihood -70771.77	Number of observations	18,519															
		Log-Likelihood	-70771.7.	2														

•	Series
	trat
	ŝ
	tormat
	50
•	ninc
•	Ĕ
•	\overline{S}
	repo
	tes:
•	tıma
	ŝ
	fer
	ume
	Dar
	þ
	Ĕ
¢	ĭ
1	-
1	-
_	e
1	õ

	Entry	Exit
	probit	probit
	P(entry X)	P(exit X)
Quality		-0.049
		(0.006)
Competitors's quality in the same format	0.028	0.002
	(0.014)	(0.001)
Age		0.163
		(0.026)
Distance	0.025	0.021
	(0.012)	(0.002)
Population	1.904	-0.199
	(0.903)	(0.046)
Number of stores commonly owned in format	-0.061	-0.002
	(0.017)	(0.001)
Number of stores owned by other firms	-0.049	-0.0001
	(0.023)	(0.0001)

Table 12: Policy parameter estimates: entry and exit

NOTES: Standard errors are in parentheses. Specifications include format market-year fixed effects. Population is in logs at municipality level.

Table 13: Parameter estimates: sunk repositioning costs and sell-o	f values	s
--------------------------------------------------------------------	----------	---

		Market Size	
Parameter	Population	Population	Population
	< 20,000	20,000-60,000	> 60,000
Repositioning costs (γ_1)	[78.73 390.41]	[403.00 637.13]	[490.02 738.35]
PPHI outer - 95%	[62.41 416.23]	[384.45 701.92]	[481.52 793.32]
Sell-off values (γ_4)	[680.34 990.98]	[736.02 1203.92]	[920.97 1890.37]
PPHI outer - 95%	[603.72 1010.39]	[701.83 1308.62]	[839.82 1983.08]
Median sales of repositioning stores	8,000	17,500	17,500
Median sales/upper bound of repositioning costs	21	27	24
Mean no. of repositionings per year-market	1	2	11

NOTES: Repositioning costs are in thousands of 2001 SEK. The Pakes et al. (2007b) estimator is used. 100 simulations are used to construct the outer intervals.

Table 14:	Sunk entry	· costs	distribution	results

•			
		Market Size	
Parameter	Population	Population	Population
	< 20,000	20,000-60,000	> 60,000
Mean	5,405	20,357	37,985
Variance	5E4	7E4	10E4
Median Sales of Entrants	2,500	4,500	10,750

NOTES: Median sales and costs are in thousands of 2001 SEK. The parameters were estimated by matching the cumulative distribution function of a normal distribution to the empirical probability of entry. The expected value of entry was computed using 100 replications of each state.

Appendix A. Sources of DELFI's data. DELFI Marknadspartner AB includes daily covers data on retail food stores from: (1) public registers, trade press, and daily press; (2) the Swedish Retailers Association (SSLF); (3) Kuponginlösen AB (a firm that deals with customers coupons); (4) each chains' headquarters; (5) matching customer registers from suppliers (customers); (6) telephone interviews, (7) annual surveys; and (8) The Swedish Retail Institute (HUI). In addition, DELFI verifies location, store-type, owner, and chain affiliation in annual reports.

Each firm has an identification number linked to it address. There are 11 store-types, based on size, location, product assortment, etc.: hypermarkets, department stores, large supermarkets, large grocery stores, other stores, small supermarkets, small grocery stores, convenience stores, gas-station stores, mini markets and seasonal stores.

Appendix B. Forward simulation steps that used to compute expected **payoffs.** The steps each year are :

- 1. Given store characteristics and market demographics, estimate the store quality from the random-coefficients demand model. I use 20 Halton draws for each unobserved characteristics.
- 2. Using the estimated sales generating function, compute discounted revenues for each store. I assume that the store are independently owned.
- 3. Compute the competition and demographics variables, which change every year and affect the multinomial (nested) logit format choice. Again, I assume that the stores are independently owned.
- 4. Simulate a choice for each store, i.e., compute the multinomial (nested) logit choice probabilities and compare then with a random random draw from a uniform distribution. If the store repositions I count how many times it does.
- 5. Conditional on current and previous, simulate the evolution of store-quality using draws from the empirical distribution of observed quality-innovations for the random components. In addition, simulate the evolution of demographics.
- 6. Update store formats.

Appendix C. Alternative estimators of sunk repositioning costs. An alternative estimator for the second step is the minimum-distance estimator, constructed using the set of inequalities below. Due to linearity in the cost function, the optimality conditions (1) can be re-written as

(36)
$$[W_j(\boldsymbol{\omega},\sigma_j,\sigma_{-j}) - W_j(\boldsymbol{\omega},\sigma'_j,\sigma_{-j})]\boldsymbol{\gamma} \ge 0$$

This can be written in terms of profitable deviations from optimal policy,

(37)
$$g(\sigma'_{j}; \boldsymbol{\gamma}, \boldsymbol{\alpha}) = [W_{j}(\boldsymbol{\omega}, \sigma_{j}, \sigma_{-j}) - W_{j}(\boldsymbol{\omega}, \sigma'_{j}, \sigma_{-j})]\boldsymbol{\gamma}$$

where α represents parameterization of the policy functions. More specifically, alternative policies can be drawn to generate a set of inequalities indexed by x. The estimates of W_j , denoted \tilde{W}_j , are obtained using forward simulation, and I use them in the sample analog of the objective function

(38)
$$Q_n(\boldsymbol{\gamma}, \boldsymbol{\alpha}) = \frac{1}{n_I} \sum_{x=1}^{n_I} (\min\{\tilde{g}(x, \boldsymbol{\gamma}, \boldsymbol{\alpha}), 0\})^2$$

I use the Nelder-Mead method to obtain the starting values. Then I plug the estimated parameters as started values in the Uncmin optimization routine.¹⁸ The later gives me the final estimates and the standard errors. Another alternative is to use the Laplace-type estimator (Chernozhukov and Hong, 2003).

Table 15: Alternative parameter estimates: sunk repositioning costs

		Market Size	
Parameter	Population	Population	Population
	< 20,000	20,000-60,000	> 60,000
Repositioning costs (γ_1)	353	637	718
Std.	(102.34)	(223.26)	(287.03)
Median sales of repositioning stores	8,000	17,500	17,500
Median sales/repositioning costs	22	27	24
Mean no. of repositioning per year	1	2	11

NOTES: Median sales and repositioning costs are in thousands of 2001 SEK. Standard errors are in parentheses. The standard errors are not corrected for first-stage estimation errors.

¹⁸Uncmin performs unconstrained nonlinear optimizations (http://www1.fpl.fs.fed.us/optimization.html).

Paper IV

From Boom to Bust: A Dynamic Analysis of IT Services^{*}

Florin G. Maican[†]

Draft: April 30, 2010

Abstract

Aggregate shocks in demand such as the burst of the 2001 dot-com bubble affect firms' behavior and, therefore, the market structure. This paper proposes a fully dynamic structural model to analyze the impact of the 2001 dot-com bust on the productivity dynamics and the cost structure for IT services. The empirical application builds on an eight year panel dataset that includes every IT service firm in Sweden and is representative for many other European countries. Entrants are found less productive than incumbents and net exit contributed the most to productivity growth in the IT services after the dot-com bust. The paper finds higher fixed investment and labor costs for software but lower for operational services after the dot-com bust. Finding the relative importance of fixed costs is a step closer to being able to link policies that affect adjustment costs in IT services.

Keywords: IT services; imperfect competition; dynamic estimation; industry dynamics; strategic interactions

JEL Classification: L86, L13, L44, L52, C1, C3, C5, C7

^{*}I would like to thank Nancy Lutz, Lennart Hjalmarsson, Cristian Huse, Matilda Orth, Helena Perrone, Ariel Pakes, and Catherine Schaumans for very useful comments. I would also like to thank participants at EARIE (Valencia) and Knowledge for Growth: European Strategies in the Global Economy (Toulouse), Advancing the Study of Innovation and Globalization in Organizations (Nuremberg), and XXIV Journadas de Economia Industrial (Vigo) for helpful comments and suggestions.

[†]Box 640, SE 405 30, Gothenburg, Sweden, phone +46-31-786 4866, Fax: +46-31-786 4154, E-mail: florin.maican@handels.gu.se

1 Introduction

The IT industry contributes significantly to increased productivity and improved service quality in virtually all sectors of the economy (Jorgenson et al., 2005, 2008). IT markets confirm strong recovery from the 2001-2003 slowdown until 2008.¹ The IT services industry, including software, has the highest contribution to total IT growth (Figure 2), e.g., 5.8 percent for software and 5.3 percent for operational services, maintenance, and repair.² Being affected differently by negative aggregate shocks in demand, such as the 2001 dot-com bust, firms change their behavior. Changes in firms' behavior regarding investment and labor affect the market structure. Using a fully dynamic model, this paper investigates the impact of the 2001 dot-com bust on productivity dynamics and the cost structure in Swedish IT services — an industry characterized by substantial entry and exit. The created selection process, induced by the impact of the 2001 dot-com bust, affected competition and productivity dynamics.³ The direct effect of the dot-com bust was a decrease in the labor productivity dispersion, which was caused by an increase in the 25th percentile and a decrease in the 75th percentile.

The IT services firms are clustered around large cities that are characterized by dynamic labor markets. Some IT services grow faster in some regions than in others, i.e., there are some sources of exogenous variation (from local markets) in firms' incentives to invest in labor and capital. IT services are considered sophisticated because the products are often highly user-specific and non-standardized. The industry is characterized by high heterogeneity where the services produced include more than one type of activity and the consumer is extremely important. Firms respond to aggregate shocks in demand such as the 2001 dot-com bust by improving the quality of their services. Adjustments in labor might be costly if IT firms have to invest in redesign or have to change their service practices to suit

¹While Western European IT markets were expected to grow at an annual average rate of 6.1 percent until 2008, the Central and East European markets were expected to grow by 13 percent (EU ICT Task Force Report, 2006). Figure 1 presents the evolution of the Western European ICT market growth from 1997 to 2007.

 $^{^{2}}$ EU market growth in this sector is principally driven by computer services. The EU IT market growth by segment in 2007 was as follows: software 5.8 percent, IT services 5.3 percent, computer hardware 2.4 percent, telecommunications equipment 2.0 percent, and carrier services 1.6 percent. IT services are important because companies and organizations modernize their IT infrastructure, i.e., consumers get lower prices and higher quality. Furthermore, IT services are highly dynamic due to the outsourcing of IT functions. The security of IT systems remains an important sector segment.

 $^{^{3}\}mathrm{Bartelsman}$ and Doms (2000) and Syverson (2010) survey empirical work on productivity changes using micro data.

new customers. The direct cost of hiring a new employee is likely smaller than the cost involved in direct work with a new environment, i.e., there is an unobserved cost when firms hire a new employee. For small tasks, IT firms might hire external consultants and therefore increase fixed labor costs.

The present paper uses a dynamic structural model where firms make optimal decisions on entry, exit, and investment given the strategies of their competitors.⁴ In IT services, the types of services offered and their quality are important aspects and depend on location. Since prices and other more detailed data on the IT services are not available, an accurate estimation of the quality of firms' services can not be obtained from a demand model. Instead, this paper estimates firms' productivity and assumes that there is a direct link between productivity and quality, i.e., a highly productive firm offers high-quality services. Firm productivity is estimated using an extension of Olley and Pakes (1996) framework that allows for lumpy investment and controls for unobserved prices. Since labor is a key factor for service quality in the IT industry, productivity is backed out from labor demand (Doraszelski and Jaumandreu, 2009; and Maican and Orth, 2009).

I assume that all relevant features of the IT services industry can be en-counted into a state vector that includes firms' perceived levels of productivity, local market demographics, and private shocks to profits. Firms receive states that depend on the payoffs in the product market. The evolution of productivity is influenced by entry, exit, and investment decisions. Firms' actions are subject to idiosyncratic shocks that are treated as private information, and they choose strategies that maximize their discounted profits, given the expected strategies of their rivals.

Using 1996-2002 panel data of the Swedish IT services industry, the present paper estimates productivity, and recovers both revenues and optimal policy functions for investment and labor consistent with the underlying model. The theoretical model is then used to simulate market outcomes with the cost structures before and after the dot-com bust.⁵ Jorgenson et al. (2005, 2008) find higher productivity

 $^{^4\}mathrm{The}$ theoretical framework is based on the Markov Perfect Equilibrium (MPE) framework of Ericson and Pakes (1995). Ericson and Pakes' framework assumes that firms make competitive investments that increase their productivity. Ackerberg et al. (2008) and Pakes (2008) review recent methodological developments in the empirical literature of imperfectly competitive markets.

 $^{^{5}}$ By comparing the predictions of the model under the different cost structures, this paper recovers the changes mode to a number of relevant policy measures. A more detailed discussion on the counterfactual analysis will be presented in future versions of the paper. It is important to understand how different ways to obtain perturbed policy functions affect the market structure. For example, we might generate policies that imply negative investments that make firms exit

growth in IT-producing industries than in IT-using industries. The IT services analyzed include all firms in software, operational services, and maintenance and repair. About 25 percent and 12-18 percent entered and exited the market during the period, respectively. The Swedish IT services market is representative of a majority of all IT markets in the EU.

The paper contributes to the previous literature by recovering both investment and labor adjustment costs in an innovative service industry before and after an aggregate negative demand shock. To my knowledge, this is also the first paper that analyzes productivity in IT services and provides estimates for demand elasticities and mark-ups for the IT services sub-sectors. The findings give information about the cost differences across sub-sectors and size groups of IT services firms that can be used, e.g., when subsidies are allocated for different groups of firms in this industry.

This paper uses a two-step procedure to recover the costs structure (Bajari et al., 2007-BBL). Recent empirical literature uses the BBL approach in a similar context (Ryan, 2009; Beresteanu and Ellickson, 2006; Ryan and Tucker, 2006; and Sweeting, 2007).⁶ Pakes et al. (2007a) (POB), Aguirregabiria and Mira (2007), and Pesendorfer and Schmidt-Dengler (2003) develop alternative extensions to the Hotz and Miller (1993) approach to estimate dynamic games where actions have a discrete choice structure. Using the POB estimator, Dunne et al. (2005) evaluate the costs of entry and exit in isolated markets in the US for dentists and chiropractors. Pakes et al. (2007b) and Pakes (2010) show how the inequalities generated by behavior choice models can be used as a basis in estimation.

By estimating firms' productivity controlling for possible unobserved demand shocks at the local market level, productivity is the only serially correlated state variable that helps for consistency in estimation of continuation values and policy functions in case of fully dynamic models. Controlling for selection when estimating productivity is important in the IT industry. The exit and entry in my data is based on organizational number. There is a high likelihood of sell-offs of small firms to large firms since small firms have been successful. IT services offer specialized product services, and improved use of IT tools can raise the average prices and therefore increase revenues and productivity. However, since price variation

early.

 $^{^{6}}$ Ryan (2009) evaluates the welfare costs of the 1990 Amendaments to the Clean Air Act on the US Portland cement industry using a dynamic model of oligopoly in the tradition of Ericson and Pakes (1995). Benkard (2004) examines the wide-body aircraft industry but does not recover estimates of fixed costs.

can also be due to local market power or demand shocks, it is important to control for demand when estimating productivity in this industry.

I find that the estimated elasticity of demand for the software industry is about -4.6, i.e., a mark-up of 1.277. For grouped operational services and maintenance firms, the estimated demand elasticity is about -5.96, yielding a mark-up of 1.52. For software, the productivity growth was around 21 percent from 1997 to 2000, but only about 6 percent from 1997 to 2002. After the 2001 dot-com bust, exit firms contributed more to productivity growth (12 percent) than continuing firms (7.5 percent). For operational services and maintenance, the productivity growth was about 70 percent from 1997 to 2000 and about 32 percent from 1997 to 2002. In the period 1997-2000, almost all productivity growth came from continuing firms. However, exit firms contributed the most (50 percent) to productivity growth from 1997 to 2002. This emphasizes the importance of selection effect in this industry. Entrants are found less productive than continuing firms.

The results indicate that doubling the capital stock increases revenues slightly over 30 percent. For software and operational services, foreign IT firms have about 19 percent higher revenues than domestic IT firms. The geographical location of owing firm has been found to be more important for productivity growth than the location of IT firms (Bloom et al., 2009).⁷ On average, the impact of the 2001 dot-com bust on revenues was a decrease of about 20 percent for software and operational services and of about 34 percent for maintenance and repair. Furthermore, firms reduced the number of employees by about 25 percent. After the dot-com bust, firms were more likely to exit in all sub-sectors. I also find that foreign IT firms were more likely to exit.

The impact of dot-com bust on investment and labor adjustment costs varies significantly depending on firm productivity and firm size. Lower adoption and smaller size IT investments in Europe are found to be responsible for the lower productivity growth in Europe than in the US over 1990s (van Ark et al., 2008). The present paper models fixed adjustment labor and capital costs to depend on the likelihood to adjust positively or negatively, and this propensity for adjustment depends on firms' state variables. My findings suggest that fixed and variable adjustment costs are important determinants of investment and labor decisions in the Swedish IT services industry. In addition to the lack of demand, they also

⁷They find that productivity from IT capital plays a key role in explaining higher productivity of US-based multinationals operating in the EU compared to EU firms. This advantage is explained by the evidence of complementarity between IT capital and human resources.

explain the downturn in productivity after the 2001 dot-com bubble. When there are fixed costs, a static evaluation ignores important economic penalties associated with the dot-com bust costs. The paper finds that, after the burst of the 2001 dot-com bubble, there was an increase in fixed (setup) investment costs for software but a decrease for operational services and maintenance and repair firms. From 2000 to the end of studied period, there were higher fixed costs for positive labor adjustment for software compared to 1996-1999 (about 4 times), lower for operational services (about 4 times), and about the same for maintenance and repair. For negative labor adjustment, the findings indicate higher fixed costs for software but lower for operational services after the dot-com bust. I find that the entry costs for software were six times lower than for operational services, which might explain the large number of entrants in software. In addition, while firms in software and operational services had higher scrap (sales-off) values after 2000, the maintenance and repair firms had lower scrap values.

The paper is organized as follows. Section 2 gives a brief overview of the Swedish IT services industry and relevant events over the last 10 years. It also includes a discussion on the data sources. Section 3 presents the theoretical model and Section 4 discusses the estimation details. The empirical results are presented in Section 5, whereas Section 6 concludes the paper.

2 Overview of the Swedish IT Services Industry

The Swedish IT industry is in better shape than it has been for many years. At the beginning of 2006, IT stocks had a 52 percent 12-month growth rate. The Swedish IT industry had 48 firms among Europe's 500 fastest growers in Deloitte's Technology European Fast 2006. In contrast to the late 1990s IT boom profit growth continues to rise due to better business models and high demand.

Data. This paper draws on a census of the Swedish IT services industry, provided by Statistics Sweden, Financial Statistics(FS) and Regional Labor Statistics (RAMS). The Swedish industrial classification code (SNI) for this industry is 72.⁸ The IT services industry includes the following subgroups: hardware consultancy (code 7210); software consultancy (code 7220) - customized software and packages software; data processing (code 7230); database activities (code 7240);

 $^{^{8}\}mathrm{The}$ SNI standard builds on the Statistical Classification of Economic Activities in the European Community (NACE).

maintenance and repair of office, accounting and computing machinery and data processing equipment (code 7250); and operational service activities (code 7260). Because it is difficult to divide IT consultancy services for hardware and software, I keep them in one group called software. In addition, there are few observations for hardware consultancy. On the other hand, data processing, database activities, and other computer-related services can be grouped into *operational service activities*.⁹ New firms have appeared while others have exited or merged. FS contains information on firm input and output and RAMS contains information on employee education and wages. The dataset covers the period 1996-2002. A unit of observation is a firm with one or many establishments. The computer consultancy was affected by some major changes in the last few years of the period. It is important to note that large firms can have many subsidiaries in the same sector, although I cannot observe this in my data. Appendix A provides additional information on the data and variable definitions.

According to the Swedish Business Statistics 1999, the Swedish industrial classification group 72 consists of 19,045 establishments (5,625 firms in my data) and around 71,000 employees (Table 1). The total net turnover was SEK 83.3 billion and value added was SEK 38.3 billion (values are 1996 SEK). Table 1 presents characteristics of the Swedish IT services during 1996-2002. From 1996 to 2002 at the industry level, the number of firms grew by 60 percent, the industry value-added by 100 percent, the number of employees by around 100 percent as well, total wages by 147 percent, and investments by 99 percent. Most of the growth occurred from 1996 to 2000. From 2000 to 2001 at the industry level, the number of firms grew by 3 percent, value-added by 22 percent, wages by 15 percent, the number of employees by 10 percent, and investments by 8 percent. However, the burst of the 2001 dot-com bubble induced a negative growth of about 2 percent in number of firms, about 8 percent in value-added, about 7 percent in total wages and labor, and 10 percent in investment.

Software consultancy is the sub-sector with the largest share of firms, employees, turnover and value added in relation to the total value for each of these variables, e.g., there are about 10 times more firms active in software than in operational services (Panels B and C). Software has net entry over the study period and the largest numbers of entrants (1,532) and exits (1,017) in 2000. Operational services had net entry until the burst of the 2001 dot-com bubble. Maintenance and repair is the smallest sub-sector – about 110 firms.

⁹Statistics Sweden (SCB), a Swedish government office, also uses this grouping.

Table 2 shows the impact of the 2001 dot-com bust on the growth rates by sub-sector between 2000-2001 and 2001-2002. The IT sub-sectors were affected differently. Operational service firms were more affected between 2000-2001, e.g., the number of firms decreased by around 20 percent, sales by 27 percent, and investments by 19 percent. Software firms were most affected from 2001 to 2002, i.e., sales decreased by 18 percent and investments by 10 percent.

IT service firms are also found in the following sectors: retail trade in computers; office machinery and equipment wholesale; and telecom products and electronic components wholesale. It is hard to specify the activities of these firms. Therefore, they are not included in the study.¹⁰ They represent 0.2 percent of the total number of companies and their net turnover represents 41 percent of the total net turnover in the industry. Apart from analyzing different sub-sectors, the paper also groups the firms into three classes according to number of employees: (i) small – 0-19 employees; (ii) medium – 20-99 employees; and (iii) large – over 100 employees.

In Sweden, IT services are concentrated to the three largest cities, i.e., Stockholm, Gothenburg, and Malmö. The Swedish government focuses on the IT sector and pays close attention to firm entry and exit.¹¹ Lundmark (1995) studies the patterns of growth and location of computer services in Sweden. More specifically, he analyzes location patterns of IT services in local markets. He emphasizes that the market structure of Swedish IT services is characterized by a large degree of local and regional sales, indicating the importance of proximity to customers. The Swedish IT industry is characterized by large heterogeneity. Most of the firms are small – around 90 percent of the firms in my data had fewer than 20 employees in 2000. Yet, despite the large proportion, small firms only generated about 25 percent of total employment and sales in 2000. Therefore, large firms that operate on both national and international markets are important for the overall performance.

Market definition. Information is what is demanded in the IT services industry. How much and from who depend on the type of activity carried out (in Sweden),

 $^{^{10}{\}rm However},$ the share of total turnover in the sectors that represents IT consultancy activities cannot be determined from the survey or from Swedish Business Statistics in 1999.

¹¹The Swedish Agency for Economic and Regional Growth (NUTEK) contributes to the creation of new enterprises, more growing enterprises, stronger regions, and consequently to promote sustainable economic growth and prosperity throughout the country. Another Swedish government agency for innovation, Vinnova, elaborates strategies and forms reference groups with key players from the industry, government agencies, and universities to improve the competitiveness of the IT industry.

price, training effort, and the level of learning.¹² Statistics Sweden (SCB) conducted a survey about demand structure in the Swedish IT services industry in 2001. They found that the customers of Swedish IT services are as follows: firms and public utilities around 76 percent; central government and municipal authorities 14 percent; households and individuals 0.2 percent; and exports around 10 percent. Only firms that are in the SNI group 72 were included in the survey. The customers of small firms are households and private individuals. Large and medium IT firms commonly have business enterprises as customers. While large companies dominate the Swedish IT services in terms of market share, small and medium companies dominate the market with respect to number of firms.¹³ Moreover, 50 percent of firms say that 75-100 percent of their sales come from neighboring municipalities and 35 percent of firms do not make sales in neighboring municipalities.

The paper uses Statistics Sweden's county definition to define markets. A county consists of a collections of municipalities. This classification groups the Swedish municipalities (290) into 25 markets that are mutually exclusive and exhaustive of the land mass of Sweden.¹⁴ The county-based market definition is a compromise between contradictory requirements. The theoretical model assumes that IT service markets are isolated geographic units; firms in one market interact competitively only with other firms in the same county market. Firms placed in too large markets may not all respond to the same market forces (external or actions of industry competitors). Counties are a suitable compromise to resolve the tension between isolating markets yet ensuring that the IT service firms within them are interconnected. IT service firms should, however, be close to their customers. Large firms in this sector may face international competition if they sell for example software.

Tables 3 and 4 present the summary statistics at the local market level for the Swedish IT service industry from 1996 to 2002, for all firms (Table 3) and grouped by size (Table 4). An average local market (county) has about 255 IT service firms; 3,100 employees; 7,225 non-IT firms; and a population of about 400,000 people (Table 3). Table 4 shows that an average market has about 230 small, 22 medium, and about 7 large IT firms. The counties that include Stockholm, Gothenburg, or

¹²Bower (1973) discusses the specificity of demand in IT services.

 $^{^{13}{\}rm Firms}$ that are in other SE-SIC 92 groups and provide IT services are not included in the survey due to the difficulties in measuring their activities. Cerda and Glanzelius (2003) provide more details about Swedish IT services.

¹⁴Statistics Sweden provides more detailed information, <u>www.scb.se</u>.

Malmö have about 10 times more firms than does an average county (Table 4).

Having access to detailed data on individual counties and information on demand based on surveys, demographic characteristics, population and number of firms (other than IT service firms) are good proxies for local demand.

3 The modeling approach

The model. To evaluate the impact of 2001 dot-com bust on cost structure it is necessary to have a theoretical model of the IT services industry. The model builds on the work of Ericson and Pakes (1995), who provide a theoretical framework of industry dynamics on imperfectly competitive markets. The IT industry is characterized by simultaneous entry, exit, investment, and production service decisions of firms in each local market (county). The structure within each county market is primarily determined by the distribution of capacities (IT labor) and the industrial structure of the market. The IT industry is characterized by heterogeneous firms, where skilled labor, demand, and the efficiency of using new technologies are the most important aspects. Investments in knowledge and technology change the quality of IT services tomorrow and firms pay both fixed and variable adjustment costs.

Timing. There is a number of firms in a set of markets in an infinite sequence of years. In each year, the timing of the game is as follows:

- 1. Each firm observes its current firm productivity and market demographics.
- 2. Each potential entrant receives a draw from the distribution of entry values and makes its entry decision; each incumbent firm makes its investment decision.
- 3. Each firm receives a private productivity shock and then firms compete in the product market.
- 4. Each incumbent that chooses to leave the market exits and receives its scrap payment; each entrant pays its entry fee. Firms decide on investments in labor and capital without knowing the decisions of their competitors.
- 5. The state vector adjusts and firms enter and exit.

Firms observe the state variables at the beginning of each period along with the entry, exit, investment, and production decisions of their rivals in the previous period. Private information shocks are drawn independently across periods from a known distribution. Firms do not update their expectations of future behavior after observing the actions of their rivals.

State space. All economically important characteristics of firms are incorporated into a state vector that includes productivity, market demographics, and a set of private information payoff "shocks" that affect firms' payoffs. Firms' state variables are grouped in the vector s. Firms receive state-dependent revenues from the product (service) market in each period. Entry, exit, and investments (labor and technology) influence the evolution of the state vector. The most important component of the state space is productivity, ω . Firm j's productivity in market m, ω_{jmt} , is not directly observable in the data, but is obtained by estimation of a value-added generating function model. The paper assumes that the productivity evolves stochastically according to the following Markov process:

(1)
$$\omega_{jmt} = \tilde{g}(\omega_{jmt-1}) + \xi_{jmt},$$

where $\xi_{jmt} \in N(0, \eta^{\omega})$ and $\tilde{g}(\cdot)$ is an unknown function. Thus, firms' actual productivity ω_{jmt} in period t can be decomposed into expected productivity $\tilde{g}(\omega_{jt-1})$ and a private productivity shock, ξ_{jt} . The private shock, ξ_{jt} , may be thought of as the realization of uncertainties that are naturally linked to productivity. The conditional expectation function $\tilde{g}(\cdot)$ is unobserved by the econometrician (though known to the firm) but can be estimated non-parametrically. Furthermore, I assume that ω_{jmt} evolves independently across markets.

Each market m is defined by its characteristics: the total number of firms (other than IT) in the market and population. Since population is highly correlated with number of firms (0.99), only the number of firms is used in the empirical part. The growth rates for population and number of non-IT firms evolve according to the following AR(1) process:

(2)
$$firms_{mt} = \delta_1^{firms} firms_{mt-1} + \delta_0^{firms} + v_{mt}^{firms}$$
, where $v_{mt}^{firms} \sim N(0, \eta^{firms})$.

Equilibrium concept. Equilibrium is obtained when firms follow strategies that maximize the expected discounted present value of their stream revenues given the expected strategies of their competitors. The paper assumes that firms' strategies

depend only on the current state vector and generate a Markov Perfect Nash Equilibrium (MPNE). The MPNE consists of a set of best response strategies governing entry, services production, exit, and investment.

Firm j makes decisions regarding, e.g., entry, exit, and investments collectively denoted by Γ_j . Since the full set of dynamic Nash equilibria is unbounded, I restrict firms' strategies to be anonymous, symmetric, and Markovian. Therefore, a firm's strategy, σ_{it} , can be written as a mapping from states to actions:

 $\sigma_{jt}: S_{jt} \to \Gamma_{jt}.$

A vector of strategies is a mapping of the current state of the system for each firm's strategy. The time horizon is infinite, payoffs are bounded, firms have Markovian strategies, and the discount factor β is positive and less than one. The value of a firm in state $s \in S$ is

(3)
$$V_j(s|\sigma(s)) = \pi_j(\sigma(s)) + \beta \int V_j(s'|\sigma) dP(s'|\sigma(s), s),$$

where $\sigma(s)$ is the vector of strategies, $\pi_j(\sigma(s))$ is the per-period payoff function, and $P(\cdot)$ is the conditional probability distribution governing the transition between states. A strategy profile σ is an MPNE giving competitors profile σ_{-j} if each firm j prefers strategy σ_j to all Markov strategies σ'_j :

(4)
$$V_j(s|\sigma_j^*, \sigma_{-j}) \ge V_j(s|\sigma_j', \sigma_{-j})$$

for all j, s, and σ'_{j} . I assume that such an equilibrium exists and is unique. Doraszelski and Satterthwaite (2010) discuss the details on existence and uniqueness.

I describe each component of the model in detail in the following sections by deriving the ex-ante value functions for potential entrants and incumbents. These value functions are important in the counterfactual simulations when the costs of the dot-com bust are evaluated.

4 Estimation

The estimation is made in two steps. In the first step, I estimate a value added generating function to obtain an estimate of firms' perceived productivity. Knowing how the state space evolves over time, the revenue generating function and the policy functions, which describe the optimal strategy profile for each firm, can be estimated. In the second step, I estimate the dynamic parameters governing investment, scrap values, and sunk costs.

Firm productivity. The present paper assumes a Cobb-Douglas technology where IT service firms sell a homogeneous product (at the subsector level) and that the factors underlying profitability differences among firms are neutral efficiency differences. Allowing for heterogeneity in the dynamic model makes this assumption not so restrictive. The services production function can be specified as

(5) $q_{jt} = \beta_0 + \beta_l l_{jt} + \beta_k k_{jt} + \omega_{jt} + u_{jt}^p,$

where q_{jt} is the log of service output sold by firm j at time t; l_{jt} is the log of labor input; and k_{jt} is the log of capital input. The unobserved factor, ω_{jt} , measures productivity, and u_{jt}^p is either measurement error (which can be serially correlated) or a shock to productivity that is not predictable during the period in which labor can be adjusted.

Specification (5) assumes that prices are constant across firms. When firms have some market power, prices set by individual firms influence their productivity. The negative correlation between input and prices leads to underestimation of the labor and capital parameters in the production function (Klette and Griliches, 1996; Melitz, 2000; and De Loecker, 2009). If the products are perfect substitutes, deflated sales are a perfect proxy for unobserved quality adjusted output. Following the recent literature, it is possible to correct for bias in elasticities by introducing the following downward sloping demand function:

(6)
$$p_{jt} = p_{It} + \frac{1}{\eta}q_{jt} - \frac{1}{\eta}q_{It} - \frac{1}{\eta}\lambda_{jt} - \frac{1}{\eta}u_{jt}^d$$

where p_{jt} is output price, p_{It} and q_{It} are IT service output price and quantity, λ_{jt} is demand shifters (observed and unobserved), and u_{jt}^d is simple i.i.d shock to demand.¹⁵ The demand specification assumes that firms operate in a market with horizontal product differentiation, where η (< -1 and finite) captures the

¹⁵There is no price index for IT services from 1996 to 2002. From 2002, Statistics Sweden has started to construct a price index for IT services. In the empirical part, I use the consumer index price. For robustness, I have constructed a backward price index (1996-2002) from new IT services price index (2002-2009). Even if this construction is problematic (small sample errors) it can be informative. Because there are no substantial changes in the elasticities, the results are not reported.

elasticity of substitution among IT services. Due to data constraints, the demand system is quite restrictive, implying a single elasticity of substitution for all IT services and that there are no differences in cross price elasticities.

I decompose demand shifters into observed local market characteristics $z_m t$, i.e., number of non-IT firms and population, and unobserved demand shock ν_{jt} :

$$\lambda_{jt} = z'_{mt}\beta_z + \nu_{jt},$$

where ν_{jt} are either correlated unobserved shocks to demand or i.i.d. If ν_{jt} are correlated unexpected shocks, they enter into productivity measure. Therefore, it is not possible to use a more sophisticated demand model that allows for product differentiation (Berry, 1994; Berry et al., 1995; Nevo, 2001). Since the IT service prices of individual firms are unobserved, the deflated output is defined as $y_{it} = q_{it} - p_{It}$. Firm productivity follows a first order Markov process (equation 1) and takes the following form: $\omega_{jt} = \tilde{g}(\omega_{jt-1}) + \xi_{jt}$. Controlling for price and demand shocks in the value-added generating function (5) yields

(7)
$$y_{jt} = \left(1 + \frac{1}{\eta}\right) \left[\beta_0 + \beta_l l_{jt} + \beta_k k_{jt}\right] - \frac{1}{\eta} q_{It} - \frac{1}{\eta} z'_{mt} \beta_z + g(\omega_{jt-1}) + \left(1 + \frac{1}{\eta}\right) \xi_{jt} \\ - \frac{1}{\eta} \nu_{jt} - \frac{1}{\eta} u^d_{jt} + \left(1 + \frac{1}{\eta}\right) u^p_{jt},$$

where $g(\cdot) = \left(1 + \frac{1}{\eta}\right) \tilde{g}(\cdot)$. The value of k_{jt} is determined by previous investment i_{jt-1} . Labor l_{jt} is correlated with the random shock in productivity ξ_{jt} . The inverse labor demand helps us to recover unobserved productivity ω_{jt-1} rather than recovering from the unknown policy function of investment in capital and labor/materials (Olley and Pakes, 1996; Ackerberg et al., 2006). The main advantage is that zero investments are included in the analysis, which is important because IT firms often invest one year, followed by several years without investment. In year t-1, firms chose current labor l_{jt-1} based on current productivity ω_{jt-1} , which gives demand for labor as

$$l_{jt-1} = \frac{1}{1 - \beta_l} [\beta_0 + \ln(\beta_l) + \beta_k k_{jt-1} + \omega_{jt-1} - (s_{jt-1} - p_{jt-1}) + \ln(1 + \frac{1}{\eta})],$$

where w_{jt-1} is total wages paid. Solving for ω_{jt-1} yields

(8)
$$\omega_{jt-1} = \frac{\eta}{1+\eta} \left[\delta_0 + \left[(1-\beta_l) - \frac{1}{\eta} \beta_l \right] l_{jt-1} + w_{jt-1} - p_{It-1} - \left(1 + \frac{1}{\eta} \right) \beta_k k_{jt-1} + \frac{1}{\eta} q_{mt-1} + \frac{1}{\eta} z'_{mt-1} \beta_z \right],$$

where $\delta_0 = -ln(\beta_l) - ln(1+1/\eta) - \beta_0(1+1/\eta) - lnE[e^{u_{jt}^p}] + \frac{1}{\eta}lnE[e^{u_{jt}^d}] + \frac{1}{\eta}lnE[e^{\nu_{jt}}].$ Appendix B presents the productivity estimation details.

Static firm payoffs. A firm's payoff in one period depends on its productivity, ω_{jmt} ; competitors' productivity, ω_{-jmt} ; local market characteristics, x_{mt} ; and the firm's investment and labor decisions. Therefore, the payoff of firm j in market m in period t is

(9)
$$\pi_{jmt}(\omega_{jmt}, \omega_{-jmt}, x_{mt}, \varepsilon_{jmt}^{r}; \boldsymbol{\beta}, \boldsymbol{\theta}) = r_{jmt}(\omega_{jmt}, \omega_{-jmt}, x_{mt}, \varepsilon_{jmt}^{r}) - c_{i}(i_{jmt}; \theta^{i}) - c_{l}(\Delta l_{jmt}; \theta^{l}),$$

where ε_{jmt}^r denotes the private shocks to profits; $c_i(\cdot; \theta^i)$ the cost associated with investment in technology (machinery); and $c_l(\cdot; \theta^l)$ the cost of adjusting the number of employees. In the forward simulations, the revenue generating function $r_{jmt}(\cdot)$ is estimated using the following form:

(10)

$$\begin{aligned} r_{jmt} &= \beta_0 + \boldsymbol{\beta}_1' \mathbf{bs}_1(\omega_{jmt}) + \boldsymbol{\beta}_2' \mathbf{bs}_2(\sum_{h \neq j} \omega_{hmt}) + \boldsymbol{\beta}_3' \mathbf{bs}_3(k_{jmt}) + \boldsymbol{\beta}_4' \mathbf{bs}_4(\sum_{h \neq j} k_{hmt}) \\ &+ \boldsymbol{\beta}_5' \mathbf{bs}_5(firms_{mt}) + \beta_6 a fter_{2000} + \beta_7 foreign_{jmt} \\ &+ \beta_8 medium_{jmt} + \beta_9 large_{jmt} + \boldsymbol{\beta}_m + \varepsilon_{jmt}^r, \end{aligned}$$

where $\mathbf{bs}(\cdot)$ is the basis function of cubic b-splines (Chen, 2007; Eubank, 1988; and Coppejans, 2004); $\boldsymbol{\beta}_m$ is the set of market effects introduced to capture differences in other unobserved factors that are common across all firms in a market; $firms_{mt}$ is the number of firms, other than IT, at the market level; and $medium_{imt}$ and $large_{imt}$ are dummy variables for medium and large firms.

Investment and labor costs. The cost function associated with investment in technology is:

$$c_{i}(i_{jmt};\boldsymbol{\theta}^{i}) = 1(i_{jmt} > 0)(\tilde{\theta}_{0}^{i,+} + \theta_{1}^{i,+}i_{jmt} + \theta_{2}^{i,+}(i_{jmt})^{2}) + 1(i_{jmt} < 0)(\tilde{\theta}_{0}^{i,-} + \theta_{1}^{i,-}i_{jmt} + \theta_{2}^{i,-}(i_{jmt})^{2}).$$

Fixed and variable adjustment costs vary separately for positive and negative investments. Setup costs from installing new equipment are covered by the fixed costs, $\tilde{\theta}_0$. Fixed costs of investment are private information to the firm and are drawn each period from a known distribution, $F^{i,+}(\cdot; \gamma^{i,+})$. Since the firm can sell old IT equipment, sunk costs associated with negative investment can be positive. These costs are private information and drawn each period from a common distribution.

bution, $F^{i,-}(\cdot; \gamma^{i,-})$. The cost function associated with labor adjustment is given by:

$$c_{l}(\Delta l_{jmt}; \boldsymbol{\theta}^{l}) = 1(\Delta l_{jmt} > 0)(\tilde{\theta}_{0}^{l,+} + \theta_{1}^{l,+} \Delta l_{jmt}) + 1(\Delta l_{jmt} < 0)(\tilde{\theta}_{0}^{l,-} + \theta_{1}^{l,-} \Delta l_{jmt}).$$

For example search and recruiting, training, explicit firing costs are covered by the $c^{l}(\cdot)$ function. Reorganization of services and consulting activities are also included. Fixed costs associated with positive and negative labor adjustment are drawn from the distributions $F^{l,+}(\cdot;\gamma^{l,+})$ and $F^{l,-}(\cdot;\gamma^{l,-})$.

Entry, exit, and fixed costs of operation. IT firms also have different costs that are not related to service production. To enter the market, firms pay an entry (sunk) cost, f_j^e . The entry cost is drawn from the common distribution, $F^e(\cdot; \gamma_e)$. Firms that exit the market receive the sell-off value associated with closing down the firm, f_j^x , which is commonly drawn from the common distribution, $F^x(\cdot; \gamma_x)$. Summarizing, the costs that depend on the status of the firm are:

$$f_j(\sigma(s)) = \begin{cases} f_j^e & \text{if the firm is an entrant,} \\ f_j^{in} & \text{if the firm is an incumbent,} \\ f_j^x & \text{if the firm exits the market.} \end{cases}$$

The ex-ante value functions for both potential entrants and incumbents can be written down. The value functions that give the expected discounted present value, in Swedish krona (SEK), of being at a given state vector, have two components:¹⁶ (i) the per-period payoff function and (ii) the continuation value, i.e., the expected value of next period's state. Firms use their value function to find their optimal entry, exit, investment, and labor policies.

The value function for the potential entrant j who decides to enter in the next period conditional on the current state and the draw from the distribution of the sunk cost of entry, f_j^e , can be written as:

(11)
$$V_j^e(s, f_j^e) = \max_{\substack{i_j^e, l_j^e \\ i_j^e, l_j^e}} \left\{ -f_j^e - \tilde{\theta}_0^i - \theta_1^i i_j^e - \theta_2^i (i_j^e)^2 - \theta_1^l \Delta l_j^e + \beta E(V(s')|s) \right\}.$$

The value function for an incumbent has two parts. The first part corresponds to whether the firm decides to exit the industry. If it does, it receives its services-market payoffs $\pi_j(s)$ and its sell-off payment f_j^x . If it remains active, it receives

¹⁶At the beginning of the study period (1996), 1 USD=6.71 SEK and 1 EUR=8.63 SEK.

service-market revenues. Therefore, if firm j continues, it obtains the following payoff:

(12)

$$V_j^{stay}(s) = \max_{i_j, l_j} -1(i_j > 0)(\tilde{\theta}_0^{i,+} + \theta_1^{i,+}i_j + \theta_2^{i,+}(i_j)^2) -1(\Delta l_j > 0)(\tilde{\theta}_0^{l,+} + \theta_1^{l,+}\Delta l_j) - 1(i_j < 0)(\tilde{\theta}_0^{i,-} + \theta_1^{i,-}i_j + \theta_2^{i,-}(i_j)^2) -1(\Delta l_j < 0)(\theta_0^{l,-} - \theta_1^{i,-}\Delta l_j) + \beta E(V(s')|s)$$

The ex-ante value function for an incumbent is a combination of the payoffs if the firm stays or exits:

(13)
$$V_j(s) = \int \pi_j(s_j) dS + (1 - p^x(s_j)) V_i^{stay}(s) + p^x(s_j) f_j^x.$$

In (13), $p^{x}(s_{j})$ is the probability that firm j exits the market. It is given by

(14)
$$p^{x}(s_{j}) = Pr(f_{j}^{x} > V_{j}^{stay}(s))$$
$$= 1 - F^{x}(V_{j}^{stay}(s); \gamma^{x})$$

The continuation value, $V_j^{stay}(s)$, can be obtained by inverting equation (14), $V_j^{stay}(s) = (F^x)^{-1}(1 - p^x(s); \gamma^x)$. The expected sell-off value, \tilde{f}_j^x , conditional on exit is $E[f_j^x|f_j^x > (F^x)^{-1}(1 - p^x(s); \gamma^x)]$, i.e., it is a function of the probability of exit and the parameters of the exit distribution, γ^x . The recovered values $\tilde{\theta}_0^{i,+}$, $\tilde{\theta}_0^{i,-}$, $\tilde{\theta}_0^{l,+}$, $\tilde{\theta}_0^{l,-}$, and \tilde{f}_j^x are the means of the distributions $F^{i,+}$, $F^{i,-}$, $F^{l,+}$, $F^{l,-}$, and F^x only when firms receive favorable draws. To avoid this problem, the fixed costs can be recovered using linear sieve (Ryan, 2009):

$$\begin{split} \tilde{\theta}_{0}^{i,+}(p_{j}^{i,+}) &= \pmb{\delta}^{i,+}\pmb{b}\pmb{s}(p_{j}^{i,+}(s)), \quad \tilde{\theta}_{0}^{i,-}(p_{j}^{i,-}) &= \pmb{\delta}^{i,-}\pmb{b}\pmb{s}(p_{j}^{i,-}(s)), \\ \tilde{\theta}_{0}^{l,+}(p_{j}^{l,+}) &= \pmb{\delta}^{l,+}\pmb{b}\pmb{s}(p_{j}^{l,+}(s)), \quad \tilde{\theta}_{0}^{l,-}(p_{j}^{l,-}) &= \pmb{\delta}^{l,-}\pmb{b}\pmb{s}(p_{j}^{l,-}(s)), \\ \tilde{f}_{j}^{x}(p_{j}^{x}) &= \pmb{\delta}^{x}\pmb{b}\pmb{s}(p_{j}^{x}(s)), \end{split}$$

where δ parameters are finite and $\mathbf{bs}(\cdot)$ are basis functions defined from the probability of positive investment, $p^{i,+}$; the probability of negative investment, $p^{i,-}$; the probability of positive labor adjustment, $p^{l,+}$; the probability of negative labor adjustment, $p^{l,-}$; and the probability of exit, p^x . The distribution of sunk entrycosts can be recovered by matching its cumulative distribution to the predicted probability of entry. A firm enters when the value of doing so, $EV^e(\mathbf{s})$, is larger than f_j^e . By simulating many forward paths of possible outcomes given that the firm entered, and averaging over those paths, I obtain the expected value of entry, which I then match against observed rates of entry. Therefore, the probability that a firm enters is given by

(15)
$$Pr(f_j^e \leq EV_j^e(\boldsymbol{s})) = F^e(EV^e(\boldsymbol{s}); \gamma^e),$$

where $F^{e}(\cdot; \gamma^{e})$ is the cumulative distribution of sunk entry-costs. The entry probability, estimated by logit, gives Pr(entry|s). If ns is the number of simulated states from which EV^{e} is recovered, then the parameters of the distribution are estimated from the following optimization problem:

(16)
$$\min_{\gamma^e} \frac{1}{ns} \sum_{k}^{ns} [Pr(entry|\boldsymbol{s}) - F^e(EV^e(\boldsymbol{s});\gamma^e)]^2.$$

The paper uses logit approximation to estimate entry and exit probabilities.¹⁷ To be more precise, I estimate the following entry and exit policies for all states:

$$Pr(entry|s) = \phi(\alpha_0 + \alpha_1 \sum_{h \neq j} \omega_{jmt} + \alpha_2 k_{jmt} + \alpha_3 \sum_{h \neq j} k_{hmt} + \alpha_4 firms_{mt} + \alpha_5 after_{2000} + \alpha_6 foreign_{jmt} + \alpha_7 medium_{jmt} + \alpha_8 large_{jmt} + \alpha_m)$$

$$Pr(exit|s) = \phi(\alpha_0 + \alpha_1 \omega_{jmt} + \alpha_2 \sum_{h \neq j} \omega_{jmt} + \alpha_3 k_{jmt} + \alpha_4 \sum_{h \neq j} k_{hmt} + \alpha_5 firms_{mt} + \alpha_6 after_{2000} + \alpha_7 foreign_{jmt} + \alpha_8 medium_{jmt} + \alpha_9 large_{jmt} + \alpha_m).$$

Both policy functions contain a dummy variable for before and after the dot-com bust.

Estimating structural parameters. The evolution process of the state vector and the level of payoff associated with each state are described by the first step estimation of productivity, policy functions, and evolution of demographic characteristics. In the second step of the estimation, I recover the rest of the parameters of cost functions by finding the set of parameters that make the firm's policy function optimal. Having the estimates from the first stage, I simulate the evolution of the market under different conditions. This is possible because the first stage estimates characterize what each firm would do in all possible situations. Using forward simulation, I find parameters of the optimal policy function that minimize the profitable deviations from these observed strategies.

¹⁷In many cases, entry and exit strategies take the form of simple cutoff rules in dynamic oligopoly models.

Firm behavior is simulated under two alternative strategies in order to identify the investment cost parameters. The first scenario implies that all firms use the optimal strategies recovered in the first stage; this strategy is denoted σ . The second scenario implies that a single firm deviates from the optimal strategy while all other firms use the optimal strategies. The strategy profile σ is an MPNE if and only if

(17)
$$V_j(s, \sigma_j, \sigma_{-j}; \theta) \ge V_j(s, \sigma'_j, \sigma_{-j}; \theta)$$

for all states s, all firms j, and alternative profiles σ'_{j} . The minimum distance estimator is constructed using this set of inequalities. Due to the linearity in the cost functions, the optimality conditions (17) can be re-written as $[W_j(s, \sigma_j, \sigma_{-j}; \theta) - W_j(s, \sigma'_j, \sigma_{-j}; \theta)]\theta \ge 0$. The above equation can be written in terms of profitable deviations from the optimal policy

(18)
$$g(x;\theta,\alpha) = [W_j(s,\sigma_j,\sigma_{-j};\theta) - W_j(s,\sigma'_j,\sigma_{-j};\theta)]\theta,$$

where α represents the parametrization of the policy functions. More specifically, alternative policies are drawn from a distribution F of all policies to generate a set of inequalities indexed by x. The estimates of W_j , denoted \tilde{W}_j , are obtained using forward simulation. They are used in the sample analog of the objective function

(19)
$$Q_n(\theta, \alpha) = \frac{1}{n_I} \sum_{k=1}^{n_I} (\min\{\tilde{g}(x, \theta, \alpha), 0\})^2.$$

I use the Nelder-Mead method to obtain the starting values. Then I plug the estimated parameters as started values in the Uncmin optimization routine.¹⁸ The later gives me the final estimates and the standard errors. Another alternative is to use the Laplace-type estimator (Chernozhukov and Hong, 2003). The present paper estimates the distribution of entry costs using a procedure that matches the observed entry rates to the simulated values of entering at each state.

Standard errors. The first-stage errors affect the standard errors in the second stage. The variance of the parameters is obtained directly from the Laplace estimator (equation 19). Due to the computation burden in forward simulations,

¹⁸Uncmin performs unconstrained nonlinear optimizations (http://www1.fpl.fs.fed.us/optimization.html).

I did not correct the second stage errors, i.e., the actual errors are downward biased. However, recent econometric literature suggests potentially easy computation alternatives to consider for future research. Ackerberg et al. (2009) propose a numerical equivalence between asymptotic variance for two-step semiparametric estimators when the sieves method is used in the first stage.

5 Results

This section presents the results of estimates of productivity, revenue-generating function, and optimal firm policies, i.e., in terms of entry, exit, investment in technology, and labor. The estimates of cost parameters are discussed in the second part of this section.

Before I discuss the estimated productivity results, I would like to summarize the results regarding labor productivity and capital intensity. Figures 3 and 4 present the evolution of the labor productivity distribution and capital intensity for the three IT services sectors. Labor productivity is measured as value added per number of employees. The firms in software and operational services with low labor productive (10th percentile) experienced a decrease in labor productivity in 2000, but then started to recover in 2001. The peak of median labor productivity occurred in 1999 for software and operational services and in 2000 for maintenance. While the labor productivity of median software and operational firms shows a weak but positive trend, the median maintenance firms had a negative trend after 2000. The highly labor-productive firms (90th percentile) increased their labor productivity from 1997 and 2000 (software and maintenance), but then those in software stagnated and remained fairly constant and those in operational services actually went down. The labor productivity dispersion decreased in all sectors after 2000 (particularly quickly in operational services). To avoid possible outliers, I measure productivity dispersion as the interquartile range over median. Software and operational services sectors have larger labor productivity dispersion than does the maintenance sector.

The next step is to look into capital intensity. Median firms and firms in the 90th percentile of capital intensity had an upward trend in all three sectors, but those at the 10th percentile decreased only in the maintenance sector after 2000 (Figure 4).¹⁹ The capital intensity dispersion increased for maintenance and for software (small slope of the trend). For operational services, the capital intensity dispersion decreased until 2001 and then started to increase.

Productivity estimates. The theoretical model assumes that productivity is the state variable that captures all important aspects of an IT firm and that there is a direct link between productivity and quality. So, I assume that IT firms that offer high quality services have high productivity. Table 5 presents the results from estimating the value-added generating function using OLS and the semiparametric estimator presented in Section $4.^{20}$ Firm productivity is recovered from the estimation of the value added generating function.

By using the OLS estimator, the coefficient of labor is around 1, suggesting presence of a simultaneity problem. Since firms productivity is positively correlated with labor, a large labor coefficient is not a surprise. Furthermore, it is expected that firms with large capital stock (large firms) stay in the market even if they have low productivity, i.e., the coefficient of the capital is downward biased for the OLS estimator (selection bias). Furthermore, the results show that the null hypothesis of constant returns to scale is accepted using the OLS estimator.

The last two columns of Table 5 show the estimates of the value-added generating function using the extended Olley and Pakes (1996) estimator (EOP) presented in Section 4. In addition to controlling for endogeneity and selection, the main advantage of this estimator is that it to some extent controls for a price bias by introducing a simple demand function. This allows me to estimate mark-ups for the IT services. Since we expect different demand elasticities for the different IT sectors, I make separate estimations for software and for operational services and maintenance. Column 3 (Table 5) presents the estimates for the software subsector. Compared to OLS, the labor coefficient goes in the right direction using EOP, i.e., it decreases to 0.680, and the capital coefficient increases to 0.374. The estimated elasticity of demand for software is about -4.6, implying a mark-up of 1.277. For grouped operational services and maintenance firms (column 5), the estimated labor coefficient decreases to 0.789 and the capital coefficient increases to 0.208 compared EOP and OLS. The estimated demand elasticity is about -5.96,

 $^{^{19}\}rm{Using}$ UK data, Faggio et al. (2007) find that industries with high productivity growth have a large increase in IT capital intensity.

 $^{^{20}\}mathrm{The}$ results using the Ackerberg et al. (2006) estimator (ACF) are available from the author. In the ACF estimator, I control for both endogeneity and selection. The ACF estimator controls for investment in the market threshold function that affects the likelihood of exit, but does not control for prices or wages.
yielding a mark-up of 1.52.²¹ The estimated productivity using the EOP estimator is used in the rest of the paper.

Figure 5 shows the evolution of different parts of the productivity distribution for different size classes. It does not distinguish between what IT sector the firms belong to and type of firm, e.g., an entrant, an exit, or an incumbent. The results suggest that scale matters: large firms are the most productive, followed by medium-sized and small firms. This holds for the entire productivity distribution. Low productivity firms (10th percentile) increased their productivity (small positive slope) until 2001. For the median and the high productivity firms (90th percentile), there are three distinct periods. Their productivity was rather constant from 1997 to 1999, developed positively from 1999 to 2000, but negatively starting in 2001. These periods are also important for the dispersion trend, a decrease from 1999 to 2001 for large and medium firms. Medium-sized firms show the largest decrease in productivity dispersion. Small firms have a constant productivity dispersion over time. A decrease in productivity dispersion can be interpreted as an increase in competition, i.e., firms increase their quality and become closer to each other.

Summarizing, the paper finds that the 2001 dot-com bubble bust has affected firms differently depending on productivity and size. There is a smaller difference in productivity levels among large and median low productive firms (10th percentile) than among high productive firms (90th percentile). On the other hand, the gap between small and medium-sized firms decreases in the upper part of the productivity distribution (90th percentile).

Dynamic productivity decomposition. To analyze the productivity dynamics at the sector level, the present paper uses a dynamic productivity decomposition. Olley and Pakes (1996) propose a static decomposition of aggregate productivity where the weighted productivity for continuing stores, Ω_t , has two components: (1) unweighted contribution of productivity improvements, $\overline{\Omega}_t$ and (2) market share reallocations for the continuing firms $cov(ms_{jt}, \omega_{jt}) \equiv \sum_j (ms_{jt} - \overline{ms}_t)(\omega_{jt} - \overline{\Omega}_t)$. The change in the productivity index from period t to period t', $\Delta\Omega_{t,t'}$, can be written as

(20) $\Delta\Omega_{t,t'} = \Delta\overline{\Omega}_{t,t'} + \Delta cov_{t,t'}.$

 $^{^{21}}$ It would have been more informative to estimate the mark-ups before and after the dot-com bust. Unfortunately, this is not possible due to data constraint.

Since the OP decomposition ignores entry and exit, Melitz and Polanec (2009) (MP) suggest a dynamic OP decomposition where there is a positive contribution for entering and exiting firms only when the aggregate productivity of these firms is larger than that of continuing firms in corresponding periods. The aggregate productivity in periods t and t' can be decomposed as

(21)
$$\begin{aligned} \Omega_t &= ms_{C_{t,t'},t}\Omega_{C_{t,t'},t} + ms_{X_{t,t'},t}\Omega_{X_{t,t'},t} \\ \Omega_{t'} &= ms_{C_{t,t'},t'}\Omega_{C_{t,t'},t'} + ms_{E_{t,t'},t'}\Omega_{E_{t,t'},t'}, \end{aligned}$$

where $ms_{C_{t,t'},t}$, $ms_{E_{t,t'},t'}$, and $ms_{X_{t,t'},t}$ are the aggregate market shares of incumbents in period t, of entrants in period t', and of exits firms in period t, respectively. Thus, the change in aggregate productivity can be written as

(22)
$$\Delta\Omega_{t,t'} = \Delta\overline{\Omega}_{C_{t,t'}} + \Delta cov_{C_{t,t'}} + ms_{E_{t,t'},t'}(\Omega_{E_{t,t'},t'} - \Omega_{C_{t,t'},t'}) + ms_{X_{t,t'},t}(\Omega_{C_{t,t'},t} - \Omega_{X_{t,t'},t})$$

Table 6 presents the MP productivity decomposition from 1997 to 2002 for the software and operational services sub-sectors using 1997 as the base year. For software, the productivity growth is around 21 percent from 1997 to 2000 but only 6 percent from 1997 to 2002. The largest growth occurred from 1997 to 2001 (23 percent). Entrants contributed negatively to productivity growth, i.e., entrants were less productive than continuing firms. On the other hand, the exit firms contributed positively to productivity growth, and the contribution increased over time, e.g., from 5 percent in 1997 to around 18 percent in 2001. From 1997 to 2001, the continuing software contributed the most to growth: 35 percent (1997-2000) and 45 percent (1997-2001). From 1997 to 2002, the software firms that exited contributed more to productivity growth than did continuing firms (12 percent versus 7.5 percent).

For operational services and maintenance, the productivity growth was about 70 percent from 1997 to 2000 and about 32 percent from 1997 to 2002. Continuing firms accounted for almost all productivity growth from 1997 to 2000. Yet, exit firms contributed the most (50 percent) to productivity growth from 1997 to 2002.

Summarizing, the decomposition results emphasize the importance of net exit for productivity growth in the IT services after the 2001 dot-com bust. This suggests important changes in the market dynamics after the impact of aggregate shocks in the market, e.g., less productive firms exit. **Revenue-generating function.** The estimated productivity is used to obtain a revenue-generating function, needed to evaluate the value functions, for each subsector. Table 7 presents two specifications. The first is a simple linear regression estimated by OLS. A flexible way to model firms' revenues as a function of the state variables is to use the method of linear sieves, i.e., a simple seminonparametric approach to estimate unknown functions. The second specification uses a non-parametric cubic b-splines approximation and is estimated by the OLS estimator. A nice feature of the linear sieves is their simple analytical form. This paper uses cubic b-splines as basis functions, denoted $bs(\cdot)$, which are finite dimensional piecewise polynomials (Appendix C provides a short description of the cubic b-splines).

The revenues are a function of firms' productivity, rivals' productivity, own capital, rivals' capital, number of non-IT firms in the local market, firm size (medium or large), and type of ownership (domestic or foreign). The variables are in log form. Rivals' productivity captures the effect of the competitive pressure on firms' revenues. The impact of rival size on revenues is captured by rivals' capital, which to some extent also captures competition, i.e., the number of IT firms. At the local market, number of non-IT firms measures demand for IT services.²²

The OLS results (Panel A) show that doubling productivity increases revenues about 58 percent for software, and 53 percent for operational services and for maintenance and repair. Doubling the capital stock increases revenues by about 37 percent for software, 29 percent for operational services, and 33 percent for maintenance and repair firms. If the number of non-IT firms at the local market level doubles, then revenues go up about 12 percent for software, 33 percent for operational services, and 22 percent for maintenance and repair firms. Rivals' productivity and capital have a negative impact on software and operational firms' sales. For software and operational services, foreign IT firms have revenues about 19 percent higher than domestic IT firms. The dot-com bubble bust decreases revenues by about 20 percent for software and operational services and by 34 percent for maintenance and repair.

Panel B (Table 7) presents only a summary of the estimation results using b-splines with 6, 9, and 14 knots. For software and operational services, the adjusted R^2 increases, the root of mean squared errors (RMSE) and absolute mean errors (MAE) decrease using b-splines as basis functions for firms' own productiv-

 $^{^{22}}$ There is a high correlation (0.99) between number of firms and population at the county level.

ity, firms' own capital, rivals' productivity, and rivals' capital. This suggests that there is no need to use nonlinear approximation for operational services' revenues. Even if the simple linear regression does a good job estimating revenues, there is a significant increase in adjusted R^2 , about 13 percentage points, and the RMSE decreases from 0.622 (OLS) to 0.599 when cubic b-splines are used for the software industry, for example. In the forward simulations, B-spline specification with 14 knots is used to estimate the value functions for software and operational services. **Policy functions.** The next step is to estimate investment and labor policy functions for all firms. In addition, I estimate the entry and exit policies. All these policies are estimated for each IT sub-sector. Table 8 presents the logit estimates for the exit (Panel A) and entry (Panel B) policies. In my dataset, entry and exit are based on organization number.

For all sub-sectors, high productivity firms and firms located in markets with a large number of firms (only software and operational services) are less likely to exit. In all sub-sectors, firms are also more likely to exit after the 2001 dot-com bust. For software and operational services, firms are less likely to exit in markets where rivals have large capital stocks and high productivity. In the Swedish case, large firms are located in large markets and might have subsidiary (technology and innovation clusters). However, markets with large capital stocks imply high demand, i.e., there is still room to differentiate in these sub-sectors. For software and operational services, I find that firms with large capital are more likely to exit. There have been many technological innovations in this industry in recent years. Failing to update utilized technology has a negative impact on firm performance. Acquisitions were made during the studied period. This may explain the findings that software and operational service firms are more likely to exit if the firm has over 20 employees.

My findings indicate that IT firms are more likely to enter if rivals have high productivity, i.e., if there is sufficient demand. Software firms are less likely to enter markets where rivals have extensive capital, i.e., markets with large firms, but more likely to enter markets with a large number of firms. Hence, these markets offer sufficient demand and skilled labor. It is less likely to have foreign entrants (software and operational services) and to observe entry after the dot-com bust.

Table 9 shows the investment policy function estimates for all IT firms by subsector. Panel A presents the estimates from a simple linear investment specification. Panel B shows summary results from non-parametric regressions using cubic b-splines as basis functions for linear sieves approximations of unknown functions in own productivity, rivals' productivity, and rivals' capital. Both regression specifications use the OLS estimator. For software and operational services, productivity has a positive and significant effect on investment, i.e., firms with high productivity invest more in capital. For maintenance and repair, firms invest more if rivals reduce their capacity. For all IT sub-sectors, firms with large capital stock invest more, but they invest less after 2000. For software and operational services, increasing the business opportunities, i.e., increasing the number of non-IT firms, has a positive impact on investment. Allowing the marginal effects to depend on the size of the variables (b-splines specifications), the accuracy of recovering the observed investment increases. By using the non-parametric specification, the adjusted \mathbb{R}^2 increases from 17 to over 86 percent for software, from 37 to 85 percent for operational services, and from 58 to 95 percent for maintenance and repair. Allowing for non-linearities in productivity and capital reduces the RMSE at least two times, and the correlation between observed and predicted investment increases from 0.41 to 0.93 for software, from 0.61 to 0.93 for operational services, and from 0.77 to 0.98 for maintenance and repair firms.

Table 10 presents the labor policy function results for all IT firms. A linear specification does a good job fitting the observed number of employees. For all IT sectors, the adjusted R^2 is about 90 percent. Allowing for non-linearities gives a better fit only for software and maintenance and repair labor policies.²³

By doubling their productivity, the number of employees increases by about 52 percent for software and by about 48 percent for operational services and maintenance and repair firms. A double capital stock increases the number of employees about by 27 percent for software, by 20 percent for operational services, and by about 15 percent for maintenance and repair. If the number of firms doubles (double potential demand), operational services firms increase labor by about 33 percent and maintenance and repair firms by about 50 percent. On average, foreign firms have about 10 percent more employees than domestic ones in software. The corresponding numbers for operational services and maintenance and repair are around 8 percent and 37 percent, respectively. After the dot-com bust, IT firms reduced the number of employees by about 28 percent in software, by 24 percent in operational services, and by about 22 percent in maintenance and repair.

Recovering cost parameters. In the second step, I obtain the cost parameters

 $^{^{23}\}mathrm{A}$ negative adjusted R^2 obtained for operational services regressions suggests that there is no need for non-linearities.

for each IT sub-sector before and after the IT dot-com bust. First, the value functions are estimated using the policy functions estimated in the previous subsection. The value functions are the expectations of discounted profits over current and future states, and profit shocks. In the estimation, 100 forward simulations are used and the discount parameter, β , is fixed to 0.95. For policy functions, the cubic b-spline estimates are used to extend the panel (forward simulations).

Since IT firms compete at the local market level, the order of generating the policy functions is very important. First, future productivity and capital stock are generated for each firm and year. To generate future productivity previous investment is also added as variable in the $q(\cdot)$ function. Second, the rivals' productivity and capital are computed for each market and year. Third, the revenues and required labor are generated knowing that the population and the number of other firms evolve as exogenous processes. Fourth, the exit estimates are used to simulate whether firms exit or continue. If a firm continues, the investment and setup cost components for investment and labor are generated using cubic bsplines with 14 knots. The setup cost components are the basis cubic b-splines in the estimated probability to invest, dis-invest, hire, or fire. The high setup costs might cause the observed lumpy investment and lack of adjustment in employment, i.e. the dynamics of investment and labor, to depend on the setup costs. Hamermesh (1989) finds empirical evidence that the adjustment labor costs, which are independent of the level changes, are determinants of lumpy adjustment.²⁴ I assume that employment dynamics in IT are generated by a process that distinguishes between hiring and firing costs.²⁵

Table 11 shows the estimated cost parameters for each sub-sector before (1997-2000) and after (2001-2002) the 2001 dot-com bust. Panel A presents the estimated results for investment. The quadratic cost of adjustment implies that the future value of additional capital depends on the choice with respect to adjustment. I present only the results for positive investment. After 2003, IT firms started to invest again (Section 2). Therefore, to be close to how industry behave I use the investment policy function from 1996-2000 in the forward simulation after 2003. Since capital stock is a state variable, allowing for excessive negative investment

 $^{^{24}\}mathrm{Hamermesh}$ and Pfann (1996) find evidence of asymmetric adjustment costs, e.g., the cost of advertising might be proportional to the number of hired employees but not firing cost (Pfann and Palm, 1993). Since the asymmetry implies non-linearity in the shape of the cost function, it would be impossible to estimate the cost parameters without additional approximations to reach linearity.

²⁵In my case, I observe the number of employees in November.

makes smaller firms have a short life. For this reason, the results for negative investment are not significant. This paper finds that setup investment costs are higher for software but lower for operational services and maintenance after the dot-com bust. In the IT services case, the setup investment cost is associated with system and network configuration costs when firms change technology. Operational service firms have at least twice the setup costs of software and maintenance and repair. This sub-sector includes data processing and database activities, which require large costs in case of migration from one system to another even if the machines get cheaper due to technological innovations. However, the implied distribution of investment costs indicates higher investment costs, on average, for operational services after the dot-com bust even if the setup costs are lower (Panel B). During the study period, there was significant innovation with respect to on both hardware and software in the database management area, e.g., integration of Oracle (commercial product) and MySQL (free) on Linux. Fast access to information became very important at the same time as the complexity of the information stored increased. Therefore, firms had to invest in advanced technologies, which might explain the larger cost after the dot-com bust in this sub-sector. This contrasts the maintenance and repair sub-sector, where investment costs decreased after dot-com. Since advances in technology make hardware cheaper, firms prefer to buy rather than repair.

Panel C shows the estimates for labor adjustment costs. For software, the setup costs of positive labor adjustment are about 4 times larger after the dot-com bust. The variable marginal labor adjustment cost is about SEK 322,000 before and about SEK 271,000 after the dot-com bubble. The findings of larger setup costs and smaller variable adjustment labor costs after the dot-com bust suggest that firms face uncertainties regarding demand and might rather work with external consultants than hire new staff. The firing setup costs (about SEK 366,000) are about 2 times larger after the dot-com bust.

For operational services, the results indicate larger setup costs for positive labor adjustment before the dot-com bust. However, the setup costs are larger for negative than for positive adjustment. Here, setup labor costs might also include expenses in connection with employees' training, e.g., training to become a certified expert. This might explain the decrease in labor adjustment setup costs after the dot-com bust, when the firms were focused on reducing the costs due to the aggregate decrease in demand. Operational services has larger marginal cost of adjustment after the dot-com bust, i.e., about SEK 379,000 before and SEK 432,708 after. The marginal labor adjustment costs are larger for operational services than for software firms.

For maintenance and repair, the estimates indicate about the same positive labor adjustment setup cost (SEK 400,000) before and after the dot-com bust. Furthermore, the marginal cost of positive labor adjustment is with about SEK 50,000 less (SEK 245,000). The parameters for the negative adjustment costs could not be identified due to too few observations.

Having the estimated labor parameters, the implied distributions of cost for labor adjustment can be computed (Panel D). The 2001 dot-com bust implies higher positive labor adjustment costs for software but lower positive labor adjustment cost for operational services and maintenance and repair.

Distributions of exit and entry sunk costs. A median firm that exits has one employee for software and two employees for operational services and maintenance and repair. After the burst of the dot-com bubble, firms in software and operational services have higher (sell-off) while the maintenance and repair firms have lower scrap values (Table 11, Panel E).

To estimate entry cost, I assume that it follows a normal distribution. Using the minimum distance estimator, I recover mean and standard deviation for each industry before and after the dot-com bust. A median entrant in software or operational services has two employees, and a median entrant in maintenance and repair has three. The mean entry cost for software and operational services is estimated to be about SEK 19,000 and SEK 120,000, respectively. However, I find no significant difference between entry costs before and after the dot-com bust for software and operational services, i.e., demand uncertainty and large setup costs might explain the decrease in the number of entrants and their size. For maintenance and repair, the mean entry cost (about SEK 135,000) is not significant. One possible explanation to this is that this industry had few entrants. The low value of sunk entry costs for software firms – about 6 times lower than for operational services – explains the observed differences in the number of entrants; i.e., the yearly number of entrants in software was about 8 times the number of entrants in operational services.

6 Conclusions

This paper analyzes the impact of the 2001 dot-com bust on productivity and cost structure in the Swedish IT services. To understand the differences in productivity among the IT service sub-sectors and how the firms change their behavior when facing demand shocks, the paper analyses the possible changes in cost structure caused by the burst of the 2001 dot-com bubble. Since changes in the cost structure impact the market dynamics and therefore productivity growth, they are important both for market structure and agencies that support this industry, which is dominated by small firms (number of firms).

The results show that from 1997 to 2002, the productivity growth was about 6 percent for software and about 32 percent for operational services. After the dot-com bust, net exit contributed the most to productivity growth, suggesting important changes in the market dynamics after this aggregate shock.

Entrants are less productive than incumbent IT firms. Software firms invest more if there is an increase in business opportunities at the local market level. On average for the IT sub-sectors, firms with large capital stocks invest more. Yet, they invested less after the 2001 dot-com bust. This study finds that, among the low productivity IT firms, medium and large firms were affected the most by the dot-com bust.

The findings indicate that differences between setup (fixed) and variable costs help explain observed behavior in investment and labor policies in the IT services industry. Since the relative importance of setup and variable adjustment costs can not be measured directly from the observed data, they are inferred from the model. The downturn in productivity growth after the dot-com bubble can be explained not only by reduced demand but also by large adjustment costs.

Changes in cost structure cause changes in prices (and vice-versa), but this important aspect is not explicitly modeled here due to difficulties finding price data for IT services. Even if the paper controls for unobserved prices in an indirect way, there is still possible to have correlated unobserved demand shocks in estimated productivity (Ackerberg et al., 2008; Foster et al., 2008; and De Loecker, 2009). A detailed investigation of demand and a better understanding of the entry process would be interesting for future research.

References

- ACKERBERG, D., L. BENKARD, S. BERRY, AND A. PAKES (2008): Handbook of Econometrics, vol. 6, chap. Econometric Tools for Analyzing Market Outcomes, pp. 4171–4276. Elsevier.
- ACKERBERG, D., K. CAVES, AND G. FRASER (2006): "Structural Identification of Production Functions," Mimeo, UCLA.
- ACKERBERG, D. A., X. CHEN, AND J. HAHN (2009): "A Practical Asymptotic Variance Estimator for Two-Step Semiparametric Estimators," mimeo, UCLA.
- AGUIRREGABIRIA, V., AND P. MIRA (2007): "Sequential Estimation of Dynamic Discrete Games," *Econometrica*, 75(1), 1–53.
- AI, C., AND X. CHEN (2003): "Efficient Estimation of Models with Conditional Moment Restrictions Containing Unknown Functions," *Econometrica*, 71(6), 1795–1843.
- BAJARI, P., L. BENKARD, AND J. LEVIN (2007): "Estimating Dynamic Models of Imperfect Competition," *Econometrica*, 75(5), 1331–1370.
- BARTELSMAN, E., AND M. DOMS (2000): "Understanding Productivity: Lessons from Longitudinal Microdata," *Journal of Economic Literature*, 38(3), 569–594.
- BENKARD, C. (2004): "A Dynamic Analysis of the Market for Widebodied Commercial Aircraft," *Review of Economic Studies*, 71, 581–611.
- BERESTEANU, A., AND P. ELLICKSON (2006): "The Dynamics of Retail Oligopoly," Mimeo, Duke University.
- BERRY, S. (1994): "Estimating Discrete-Choice Models of Product Differentiation," The RAND Journal of Economics, 25(2), 242–262.
- BERRY, S., J. LEVINSOHN, AND A. PAKES (1995): "Automobile Prices in Market Equilibrium," *Econometrica*, 63(4), 841–890.
- BLOOM, N., R. SADUN, AND J. VAN REENEN (2009): "Americans Do I.T. Better: US Multinationals and the Productivity Miracle," Discussion paper, LSE Working Paper.
- BOWER, R. (1973): "Market changes in the Computer Services Industry," Bell Journal of Economics and Management Science, 4(2), 539–590.
- CERDA, M., AND M. GLANZELIUS (2003): "Service Price Index for Computer Services. Industry description for SE-SIC 92 group 72 TPI project report no.2," Discussion paper, Statistics Sweden.

- CHEN, X. (2007): "Large Sample Sieve Estimation of Semi-Nonparametric Models," in *The Handbook of Econometrics*, ed. by J. Heckman, and E. Leaner, vol. 6.
- CHERNOZHUKOV, V., AND H. HONG (2003): "A MCMC Approach to Classical Estimation," *Journal of Econometrics*, 115(2), 293–346.
- COPPEJANS, M. (2004): "On Kolmogorov's Representation of Functions of Several Variables by Functions of One Variable," *Journal of Econometrics*, 123, 1–31.
- DE BOOR, C. (1978): A Practical Guide to Splines. Springer, Berlin.
- DE LOECKER, J. (2009): "Product Differentiation, Multi-Product Firms and Structural Estimation of Productivity," Mimeo, Princeton University.
- DORASZELSKI, U., AND J. JAUMANDREU (2009): "R&D and Productivity: Estimating Production Functions when Productivity is Endogenous," Mimeo, Harvard University.
- DORASZELSKI, U., AND M. SATTERTHWAITE (2010): "Computable Markov-Perfect Industry Dynamics," *Rand Journal of Economics*, 41(2), 215–243.
- DUNNE, T., S. KLIMEK, M. ROBERTS, AND Y. XU (2005): "Entry and Exit in Geographical Markets," Mimeo, Pennsylvania State University.
- ERICSON, R., AND A. PAKES (1995): "Markov-Perfect Industry Dynamics: A Framework for Empirical Work," *Review of Economic Studies*, 62, 53–83.
- EU ICT TASK FORCE REPORT (2006): "Fostering the Competitiveness of Europe's ICT Industry," Discussion paper, European Commission.
- EUBANK, R. L. (1988): Spline Smoothing and Nonparametric Regression. Markel Kedder, New York.
- FAGGIO, G., G. SALVANES, AND J. VAN REENEN (2007): "The Evolution of Inequality in Productivity and Wages," Discussion Paper 13351, NBER Working Paper Series.
- FOSTER, L., J. HALTIWANGER, AND C. SYVERSON (2008): "Reallocation, Firm Turnover, and Efficiency: Selection on Productivity or Profitability?," *American Economic Review*, 98(1), 394–425.
- HAMERMESH, D. A. (1989): "Labor Demand and the Structure of Adjustment Costs," American Economic Review, 79, 674–689.
- HAMERMESH, D. A., AND G. A. PFANN (1996): "Adjustment Costs in Factor Demand," Journal of Economic Literature, 34, 1264–1292.

- HOTZ, V., AND R. MILLER (1993): "Conditional Choice Probabilities and the Estimation of Dynamic Models," *Review of Economic Studies*, 60(3), 497–529.
- HULTEN, C., AND F. WYKOFF (1981): chap. The measurement of Economic Depreciation, in C. Hulten (ed.), *Depreciation, Inflation, and Taxation of Income* from Capital, Urban Institute Press, Washington, D.C.
- JORGENSON, D. W., M. S. HO, AND K. J. STIROH (2005): Information Technology and the American Growth Resurgence. Cambridge: MIT Press.

(2008): "A Retrospective Look at the U.S. Productivity Growth Resurgence," *Journal of Economic Perspective*, 22, 1.

- KLETTE, T., AND Z. GRILICHES (1996): "The Inconsistency of Common Scale Estimators when Output Prices are Unobserved and Endogenous," *Journal of Applied Econometrics*, 11(4), 343–361.
- LUNDMARK, M. (1995): "Computer Services in Sweden: Markets, Labor Qualifications and Patterns of Location," Geografiska Annaler. Series B, Human Georgaphy, 77(2), 125–139.
- MAICAN, F., AND M. ORTH (2009): "Productivity Dynamics and the Role of "Big-Box" Entrants in Retailing," Working Paper no. 328, Department of Economics, University of Gothenburg.
- MELITZ, M. (2000): "Estimating Firm Level Productivity in Differentiated Product Industries," Mimeo, Harvard University.
- MELITZ, M., AND S. POLANEC (2009): "Dynamic Olley-Pakes Decomposition with Entry and Exit," Mimeo, Princeton University.
- NEVO, A. (2001): "Mergers with Differentiated Products: The Case of the Readyto-Eat Cereal Industry," *Econometrica*, 69(2), 307–342.
- NEWEY, W., AND J. POWELL (2003): "Instrumental Variables Estimation of Nonparametric Models," *Econometrica*, 71(3), 1557–1569.
- NEWEY, W., J. POWELL, AND F. VELLA (1999): "Nonparametrric Estimation of Triangular Simultaneous Equations Models," *Econometrica*, 67(3), 565–603.
- OLLEY, S., AND A. PAKES (1996): "The Dynamics of Productivity in the Telecommunications Equipment Industry," *Econometrica*, Vol. 64(No. 6), 1263– 1297.
- PAKES, A. (2008): "Theory and Empirical Work on Imperfectly Competitive Markets," Discussion Paper 14117, NBER.

(2010): "Alternative Models for Moment Inequalities," Discussion paper, Harvard University.

- PAKES, A., M. OSTROVSKY, AND S. BERRY (2007): "Simple Estimators for the Parameters of Discrete Dynamic Games, with Entry/Exit Examples," *RAND Journal of Economics*, 38(2), 373–399.
- PAKES, A., J. PORTER, J. HO, AND J. ISHII (2007): "Moment Inequalities and their Application," Discussion paper, mimeo, Harvard University.
- PESENDORFER, M., AND SCHMIDT-DENGLER (2003): "Identification and Estimation of Dynamic Games," Mimeo, London School of Economics.
- PFANN, G. A., AND F. PALM (1993): "Asymmetric Adjustment Cost in Nonlinear Labor Demand Models for the Netherlands and U.K. Manufacturing Sectors," *Review of Economic Studies*, 60(2), 397–412.
- ROBINSON, P. (1988): "Root-N-Consistent Semiparametric Regression," Econometrica, 56(4), 931–954.
- RYAN, S. (2009): "The Costs of Environmental Regulation in a Concentrated Industry," Mimeo, MIT.
- RYAN, S., AND C. TUCKER (2006): "Heterogeneity and the Dynamics of Technology Adoption," Mimeo, MIT.
- SCHUMAKER, L. L. (1981): Spline Functions Basic Theory. Wiley, New York.
- SWEETING, A. (2007): "Dynamic Product Repositioning in Differentiated Product Markets: The Case of Format Switching in the Commercial Radio Industry," Discussion Paper 13522, NBER.
- SYVERSON, C. (2010): "What Determines Productivity?," NBER Working Paper Series, (15712).
- VAN ARK, B., M. O'MAHONY, AND M. P. TIMMER (2008): "The Productivity Gap between Europe and the United States: Trends and Causes," *Journal of Economic Perspectives*, 22(1), 25–44.



Figure 1: Western European ICT market growth, 1997-2007, in percent. Source: EITO 2006 in cooperation with IDC.



Figure 2: EU ICT, excluding Cyprus and Malta, market growth by segment, 2005-2007, in percent. Source: EITO 2006 in cooperation with IDC.

Vear Vear	Firms	Entry	Exit	Sales	Value	Total	Employees	Investment
rear	1 11 11 15	Lintry	LIAIO	04165	added	wages	Employees	investment
1996	4 116		543	48 320 538	21 459 073	12 070 602	42 686	1 774 708
1007	4,110	1 166	551	53 200 042	25 707 591	14 506 458	49.883	2 038 002
1998	5 109	1 149	644	66 707 752	31 395 570	17 851 110	59 208	2,050,502
1999	5 625	1 185	820	83 369 434	38 282 741	22 787 128	71 133	3 022 545
2000	6 523	1 694	1 203	96 284 420	39 710 885	28 085 689	85 928	3 668 669
2001	6 749	1 336	1 326	112 979 488	48 588 640	32 312 569	94 096	3 947 814
2001	6 623	962	1,020	100 931 185	44 672 840	29 804 633	87 567	3 539 552
D. C.f	0,020	001		100,001,100	11,012,010	20,001,000	01,001	0,000,002
1006	2 474		445	25 690 017	16 449 094	0.276.169	20 590	1 110 591
1990	3,474	072	440	49 241 881	20 666 581	9,570,102	32,000	1,110,561
1997	3,002	1 000	441 596	42,341,001	20,000,581	11,901,720	40,393	1,400,592
1998	4,398	1,022	530 700	20,348,497	20,907,847	10,241,528	49,007	1,741,335
1999	4,908	1,051	1.017	67,495,845	32,512,499	19,477,029	59,823	2,299,510
2000	5,742	1,532	1,017	76,266,817	32,079,924	23,499,892	70,792	2,937,217
2001	5,043	1,222	1,168	97,738,433	42,385,864	28,361,781	81,518	3,355,883
2002	5,932	856		80,063,479	36,335,808	24,621,346	71,526	3,007,020
C: Op	erational	services						
1996	527		88	11,114,449	4,374,038	2,294,675	8,580	646,102
1997	583	174	100	9,435,368	4,379,401	2,139,978	7,996	549,014
1998	601	122	88	8,888,705	3,781,396	2,199,835	7,939	493,259
1999	616	124	102	13,227,458	4,896,611	2,774,268	9,497	678,515
2000	671	147	168	17,881,246	6,739,992	4,039,204	13,299	709,624
2001	602	101	141	12,969,458	5,325,738	3,305,638	10,298	574,791
2002	592	93		18,956,159	7,454,423	4,606,482	13,932	518,766
D: Ma	aintenance	e and rep	air					
1996	115		10	1,567,272	637,011	399,765	1,526	18,025
1997	116	19	10	1,432,693	661,608	384,756	1,494	23,295
1998	110	5	20	1,470,551	706,326	409,747	1,612	23,191
1999	101	10	9	2,646,131	873,630	535,830	1,813	44,520
2000	110	15	18	2,136,357	890,968	546,592	1,837	21,828
2001	104	13	17	2,271,597	877,038	645, 150	2,280	17,139
2002	99	13		1,911,548	882,608	576,804	2,109	13,764

Table 1: Characteristics of the data for the Swedish IT service industry 1996-2002.

NOTE: The data are from the merge between Financial Statistics(FS) and Regional Labor Statistics(RAMS) databases. Sales, value-added, wages, investment are measured in thousand 1996 SEK. 1 USD=6.71 SEK and 1 EUR=8.63 SEK.

Table 2:	The 2001	$\operatorname{dot-com}$	bust:	growth rates	(percent)	2000-2001	and	2001-2002.
----------	----------	--------------------------	-------	--------------	-----------	-----------	-----	------------

		-	(=	,				
	Fii	ms	Sa	les	Empl	oyees	Inves	tment
	2000-2001	2001-2002	2000-2001	2001-2002	2000-2001	2001-2002	2000-2001	2001-2002
Software	5	-2	28	-18	15	-10	14	-10
Operational services	-10	-2	-27	46	-25	58	-19	-10
Maintenance and repair	-5	-5	6	-16	17	-7	21	-20
				((

NOTE: The data come from the merge between Financial Statistics(FS) and Regional Labor Statistics(RAMS) databases. The growth rates are computed at the sub-sector level.

Variable	Minimun	Mean	Median	Maximum	Standard
					Deviation
Services production					
Sales	257	36,403	6,569	798,279	118,663
Value added	70	16, 196	3,082	333,439	51,010
Capital	2,189	201,392	32,373	4,745,895	667,003
Employees	24	3,179.99	773	62,314	9,534
Wages	40	10,203	1,891	226,977	33,383
Demand					
Other firms	916	7,225	4,338	42,477	8,809
Population	57,313	399,814	269,699	1,838,882	416,484
Competition					
IT firms	6	255	87	3,490	593
Investment					
Investment	-4,307	1,068	168	23,349	3,594

 Table 3: Summary statistics at the local market level for the Swedish IT service industry 1996-2002.

NOTE: The data come from the merge between Financial Statistics(FS) and Regional Labor Statistics(RAMS) databases. There are 160 observations in 25 regional markets. The variables are aggregated at the county level. Sales, value-added, wages, capital, and investment are measured in thousand 1996 SEK. 1 USD=6.71 SEK and 1 EUR=8.63 SEK.

A. Small size IT servi	ce firms: 0-19	employees			
Variable	Minimun	Mean	Median	Maximum	Standard
					Deviation
Services production	n				
Sales	164	8,908	2,389	137,769	23,116
Value added	70	3,721	1,102	52,299	9,324
Capital	1,263	47,820	14,169	739,642	118,627
Employees	17	799	273	10,861	1,855
Wages	40	2,259	627	37,211	5,989
Competition					
IT firms	6	230	81	3,061	527
Investment					
Investment	-234	249	75	4,771	660
B. Medium size IT set	rvice firms: 20-	99 employe	ees		
Services production	n				
Sales	29	8,616	2,280	145,898	23,682
Value added	-55	3,782	1,102	61,821	9,837
Capital	341	41,397	8,994	828,553	115,150
Employees	20	846	260	13,128	2,102
Wages	38	2,771	736	50,442	7,657
Competition					
IT firms	1	22	7	356	55
Investment					
Investment	-183	245	44	5,753	703
C. Large size IT servi	ce firms: over i	100 employ	ees		
Services production	n				
Sales	314	33,116	6,125	514,612	93,380
Value added	217	15,211	3,329	224,736	41,269
Capital	0	195,859	46,654	3,610,016	579,167
Employees	104	2,707	596	38,325	7,248
Wages	184	9,107	1,746	139,325	25,648
Competition					
IT firms	1	7	2	90	17
Investment					
Investment	-4,393	1,006	184	19,792	3,155

 Table 4: Summary statistics of Swedish IT service grouped by size

NOTE: The data are from the merge between Financial Statistics(FS) and Regional Labor Statistics(RAMS) databases. There are 160 observations in 25 regional markets. The variables are aggregated at the county level. Sales, value-added, wages, capital, and investment are measured in thousand 1996 SEK. 1 USD=6.71 SEK and 1 EUR=8.63 SEK.



Figure 3: Evolution of the labor productivity percentiles and dispersion from 1996 to 2002.



Figure 4: Evolution of the capital intensity percentiles and dispersion from 1996 to 2002.

	OLS	EOP	OLS	EOP
	software	software	Op. services and maintenance	Op. services and maintenance
Log No Emp.	1.017	0.680	0.995	0.789
	(0.006)	(0.0004)	(0.0163)	(0.031)
Log capital	0.118	0.374	0.169	0.208
	(0.004)	(0.004)	(0.010)	(0.003)
Market output		0.217		0.168
		(0.004)		(0.010)
Scale	1.135	1.347	1.164	1.196
Demand		-4.609		-5.96
Mark-up		1.277		1.524
Sargan (p-value)		0.101		0.125
No. obs.	28,277	28,277	4,028	4,028

 Table 5: Estimates of value-added generating function parameters in Swedish IT

 services

NOTE: OLS is ordinary least square regression including year dummies; EOP is the semiparametric estimation of equation (23) specified in Section 4, including selection. Two-stage GMM is used in the EOP estimation. Reported standard errors (in parentheses) are robust to heteroscedasticity.



Figure 5: Evolution of the total factor productivity percentiles and dispersion for different size classes from 1996 to 2002.

$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$	A. Software							
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$		Surviving firms	Difference	Difference	Difference	Change in	Difference	Difference
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$		change in	between entering	between surviving	between surviving	covariance for	between entering	between surviving
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$		weighted	and surviving	and exiting	and exiting	surviving firms	and surviving	and exiting
Surviving firms productivity productivity productivity productivity Year Prod. Growth $(w_{C_{\ell'}} - w_{C_{\ell}})$ $s_{E_{\ell}}(w_{E_{\ell'}} - w_{C_{\ell}})$ $s_{E_{\ell}}(w_{E_{\ell'}} - w_{C_{\ell'}})$ $s_{E_{\ell'}}(w_{E_{\ell'}} - w_{C_{\ell'}})$ <		productivity	in weighted	in weighted	in unweighted		in unweighted	in unweighted
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$		Surviving firms	productivity	productivity	productivity		productivity	productivity
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$	Year Prod. Growth	$(\omega_{C_t}, -\omega_{C_t})$	$s_{E_{\star}\prime}\left(\omega_{E_{\star}\prime}-\omega_{C_{\star}\prime} ight)$	$s_{X_t}(\omega_{C_t}-\omega_{X_t})$	$(\overline{\omega}_{C_t} - \overline{\omega}_{X_t})$	$(cov_{C_t}, - cov_{C_t})$	$s_{E_{+\prime}}(\overline{\omega}_{E_{+\prime}}-\overline{\omega}_{C_{+\prime}})$	$s_{X_t}(\overline{\omega}_{C_t} - \overline{\omega}_{X_t})$
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	(1)	(2)	(3) (3)	(4)	(5)	(9)	(<u>L</u>)	(8)
$ \begin{array}{ccccccccccccccccccccccccccccccccccc$	1998 -0.005	0.062	-0.123	0.055	-0.020	0.083	-0.033	0.036
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	1999 -0.313	-0.205	-0.211	0.102	-0.284	0.078	-0.055	0.063
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	2000 0.208	0.353	-0.235	060.0	0.276	0.077	-0.024	0.087
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	2001 0.231	0.457	-0.401	0.175	0.318	0.139	-0.033	0.072
B. Operational services and repair 0.128 0.083 -0.189 0.234 0.171 -0.087 -0.031 1998 0.128 0.083 -0.189 0.234 0.171 -0.087 -0.031 1999 0.138 0.275 -0.108 -0.028 0.002 0.273 -0.074 2000 0.707 0.028 -0.033 0.461 0.246 -0.053 2001 0.663 0.990 -0.537 0.209 0.594 -0.017 2002 0.390 -0.166 0.209 0.535 -0.177 -0.013	2002 0.063	0.075	-0.135	0.123	-0.049	0.124	-0.059	0.078
$ \begin{array}{ c c c c c c c c c c c c c c c c c c c$	B. Operational service	s and repair						
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	1998 0.128	0.083	-0.189	0.234	0.171	-0.087	-0.031	0.199
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	1999 0.138	0.275	-0.108	-0.028	0.002	0.273	-0.074	0.102
2001 0.663 0.990 -0.537 0.209 0.595 0.394 -0.017 2002 0.323 0.004 -0.186 0.505 -0.172 0.177 -0.013	2000 0.702	0.707	0.028	-0.033	0.461	0.246	-0.053	0.097
2002 0.323 0.004 -0.186 0.505 -0.172 0.177 -0.013	2001 0.663	0.990	-0.537	0.209	0.595	0.394	-0.017	0.031
	2002 0.323	0.004	-0.186	0.505	-0.172	0.177	-0.013	0.062

	Ψ
	and
	entry
	with
	position
	decom
	productivity
:	Ulley-Pakes 1
	ynamic
ĥ	с С
	ole

A. Linear parametric specification	1		
Variable	Software	Operational	Maintenance
		services	and repair
Intercept	6.371	5.956	7.075
	(0.102)	(0.345)	(0.686)
Own productivity	0.585	0.522	0.534
	(0.004)	(0.015)	(0.028)
Own capital	0.367	0.292	0.326
	(0.004)	(0.013)	(0.025)
Rivals' productivity	-6.63E-05	-1.85E-04	-3.45E-04
	(6.38E-06)	(7.86E-05)	(0.001)
Rivals' capital	-3.22E-08	-1.05E-08	-3.24E-06
	(4.57E-09)	(6.13E-08)	(3.79E-06)
Number of firms (other than IT)	0.119	0.327	0.222
	(0.011)	(0.039)	(0.083)
Foreign owner	0.186	0.198	0.389
	(0.032)	(0.088)	(0.276)
Medium size (20-99 employees)	0.199	0.042	-0.054
	(0.024)	(0.073)	(0.170)
Large size (over 100 employees)	0.458	0.304	-0.098
	(0.045)	(0.139)	(0.359)
After 2000	-0.202	-0.194	-0.335
	(0.017)	(0.052)	(0.090)
Adjusted B^2	0.821	0.814	0.808
Boot of mean squares errors	0.622	0.708	0.583
Absolute mean errors	0.387	0.708	0.340
	0.001	0.001	0.040
B. Linear non-parametric specifica	ation using cu	ubic b-splines	
Adjusted P2	$er \text{ or knots}, \kappa$	$x_n = 0$	0.812
Registed A	0.652		0.615
Absolute mean emerg	0.004		0.302
Absolute mean errors	0.305		0.510
A directed D2	$er \text{ or knots}, \kappa$	$c_n \equiv 9$	0.000
Aujusteu A	0.833		0.822
Absolute mean errors	0.600		0.524
Absolute mean errors	0.300 mofilmoto <i>h</i>	- 14	0.275
Adjusted P^2	$r \text{ or knots}, \kappa$	n = 14	0.650
Reat of mean squares arrors	0.834		0.829
Absolute mean errors	0.399		0.490
Absolute mean errors	0.559		0.240

Table 7: Revenues generation function estimates

NOTE: The dependent variable is the log of revenues. The independent variables are as follows: Own productivity measures the firm's productivity; own capital is the firm's capital stock; rivals' productivity is the log of sum of rivals' productivity at the county level; rivals' capital is the log of rivals' productivity at the county level; rivals' capital is the log of the number of non-IT firms at the county level; population is the log of population at the county level; foreign owner is a dummy variable indicating whether the firm has foreign ownership; medium size (20-99 employees), large size (over 100 employees), and after 2000 are dummy variables for the firm's are in parentheses.

A. Exit policies			
Variable	Software	Operational	Maintenance
		services	and repair
Intercept	14.292	-0.744	-0.431
	(0.479)	(0.948)	(2.32)
Own productivity	-0.321	-0.389	-0.285
	(0.014)	(0.036)	(0.087)
Own capital	0.027	0.084	0.021
	(0.012)	(0.032)	(0.086)
Rivals' productivity	-0.771	0.022	-0.350
	(0.022)	(0.042)	(0.179)
Rivals' capital	-0.646	0.093	-0.204
	(0.035)	(0.043)	(0.116)
Number of firms (other than IT)	-0.208	-0.496	-0.033
	(0.035	(0.120)	(0.268)
Foreign owner	0.628	0.730	1.122
	(0.102)	(0.226)	(0.635)
Medium size (20-99 employees)	0.015	0.795	0.687
,	(0.088)	(0.200)	(0.601)
Large size (over 100 employees)	1.484	0.548	0.614
	(0.149)	(0.415)	(1.004)
After 2000	1.978	2.079	2.603
	(0.035)	(0.093)	(0.247)
Log-likelihood	-11851.188	-1635.361	-270.744
The likelihood ratio index	0.389	0.289	0.390
B. Entry policies			
Intercept	-1.051	-3.144	-4.822
	(0.353)	(0.860)	(1.931)
Rivals' productivity	0.068	0.184	0.434
* 0	(0.015)	(0.036)	(0.150)
Rivals' capital	-0.192	-0.016	0.100
	(0.028)	(0.039)	(0.110)
Number of firms (other than IT)	0.265	0.177	0.134
,	(0.028)	(0.116)	(0.232)
Foreign owner	-0.581	-0.468	0.334
0	(0.115)	(0.287)	(0.744)
Medium size (20-99 employees)	-0.867	-0.902	-0.784
· · · · · · · · · · · · · · · · · · ·	(0.064)	(0.168)	(0.631)
Large size (over 100 employees)	-1.307	-1.665	-0.234
3 · · · (· · · · · · · · · · · · · · ·	(0.165)	(0.478)	(1.016)
After 2000	-0.760	-1.240	-1.416
	(0.034)	(0.107)	(0.282)
Log-likelihood	-15849.964	-1852.225	-330.732
The likelihood ratio index	0.184	0.194	0.255

Table 8: Entry and exit policy functions estimation results

NOTE: The estimations are done using logit estimator. The independent variables are as follows: Own productivity measures the firm's productivity; own capital is the firm's capital stock; rivals' productivity is the log of sum of rivals' productivity at the county level; rivals' capital is the log of rivals' productivity at the county level; Number of firms other than IT is the log of the number of non-IT firms at the county level; population is the log of population at the county level; foreign owner is a dummy variable indicating whether the firm has foreign ownership; medium size (20-99 employees), large size (over 100 employees), and after 2000 are dummy variables for the firm's size and the period following the 2001 dot-com bust. The standard errors are in parentheses.

A. Investment policies for IT servi	ces		
Variable	Software	Operational	Maintenance
		services	and repair
Intercept	-1.565	-3.635	-0.778
	(3.682)	(5.385)	(0.543)
Own productivity	0.078	0.201	-0.003
	(0.032)	(0.179)	(0.019)
Own capital	0.474	0.947	0.105
	(0.027)	(0.165)	(0.017)
Rivals' productivity	-0.067	1.898	0.251
	(0.268)	(4.155)	(0.446)
Rivals' capital	-1.732	-4.728	-0.318
	(4.573)	(6.855)	(0.175)
Number of firms (other than IT)	0.108	0.245	0.041
	(0.057)	(0.425)	(0.039)
Foreign owner	0.632	1.940	0.631
	(0.212)	(1.059)	(0.192)
Medium size (20-99 employees)	-0.551	-2.328	0.373
	(0.157)	(0.881)	(0.118)
Large size (over 100 employees)	9.192	27.088	2.120
	(0.295)	(1.672)	(0.253)
After 2000	-0.188	-0.466	-0.127
	(0.089)	(0.553)	(0.061)
Adjusted R^2	0.174	0.367	0.583
Root of mean squares errors	4.085	8.501	0.405
Absolute mean errors	16.688	72.280	0.165
Correlation (observed, predicted)	0.418	0.609	0.770
B. Linear non-parametric specifica	tion using c	ubic b-splines	
Number	of knots, k_n	$_{1} = 6$	
Adjusted R^2	0.729	0.743	0.877
Root of mean squares errors	2.335	5.382	0.213
Absolute mean errors	5.451	28.966	0.045
Correlation (observed, predicted)	0.854	0.883	0.942
Number	of knots, k_n	$_{i} = 9$	
Adjusted R^2	0.836	0.767	0.945
Root of mean squares errors	1.818	5.075	0.136
Absolute mean errors	3.305	25.763	0.018
Correlation (observed, predicted)	0.915	0.887	0.976
Number	of knots, k_n	= 14	
Adjusted R^2	0.869	0.854	0.950
Root of mean squares errors	1.618	3.997	0.125
Absolute mean errors	2.619	15.982	0.015
Correlation (observed, predicted)	0.933	0.930	0.980

Table 9: Investment policy functions estimation results

NOTE: The dependent variable is the investment expenditure in ten thousand Swedish krona. The independent variables are as follows: Own productivity measures the firm's productivity; own capital is the firm's capital stock; rivals' productivity is the log of sum of rivals' productivity at the county level; rivals' capital is the log of rivals' productivity at the county level; Number of firms other than IT is the log of the number of non-IT firms at the county level; population is the log of population at the county level; foreign owner is a dummy variable indicating whether the firm has foreign ownership; medium size (20-99 employees), large size (over 100 employees), and after 2000 are dummy variables for the firm's size and the period following the 2001 dot-com bust. For Panel A, the standard errors are in parentheses.

A. Labor policies for IT services			
Variable	Software	Operational	Maintenance
		services	and repair
Intercept	1.012	1.603	1.620
	(0.423)	(0.619)	(0.824)
Own productivity	0.522	0.482	0.485
	(0.002)	(0.006)	(0.010)
Own capital	0.267	0.199	0.151
	(0.002)	(0.006)	(0.016)
Rivals' productivity	-1.261	-2.542	-1.069
	(0.174)	(0.387)	(0.568)
Rivals' capital	-1.224	-0.339	-0.302
	(0.725)	(0.680)	(0.151)
Number of firms (other than IT)	0.017	0.327	0.500
	(0.041)	(0.105)	(0.158)
Population	0.122	-0.025	-0.177
	(0.045)	(0.116)	(0.173)
Foreign owner	0.098	0.078	0.372
	(0.018)	(0.041)	(0.092)
After 2000	-0.278	-0.243	-0.219
	(0.009)	(0.022)	(0.036)
Adjusted R^2	0.899	0.929	0.939
Root of mean squares errors	0.351	0.329	0.245
Absolute mean errors	0.123	0.108	0.060
Correlation (observed, predicted)	0.949	0.964	0.970
B. Linear non-parametric specifica	tion using c	ubic b-splines	
Number	of knots, k_n	a = 6	
Adjusted R^2	0.943	-1.272	0.960
Root of mean squares errors	0.069	1.865	0.193
Absolute mean errors	0.263	3.480	0.037
Correlation (observed, predicted)	0.971	0.584	0.981
Number	of knots, k_n	$_{a} = 9$	
Adjusted R^2	0.945	-8.892	0.963
Root of mean squares errors	0.067	3.858	0.178
Absolute mean errors	0.259	14.888	0.031
Correlation (observed, predicted)	0.972	0.507	0.984
Number	of knots, k_n	= 14	
Adjusted R^2	0.946	-2.010	0.964
Root of mean squares errors	0.065	2.112	0.169
Absolute mean errors	0.256	4.461	0.028
Correlation (observed, predicted)	0.973	0.475	0.985

Table 10: Labor policy functions estimation results

NOTE: The dependent variable is the log of number of employees. The independent variables are as follows: Own productivity measures the firm's productivity; own capital is the firm's capital stock; rivals' productivity is the log of sum of rivals' productivity at the county level; rivals' capital is the log of rivals' productivity at the county level; Number of firms other than IT is the log of the number of non-IT firms at the county level; population is the log of population at the county level; foreign owner is a dummy variable indicating whether the firm has foreign ownership; medium size (20-99 employees), large size (over 100 employees), and after 2000 are dummy variables for the firm's size and the period following the 2001 IT bubble burst. For Panel A, the standard errors are in parentheses.

Variable	Software		Operational		Mainte	Maintenance	
	Before	After	Before	After	Before	After	
A: Investment cost	Defore	ritter	Defore	111001	Defore	111001	
Setup investment	170.61	323.68	667.13	645.98	193.01	148.23	
Std.	(9.60)	(7.63)	(5.81)	(9.51)	(8.59)	(5.27)	
Variable investment	6.26	5.36	10.02	13.80	34.39	14.07	
Std.	(0.16)	(0.60)	(0.33)	(0.70)	(4.57)	(9.36)	
Variable investment squared	6.76e-6	295e-5	4.09e-5	8.60e-5	0.003	0.06	
Std.	(8.03e-7)	(4.32e-6)	(2.10e-6)	(5.37e-6)	(0.0008)	(0.04)	
B: Implied distributions of inve	estment cos	ts		. ,		. ,	
Investment - mean	4.293.81	4.673.79	3.015.53	3,491.63	1.043.34	787.17	
Investment - std.	2,992.91	44.75	241.43	453.68	50.69	75.27	
C: Labor adjustment cost							
Setup positive adjustemnt	141.89	666.67	413.50	89.63	400.47	404.28	
Std.	(78.92)	(347.88)	(151.26)	(34.89)	(60.20)	(87.47)	
Variable positive adjustment	322.58	271.27	379.26	432.708	294.36	245.14	
Std.	(2.81)	(2.17)	(2.06)	(4.34)	(9.76)	(2.20)	
Setup negative adjustment	167.12	366.72	562.01	210.55	337.74		
Std.	(67.37)	(89.02)	(87.50)	(93.33)	(99.74)		
Variable negative adjustment	285.14	285.82	324.05	279.41	262.132		
Std.	(1.47)	(4.57)	(3.959)	(26.38)	(2.01)		
D: Implied distributions of lab	or adjustme	ent costs					
Positive adjustment - mean	2,587.32	2,890.39	2,000.37	1,515.04	2,534.41	2,089	
Positive adjustment - std.	1,039.67	1,943.26	171.25	671.25	475.75	319.42	
Negative adjustment - mean	1,262	1,429.50	1,650.87	927.97	1,003.73		
Negative adjustment - std.	478.26	631,68	287.41	150.04	66.98		
E: Exit costs							
Scrap (sell-of values)	170.01	175.01	230.00	255.15	260.11	248.04	
Std.	(61.02)	(50.29)	(27.24)	(23.45)	(40.57)	(34.78)	
F: Entry costs							
Sunk cost	18.69	19.92	120.61	120.33	135.96	117.14	
Std.	(4.08)	(4.66)	(34.63)	(34.53)	(146.97)	(104.30)	

Table 11: Cost estimates by sub-sector before and after the 2001 dot-com bust

NOTE: The estimates are obtained using 2,000 simulations with 100 years each, where the initial states are held constant across simulations. Standard errors in parentheses.

Appendix A: Data. This section describes the variables in the data. Value added is total shipments, adjusted for changes in inventories, minus the cost of materials. Real value added is constructed by deflating value added by a fivedigit industry output deflater. The deflectors are taken from Statistics Sweden. The labor variable is the total number of employees. The total wages come from RAMS. I deflated sales, wages, and investment by the consumer price index (CPI) from IMF-CDROM 2005. The capital measure is constructed using a perpetual inventory method, $k_{t+1} = +(1 - \delta)k_t + i_t$. Since the capital data distinguish between buildings and equipment, all calculations of the capital stock are done separately for buildings and equipment. As suggested by Hulten and Wykoff (1981), buildings are depreciated at a rate of 0.0361 and equipment at a rate of 0.1179. In the empirical part, the paper only uses machinery and equipment capital stock.

Appendix B: Productivity estimation. Considering the high turnover in the industry, it is important to control for selection. Olley and Pakes' (1996) approach to control for selection is to substitute the predicted survival probability into $g(\cdot)$. Thus, the final value-added generating function to be estimated is

(23)
$$y_{jt} = \left(1 + \frac{1}{\eta}\right) \left[\beta_0 + \beta_l l_{jt} + \beta_k k_{jt}\right] + -\frac{1}{\eta} q_{mt} - \frac{1}{\eta} z'_{mt} \beta_z + g(\mathscr{P}_{t-1}, \omega_{jt-1}) \\ + \left(1 + \frac{1}{\eta}\right) \xi_{jt} - \frac{1}{\eta} \nu_{jt} - \frac{1}{\eta} u^d_{jt} + \left(1 + \frac{1}{\eta}\right) u^p_{jt},$$

where ω_{it-1} comes from (8).²⁶

The value-added generating function (23) is estimated using the sieve minimum distance (SMD) procedure proposed by Newey and Powell (2003) and Ai and Chen (2003) for independent and identically distributed (i.i.d) data. The goal is to obtain an estimable expression for the unknown parameters β and g_{K_T} where K_T indicates all parameters in $g(\cdot)$. To approximate $g(\cdot)$, a third order polynomial in ω_{t-1} is used.²⁷ A tensor product polynomial series of labor, capital, large entrants and local market conditions are used as instruments. To compute ω_{t-1} and, hence, to approximate $g(\cdot)$, I use the following instruments. This set of instruments is also used to estimate (23) using the optimal weighting matrix are the

²⁶The condition for identification is that the variables in the parametric part of the model are not perfectly predictable (in the least square sense) by the variables in the non-parametric part (Robinson, 1988). Hence, there cannot be a functional relationship between the variables in the parametric and non-parametric part (Newey et al., 1999). Including additional variables that affect productivity guarantees the identification.

 $^{^{27}\}text{For robustness, the expand }g(\cdot)$ using a 4th order polynomial was also used. Yet, the results were similar.

following: $\{1, l_{t-1}, s_{t-1}, k_{t-1}, p_{It-1}, w_{mt-1}, pop_{mt-1}, firms_{mt-1}, \}$. Using the specified GMM implementation, the parameter values (β, g_{K_T}) are jointly estimated. The Nelder-Mead numerical optimization method is used to minimize the GMM objective function.

Appendix C: B-Splines. We consider a cubic spline, $f(x), x \in [a, b]$, with q interior knots, $\xi_i, i = 1, \dots, q$, that can be written as a sum of piecewise polynomials of order 4 (degree 3) on the any interval $[\xi_{i-1}, \xi_i]$,

$$f(x) = \sum_{m=0}^{3} \delta_{mi} x^{m}, \quad x \in [\xi_{i-1}, \xi_i), \quad i = 1, \cdots, q, \quad or \quad x \in [\xi_q, \xi_{q+1}], i = q+1.$$

The function f(x) is assumed to be twice continuously differentiable. Collinearity is a potential problem when using cubic spline in regressions. For this reason, the b-splines are preferred because of their numerical properties. For b-splines, the basis is derived recursively.²⁸ To do this, additional knots, such as $\xi_{-3} = \xi_{-2} =$ $\xi_{-1} = a$ and $\xi_{q+2} = \xi_{r+3} = \xi_{r+4} = b$, have to be added. The basis for b-splines, $\{B_{i,4}\}_{i=-3}^{q}$ is given by

$$B_{i,n} = \frac{x - \xi_i}{\xi_{i+n-1} - \xi_i} B_{i,n-1}(x) + \frac{\xi_{i+n} - x}{\xi_{i+n} - \xi_{i+1}} B_{i+1,n-1}(x), \quad \text{if} \quad n=2,3,4,$$

where $B_{i,1}(x)$ is equal to 1 if $x \in [\xi_i, \xi_{i+1})$ and 0 otherwise. Having the basis, the cubic b-spline is given by

$$bs(x) = \sum_{i=1}^{q+4} \alpha_i B_{i-4,4}(x),$$

where $\alpha_i, i = 1, \cdots, q + 4$ is the set of coefficients.

²⁸Schumaker (1981) provides a detailed overview on spline theory. de Boor (1978) and Eubank (1988) provide detailed information on b-splines.

Previous doctoral theses in the Department of Economics, Gothenburg

Avhandlingar publicerade innan serien Ekonomiska Studier startades (Theses published before the series Ekonomiska Studier was started):

Östman, Hugo (1911), Norrlands ekonomiska utveckling Moritz, Marcus (1911), Den svenska tobaksindustrien Sundbom, I. (1933), Prisbildning och ändamålsenlighet Gerhard, I. (1948), Problem rörande Sveriges utrikeshandel 1936/38 Hegeland, Hugo (1951), The Quantity Theory of Money Mattsson, Bengt (1970), Cost-Benefit analys Rosengren, Björn (1975), Valutareglering och nationell ekonomisk politik Hialmarsson, Lennart (1975), Studies in a Dynamic Theory of Production and its Applications Örtendahl, Per-Anders (1975), Substitutionsaspekter på produktionsprocessen vid massaframställning Anderson, Arne M. (1976), Produktion, kapacitet och kostnader vid ett helautomatiskt emballageglasbruk Ohlsson, Olle (1976), Substitution och odelbarheter i produktionsprocessen vid massaframställning Gunnarsson, Jan (1976), Produktionssystem och tätortshierarki – om sambandet mellan rumslig och ekonomisk struktur Köstner, Evert (1976), Optimal allokering av tid mellan utbildning och arbete Wigren, Rune (1976), Analys av regionala effektivitetsskillnader inom industribranscher Wästlund, Jan (1976), Skattning och analys av regionala effektivitetsskillnader inom industribranscher

Flöjstad, Gunnar (1976), Studies in Distortions, Trade and Allocation Problems Sandelin, Bo (1977), Prisutveckling och kapitalvinster på bostadsfastigheter Dahlberg, Lars (1977), Empirical Studies in Public Planning

Lönnroth, Johan (1977), Marxism som matematisk ekonomi

Johansson, Börje (1978), Contributions to Sequential Analysis of Oligopolistic Competition

<u>Ekonomiska Studier</u>, utgivna av Nationalekonomiska institutionen vid Göteborgs Universitet. Nr 1 och 4 var inte doktorsavhandlingar. (The contributions to the department series 'Ekonomiska Studier' where no. 1 and 4 were no doctoral theses):

- 2. **Ambjörn, Erik** (1959), Svenskt importberoende 1926-1956: en ekonomiskstatistisk kartläggning med kommentarer
- Landgren, K-G. (1960), Den "Nya ekonomien" i Sverige: J.M. Keynes, E. Wigfors och utecklingen 1927-39
- 5. **Bigsten, Arne** (1979), Regional Inequality and Development: A Case Study of Kenya
- 6. **Andersson, Lars** (1979), Statens styrning av de kommunala budgetarnas struktur (Central Government Influence on the Structure of the Municipal Budget)
- 7. **Gustafsson, Björn** (1979), Inkomst- och uppväxtförhållanden (Income and Family Background)

- 8. **Granholm, Arne** (1981), Interregional Planning Models for the Allocation of Private and Public Investments
- 9. **Lundborg, Per** (1982), Trade Policy and Development: Income Distributional Effects in the Less Developed Countries of the US and EEC Policies for Agricultural Commodities
- 10. **Juås, Birgitta** (1982), Värdering av risken för personskador. En jämförande studie av implicita och explicita värden. (Valuation of Personal Injuries. A comparison of Explicit and Implicit Values)
- 11. **Bergendahl, Per-Anders** (1982), Energi och ekonomi tillämpningar av inputoutput analys (Energy and the Economy - Applications of Input-Output Analysis)
- 12. **Blomström, Magnus** (1983), Foreign Investment, Technical Efficiency and Structural Change Evidence from the Mexican Manufacturing Industry
- 13. Larsson, Lars-Göran (1983), Comparative Statics on the Basis of Optimization Methods
- 14. **Persson, Håkan** (1983), Theory and Applications of Multisectoral Growth Models
- 15. Sterner, Thomas (1986), Energy Use in Mexican Industry.
- 16. **Flood, Lennart** (1986), On the Application of Time Use and Expenditure Allocation Models.
- 17. **Schuller, Bernd-Joachim** (1986), Ekonomi och kriminalitet en empirisk undersökning av brottsligheten i Sverige (Economics of crime an empirical analysis of crime in Sweden)
- 18. **Walfridson, Bo** (1987), Dynamic Models of Factor Demand. An Application to Swedish Industry.
- Stålhammar, Nils-Olov (1987), Strukturomvandling, företagsbeteende och förväntningsbildning inom den svenska tillverkningsindustrin (Structural Change, Firm Behaviour and Expectation Formation in Swedish Manufactury)
- 20. **Anxo, Dominique** (1988), Sysselsättningseffekter av en allmän arbetstidsförkortning (Employment effects of a general shortage of the working time)
- 21. **Mbelle, Ammon** (1988), Foreign Exchange and Industrial Development: A Study of Tanzania.
- 22. **Ongaro, Wilfred** (1988), Adoption of New Farming Technology: A Case Study of Maize Production in Western Kenya.
- 23. Zejan, Mario (1988), Studies in the Behavior of Swedish Multinationals.
- 24. **Görling, Anders** (1988), Ekonomisk tillväxt och miljö. Förorenings-struktur och ekonomiska effekter av olika miljövårdsprogram. (Economic Growth and Environment. Pollution Structure and Economic Effects of Some Environmental Programs).
- 25. **Aguilar, Renato** (1988), Efficiency in Production: Theory and an Application on Kenyan Smallholders.
- 26. Kayizzi-Mugerwa, Steve (1988), External Shocks and Adjustment in Zambia.

- 27. **Bornmalm-Jardelöw, Gunilla** (1988), Högre utbildning och arbetsmarknad (Higher Education and the Labour Market)
- 28. **Tansini, Ruben** (1989), Technology Transfer: Dairy Industries in Sweden and Uruguay.
- 29. Andersson, Irene (1989), Familjebeskattning, konsumtion och arbetsutbud En ekonometrisk analys av löne- och inkomstelasticiteter samt policysimuleringar för svenska hushåll (Family Taxation, Consumption and Labour Supply An Econometric Analysis of Wage and Income Elasticities and Policy Simulations for Swedish Households)
- 30. **Henrekson, Magnus** (1990), An Economic Analysis of Swedish Government Expenditure
- 31. **Sjöö, Boo** (1990), Monetary Policy in a Continuous Time Dynamic Model for Sweden
- 32. Rosén, Åsa (1991), Contributions to the Theory of Labour Contracts.
- 33. **Loureiro, Joao M. de Matos** (1992), Foreign Exchange Intervention, Sterilization and Credibility in the EMS: An Empirical Study
- 34. **Irandoust, Manuchehr** (1993), Essays on the Behavior and Performance of the Car Industry
- 35. **Tasiran, Ali Cevat** (1993), Wage and Income Effects on the Timing and Spacing of Births in Sweden and the United States
- 36. **Milopoulos, Christos** (1993), Investment Behaviour under Uncertainty: An Econometric Analysis of Swedish Panel Data
- 37. Andersson, Per-Åke (1993), Labour Market Structure in a Controlled Economy: The Case of Zambia
- 38. Storrie, Donald W. (1993), The Anatomy of a Large Swedish Plant Closure
- 39. Semboja, Haji Hatibu Haji (1993), Energy and Development in Kenya
- 40. **Makonnen, Negatu** (1993), Labor Supply and the Distribution of Economic Well-Being: A Case Study of Lesotho
- 41. Julin, Eva (1993), Structural Change in Rural Kenya
- 42. Durevall, Dick (1993), Essays on Chronic Inflation: The Brazilian Experience
- 43. **Veiderpass, Ann** (1993), Swedish Retail Electricity Distribution: A Non-Parametric Approach to Efficiency and Productivity Change
- 44. **Odeck, James** (1993), Measuring Productivity Growth and Efficiency with Data Envelopment Analysis: An Application on the Norwegian Road Sector
- 45. **Mwenda, Abraham** (1993), Credit Rationing and Investment Behaviour under Market Imperfections: Evidence from Commercial Agriculture in Zambia
- 46. **Mlambo, Kupukile** (1993), Total Factor Productivity Growth: An Empirical Analysis of Zimbabwe's Manufacturing Sector Based on Factor Demand Modelling
- 47. Ndung'u, Njuguna (1993), Dynamics of the Inflationary Process in Kenya
- 48. **Modén, Karl-Markus** (1993), Tax Incentives of Corporate Mergers and Foreign Direct Investments
- 49. Franzén, Mikael (1994), Gasoline Demand A Comparison of Models
- 50. **Heshmati, Almas** (1994), Estimating Technical Efficiency, Productivity Growth And Selectivity Bias Using Rotating Panel Data: An Application to Swedish Agriculture

- 51. **Salas, Osvaldo** (1994), Efficiency and Productivity Change: A Micro Data Case Study of the Colombian Cement Industry
- 52. **Bjurek, Hans** (1994), Essays on Efficiency and Productivity Change with Applications to Public Service Production
- 53. **Cabezas Vega, Luis** (1994), Factor Substitution, Capacity Utilization and Total Factor Productivity Growth in the Peruvian Manufacturing Industry
- 54. **Katz, Katarina** (1994), Gender Differentiation and Discrimination. A Study of Soviet Wages
- 55. **Asal, Maher** (1995), Real Exchange Rate Determination and the Adjustment Process: An Empirical Study in the Cases of Sweden and Egypt
- 56. Kjulin, Urban (1995), Economic Perspectives on Child Care
- 57. **Andersson, Göran** (1995), Volatility Forecasting and Efficiency of the Swedish Call Options Market
- 58. **Forteza, Alvaro** (1996), Credibility, Inflation and Incentive Distortions in the Welfare State
- 59. Locking, Håkan (1996), Essays on Swedish Wage Formation
- 60. Välilä, Timo (1996), Essays on the Credibility of Central Bank Independence
- 61. **Yilma, Mulugeta** (1996), Measuring Smallholder Efficiency: Ugandan Coffee and Food-Crop Production
- 62. **Mabugu, Ramos E.** (1996), Tax Policy Analysis in Zimbabwe Applying General Equilibrium Models
- 63. **Johansson, Olof** (1996), Welfare, Externalities, and Taxation; Theory and Some Road Transport Applications.
- 64. **Chitiga, Margaret** (1996), Computable General Equilibrium Analysis of Income Distribution Policies in Zimbabwe
- 65. **Leander, Per** (1996), Foreign Exchange Market Behavior Expectations and Chaos
- 66. Hansen, Jörgen (1997), Essays on Earnings and Labor Supply
- 67. **Cotfas, Mihai** (1997), Essays on Productivity and Efficiency in the Romanian Cement Industry
- 68. **Horgby, Per-Johan** (1997), Essays on Sharing, Management and Evaluation of Health Risks
- 69. **Nafar, Nosratollah** (1997), Efficiency and Productivity in Iranian Manufacturing Industries
- 70. **Zheng, Jinghai** (1997), Essays on Industrial Structure, Technical Change, Employment Adjustment, and Technical Efficiency
- 71. **Isaksson, Anders** (1997), Essays on Financial Liberalisation in Developing Countries: Capital mobility, price stability, and savings
- 72. Gerdin, Anders (1997), On Productivity and Growth in Kenya, 1964-94
- 73. **Sharifi, Alimorad** (1998), The Electricity Supply Industry in Iran: Organization, performance and future development
- 74. **Zamanian, Max** (1997), Methods for Mutual Fund Portfolio Evaluation: An application to the Swedish market
- 75. **Manda, Damiano Kulundu** (1997), Labour Supply, Returns to Education, and the Effect of Firm Size on Wages: The case of Kenya
- 76. **Holmén, Martin** (1998), Essays on Corporate Acquisitions and Stock Market Introductions

- 77. Pan, Kelvin (1998), Essays on Enforcement in Money and Banking
- 78. **Rogat, Jorge** (1998), The Value of Improved Air Quality in Santiago de Chile
- 79. Peterson, Stefan (1998), Essays on Large Shareholders and Corporate Control
- 80. **Belhaj, Mohammed** (1998), Energy, Transportation and Urban Environment in Africa: The Case of Rabat-Salé, Morocco
- 81. **Mekonnen, Alemu** (1998), Rural Energy and Afforestation: Case Studies from Ethiopia
- 82. **Johansson, Anders** (1998), Empirical Essays on Financial and Real Investment Behavior
- 83. Köhlin, Gunnar (1998), The Value of Social Forestry in Orissa, India
- 84. Levin, Jörgen (1998), Structural Adjustment and Poverty: The Case of Kenya
- 85. Ncube, Mkhululi (1998), Analysis of Employment Behaviour in Zimbabwe
- 86. Mwansa, Ladslous (1998), Determinants of Inflation in Zambia
- 87. **Agnarsson, Sveinn** (1998), Of Men and Machines: Essays in Applied Labour and Production Economics
- 88. **Kadenge, Phineas** (1998), Essays on Macroeconomic Adjustment in Zimbabwe: Inflation, Money Demand, and the Real Exchange Rate
- 89. Nyman, Håkan (1998), An Economic Analysis of Lone Motherhood in Sweden
- 90. Carlsson, Fredrik (1999), Essays on Externalities and Transport
- 91. Johansson, Mats (1999), Empirical Studies of Income Distribution
- 92. Alemu, Tekie (1999), Land Tenure and Soil Conservation: Evidence from Ethiopia
- 93. Lundvall, Karl (1999), Essays on Manufacturing Production in a Developing Economy: Kenya 1992-94
- 94. Zhang, Jianhua (1999), Essays on Emerging Market Finance
- 95. **Mlima, Aziz Ponary** (1999), Four Essays on Efficiency and Productivity in Swedish Banking
- 96. **Davidsen, Björn-Ivar** (2000), Bidrag til den økonomisk-metodologiske tenkningen (Contributions to the Economic Methodological Thinking)
- 97. Ericson, Peter (2000), Essays on Labor Supply
- 98. **Söderbom, Måns** (2000), Investment in African Manufacturing: A Microeconomic Analysis
- 99. **Höglund, Lena** (2000), Essays on Environmental Regulation with Applications to Sweden
- 100. Olsson, Ola (2000), Perspectives on Knowledge and Growth
- 101. Meuller, Lars (2000), Essays on Money and Credit
- 102. Österberg, Torun (2000), Economic Perspectives on Immigrants and Intergenerational Transmissions
- 103. **Kalinda Mkenda, Beatrice** (2001), Essays on Purchasing Power Parity, RealExchange Rate, and Optimum Currency Areas
- 104. **Nerhagen, Lena** (2001), Travel Demand and Value of Time Towards an Understanding of Individuals Choice Behavior

- 105. **Mkenda, Adolf** (2001), Fishery Resources and Welfare in Rural Zanzibar
- 106. Eggert, Håkan (2001), Essays on Fisheries Economics
- 107. **Andrén, Daniela** (2001), Work, Sickness, Earnings, and Early Exits from the Labor Market. An Empirical Analysis Using Swedish Longitudinal Data
- 108. Nivorozhkin, Eugene (2001), Essays on Capital Structure
- 109. **Hammar, Henrik** (2001), Essays on Policy Instruments: Applications to Smoking and the Environment
- 110. **Nannyonjo, Justine** (2002), Financial Sector Reforms in Uganda (1990-2000): Interest Rate Spreads, Market Structure, Bank Performance and Monetary Policy
- 111. Wu, Hong (2002), Essays on Insurance Economics
- 112. Linde-Rahr, Martin (2002), Household Economics of Agriculture and Forestry in Rural Vienam
- 113. **Maneschiöld, Per-Ola** (2002), Essays on Exchange Rates and Central Bank Credibility
- 114. Andrén, Thomas (2002), Essays on Training, Welfare and Labor Supply
- 115. **Granér, Mats** (2002), Essays on Trade and Productivity: Case Studies of Manufacturing in Chile and Kenya
- 116. Jaldell, Henrik (2002), Essays on the Performance of Fire and Rescue Services
- 117. Alpizar, Francisco, R. (2002), Essays on Environmental Policy-Making in Developing Countries: Applications to Costa Rica
- 118. Wahlberg, Roger (2002), Essays on Discrimination, Welfare and Labor Supply
- 119. Piculescu, Violeta (2002), Studies on the Post-Communist Transition
- 120. **Pylkkänen, Elina** (2003), Studies on Household Labor Supply and Home Production
- 121. Löfgren, Åsa (2003), Environmental Taxation Empirical and Theoretical Applications
- 122. **Ivaschenko, Oleksiy** (2003), Essays on Poverty, Income Inequality and Health in Transition Economies
- 123. Lundström, Susanna (2003), On Institutions, Economic Growth and the Environment
- 124. Wambugu, Anthony (2003), Essays on Earnings and Human Capital in Kenya
- 125. Adler, Johan (2003), Aspects of Macroeconomic Saving
- 126. Erlandsson, Mattias (2003), On Monetary Integration and Macroeconomic Policy
- 127. Brink, Anna (2003), On the Political Economy of Municipality Break-Ups
- 128. Ljungwall, Christer (2003), Essays on China's Economic Performance During the Reform Period
- 129. Chifamba, Ronald (2003), Analysis of Mining Investments in Zimbabwe
- 130. **Muchapondwa, Edwin** (2003), The Economics of Community-Based Wildlife Conservation in Zimbabwe
- 131. Hammes, Klaus (2003), Essays on Capital Structure and Trade Financing
- 132. Abou-Ali, Hala (2003), Water and Health in Egypt: An Empirical Analysis
- 133. **Simatele, Munacinga** (2004), Financial Sector Reforms and Monetary Policy in Zambia
- 134. Tezic, Kerem (2004), Essays on Immigrants' Economic Integration
- 135. INSTÄLLD

- 136. Gjirja, Matilda (2004), Efficiency and Productivity in Swedish Banking
- 137. Andersson, Jessica (2004), Welfare Environment and Tourism in Developing Countries
- 138. **Chen, Yinghong** (2004), Essays on Voting Power, Corporate Governance and Capital Structure
- 139. **Yesuf, Mahmud** (2004), Risk, Time and Land Management under Market Imperfections: Applications to Ethiopia
- 140. **Kateregga, Eseza** (2005), Essays on the Infestation of Lake Victoria by the Water Hyacinth
- 141. **Edvardsen, Dag Fjeld** (2004), Four Essays on the Measurement of Productive Efficiency
- 142. Lidén, Erik (2005), Essays on Information and Conflicts of Interest in Stock Recommendations
- 143. Dieden, Sten (2005), Income Generation in the African and Coloured Population– Three Essays on the Origins of Household Incomes in South Africa
- 144. Eliasson, Marcus (2005), Individual and Family Consequences of Involuntary Job Loss
- 145. **Mahmud, Minhaj** (2005), Measuring Trust and the Value of Statistical Lives: Evidence from Bangladesh
- 146. **Lokina, Razack Bakari** (2005), Efficiency, Risk and Regulation Compliance: Applications to Lake Victoria Fisheries in Tanzania
- 147. **Jussila Hammes, Johanna** (2005), Essays on the Political Economy of Land Use Change
- 148. **Nyangena, Wilfred** (2006), Essays on Soil Conservation, Social Capital and Technology Adoption
- 149. **Nivorozhkin, Anton** (2006), Essays on Unemployment Duration and Programme Evaluation
- 150. Sandén, Klas (2006), Essays on the Skill Premium
- 151. **Deng, Daniel** (2006), Three Essays on Electricity Spot and Financial Derivative Prices at the Nordic Power Exchange
- 152. **Gebreeyesus, Mulu** (2006), Essays on Firm Turnover, Growth, and Investment Behavior in Ethiopian Manufacturing
- 153. **Islam, Nizamul Md.** (2006), Essays on Labor Supply and Poverty: A Microeconometric Application
- 154. **Kjaer, Mats** (2006), Pricing of Some Path-Dependent Options on Equities and Commodities
- 155. **Shimeles, Abebe** (2006), Essays on Poverty, Risk and Consumption Dynamics in Ethiopia
- 156. Larsson, Jan (2006), Four Essays on Technology, Productivity and Environment
- 157. **Congdon Fors, Heather** (2006), Essays in Institutional and Development Economics
- 158. **Akpalu, Wisdom** (2006), Essays on Economics of Natural Resource Management and Experiments
- 159. **Daruvala, Dinky** (2006), Experimental Studies on Risk, Inequality and Relative Standing
- 160. **García, Jorge** (2007), Essays on Asymmetric Information and Environmental Regulation through Disclosure

- 161. **Bezabih, Mintewab** (2007), Essays on Land Lease Markets, Productivity, Biodiversity, and Environmental Variability
- 162. **Visser, Martine** (2007), Fairness, Reciprocity and Inequality: Experimental Evidence from South Africa
- 163. **Holm, Louise** (2007), A Non-Stationary Perspective on the European and Swedish Business Cycle
- 164. Herbertsson, Alexander (2007), Pricing Portfolio Credit Derivatives
- 165. **Johansson, Anders C.** (2007), Essays in Empirical Finance: Volatility, Interdependencies, and Risk in Emerging Markets
- 166. **Ibáñez Díaz, Marcela** (2007), Social Dilemmas: The Role of Incentives, Norms and Institutions
- 167. **Ekbom, Anders** (2007), Economic Analysis of Soil Capital, Land Use and Agricultural Production in Kenya
- 168. Sjöberg, Pål (2007), Essays on Performance and Growth in Swedish Banking
- 169. **Palma Aguirre, Grisha Alexis** (2008), Explaining Earnings and Income Inequality in Chile
- 170. **Akay, Alpaslan** (2008), Essays on Microeconometrics and Immigrant Assimilation
- 171. Carlsson, Evert (2008), After Work Investing for Retirement
- 172. Munshi, Farzana (2008), Essays on Globalization and Occupational Wages
- 173. **Tsakas, Elias** (2008), Essays on Epistemology and Evolutionary Game Theory
- 174. Erlandzon, Karl (2008), Retirement Planning: Portfolio Choice for Long-Term Investors
- 175. **Lampi, Elina** (2008), Individual Preferences, Choices, and Risk Perceptions Survey Based Evidence
- 176. **Mitrut, Andreea** (2008), Four Essays on Interhousehold Transfers and Institutions in Post-Communist Romania
- 177. **Hansson, Gustav** (2008), Essays on Social Distance, Institutions, and Economic Growth
- 178. **Zikhali, Precious** (2008), Land Reform, Trust and Natural Resource Management in Africa
- 179. **Tengstam, Sven** (2008), Essays on Smallholder Diversification, Industry Location, Debt Relief, and Disability and Utility
- 180. Boman, Anders (2009), Geographic Labour Mobility Causes and Consequences
- 181. **Qin, Ping** (2009), Risk, Relative Standing and Property Rights: Rural Household Decision-Making in China
- 182. Wei, Jiegen (2009), Essays in Climate Change and Forest Management
- Belu, Constantin (2009), Essays on Efficiency Measurement and Corporate Social Responsibility
- 184. Ahlerup, Pelle (2009), Essays on Conflict, Institutions, and Ethnic Diversity
- 185. **Quiroga, Miguel** (2009), Microeconomic Policy for Development: Essays on Trade and Environment, Poverty and Education
- 186. Zerfu, Daniel (2010), Essays on Institutions and Economic Outcomes
- 187. Wollbrant, Conny (2010), Self-Control and Altruism
- 188. Villegas Palacio, Clara (2010), Formal and Informal Regulations: Enforcement and Compliance

189. **Maican, Florin G.** (2010), Essays in Industry Dynamics on Imperfectly Competitive Markets